

International Labour Office Geneva

# Women and men in the informal economy: A statistical picture



# Women and Men in the Informal Economy: A Statistical Picture

Third Edition

# Women and Men in the Informal Economy: A Statistical Picture

Third Edition

International Labour Office, Geneva

Copyright © International Labour Organization 2018 First published 2018

Publications of the International Labour Office enjoy copyright under Protocol 2 of the Universal Copyright Convention. Nevertheless, short excerpts from them may be reproduced without authorization, on condition that the source is indicated. For rights of reproduction or translation, application should be made to ILO Publications (Rights and Licensing), International Labour Office, CH-1211 Geneva 22, Switzerland, or by email: rights@ilo.org. The International Labour Office welcomes such applications.

Libraries, institutions and other users registered with a reproduction rights organization may make copies in accordance with the licences issued to them for this purpose. Visit www.ifrro.org to find the reproduction rights organization in your country.

Women and men in the informal economy: a statistical picture (third edition) / International Labour Office – Geneva: ILO, 2018

ISBN 978-92-2-131580-3 (print) ISBN 978-92-2-131581-0 (web pdf)

International Labour Office informal employment / women workers / men workers / temporary employment / part time employment / informal economy / urban area / data collecting / measurement / developed countries 13.01.3

ILO Cataloguing in Publication Data

The designations employed in ILO publications, which are in conformity with United Nations practice, and the presentation of material therein do not imply the expression of any opinion whatsoever on the part of the International Labour Office concerning the legal status of any country, area or territory or of its authorities, or concerning the delimitation of its frontiers.

The responsibility for opinions expressed in signed articles, studies and other contributions rests solely with their authors, and publication does not constitute an endorsement by the International Labour Office of the opinions expressed in them.

Reference to names of firms and commercial products and processes does not imply their endorsement by the International Labour Office, and any failure to mention a particular firm, commercial product or process is not a sign of disapproval.

Information on ILO publications and digital products can be found at: www.ilo.org/publns.

This publication was produced by the Document and Publications Production, Printing and Distribution Branch (PRODOC) of the ILO.

Graphic and typographic design, layout and composition, proofreading, printing, electronic publishing and distribution.

PRODOC endeavours to use paper sourced from forests managed in an environmentally sustainable and socially responsible manner.

Code: JMB-CORR-ICA

# Preface

More than 60 per cent of the world's employed population earn their livelihoods in the informal economy. Informality exists in all countries regardless of the level of socio-economic development, although it is more prevalent in developing countries. The 2 billion women and men who make their living in the informal economy are deprived of decent working conditions. Evidence shows that most people enter the informal economy not by choice, but as a consequence of a lack of opportunities in the formal economy and in the absence of other means of livelihood.

The main challenge for the transition to the formal economy is finding the right policy mix that corresponds to the diversity of characteristics and drivers of informality. Reliable and relevant statistics are needed to better understand these complex aspects of informality and monitor progress towards formalization. In June 2015, the International Labour Conference adopted the Transition from the Informal to the Formal Economy Recommendation (No. 204), the first international labour standard which focuses on the informal economy in its entirety. That same year, in September, the United Nations adopted the 2030 Agenda for Sustainable Development, which included the transition to formality in the targets for Sustainable Development Goal 8. These two instruments represent major milestones in the global approach to formalization, particularly by providing guidance on the process. The ILO has made the formalization of the informal economy one of its strategic outcomes and supports tripartite constituents in facilitating the transition to the formal level.

This report forms part of the ILO follow-up plan of action to support the implementation of Recommendation No. 204. It is all the more timely given that the United Nations General Assembly has recently approved the Global Indicator Framework to monitor the Sustainable Development Goals, including a specific global statistical indicator on informal employment (8.3.1). The report provides for the first time a statistical profile of the informal economy at the global level using a common set of operational criteria to measure informal employment and employment in the informal sector for more than 100 countries, including both developed and developing countries. Statistics on informal employment are disaggregated by sex, age, level of education, status in employment and other socio-economic characteristics.

This is the outcome of joint collaboration by the Employment Policy Department (led by Azita Berar Awad, former Director), the Inclusive Labour Markets, Labour Relations and Working Conditions Branch and the Department of Statistics. It has benefited from valuable inputs from Joann Vanek of the Women in Informal Employment: Globalizing and Organizing (WIEGO) who also contributed to the two previous editions of the *Women and Men in the Informal Economy* statistical reports.

This publication was prepared by Florence Bonnet, Vicky Leung and Juan Chacaltana. It was extensively reviewed in the ILO, benefiting from comments by the following colleagues: Mariya Aleksynska, Coffi Agossou, Yacouba Diallo, Steven Kapsos, David Kucera, Fatime Ndiaye, Frédéric Lapeyre, Jon Messenger, Naoko Otobe, Yves Perardel, Ken Shawa and Judith Van Doorm. Our special appreciation goes to Chris Edgar for his support during the editorial process.

Sangheon Lee Director	Philippe Marcadent Chief	Rafa
Employment Policy	Inclusive Labour Markets,	
Department	Labour Relations and Working	
	Conditions Branch	

Rafael Diez de Medina Director Department of Statistics

# Contents

Pref	face		V
Abb	reviatio	ons and acronyms	xiii
		n: Sustainable development, decent work for all and the transition formal to the formal economy	1
	pter 1: n the in	Measuring the informal economy with a view to facilitating the transition formal to the formal economy	3
1.1		tion from the Informal to the Formal Economy Recommendation, No. 204)	3
1.2		is new in this publication?	6
Cha	pter 2:	Statistical overview relating to the informal economy	13
2.1	Globa	overview	13
	Magn	tude of the informal economy	13
	Diverg	ing trends in the share of informal employment	15
	Comp	osition of the informal economy	16
	Inform	al economic units	17
	Inform	ality and status in employment	17
	Age p	ofile of formality	19
	Educa	tion and informality	19
	Inform	nality and rural/ urban location	20
	Sector	al dimension of informality	20
	Gende	r dimension of informality	20
2.2	Region	nal overview	27
	2.2.1	Africa	27
		Magnitude	29
		Composition	30
		Characteristics	30
	2.2.2	Americas	31
		Magnitude	33
		Composition	34

		Characteristics	34
	2.2.3	Asia and the Pacific	35
		Magnitude	37
		Composition	38
		Characteristics	38
	2.2.4	Europe and Central Asia	39
		Magnitude	41
		Composition	42
		Characteristics	42
Cha	pter 3:	Thematic issues in the informal economy	45
3.1	Inform	al employment and level of economic and social development	45
3.2	Inform	al employment and poverty	48
3.3	Inform	al employment and education	51
3.4	Inform	al employment and working conditions	58
	3.4.1	Informal employment and non-standard forms of employment among employees.	58
	3.4.2	Informal employment and working time	61
Cha	pter 4:	Main findings	67
Refe	erences		71
Арр	endix A	Country groupings, sources and method	73
Арр	endix B	Statistical appendix: Country data. Key indicators on the size and composition of the informal economy	85
Арр	endix C	Statistical appendix: Global and regional estimates Micro determinants of informality and informality and working conditions	132
List	of figur	es	
Figu	ire 1.	Intervention model for integrated strategies	4
Figu	ire 2.	Overview of main steps and possible sequence of the diagnostics	5
Figu	ire 3.	Criteria and sequence used in the common operational definition of the informal sector (main job)	9
Figu	ire 4.	Criteria used in the common operational definition of the informal employment (main job)	11
Figu	ire 5.	Share of informal employment in total employment, including and excluding agriculture (percentages, 2016)	13
Figu	ire 6.	Trends in the share of informal employment in total employment (including and excluding agriculture, percentages)	15

Figure 7.	Percentage of economic units in the informal sector, the formal sector and household sector (percentages, 2016)	1
Figure 8.	Informality and status in employment	1
Figure 9.	Share of formal employment in total employment by age (percentages, 2016)	1
Figure 10.	Share of informal employment in total employment by level of education (percentages, 2016)	2
Figure 11.	Gender gap in the share of informal employment in total employment in total employment including agriculture (percentage points, latest available year)	2
Figure 12.	Distribution of the employed population in informal employment by categories of employment status and sex (percentages, 2016)	2
Figure 13.	Key indicators on the size and composition of the informal economy in Africa (percentages, 2016)	2
Figure 14.	Key indicators on the size and composition of the informal economy in the Americas (percentages, 2016)	3
Figure 15.	Key indicators on the size and composition of the informal economy in Asia and the Pacific (percentages, 2016)	3
Figure 16.	Key indicators on the size and composition of the informal economy in Europe and Central Asia (percentages, 2016)	3
Figure 17.	Shares of informal employment in total employment and Human Development Index values (latest available year)	4
Figure 18.	Share of informal employment in total employment and GDP per capita (latest available year)	4
Figure 19.	The gender dimension of informality and development: gender gap in the share of informal employment in total employment (percentage points, latest available year)	4
Figure 20.	The sectoral composition of GDP (contribution of main industry sectors to GDP) and the share of informal employment in total employment (percentages, latest available year)	4
Figure 21.	Share of status in employment and share of informal employment in total employment (percentages, latest available year)	4
Figure 22.	Shares of informal employment among the poor (horizontal axis) and non-poor (vertical axis) (percentages, latest available year)	4
Figure 23.	Poverty rates and the informal or formal nature of worker's main job (percentages, latest available year)	5
Figure 24.	Working poverty rates among women and men in informal and in formal employment (percentages, latest available year)	5
Figure 25.	Share of informal employment (percentage of total employment) by level of education (percentages, latest available year)	5
Figure 26.	Distribution of workers in informal and formal employment by level of education (percentages, latest available year)	5
Figure 27.	Share of informal employment (percentage of total employment) by level of education and status in employment (percentages, latest available year)	5

Figure 28.	Share of employment in informal economic units and the level of education and status in employment (percentages, latest available year)	55
Figure 29.	Educational composition of employment in the informal and formal sector (percentages, latest available year)	56
Figure 30.	Share of informal employment (percentage of total employment) by level of education and sex (percentages, latest available year)	57
Figure 31.	Share of informal employment among employees in permanent full-time employment compared to non-standard forms of employment by sex (percentages, latest available year)	59
Figure 32.	Share of informal employment among workers in marginal employment (less than 20 hours a week), in part-time employment (less than 35 hours of week) compared to those working more than 35 hours a week (percentages, latest available year)	61
Figure 33.	Proportion of workers with less than 20 hours a week and the formal or informal nature of worker's main job (percentages, based on latest available year)	63
Figure 34.	Incidence of time-related underemployment and the informal or formal nature of employment (percentages, latest available year)	64
Figure 35.	Proportion of workers working more than 48 hours a week and the formal or informal nature of workers' main job (percentages, latest available year)	65
Figure C.1	Share of informal employment by level of education, status in employment and region	147
Figure C.2	Proportion of workers working more than 60 hours a week, depending of the formal or informal nature of main job	156

# List of tables

Table 1	Key indicators on the size and composition of the informal economy: a global picture (percentages, 2016)	23
Table 2	Key indicators on the size and composition of the informal economy in Africa (percentages, 2016)	28
Table 3	Key indicators on the size and composition of the informal economy in the Americas (percentages, 2016)	32
Table 4	Key indicators on the size and composition of the informal economy in Asia and the Pacific (percentages, 2016)	36
Table 5	Key indicators on the size and composition of the informal economy in Europe and Central Asia (percentages, 2016)	40
Table B.1	Share of informal employment in total employment and in non-agricultural employment by sex	85
Table B.2	Share of informal employment in total employment and in non-agricultural employment by urban/rural location	91
Table B.3	Share of informal employment in total employment by status in employment (including agriculture)	97

Table B.4	Share of informal employment in total employment by status in employment (excluding agriculture)	103
Table B.5	Distribution of workers in informal employment and in formal employment by employment status and sex (including agriculture)	109
Table B.6	Share of informal employment in agriculture, industry and services by sex $\ldots$ .	115
Table B.7	Distribution of workers in informal employment and in formal employment by sector.	120
Table B.8	Share of employment in the informal sector, in the formal sector and in households by sex	126
Table C.1	Share of informal employment by level of education, status in employment, sex and region	134
Table C.2	Distribution of informal employment and formal employment by level of education, status in employment, sex and region	138
Table C.3	Share of informal employment by age group, status in employment, sex and region	148
Table C.4	Informality and working time by sex for total employment and for employees	152

# List of boxes

Box 1	National diagnostics and statistics	4
Box 2	International and operational definitions of informal employment and employment in the informal sector	7
Box 3	Women and men in the informal economy	21
Box 4	Are educated (and more productive) workers more likely to go to formal sector jobs? Is labour in informal sector enterprises less qualified?	55

# Abbreviations and acronyms

GDP	gross domestic product
GNI	gross national income
HDI	Human Development Index
ICLS	International Conference for Labour Statisticians
ILC	International Labour Conference
ILO	International Labour Organization/Office
ISSA	International Social Security Association
LAC	Latin America and the Caribbean
NGOs	non-governmental organizations
PPP	Purchasing power parity
SDGs	Sustainable Development Goals
UNDP	United Nations Development Programme
WDI	World Development Indicators
WIEGO	Women in Informal Employment: Globalizing and Organizing

# Introduction

# Sustainable development, decent work for all and the transition from the informal to the formal economy

Policy-makers, workers' and employers' organizations and academics increasingly acknowledge that the high incidence of informality in all its aspects is a major challenge for sustainable development. Informality has a harmful effect on workers' rights, including fundamental principles and rights at work, social protection, decent working conditions and the rule of law. It also has a negative impact on the development of sustainable enterprises (especially in terms of low productivity and lack of access to finance), public revenues and governments' scope of action, particularly with regard to economic, social and environmental policies, the robustness of institutions, and fair competition in national and international markets.

Statistics serve as quantifiable evidence to underpin the policy-making process. However, informality also poses another challenge to policy-makers, given that by its nature production in the informal economy is either underestimated or often not recorded in national accounts. This makes an analysis of the situation and subsequently the formulation of policies to protect rights in the sector difficult. The key challenges for measuring the informal economy include promoting a better understanding of the statistical concepts of the informal economy, guiding countries on the practical application of international standards within statistical frameworks, collecting the required data in national surveys and analysing informal economy statistics as an input to policy design and monitoring (ILO, 2013a).

As 2 billion of the world's employed population make their living in the informal economy, there is an urgent need to tackle informality. Although not everyone in the informal economy is poor and there is also poverty in the formal economy, ample empirical research has shown that workers in the informal economy face a higher risk of poverty than those in the formal economy, while informal economic units face<sup>1</sup> lower productivity and income. Indeed, most people enter the informal economy not by choice but as a consequence of a lack of opportunities in the formal economy and in the absence of any other means of earning a living. For all those reasons, transition from the informal to the formal economy is of strategic significance for hundreds of millions of workers and economic units around the world that are working and producing in precarious and vulnerable conditions.

The informal economy encompasses a huge range of situations across and within economies. From a policy-making perspective, it is important to stress that workers and economic units – including enterprises, entrepreneurs and households – in the informal economy present a broad diversity of characteristics and needs. This needs to be known and understood, together with the reasons for the persistence of the informal economy, in order to formulate effective policies for the transition to formality. Workers in the informal economy differ widely in terms of income level, age, education level, status in employment, sector, type and size of enterprise, location of workplace (rural or urban, private or public areas), degree of coverage of social and employment protection.

<sup>&</sup>lt;sup>1</sup> In line with Recommendation No. 204, "economic units" in the informal economy include: (a) units that employ hired labour; (b) units that are owned by individuals working on their own account, either alone or with the help of contributing family workers; and (c) cooperatives and social and solidarity economy units.

Economic units in the informal economy also demonstrate a vast diversity of characteristics with regard to size, sectors of concentration, degree of compliance with laws and regulations, or level of productivity, among others.

The ILO has played a leading role in placing the topic on the international policy agenda and debate for more than 40 years. It has produced data on informal employment and developing research, including on a) working conditions in the informal economy, b) the drivers of informalization and c) innovative policies to facilitate the transition from the informal to the formal economy. The adoption of the Recommendation No. 204<sup>2</sup> by the International Labour Conference in 2015 was of strategic significance for the world of work and for the future of work as it is the first international labour standard to focus on the informal economy in its entirety and diversity and to provide practical guidance to address these priorities. This Recommendation clearly stresses the need to: (a) facilitate the transition of workers and economic units from the informal to the formal economy, while respecting workers' fundamental rights and ensuring opportunities for income security, livelihoods and entrepreneurship; (b) promote the creation, preservation and sustainability of enterprises and decent jobs in the formal economy and the coherence of macroeconomic, employment, social protection and other social policies; and (c) prevent the informalization of formal economy jobs.

This new consensus recognizing that transition from the informal to the formal economy is essential to achieving inclusive development and to realizing decent work for all is reflected in the 2030 Agenda for Sustainable Development Agenda,<sup>3</sup> notably Sustainable Development Goal (SDG) 8, which is to promote sustained, inclusive and sustainable economic growth, full and productive employment and decent work for all, and, in particular, for the fulfilment of Target 8.3. This target aims to promote development-oriented policies that support productive activities and decent job creation, and to encourage the formalization and growth of micro-, small and medium-sized enterprises, including through access to financial services.

While a number of countries have a proven record of developing effective legal and policy frameworks for formalization, many others have only recently embarked on similar paths or are starting to consider the possibility of doing so. Thus one can see a trend in all regions of the world to adapt and strengthen national statistical systems to better capture informal employment in a way that enables monitoring the results of the new policy frameworks to support the process of transition to formality. There is now considerable demand from member States for evidence-based practical knowledge and exchange of experience about producing data on informal employment, as the monitoring system of the SDGs will now include a new indicator – indicator 8.3.1 ("share of informal employment in non-agriculture employment, by sex") – which will be the indicator to monitor progress for Target 8.3.

Chapter 1 of this publication explains the latest context in which statistics relating to the informal economy have been included as an important element in the policy-making process for the transition from the informal to the formal economy. This chapter also explains the methodology, using a common set of operational criteria to calculate informal employment and employment in the informal sector. Chapter 2 provides global and regional estimates using the latest available data and detailed profiles for each region. Chapter 3 focuses on thematic issues regarding the informal economy, namely the relation between informality and the level of social and economic development. This chapter examines the close relationship between informality and the level of economic development, poverty, the level of education and working conditions respectively. Chapter 4 presents the main findings.

<sup>&</sup>lt;sup>2</sup> Available at: http://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPUB:12100:0::NO:12100:P12100\_INSTRUMENT\_ ID:3243110:NO

<sup>&</sup>lt;sup>3</sup> Available at: http://www.un.org/sustainabledevelopment/

# Chapter 1

# Measuring the informal economy with a view to facilitating the transition from the informal to the formal economy

# 1.1 Transition from the Informal to the Formal Economy Recommendation, 2015 (No. 204)

Recommendation No. 204 advocates for a practical approach to achieving decent work for all and inclusive development. It puts an emphasis on integrated strategies in order to facilitate the transition to the formal economy, create new formal jobs and prevent further informalization. It also recognizes the crucial role of statistics in the policy process.

In this regard, Recommendation No. 204 encourages member States to undertake a proper assessment and diagnostics of factors, characteristics, causes and circumstances of informality in the national context to inform the design and implementation of laws and regulations, policies and other measures aiming to facilitate the transition to the formal economy.

Furthermore, the transition to formality is increasingly seen as a central goal in national employment policies (ILO, 2014a). This is in line with Recommendation No. 204, part IV of which is dedicated to providing policy guidance on employment policies. In pursuing the objective of quality job creation in the formal economy, the Recommendation advises member States to formulate and implement a national employment policy in line with the Employment Policy Convention, 1964 (No. 122), and to make full, decent, productive and freely chosen employment a central goal in their national development and growth strategy or plan.

Recommendation No. 204 provides policy guidance on data collection and monitoring in Part VIII and proposes that member States should, in consultation with employers' and workers' organizations, on a regular basis, where possible and as appropriate, collect, analyse and disseminate statistics disaggregated by sex, age, workplace, and other specific socio-economic characteristics on the size and composition of the informal economy, including the number of informal economic units, the number of workers employed and their sectors; and monitor and evaluate the progress towards formalization.

In addition, Recommendation No. 204 requests that in developing or revising the concepts, definitions and methodology used in the production of data, statistics and indicators on the informal economy, member States should take into consideration relevant guidance provided by the International Labour Organization (ILO), in particular and as appropriate, the guidelines concerning a statistical definition of informal employment adopted by the 17th International Conference of Labour Statisticians (ICLS) in 2003 and their subsequent updates.

The International Labour Conference (ILC) also adopted a Resolution concerning efforts to facilitate the transition from the informal to the formal economy, which calls for ILO tripartite constituents

to give full effect to the Recommendation and the ILO to develop a strategy and action plan for promoting and supporting the implementation of the Recommendation (ILO, 2015a).

The follow-up to this resolution adopted by the Governing Body proposed priorities for a possible ILO action strategy for the period 2016–21 to support the implementation of Recommendation No. 204's integrated approach by member States (ILO, 2015b). It proposes an intervention model to constituents for designing, implementing and monitoring integrated strategies to facilitate the transition to the formal economy according to national circumstances and priorities (figure 1). Statistics play an important role in the diagnostics (box 1) and monitoring phase to track the progress towards formalization.

### Figure 1. Intervention model for integrated strategies



**Source:** Intervention model for integrated strategies presented in ILO, Formalization of the informal economy: Follow-up to the resolution concerning efforts to facilitate the transition from the informal to the formal economy, 325th Session, Governing Body, Geneva, 29 October–12 November 2015, GB.325/POL/1/2.

### **Box 1.** National diagnostics and statistics

According to Recommendation No. 204, the main purpose of undertaking the diagnostics is to understand the factors, characteristics, causes and circumstances of informality in the national context and to inform the design and implementation of laws and regulations, policies and other measures aiming to facilitate the transition to the formal economy.

Diagnostics is composed of several components or steps to collect, analyse, share and discuss quantitative and qualitative information to create the conditions for a consensus about the situation and agree on priorities. Figure 2 below provides an overview of main steps and possible sequence of national diagnostics. In order to understand every piece of the puzzle in the informal economy, statistics play a key role especially in the first step that is to quantify the extent and qualify the nature of the informal economy, to assess working conditions and to identify some of the causes of informality.



# 1.2 What is new in this publication?

Building on the momentum created since the adoption of the Sustainable Development Goals (SDGs) and Recommendation No. 204, the ILO is supporting its member States in conducting diagnostics of the informal economy, including assisting national statistical offices with producing sex-disaggregated statistics for reporting on indicator 8.3.1 under Goal 8 of the SDGs, fostering tripartite consensus on follow-up actions and designing systems for regular monitoring of progress towards formalization.

In order to fulfil these measurement objectives, statistics on the informal sector and on informal employment need to be given high priority in national statistical policy, and the production of statistics on these topics should be integrated into the regular national statistical system. Strate-gies and programmes for the regular collection of statistics on the informal sector and on informal employment should form an integral part of every national plan for statistical development. These programmes should be based on surveys that measure informal employment directly, thus avoid-ing indirect estimation methods as much as possible as they are imprecise and too aggregated to provide relevant information for policy-making.

In that context, the growing production of statistics on the informal sector and on informal employment marks an important step towards the overall improvement of labour statistics, economic statistics and national accounts. An increasing but still insufficient number of countries are applying an appropriate methodology for the statistical measurement of the informal sector and of informal employment. Those statistics are crucial for designing and evaluating public policies and programmes that aimed at promoting and creating quality employment in the formal economy.

The first edition of *Women and men in the informal economy: A statistical picture* was published in 2002 to support the general discussion on decent work and the informal economy at the 90th Session of the ILC (Geneva, June 2002). The statistics in the first edition were prepared using indirect and residual methods (ILO, 2002) for 25 countries in the absence of reliable data collected directly. Following the 2002 ILC, the 17th ICLS took up the challenge of developing new frameworks which could better capture the phenomenon of informality and adopted guidelines endorsing the framework as an international statistical standard in 2003 (ILO, 2013a).

The conceptual framework in the 17th ICLS guidelines links the enterprise-based concept of employment in the informal sector in a coherent and consistent manner with a broader, job-based concept of informal employment (ILO, 2013b). The second edition of the publication (2013) compiled data for 47 countries, which are based on direct measures supplied by countries and using data from national surveys.

To enhance international comparability, this third edition applies, as far as possible, a systematic approach to measuring informal employment and employment in the informal sector when processing micro data for more than 100 countries representing more than 90 per cent of the world's employed population aged 15 years old and over.<sup>4</sup> All estimates are based on a common set of operational criteria to determine informal employment and employment in the informal sector as a person's main job (box 2). As a result, statistics in this edition are comparable across countries and regions, but the ILO's country estimates might differ from national ones when they exist. Countries benefit from a certain degree of flexibility to measure both employment in the informal sector and informal employment to adjust to the national context and circumstances while being in line with international standards. The main objective of this edition is to provide estimates with the best level of comparability between countries. Moreover, this edition provides global and regional estimates for 2016 based on proportions calculated from data for the latest

<sup>&</sup>lt;sup>4</sup> Missing values are estimated for countries where micro data were not available for the main indicators related to the size and composition of informal employment, in particular for indicator SDG 8.3.1 (see estimation method presented in Appendix A.3 and key indicators on the size and composition of the informal economy in Appendix B).

available year.<sup>5</sup> Statistics on the size of the informal economy are disaggregated by sex, age, level of education, workplace (rural and urban), status in employment and other specific socio-economic characteristics allowing for an analysis of the composition of the informal economy in line with the policy guidance on data collection and monitoring in Recommendation No. 204. This harmonization process and the set of indicators provided in this third edition should not, however, override the important need to improve data collection tools in countries to ensure the inclusion of the necessary questions in main national surveys<sup>6</sup> and then the analysis of data to support policies.

# Box 2. International and operational definitions of informal employment and employment in the informal sector

International standards distinguish between employment in the informal sector and informal employment. Employment in the informal sector is an enterprise-based concept and it is defined in terms of the characteristics of the place of work of the worker. By contrast, informal employment is a job-based concept and it is defined in terms of the employment relationship and protections associated with the job of the worker.

#### A. Employment in the informal sector

According to the international standards adopted by the 15th ICLS, the informal sector consists of units engaged in the production of goods or services with the primary objective of generating employment and incomes to the persons concerned. The informal sector is a subset of unincorporated enterprises not constituted as separate legal entities independently of their owners (ILO, 1993). They are owned by individual household members or several members of the same or different households. Typically, they are operating at a low level of organization, on a small scale and with little or no division between labour and capital as factors of production.

In order to allow international comparisons, the ILO adopted a common operational definition of employment in the informal sector using the following criteria:

<ul> <li>Institutional sector (government, public enterprises, non-governmental organizations (NGOs); private sector; households). The "institutional sector" is meant to separate persons working in government, public and private corporations, non-governmental and international organizations, and other institutions clearly recognized as belonging to the formal sector. It also serves to identify persons working in private households producing wholly for own final use.</li> <li>If working for:</li> <li><i>Government/ public corporations, NGOs and international NGOs</i>, the economic unit is considered as being part of the formal sector</li> <li><i>Households</i>, the economic unit is part of households</li> <li><i>Private economic units</i>, then additional criteria are needed to assess whether the unit is incorporated or constituted as a separate legal entity</li> </ul>		Criteria used to define the informal sector and employment in the informal sector	Operational definition and modalities to define formal and informal economic units
	1	non-governmental organizations (NGOs); private sector; households). The "institutional sector" is meant to separate persons working in government, public and private corporations, non-governmental and international organizations, and other institutions clearly recognized as belonging to the formal sector. It also serves to identify persons working in private	<ul> <li>Government/ public corporations, NGOs and international NGOs, the economic unit is considered as being part of the formal sector</li> <li>Households, the economic unit is part of households</li> <li>Private economic units, then additional criteria are needed to assess whether the unit is incorpo-</li> </ul>

<sup>&</sup>lt;sup>5</sup> The range of years of micro data used as a basis for the estimates is from mid-2000 to 2016. Data for more than half of the countries are from 2013 onwards and from 2010 onwards for 90 per cent of the countries considered (see original sources and years in Appendix A.2). For each indicator, global and regional estimates of proportions result from the weighted average of national proportions for the latest year available (as indicated in Appendix A.2). Those regional and global estimates are weighted by the denominator of the considered indicator using 2016 data from the ILO's Trends Econometric Models as relevant. When absolute numbers are presented in this report, they refer to 2016 by multiplying the estimated regional or global estimate by absolute numbers for 2016 from the ILO's Trends Econometric Models as appropriate according to the denominator (see Appendix A.3).

<sup>&</sup>lt;sup>6</sup> See the analysis of main gaps on Appendix A.3.

	Criteria used to define the informal sector and employment in the informal sector	Operational definition and modalities to define formal and informal economic units
2	<b>Final destination of production</b> : the purpose of the second mandatory criterion about the "destination of production" is to exclude from the scope of informal sector persons working in a farm or private business (unincorporated enterprise) where the main intended destination of the production is wholly for own final use. <sup>7</sup>	<ul> <li>If produce goods and services:</li> <li>at least partially for sale or barter, then go to the next criteria</li> <li>If final use only, the economic unit is part of households</li> </ul>
3	<b>Registration of the economic unit under national</b> <b>legislation</b> (in the process of registration is consid- ered as not registered): this includes registration with social security authorities, sales or income tax authorities and should be at national level. It identifies enterprises that are similar to corpora- tions (quasi-corporations) and therefore outside the scope of the informal sector. The appropriate forms of registration relevant to the concept of informal sector should be examined in the national context.	<ul> <li>If registered to relevant national institution(s), the economic unit is part of the formal sector</li> <li>If not registered/ in the process of registration/ missing information/ don't know, then go to the next criteria</li> </ul>
4	<b>Bookkeeping:</b> this criterion assesses whether the economic unit maintains a set of accounts required by law (e.g. balance sheets) or keeps some official accounts. The purpose of the information on bookkeeping practices of the farm or private unincorporated enterprise is to identify whether the economic unit is constituted as a separate legal entity independent of its owner(s).	<ul> <li>If <i>formal bookkeeping</i>, the economic unit is part of the <b>formal sector</b></li> <li>If no formal bookkeeping/ missing information/ don't know, then go to the <b>next criteria</b> (alternative approach)</li> </ul>
	ative criteria are <b>applied only if information about criteria</b> ia 1 and 2 are mandatory criteria.	3 and 4 is missing (no question, no answer or don't know)
5	For employees: Contribution of employer (and employee) to social security or alternatively declaration of labour income (e.g. does the employer contribute to social security or declare labour income in order to pay income tax?)	<ul> <li>If employees are contributing to social security (preferably employers' contributions on behalf of employees), then the economic unit is part of the formal sector</li> <li>Otherwise (no contribution, no answer, don't know), then go to the next criteria</li> </ul>
6	Size of the economic unit: differentiate between units with five or fewer workers and those with more than five workers engaged/employed on a continuous basis.	If the unit has: more than five workers and is located in fixed visible premises, then the economic unit is part of
7	Location of workplace: Differentiate between economic units that are located in the owners' dwelling, in the street, in construction sites, in agricultural plots, that are itinerant, etc. and economic units that have fixed visible premises, such as offices and factories.	<ul> <li>the formal sector</li> <li>fewer than six workers or not in fixed premises, then the economic unit is part of the informal sector</li> </ul>

<sup>&</sup>lt;sup>7</sup> This is true if employment is defined according to the 13th ICLS resolution. The concept of (informal) employment in the 17th ICLS goes beyond employment as defined in the more up-to-date Resolution concerning statistics of work, employment and labour underutilization adopted at the 19th ICLS. Different activities are now excluded from employment that was previously included. This concerns in particular own-use production of goods which is an important component in informal employment. Some adjustments to the definition of informal employment are needed and will be discussed in the near future while currently in the transition period.



### **B.** Informal employment

In contrast to the concept of the informal sector that refers to production units as observation units, the concept of informal employment refers to jobs as observation units (ILO, 2003; Hussmanns, 2004). In the case of own-account workers and employers, the informal employment status of the job is determined by the informal sector nature of the enterprise. Thus, own-account workers (without hired workers) operating an informal enterprise are classified as in informal employment. Similarly, employers (with hired workers) operating an informal enterprise are classified as in informal employment. All contributing family workers are classified as having informal employment, irrespective of whether they work in formal or informal sector enterprises.

In the case of employees, informal employment is defined in terms of the employment relationship. According to international standards, for a job held by an employee to be considered as informal, the employment relationship should not be, in law or in practice, subject to national labour legislation, income taxation, social protection or entitlement to certain employment benefits (advance notice of dismissal, severance pay, paid annual or sick leave, etc.). The underpinning reasons may be the non-declaration of the jobs of the employees, casual jobs or jobs of a short duration, jobs with hours of work or wages below a specified threshold (e.g. for social security contributions) or lack of application of law and regulation in practice. Employers and own-account workers are considered to be informal when their economic units belong to the informal sector. In practice, the formal or informal nature of a job held by an employee is determined on the basis of operational criteria such as social security contributions by the employer (on behalf of the employee), and entitlement to paid sick leave and paid annual leave.

Criteria adopted for harmonized ILO estimates of informal employment:

Criteria used to define informal employment	Operational definition and modalities to define formal and informal employment
<ul> <li>The starting point is the employment status of the person.</li> <li>a) If the person is reported as a contributing family worker, no further questions are required and the person is classified as having an informal job.</li> <li>b) If the person is recorded as an employer, or own-account worker, or member of a producer's cooperative, no additional questions are required and the formal or informal nature of the job is determined according to the formal or informal nature of the person's economic unit. Employers, own-account workers and members of producers' cooperatives with enterprises in the formal sector are classified as having a formal job. Similarly, employers, own-account workers and members of producers' cooperatives with enterprises in the informal sector are classified as having a formal job. Similarly, employers, own-account workers and members of producers' cooperatives with enterprises in the informal sector are classified as having an informal job. If the enterprise is a household enterprise or a private business producing wholly for own or family use, the owner is also classified as having an informal job.</li> <li>c) The statistical treatment of 'employees' and 'not stated' is different and depends on the criteria of social security contributions by the employer or alternatively to entitlements to paid annual leave and paid sick leave.</li> </ul>	<ul> <li>If Contributing family workers, all considered as being in informal employment (here defined as the main job)</li> <li>If Employers, own-account workers, members of cooperatives, the formal / informal nature of their job is determined by the formal/ informal nature of their economic unit</li> <li>X If own a formal economic unit (formal sector), classified as having a formal job</li> <li>X If own an informal economic unit (informal sector), classified as having an informal job</li> <li>X If own an informal economic unit (informal sector), classified as having an informal job</li> <li>X If in households because of producing only for final use, classified as having an informal job</li> <li>If employees, then go to the next criteria</li> </ul>
<i>For employees</i> <b>Contributions to a social security</b> (ideally for pension) scheme by the employer (on behalf of the employee and usually complementing the employee's contribution) is the option most commonly used in countries and the one applied here. It should also be noted that the question is whether the employer makes contributions on behalf of the employee to the scheme, and not whether the employee is currently benefitting from the scheme with respect to a previ- ously held job.	<ul> <li>If contributions to social security (employer's contributions on behalf of employees), then the employee is in formal employment</li> <li>If no contributions to social security, considered as being in informal employment</li> <li>If don't know or no answer, then go to the next criteria</li> </ul>



## Figure 4. Criteria used in the common operational definition of the informal employment (main job)



The criteria and sequence were applied systematically to a person's main job in 119 countries with available data from the national labour force survey or similar national household survey. The main purpose is to allow international comparisons. They may differ from the approach adopted at the national level resulting in slightly different estimates of employment in the informal sector and informal employment. Chapter 2

# Statistical overview relating to the informal economy

This chapter provides a global overview and regional and subregional profiles on statistics relating to the informal economy. The structure of each section starts with the magnitude of the informal economy and follows with its composition and characteristics at the global and regional levels. The regions are divided into Africa, the Americas, Asia and the Pacific, and Europe and Central Asia.<sup>8</sup>

# 2.1 Global overview

# Magnitude of the informal economy

Two billion of the world's employed population aged 15 and over work informally, representing 61.2 per cent of global employment (figure 5, Panel C). The proportion of informal employment varies in different regions (figure 5, panels A and C). Among the five main regions, the vast majority of employment in Africa (85.8 per cent) is informal. Asia and the Pacific (68.2 per cent) and the Arab States (68.6 per cent) have almost the same level of informality. In the Americas (40.0 per

# Figure 5. Share of informal employment in total employment, including and excluding agriculture (percentages, 2016)



<sup>8</sup> Due to insufficient coverage from existing data and the absence of subregional division, regional profile is not available for the Arab States.

<sup>&</sup>lt;sup>9</sup> Even though SDG indicator 8.3.1 refers to the proportion of informal employment in non-agricultural employment, the incidence of informality in agriculture may be large in some countries. Thus, in order to have a comprehensive picture of workers' working conditions and the overall extent of informality, especially in developing and emerging countries, indicator 8.3.1 should be supplemented by other measures of informality covering the whole economy (agriculture and non-agriculture), as presented in figure 5, panel A and throughout this publication.





Panel C. Components of informal employment as a percentage of total employment: the informal sector, formal sector and household sector (percentages, 2016)



Note: Country data for figure 5 are provided in Appendix B. Original data sources are listed in Appendix A.2.

Source: ILO calculations based on household survey micro datasets.

cent) and Europe and Central Asia (25.1 per cent), less than half of employment is informal (figure 5, panel C and table 1.1). Excluding agriculture, the global level of informal employment falls to 50.5 per cent, but non-agricultural informal employment remains high in three regions (Africa, the Arab States, and Asia and the Pacific) (figure 5, panels B and C and table 1.2).

The level of socio-economic development is positively correlated to formality (also see Chapter 3.1). Developed countries are considered as high-income countries as defined by the World Bank; emerging countries as middle-income countries, and developing countries as low-income countries.<sup>10</sup> Emerging and developing countries have higher shares of informal employment than developed countries. Emerging and developing countries represent 82 per cent of world employment, but 93 per cent of the world's informal employment is in these countries. More than two-thirds of the employed population in emerging and developing countries are in informal employment (69.6 per cent), while less than one-fifth of the employed population (18.3 per cent) are in developed countries. Regions with countries at a higher level of socio-economic development, the Americas, Asia and the Pacific, and Europe and Central Asia have lower shares of informal employment (figure 5, panel C and table 1.1).

#### Diverging trends in the share of informal employment

Comparable data available over time do not allow to draw conclusions on global or regional trends. However, there is evidence of diverging trends depending on regions or countries (figure 6). In Latin America a decrease in the share of informal employment in the recent past seems to characterize the majority of countries as, for instance, in Argentina or Peru.<sup>11</sup> Similar trends can also be observed in Viet Nam and to some extent in South Africa.<sup>12</sup> In contrast, for countries with trends data from Europe and Central Asia with the exception of Armenia, the picture shows an increase in the share of informal employment as illustrated by the Russian Federation and Serbia. Finally, several countries such as Pakistan and Côte d'Ivoire remain at a high level of informality without clear downward or upward trends.

# Figure 6. Trends in the share of informal employment in total employment, including and excluding agriculture (percentages)



<sup>&</sup>lt;sup>10</sup> Following the World Bank definition, for the current 2018 fiscal year, low-income economies are defined as those with a gross national income (GNI) per capita, calculated using the World Bank Atlas method, of US\$1,005 or less in 2016; lower middle-income economies are those with a GNI per capita between US\$1,006 and US\$3,955; upper middle-income economies are those with a GNI per capita between US\$3,956 and US\$12,235; high-income economies are those with a GNI per capita of US\$12,236 or more. See income groupings in Appendix A.1.

<sup>&</sup>lt;sup>11</sup> For the case of Argentina, it is argued that the informality incidence fell significantly over a decade since 2003 as a result of high economic growth accompanied by an integrated two-phased, multi-policy approach on various economic, labour and social dimensions (ILO, forthcoming). In the case of Peru, growth and structural transformation in the moderate reduction of informality observed between 2002 and 2012 played a major role (Chacaltana, 2016).

<sup>&</sup>lt;sup>12</sup> In South Africa, in addition to economic growth, several formalization policies were implemented including a National Informal Business Upliftment Strategy (NIBUS), increased enforcement measures, recognition of prior learning and increased access to social security. Such measures contributed to the overall trend together with the effect of other factors such as urbanisation, unemployment, taxes, costs of compliance associated with formality, enforcement power of government regulations, corruption, education level and working hours flexibility (ILO, forthcoming).



Source: ILOSTAT

#### Composition of the informal economy

Informal employment can be in the informal sector, in the formal sector or in the household sector. The 61.2 per cent of global employment that is informal is comprised of 51.9 per cent in the informal sector, 6.7 per cent in the formal sector and 2.5 per cent in households. In all regions employment in the informal sector is the largest of the three components of informal employment. Informal employment in the formal sector is a somewhat large proportion of informal employment in two regions: the Americas and Europe and Central Asia, where it represents 7.9 per cent and 5.3 per cent of total employment respectively. The share of informal employment in the formal sector concerns primarily employees and to some extent contributing family workers. The 39.7 per cent of all employees in informal employment is comprised of 10.8 per cent who are informally employed in the formal sector and 2.1 per cent who are employed as domestic workers in households. The share of employees in informal employment in the formal sector represents 27.4 per cent of total informal employment among employees worldwide and up to 42 per cent in Latin America and the Caribbean and more than half in developed countries from Europe and Central Asia. Among the five regions, the level of informal employment in households in Africa is the highest at 4.3 per cent and the global average rate is 2.5 per cent (figure 5, panel C and table 1.1).

The share of informal employment normally decreases when agriculture is excluded but the distribution of informal employment in different sectors does not necessarily follow the same pattern. In most regions, informal employment in the formal sector increases with the exclusion of agriculture and exceeds the global average (8.0 per cent), for instance from 5.7 per cent to 9.5 per cent in Africa, from 7.5 per cent to 8.7 per cent in the Arab States and from 6.9 per cent to 8.8 per cent Asia and the Pacific (figure 5, panel C and table 1.2).

#### Informal economic units

Globally, the vast majority of economic units<sup>13</sup> are informal (80.9 per cent). Informality is even higher in units in Africa (92.4 per cent) and the Arab States (90.8 per cent). The share of economic units in the informal sector for emerging and developing countries (82.5 per cent) is quite similar to the global level while fewer economic units are informal in developed countries (55.7 per cent) in relative terms (figure 7).

# Figure 7. Percentage of economic units\* in the informal sector, the formal sector and the household sector (percentages, 2016)



\* Economic units are estimated as the sum of own-account workers and employers.

Source: ILO calculations based on household survey micro datasets.

#### Informality and status in employment

Apart from contributing family workers, all of whom are considered as informal by statistical definition, the employment status category with the highest percentage of informality is own-account workers, both globally and regionally. Globally, 86.1 per cent of own-account workers are informal. Only in Europe and Central Asia (60 per cent) is the rate of informal employment among ownaccount workers lower than the global average. Informality among own-account workers is high in both emerging and developing countries (87 per cent) and developed countries (68.8 per cent) (figure 8, panel A and table 1.3).

<sup>&</sup>lt;sup>13</sup> Economic units are estimated as the sum of employers and own-account workers.







\*Entrepreneurs refer to owners of economic units and the estimates are based on the number of employers and own-account workers. Informal entrepreneurs refer to employers and own-account workers in the informal sector.



# Panel B. Composition of informal and formal employment by categories of status in employment (percentages, 2016)

Source: ILO calculations based on household survey micro datasets.

Globally, own-account workers make up 45.0 per cent of informal employment and employees 36.2 per cent. Contributing family workers represent 16.1 per cent and employers account for 2.7 per cent of total informal employment. Africa and Asia and the Pacific share a similar composition of informal employment where own-account workers are the largest group and contributing family workers represent a significant proportion compared to other regions. The situation in emerging and developing countries and developed countries is different. While own-account workers and contributing family workers represent the largest group in developing and in emerging countries, employees represent the largest group (51.3 per cent) in developed countries (figure 8, panel B and table 1.3<sup>14</sup>). The share of employees in total informal employment tends to increase with economic development and by contrast, the share of contributing family workers and of own-account workers as part of informal employment decreases.

#### Age profile of formality

The level of informality is higher among young people and older persons. Worldwide three out of four young (77.1 per cent) and older persons (77.9 per cent) are in informal employment. Informal employment is more likely for young people in emerging and developing countries. The employment of older persons is more likely to be informal than that of young people whatever the socio-economic development of a country and region (figure 9 and table 1.5).<sup>15</sup>



#### Figure 9. Share of formal employment in total employment by age<sup>16</sup> (percentages, 2016)

Source: ILO calculations based on household survey micro datasets.

#### Education and informality

The level of education is another key factor affecting the level of informality. Globally, when the level of education increases, the level of informality decreases. Those who have completed secondary and tertiary education are less likely to be in informal employment compared to workers who have either no education or completed primary education. This phenomenon is observed at the global and regional level and emerging and developing and developed countries share similar patterns (figure 10, table 1.6 and Education and informality in Chapter 3).<sup>17</sup>

<sup>&</sup>lt;sup>14</sup> See country data on informality and status in employment in Appendix B, tables B.3 and B.4.

<sup>&</sup>lt;sup>15</sup> See data on informality and broad age groups in Appendix C, table C.3.

<sup>&</sup>lt;sup>16</sup> Regional and global aggregates are available in Appendix C.3

<sup>&</sup>lt;sup>17</sup> See data on informality and education in Appendix C, tables C.1 and C.2.



Figure 10. Share of informal employment in total employment by level of education<sup>18</sup> (percentages, 2016)

Source: ILO calculations based on household survey micro datasets.

#### Informality and rural/ urban location

At the global level, persons living in rural areas (80.0 per cent) are twice as likely to be in informal employment as those in urban areas (43.7 per cent). The largest differences are in the Americas, Asia and the Pacific, and Europe and Central Asia (table 1.7).<sup>19</sup>

### Sectoral dimension of informality

Agriculture is the industry sector with the highest level of informal employment (93.6 per cent) around the world. The industry (57.2 per cent) and service (47.2 per cent) sectors are relatively less exposed to informality, especially the service sector in the Arab States and Asia and the Pacific. However, when informal employment is the main source of employment, especially in emerging and developing countries, a high level of informality is observed in all sectors (table 1.8).<sup>20</sup>

#### Gender dimension of informality

Globally, informal employment is a greater source of employment for men (63.0 per cent) than for women (58.1 per cent). This is true for both the averages for emerging and developing countries and developed countries and for agricultural and non-agricultural informal employment. Out of the 2 billion workers in informal employment worldwide, just more than 740 million are women. This global picture hides important disparities however (further discussions in box 3 and Chapter 3) and results from the influence of major countries such as China or the Russian Federation. By contrast, in low and lower-middle income countries, a higher proportion of women are in informal employment than men.<sup>21</sup> In Africa, 89.7 percent of employed women are in informal employment

<sup>&</sup>lt;sup>18</sup> Regional and global aggregates are available in Appendices C.1 and C.2.

<sup>&</sup>lt;sup>19</sup> See country data on informality and rural or urban location in Appendix B, table B.2.

<sup>&</sup>lt;sup>20</sup> See country data on informality and sector in Appendix B, tables B.6 and B.7.

<sup>&</sup>lt;sup>21</sup> In low-income countries, 92.1 per cent of employed women are in informal employment compared to 87.5 per cent of men. In lower-middle countries, 84.5 per cent of women are in informal employment compared to 83.4 per cent of men.

in contrast to 82.7 per cent of men. However, even though globally there are fewer women than men in informal employment, women in the informal economy are more often found in the most vulnerable situations, for instance as domestic workers, home-based workers or contributing family workers, than their male counterparts. Moreover, the lower the participation rate of women in the labour market, the lower the share of informal employment in total women's employment. This is particularly observed in North Africa and the Arab States (table 1.4).<sup>22</sup>

### Box 3. Women and men in the informal economy

Globally, the share of women in informal employment is lower than the share of men in informal employment but there are actually more countries (55.5 per cent) where the share of women in informal employment exceeds the share of men in informal employment (figure 11). Women are indeed more exposed to informal employment in more than 90 per cent of sub-Saharan African countries, 89 per cent of countries from Southern Asia and almost 75 per cent of Latin American countries.

A major difference between women and men in informal employment is the proportion of women contributing family workers (figure 12). This proportion is more than three times higher among women in informal employment compared to men. This status, particularly vulnerable, represents 28.1 per cent of women in informal employment compared to 8.7 per cent for men. More than 30 per cent of women in informal employment in low- and lower-middle income countries are contributing family workers, usually considered as unpaid.

# Figure 11. Gender gap in the share of informal employment in total employment including agriculture (percentage points, latest available year)



The main consequences are that, with the exception of high-income countries, a lower share of informally employed women are employees and also a lower share are own-account workers and employers, whatever the level of development. This means inevitably that informally employed women earn a lower income from labour but also have a risk of exposure to higher decent work deficits and points to a need to facilitate transition to other categories of employment status and improve working conditions as a first step towards formalization.

<sup>22</sup> See country data on the size and composition of the informal economy by sex in Appendix B, table B.1.



Source: ILO calculations based on household survey micro datasets.

Own-account workers represent the largest group of workers in informal employment globally and in lowand lower-middle income countries. Worldwide, more than half of men in informal employment are ownaccount workers and 36.3 per cent of women are. The situation differs in higher income countries where employees account for the largest part of informal employment, particularly women.

In high-income countries, the vast majority of women holding informal jobs are employees (57.4 per cent), while the most common status among men in informal employment (more than half when considering own-account workers and employers) is that of entrepreneur. More than one-third (35.7 per cent) of women employees in informal employment work on a part-time basis and more than a quarter are in temporary employment (see Chapter 3).

			Wo	World				Emergin	g and dev	Emerging and developing countries	untries			Developec	Developed countries	(0)
	sinta	Атегісая	sətst2 darA	sitice Pacific	Europe and Central Asia	Total	sintA	Americas	Arab States	office 9 and bus size	Europe and Central Asia	Total	Americas	chick Pacific	Europe and Central Asia	
Share of informal employment and its components in total employment	t and its co	mponents i	n total emp	loyment		-	-	-		-			-	-	-	-
Total employment <sup>23</sup>	85.8	40.0	68.6	68.2	25.1	61.2	85.8	53.8	68.6	71.4	36.8	69.6	19.4	21.7	15.6	
In the informal sector	76.0	29.3	60.9	58.8	19.4	51.9	76.1	37.7	60.9	61.4	28.5	59.2	16.9	16.5	11.9	
In the formal sector	5.5	7.9	7.5	6.9	5.3	6.7	5.7	11.9	7.5	7.0	7.2	7.3	1.9	4.9	3.7	
In households	4.3	2.7	0.2	2.5	0.5	2.5	4.0	4.2	0.2	2.7	1.1	2.9	0.6	0.2	0.0	
Employees	56.8	25.9	54.3	49.8	15.4	39.7	56.8	38.3	54.3	54.8	25.7	49.7	12.9	10.4	נינ	
In the informal sector	33.7	13.4	42.7	36.0	9.7	26.7	33.7	16.1	42.7	39.8	17.5	33.6	10.6	6.1	3.7	
In the formal sector	18.9	9.4	11.4	11.8	5.5	10.8	18.9	16.3	11.4	12.7	7.4	13.4	2.1	4.1	4.0	
In households	4.2	3.1	0.2	2.0	0.3	2.1	4.2	5.8	0.2	2.2	0.7	2.8	0.3	0.2	0.0	
Share of non-agricultural informal employment and its components in non	rmal emplo	yment and	its compon	ents in non-	agricultura	agricultural employment										
Total employment	71.9	36.1	63.9	59.2	20.9	50.5	71.9	49.6	63.9	62.8	30.0	59.5	18.9	19.5	14.4	
In the informal sector	59.2	26.0	55.1	48.9	15.2	40.8	59.2	33.6	55.1	52.0	21.6	48.2	16.4	14.3	10.6	
In the formal sector	9.4	7.2	8.7	8.8	5.4	8.0	9.5	11.4	8.7	9.1	7.6	9.3	1.9	5.0	3.8	
la householde	, ,	•														

Table 1. Key indicators on the size and composition of the informal economy: a global picture (percentages, 2016)

<sup>&</sup>lt;sup>23</sup> Due to rounding, some totals may not correspond with the sum of the separate figures.

			World	ld				Emergin	g and deve	Emerging and developing countries	intries		0	Developed countries	countries	
	Africa	Americas	Arab States	Asia and the Pacific	Europe and Central Asia	Total	Africa	Атегісаз	sətət2 darA	offices and the sick	Europe and Central Asia	Total	Americas	Asia and the Pacific	Europe and Central Asia	lstoT
Employees	50.1	24.0	51.7	48.2	14.7	37.7	50.1	35.7	51.7	52.2	24.8	47.2	12.8	10.9	1.1	10.2
In the informal sector	27.6	12.2	39.6	35.2	9.0	25.7	27.6	14.1	36.9	38.3	16.6	32.3	10.4	6.4	3.7	6.7
In the formal sector	18.6	8.6	11.9	11.5	5.4	10.3	18.6	15.4	11.9	12.3	7.4	12.7	2.1	4.2	4.0	3.3
In households	4.0	3.2	0.2	1.4	0.3	2.7	4.0	6.2	0.2	1.6	0.8	2.2	0.3	0.2	0.0	0.2
3 Informality and status in employment	oyment															
Percentage of informal workers by status in employment	by status ir	1 employme	nt													
Total	85.8	40.0	68.6	68.2	25.1	61.2	85.8	53.8	68.6	71.4	36.8	69.6	19.4	21.7	15.6	18.3
Employees	56.8	25.9	54.3	49.8	15.4	39.7	56.8	38.2	54.3	54.8	25.7	49.7	12.9	10.4	7.7	10.2
Employers	77.9	31.3	75.1	53.6	39.9	50.7	77.9	42.7	75.1	56.1	29.6	56.2	10.3	25.1	44.4	28.8
<b>Own-account workers</b>	94.4	82.2	94.3	86.2	60.0	86.1	94.4	83.8	94.3	86.6	52.0	87.0	72.8	63.6	68.9	68.8
Contributing family workers	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Composition of informal employment	ment															
Total	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0	100.0
Employees	29.7	49.2	54.2	34.4	56.3	36.2	29.7	45.1	54.2	34.1	64.7	35.4	66.2	46.0	40.2	51.3
Employers	2.7	3.5	4.8	2.3	5.3	2.7	2.7	3.9	4.8	2.2	2.0	2.5	2.1	5.4	11.8	6.5
Own-account workers	50.1	40.7	39.3	45.5	29.7	45.0	50.1	43.2	39.3	45.7	22.6	45.5	30.5	33.6	43.3	35.9
Contributing family workers	17.5	6.5	1.6	17.8	8.7	16.1	17.5	7.8	1.6	17.9	10.7	16.6	1.2	15.0	4.8	6.3

			Mo	World				Emergi	ig and dev	Emerging and developing countries	untries			Developed	Developed countries	
	Africa	263inemA	sətət2 dərA	office Pacific	Europe and Central Asia	Total	sjrita	8mericas	sətət2 dərA	office Pacific	Europe and Central Asia	Total	Americas	chick Pacific	Europe and Central Asia	Total
4 Share of informal employment in total employment by sex	t in total en	nployment t	by sex	-			-	-	-	-		-	-	-	-	ſ
Male (including agriculture)	82.7	40.5	70.2	70.5	26.4	63.0	82.7	52.9	70.2	73.6	37.8	70.7	20.2	20.7	17.0	18.9
Female (including agriculture)	89.7	39.2	61.8	64.1	23.6	58.1	89.7	55.0	61.8	67.4	35.7	67.5	18.5	23.1	14.0	17.6
المتناطينية مستابينا مراملا	7 73	36 A	56 F	62.0	27 E	52 1	7 73	V LV	<u>е</u> б г	GE A	301	613	10.1	10 F	15.7	7 71
		t. 00	0.00	0.20	0.22	1.00	1.10	4. / 4. r	0.00		1.20	0.10	1.0.1	0.01	1.01	1.11
Female (excluding agriculture)	/8.b	36.9	49.6	53.9	18.8	46.4	/8./	52.5	49.6	/./c	c./2	56.3	18.4	20.9	12.8	16.b
5 Share of informal employment in total employment by age	t in total en	nployment t	oy age													
Youth (15-24)	94.9	46.2	85.1	86.3	35.7	77.1	94.9	63.4	85.1	87.8	49.2	84.4	13.5	22.0	25.8	19.2
Adults (25+)	82.8	40.4	61.1	67.1	21.8	58.7	82.8	53.6	61.1	69.7	34.9	67.4	19.9	22.5	14.8	18.0
25-29	85.5	37.0	68.8	70.8	26.4	62.6	85.5	47.5	68.8	72.8	40.0	69.8	17.4	15.4	16.3	16.6
30-34	83.1	37.6	61.0	67.3	21.8	58.6	83.1	48.0	61.0	69.7	34.1	66.6	18.5	13.4	13.9	15.3
35-54	79.7	39.9	56.9	63.8	20.0	55.7	79.7	52.5	56.9	66.2	33.2	64.3	19.4	16.4	13.6	15.9
55-64	84.2	42.1	62.8	72.4	21.5	59.2	84.1	61.3	62.8	76.6	34.3	72.8	21.8	30.4	15.9	21.2
65+	96.0	54.4	71.7	86.3	40.8	77.9	96.0	79.0	71.7	91.6	46.0	88.5	25.3	55.4	37.0	38.5
		1	1													
5 Share of informal employment in total employment by highest level of ed	t in total en	nployment t	oy nignest l	_	ICATION											
No education	94.0	81.0	88.6	94.9	77.1	93.8	94.0	82.7	88.6	94.9	83.5	93.9	47.9	92.7	28.7	52.7
Primary education	88.5	71.3	77.0	89.7	40.9	84.6	88.5	73.4	77.0	89.8	46.5	86.0	44.8	73.9	29.2	40.5
Secondary education	68.1	46.9	51.9	58.9	23.3	51.7	68.1	51.3	51.9	62.0	36.1	59.1	21.7	23.7	16.6	19.2
Tertiary education	27.0	22.7	22.4	30.7	15.2	23.8	27.0	33.7	22.4	34.2	22.9	32.0	18.6	14.1	11.7	16.1
			Mo	World				Emergi	ng and de	<b>Emerging and developing countries</b>	ountries			Jeveloped	<b>Developed</b> countries	6
--	------------------	-----------	--------------	----------------	-------------------------	-------	--------	----------	-------------	--	--------------------------	-------	----------	----------------	----------------------------	-------
	Africa	Americas	sətət2 dərA	office Pacific	Europe and Central Asia	Total	Africa	Атегісая	Arab States	Asia and the Pacific	eizA lentral bue equital	Total	Americas	cifice Pacific	Europe and Central Asia	Total
Share of informal employment in total employment by area of residence	nent in total er	nployment	by area of r	esidence		L.	-		-	-	-	-	-		-	-
Rural	88.3	52.6	69.3	85.2	33.2	80.0	88.3	69.1	69.3	86.0	47.3	84.2	24.8	22.2	20.4	22.1
Urban	76.3	35.8	63.9	47.4	19.4	43.7	76.3	47.5	63.9	51.5	29.0	52.6	18.9	17.3	14.3	16.7
				24 												
8 Snare of informal employment in total employment by industrial sector-	nent in total ei	npioyment	oy industria													
Agriculture	97.9	77.5	95.6	94.7	71.6	93.6	97.9	79.9	92.6	94.9	76.5	94.3	50.2	75.5	52.5	58.7
Industry	77.4	38.4	80.9	68.8	21.9	57.2	77.4	49.8	80.9	73.4	33.5	67.2	18.2	20.3	11.5	15.8
Services	70.2	35.5	57.3	54.1	20.2	47.2	70.1	49.6	57.3	57.2	27.8	55.5	19.1	19.1	15.3	17.5

Note: Indicators 1, 2, 3, 4, 7 and 8: based on the analysis of 119 countries' micro datasets (national labour force surveys or similar national household surveys) representing 90 per cent of global employment. Values for missing countries have been estimated based on the average value for the subregion and country income group. Indicators 5 and 6 are based on available micro datasets from 110 countries that represent more than 85 per cent of the world's employed population. There is no estimate of missing values, resulting in small differences in global and regional estimates. All indicators refer to the main job of workers aged 15 years and over.

Source: ILO calculations based on household survey micro datasets.

<sup>&</sup>lt;sup>24</sup> The category "not classified" for missing data on an industrial sector is not displayed in this table.

### 2.2 Regional overview

### 2.2.1 Africa

Figure 13. Key indicators on the size and composition of the informal economy in Africa (percentages, 2016)



#### Composition of informal and formal employment **G2** by categories of status in employment



#### Share of informal employment in total employment G3 by categories of status in employment





100.0

80 100





Contributing family workers 100.0 97.5 Own-account workers 83.4 Employers Employees 57.4 20 40 60 80 100 Share of informal employment (%)



0 20 40 60 80 100 Share of informal employment (%)

Employers

Employees

60.2

62.2



#### Table 2. Key indicators on the size and composition of the informal economy in Africa (percentages, 2016)

.....

Afri	Ca	Africa	Northern Africa	Sub-Saharan Africa	Central Africa <sup>25</sup>	Eastern Africa	Southern Africa	Western Africa
1	Share of informal employment and its components in total employment <sup>26</sup>	85.8	67.3	89.2	91.0	91.6	40.2	92.4
	In the informal sector In the formal sector In households	76.0 5.5 4.3	58.1 8.4 0.7	79.2 5.0 5.0	83.5 6.9 0.6	82.9 4.8 3.8	27.6 4.9 7.7	79.6 4.4 8.3
2	Share of non-agricultural informal employment and its components in non-agricultural employment	71.9	56.3	76.8	78.8	76.6	36.1	87.0
	In the informal sector In the formal sector In households	59.2 9.4 3.4	46.2 9.2 1.0	63.3 9.5 4.2	63.4 14.5 0.9	59.6 11.4 5.8	23.5 5.0 7.6	76.9 7.0 3.2
	Informal employment rate by:							
3	Sex Male excluding agriculture Female	82.7 67.7 89.7	68.5 59.7 62.2	86.4 71.6 92.1	87.1 73.2 95.2	89.1 72.3 94.0	38.4 34.4 42.4	89.8 81.0 95.0
	excluding agriculture	89.7 78.6	41.5	92.1 82.8	95.2 88.2	94.0 82.1	42.4 38.2	95.0 92.3

<sup>&</sup>lt;sup>25</sup> Data from available datasets for this subregion are not recent for a number of countries in this subregion (Angola (2009), Cameroon (2007), Chad (2003), Congo (2009) and DRC (2005)) and care should be taken when interpreting estimates.

<sup>&</sup>lt;sup>26</sup> Due to rounding, some totals may not correspond with the sum of the separate figures.

Africa	Africa	Northern Africa	Sub-Saharan Africa	Central Africa	Eastern Africa	Southern Africa	Western Africa
4 Age							
Youth (15—24)	94.9	87.5	95.8	95.1	96.7	56.4	97.9
Adults (25+)	82.8	63.7	86.6	90.5	89.1	40.5	92.4
25–29	85.5	71.6	88.2	90.3	89.4	47.2	94.3
30–34	83.1	68.2	86.1	90.1	86.7	40.8	93.8
35–54	79.7	58.1	84.4	88.9	88.0	38.3	90.6
55–64	84.2	64.1	88.6	94.7	92.2	37.1	92.8
65+	96.0	94.0	96.2	98.0	96.8	64.2	96.7
5 Highest level of education							
No education	94.0	86.7	95.4	97.9	95.7	63.0	96.2
Primary education	88.5	78.3	89.6	96.0	94.7	48.6	92.0
Secondary education	68.1	57.4	71.9	83.5	82.0	27.2	73.7
Tertiary education	27.0	27.4	26.7	40.0	34.1	12.2	49.4
6 Residence area							
Rural	88.3	75.6	90.1	84.8	90.6	57.1	95.2
Urban	76.3	58.1	80.8	89.0	80.0	32.7	87.3
7 Industrial sectors <sup>27</sup>							
Agriculture	97.9	95.5	98.1	98.2	98.4	73.7	98.3
Industry	77.4	69.7	80.6	84.8	83.2	34.1	91.5
Services	70.2	49.4	75.8	75.9	76.2	37.8	85.4

Note: Country data are available in Appendix B: table B.1 for indicators 1. 2 and 3; table B.2 for indicator 6 and tables B.6 and B.7 for indicator 7. Global and regional estimates desegregated by sex are available in table C.3 for indicator 4; tables C.1 and C.2 for indicator 5.

Source: ILO calculations based on household survey micro datasets.

#### Magnitude

Informal employment is the main source of employment in Africa, accounting for 85.8 per cent of all employment, or 71.9 per cent, excluding agriculture. Africa is broadly divided into sub-Saharan Africa and Northern Africa, with each subregion having very different socio-economic development and different levels of informal employment: 67.3 per cent in Northern Africa and 89.2 per cent in sub-Saharan Africa. Excluding agriculture, informal employment accounts for 56.3 per cent and 76.8 per cent of total employment respectively (figure 13.G1 and table 2).

Within sub-Saharan Africa, informal employment is the main source of employment in Central Africa (91.0 per cent), Eastern Africa (91.6 per cent) and Western Africa (92.4 per cent). If agriculture is excluded, informal employment continues to dominate employment with a 78.8 per cent share in Central Africa, 76.6 per cent in Eastern Africa and 87.0 per cent in Western Africa. Southern Africa is the only subregion with less than half of the employed population in informal employment at 40.2 per cent and 36.1 per cent excluding agriculture. In this subregion, employees represent 84.3 per cent of total employment compared to 40.4 per cent for Africa and 37.2 per cent for sub-Saharan Africa (figure 13.G1 and table 2).

<sup>&</sup>lt;sup>27</sup> The category "not classified" in case of missing data on industrial sector is not displayed in this table.

There is a large variation in the share of informal employment among countries in sub-Saharan Africa. The share of informal employment reaches its highest rate in Burkina Faso (94.6 per cent) and Benin (94.5 per cent). South Africa (34.0 per cent) and Cabo Verde (46.5 per cent) are among the countries with the lowest share of informal employment (Appendix B. table B.1).

Based on the number of "entrepreneurs" (own-account workers and employers) owners of informal economic units, it is estimated that the vast majority – 92.4 per cent - of all economic units in Africa are informal (figure 13.G5).

#### Composition

In Africa, 76.0 per cent of employment is in the informal sector and a relatively small proportion is in the formal sector (5.5 per cent) and in households (4.3 per cent). Excluding agriculture, the average share of informal employment in the informal sector drops from 76.0 per cent to 59.2 per cent and also in all subregions, whereas the share of informal employment in the formal sector increases notably from 5.5 per cent of total employment to 9.4 per cent of non-agricultural employment. Non-agricultural informal employment more than doubles in Central Africa, from 6.9 per cent to 14.5 per cent, and in Eastern Africa, from 4.8 per cent to 11.4 per cent. Compared to the regional average (4.3 per cent) and other subregions, the share of informal employment in households is relatively high in Southern Africa (7.7 per cent) and Western Africa (8.3 per cent) (table 2).

Own-account workers are the largest group within informal employment in the region, apart from Southern Africa where more employees are informal than own-account workers, and Northern Africa, where employees and own-account workers account for more or less the same proportion (figure 13.G2 and table 2).

In Africa, 94.4 per cent of own-account workers are in the informal sector and this pattern is similar in all of the subregions. More employers (77.9 per cent) tend to be informal than employees (56.8 per cent). This tendency applies in all subregions and the difference is particularly large in Northern Africa, where 95.1 per cent of employers are informal compared to 46.7 per cent of employees (figure 13.G3 and table 2).

#### **Characteristics**

At the regional level, informal employment is a greater source of employment for women than for men: 89.7 per cent for women in comparison to 82.7 per cent for men, and for non-agricultural employment, 78.6 per cent for women and 67.7 per cent for men. In sub-Saharan Africa, except Southern Africa, more than 90 per cent of women are in informal employment compared to 86.4 per cent of men. Northern Africa shows an inverse situation with slightly more men (68.5 per cent) than women (62.2 per cent) in informal employment (table 2).

The young and old have especially high rates of informal employment: 94.9 per cent of persons between ages 15 and 24 in employment and 96.0 per cent of persons aged 65 and older. The figures reach as high as 97.9 per cent of young people in Western Africa and 98.0 per cent of older persons in Central Africa (table 2).

The level of education is closely linked to informality in Africa. Those with no education tend be highly informal (94.0 per cent). The rate of informality reduces to 88.5 per cent with primary education and further decreases to 68.1 per cent for those with secondary education and to 27.0 per cent among those with tertiary education. The reduction of informality with increasing education is less obvious in Western Africa where almost half of the employed population with tertiary education (49.4 per cent) are informal (see figure 13.G4 and table 2).

Informal employment dominates the labour market in both rural (88.3 per cent) and urban (76.3 per cent) areas, although informality is higher in rural areas (table 2).

Almost all of the agricultural sector in Africa is informal (97.9 per cent). The rate of informality is lower in the industry (77.4 per cent) and the service (70.2 per cent) sectors, but still is very high.

Informal employment is more common in industry than in the service sector in all subregions, apart from Southern Africa where 34.1 per cent of the industry sector and 37.8 per cent of the service sector are informal (table 2).

#### 2.2.2 Americas

Figure 14. Key indicators on the size and composition of the informal economy in the Americas (percentages, 2016)





.....



Percentage of economic units\* in the informal sector, the formal sector and the household sector



\* Economic units are estimated as the sum of own-account wokers and employers.

#### Table 3. Key indicators on the size and composition of the informal economy in the Americas (percentages, 2016)

Am	ericas	Americas	Northern America	Latin America and the Caribbean	The Caribbean	Central America	South America
1	Share of informal employment and its components in total employment <sup>28</sup>	40.0	18.1	53.1	57.6	58.0	50.8
	In the informal sector In the formal sector In households	29.3 7.9 2.7	16.0 1.7 0.5	37.4 11.6 4.1	42.4 10.2 4.9	36.5 16.5 5.0	37.3 9.8 3.7
2	Share of non-agricultural informal employment and its components in non-agricultural employment	36.1	17.7	49.0	51.6	55.4	46.4
	In the informal sector In the formal sector In households	26.0 7.2 2.9	15.6 1.6 0.5	33.3 11.1 4.6	35.5 10.8 5.3	36.7 13.4 5.4	31.9 10.3 4.2
	Share of informal employment in total employment by:						
3	Sex						
	Male excluding agriculture Female	40.5 35.4 39.2	18.9 18.3 17.3	52.3 46.8 54.3	56.7 48.6 58.7	55.6 51.4 61.8	50.5 44.9 51.3
	excluding agriculture	36.9	17.1	51.8	55.5	60.6	48.3

28 Due to rounding, some totals may not correspond with the sum of the separate figures.

Americas	Americas	Northern America	Latin America and the Caribbean	The Caribbean	Central America	South America
4 Age						
Youth (15—24)	46.2	12.8	62.4	62.6	70.5	58.4
Adults (25+)	40.4	19.5	52.5	48.3	55.5	51.5
25–29	37.0	17.5	46.4	49.0	52.3	44.0
30–34	37.6	18.5	47.0	45.2	51.9	45.2
35–54	39.9	19.1	51.3	46.3	53.3	50.7
55–64	42.1	21.2	59.9	53.5	59.3	60.3
65+	54.4	23.7	78.0	64.6	81.2	76.8
5 Highest level of education						
No education	81.0	13.6	82.2	79.4	78.5	88.3
Primary education	71.3	32.8	72.5	69.6	74.5	71.6
Secondary education	46.9	18.2	50.8	55.9	58.0	48.9
Tertiary education	22.7	18.5	33.5	26.3	33.5	33.7
6 Residence area						
Rural	52.6	23.4	68.5	68.9	67.4	69.0
Urban	35.8	17.6	47.0	55.8	50.0	45.3
7 Industrial sectors <sup>29</sup>						
Agriculture	77.5	50.7	79.2	86.9	70.5	82.3
Industry	38.4	16.7	49.1	52.1	53.7	47.0
Services	35.5	17.9	49.0	51.4	56.2	46.2

Note: Country data are available in Appendix B: table B.1 for indicators 1, 2 and 3; table B.2 for indicator 6 and tables B.6 and B.7 for indicator 7. Global and regional estimates desegregated by sex are available in table C.3 for indicator 4; tables C.1 and C.2 for indicator 5.

Source: ILO calculations based on household survey micro datasets.

#### Magnitude

In the Americas, 40.0 per cent of total employment is informal. In absolute numbers, 183 million people are in informal employment whether in formal or informal enterprises. These rates vary widely by subregion, at 18.1 per cent in Northern America and 53.1 per cent in Latin America and the Caribbean (LAC). Within LAC countries, the highest shares of informal employment are in Central America (58.0 per cent) and the Caribbean (57.6 per cent). At the country level, the share of informal employment ranges from 24.5 per cent in Uruguay to close to 80 per cent in Honduras, Guatemala and Nicaragua, and above 80 per cent in the Plurinational State of Bolivia (figure 14.G1. table 3 and Appendix B table B.1).

If agriculture is excluded from these calculations, non-agricultural informal employment rates are 36.1 per cent for the Americas, 17.7 per cent for Northern America and 49.0 per cent for LAC (figure 14.G1 and table 3).

Employment in the informal sector for own-account workers and employers can be a proxy for the number of informal economic units. An estimated 71.2 per cent of all economic units in the Americas can thus be considered as informal, including 76.6 per cent for LAC (figure 14.G5).

<sup>&</sup>lt;sup>29</sup> The category "not classified" in the case of missing data on the industrial sector is not displayed in this table.

#### Composition

Informal employment accounts for 40.0 per cent of total employment, of which 29.3 per cent is in the informal sector, 7.9 per cent in the formal sector and 2.7 per cent in households. Economic units in the informal sector provide the largest share of informal employment, but a significant proportion of employees in informal employment are also employed in the formal sector (42.6 per cent of informally employed employees are in the formal sector and 15.2 per cent in households as domestic workers). In LAC, especially Central America, the component related to informal employment in the formal sector is relatively high, reflecting the significant numbers of the wage employed who are either not protected or insufficiently protected within formal sector enterprises. This pattern is accentuated with non-agricultural informal employment (table3).

The share of informal employment is high among own-account workers (82.2 per cent) and for contributing family workers who by definition are informal. Informal employment rates are lower for employers (31.3 per cent) and employees (25.9 per cent). The share of informal employment among employees is higher in Latin America (37.2 per cent) than in Northern America (12.7 per cent). Within the LAC region, the highest proportion of informal employment among employees is in Central America (48.3 per cent) (figure 14.G3).

In LAC, employees represent the largest share of informal employment (44.8 per cent of total informal employment), influenced by the situation in Central America, where employees account for 55.1 per cent of total informal employment. In other subregions, the proportion of own-account workers exceeds that of employees. In Northern America, employees represent 70.7 per cent of total informal employment. Throughout the region, the proportion of employees is always lower among workers in informal employment compared to workers in formal employment.

#### **Characteristics**

There are some groups within the labour market where informality is especially significant. The share of informal employment is higher among men than among women, mainly in Northern America where the rate is 18.9 per cent for men and 17.3 for women. In LAC, the share of informal employment in total employment is higher for women (54.3 per cent) than for men (52.3 per cent), especially in Central America where informal employment rates for women are 61.8 per cent compared to that of men, at 55.6 per cent (table 3).

The share of informal employment is also relatively high for the youth population: 46.2 per cent in comparison to the adult's rate of 40.4 per cent. While the share of informal employment decreases to 37.6 per cent when young people reach the 30-34 age group, informality increases from the age of 35 and above to reach 54.4 per cent among workers aged 65 years old and above (table 3).

There is a clear inverse relation with the level of education. Informality decreases as education increases, from 81.0 per cent among those with no education, to 71.3 per cent for those with primary education, to 46.9 per cent for those with secondary education and 22.7 per cent with tertiary education. The inverse relation is clearly observed in LAC, with the even higher level of 33.5 per cent for those with tertiary education (figure 14.G4 and table 3).

The place of residence also influences the share of informal employment in total employment. It is higher in rural areas at 52.6 per cent, as compared to 35.8 per cent in urban areas. This difference is particularly significant in LAC, where rural informality reaches 68.5 per cent as compared to 47.0 per cent for urban areas (table 3).

Among the economic sectors, informality is highest in agriculture, where 77.5 per cent of employment is informal. In industry 38.4 per cent is informal and in services 35.5 per cent. Informality is much lower in each of the sectors in Northern America which lowers the subregional averages. For example, in the Caribbean 86.9 per cent of agriculture is informal (table 3).

#### 2.2.3 Asia and the Pacific





#### Composition of informal and formal employment **G**2 by categories of status in employment





# G3



100.0

74.4

Own-account workers

Employers Employees

> 20 40 60 80 100

43 7

Share of informal employment (%)

53.6



100.0

74.4

57.2

40 60 80 100

0 20 40 60 80 100 Share of informal employment (%)

83.4

64 5

Employe

Employees



Share of informal employment (%)

.....

Female

... excluding agriculture

100

90

80

70

60

50

40 30

20 10

0

No education

---- Asia and the Pacific

Pacific Islands

----- South-Eastern Asia and the Pacific



Primary education Second ary education

🔶 Eastern Asia

🔶 Southern Asia





 $^{\ast}$  Economic units are estimated as the sum of own-account wokers and employers.

# Table 4. Key indicators on the size and composition of the informal economy in Asia and the Pacific (percentages, 2016)

Tertiary education

Asia	a and the Pacific	Asia and the Pacific	Without China	Eastern Asia	Without China	South-Eastern Asia and the Pacific	Southern Asia
1	Share of informal employment and its components in total employment <sup>30</sup>	68.2	77.6	50.7	26.6	75.2	87.8
	In the informal sector	58.8	65.5	44.7	21.5	57.4	77.4
	In the formal sector	6.9	7.4	5.9	4.3	9.7	6.9
	In households	2.5	4.2	0.1	0.8	6.9	3.5
2	Share of non-agricultural informal employment and its components in non-agricultural employment	59.2	64.8	49.1	21.7	63.9	77.6
	In the informal sector	48.9	50.5	43.0	16.4	46.6	62.5
	In the formal sector	8.8	11.2	6.0	4.5	12.2	12.7
	In households	1.4	2.8	0.1	0.9	4.7	2.4
	Share of informal employment in total employment by:						
3	Sex						
	Male	70.5	78.4	52.2	25.4	75.2	86.8
	excluding agriculture	62.0	67.3	50.9	20.5	64.6	77.6

64.1

53.9

48.4

58.6

48.4

46.4

28.3

23.2

75.4

62.9

90.7

76.9

 $^{\scriptscriptstyle 30}$   $\,$  Due to rounding, some totals may not correspond with the sum of the separate figures.

Asia and the Pacific	Asia and the Pacific	Without China	Eastern Asia	Without China	South-Eastern Asia and the Pacific	Southern Asia
4 Age						
Youth (15–24)	86.3		70.3		87.0	95.5
Adults (25+)	67.1		49.5		76.0	89.6
25–29	70.8		53.8		76.3	90.6
30–34	67.3		46.2		74.2	89.8
35–54	63.8		47.0		74.9	88.3
55–64	72.4		56.8		79.1	90.6
65+	86.3		72.4		83.9	98.3
5 Highest level of education						
No education	94.9		89.2		95.0	95.2
Primary education	89.7		84.8		88.7	92.7
Secondary education	58.9		52.1		70.3	84.2
Tertiary education	30.7		12.8		43.5	72.0
6 Residence area						
Rural	85.2		80.4		75.5	92.8
Urban	47.4		32.9		55.3	75.1
7 Industrial sectors <sup>31</sup>						
Agriculture	94.7	96.9	82.3	88.6	91.7	99.3
Industry	68.8	71.1	49.0	25.3	67.8	81.9
Services	54.1	60.9	46.1	20.2	60.6	75.7

**Note:** Country data are available in Appendix B: table B.1 for indicators 1, 2 and 3; table B.2 for indicator 6 and tables B.6 and B.7 for indicator 7. Global and regional estimates desegregated by sex are available in table C.3 for indicator 4; tables C.1 and C.2 for indicator 5.

Source: ILO calculations based on household survey micro datasets.

#### Magnitude

In the Asia and Pacific region, more than half of the employed population engage in non-agricultural informal employment which stands at 59.2 per cent (64.8 per cent when excluding China). If agricultural employment is included, the share of informal employment reaches 68.2 per cent (77.6 per cent excluding China). Within the region, Southern Asia, South-Eastern Asia and the Pacific have higher shares of informal employment, both excluding and including agriculture, than the average. The share of informal employment in these two subregions represents 87.8 per cent and 75.2 per cent of total employment respectively. Excluding agriculture, the shares of informal employment decrease slightly to 77.6 per cent and 63.9 per cent respectively (figure 15.G1 and table 4).

The economic development of countries in the Asia and Pacific region varies considerably, and this is reflected in the proportions of informally employed. The share of informal employment ranges from the highest level of over 90 per cent (94.3 per cent in Nepal, 93.6 per cent in Lao People's Democratic Republic and 93.1 per cent in Cambodia) to the lowest with proportions below 20 per cent in Japan. The share of informal employment is on average 71.4 per cent in developing and emerging Asian countries and 21.7 per cent in developed Asian countries (Appendix B table B.1 and table 1).

<sup>&</sup>lt;sup>31</sup> The category "not classified" in the case of missing data on an industrial sector is not displayed in this table.

Focusing on economic units rather than employment, it is estimated that 81.4 per cent of all economic units in Asia and the Pacific are informal. This proportion ranges from 64.8 per cent in South-Eastern Asia and the Pacific to 91.9 per cent in Southern Asia (figure 15.G5).

#### Composition

Informal employment in the informal sector is the major component of informal employment in the region at 58.8 per cent of employment, 6.9 per cent is informal employment in the formal sector and 2.5 per cent in the household sector. In Southern Asia, 77.4 per cent of employment is in the informal sector, 6.9 per cent is informal employment in the formal sector and 3.5 per cent in the household sector. In South-Eastern Asia, the share of informal employment in the formal sector and in households is relatively high at 9.7 per cent and 6.9 per cent respectively compared to other subregions. Excluding agriculture, the share of informal employment in the formal sector is almost 13 per cent in Southern Asia (table 4).

In formal employment, employees form the overwhelming majority of the status groups across all subregions while in informal employment, own-account workers and contributing family workers are also significant components across all of the subregions (figure 15.G2).

Apart from contributing family workers, the large majority of own-account workers are informal (86.2 per cent) and in Southern Asia informality among own account workers is even higher (94.0 per cent) (figure 15.G3).

#### **Characteristics**

At the regional level, men are more likely to be in informal employment than women (70.5 per cent of all men in employment are in informal employment compared to 64.1 per cent for women) and a similar pattern is found if agricultural employment is excluded (62.0 per cent for men and 53.9 per cent for women). The two exceptions are South-Eastern Asia and Southern Asia (table 4 and box 3).

Informality is prevalent among the young population with 86.3 per cent in informal employment in the region compared with 67.1 per cent of the adult population. With a large employed population in informal employment in South Asia, almost 96 per cent of young people are informally employed. The share of informal employment decreases to 70.8 per cent when young people enter into adulthood and starts increasing again from the 35–54 age group (table 4).

The level of education is observed to affect the share of informal employment significantly. The proportion of persons employed informally decreases with a higher level of education. While there is only a slight difference in informal employment rates between those with no education (94.9 per cent) and primary education (89.7 per cent), informal employment rates drop significantly from 89.7 per cent for those with primary education to 58.9 per cent for those with secondary education and to only 30.7 per cent for those with tertiary education. These differences by educational level are greatest in Eastern Asia, where those with no education (89.2 per cent) and primary education (84.8 per cent) are overwhelmingly informal and informal employment rates reduce to half (52.1 per cent) for those with secondary education and even to only 12.8 per cent for those with tertiary education (figure 15.G4 and table 4).

Informal employment is predominant in rural areas (85.2 per cent of employment) and is almost half of the employment (47.4 per cent) in urban areas. The largest urban–rural difference in informality is in Eastern Asia, where 80.4 per cent of the rural population is in informal employment as compared with 32.9 per cent in urban areas (table 4).

Almost all of agricultural employment (94.7 per cent) is informal in the region, and it reaches a high of 99.3 per cent in Southern Asia. Informal employment represents a higher share in the industry sector (68.8 per cent) than in the service sector (54.1 per cent) (table 4).

#### 2.2.4 Europe and Central Asia





### G2 Composition of informal and formal employment by categories of status in employment









100.0

100.0



Primary education Secondary education

----- Central and Western Asia

----- Northern, Southern and Western Europe

Tertiary education

10

0

No education

---- Eastern Europe

----- Europe and Central Asia



Percentage of economic units\* in the informal sector, the formal sector and the household sector



\* Economic units are estimated as the sum of own-account wokers and employers.

# Table 5. Key indicators on the size and composition of the informal economy in Europe and Central Asia (percentages, 2016)

Eur	ope and Central Asia	Europe and Central Asia	Northern, Southern and Western Europe	Eastern Europe	Central and Western Asia
1	Share of informal employment and its components in total employment <sup>32</sup>	25. 1	14.3	31.5	43.4
	In the informal sector	19.4	11.3	21.9	37.7
	In the formal sector	5.3	2.8	9.5	3.8
	In households	0.5	0.3	0.1	2.0
2	Share of non-agricultural informal employment and its components in non-agricultural employment	20.9	13.2	28.3	31.2
	In the informal sector	15.2	10.4	18.3	25.4
	In the formal sector	5.4	2.7	9.9	3.9
	In households	0.3	0.1	0.1	1.7
	Share of informal employment in total employment by:				
3	Sex				
	Male	26.4	15.3	34.4	41.1
	excluding agriculture	22.6	14.1	31.4	31.7
	Female	23.6	13.2	28.4	47.3
	excluding agriculture	18.8	12.1	25.0	30.1

<sup>32</sup> Due to rounding, some totals may not correspond with the sum of the separate figures.

Eur	ope and Central Asia	Europe and Central Asia	Northern, Southern and Western Europe	Eastern Europe	Central and Western Asia
4	Age				
	Youth (15–24)	35.7	24.7	44.2	53.4
	Adults (25+)	21.8	13.4	31.6	39.4
	25–29	26.4	14.1	40.0	32.7
	30–34	21.8	12.2	33.5	29.3
	35–54	20.0	12.2	30.0	38.0
	55–64	21.5	14.9	26.2	62.3
	65+	40.8	38.2	29.3	79.8
5	Highest level of education				
	No education	77.1	32.3	50.0	83.7
	Primary education	40.9	27.6	44.9	49.5
	Secondary education	23.3	15.6	34.4	30.0
	Tertiary education	15.2	11.3	22.3	20.8
6	Residence area				
	Rural	33.2	17.9	36.3	73.9
	Urban	19.4	13.3	26.6	40.2
7	Industrial sectors <sup>33</sup>				
	Agriculture	71.6	47.5	64.5	86.0
	Industry	21.9	10.1	30.2	34.2
	Services	20.2	14.2	26.9	28.7

Note: Country data are available in Appendix B: table B.1 for indicators 1, 2 and 3; table B.2 for indicator 6 and tables B.6 and B.7 for indicator 7. Global and regional estimates desegregated by sex are available in table C.3 for indicator 4; tables C.1 and C.2 for indicator 5.

Source: ILO calculations based on household survey micro datasets.

#### Magnitude

In Europe and Central Asia, a quarter (25.1 per cent) of the employed population engages in informal employment and the share decreases to 20.9 per cent if agriculture is excluded. The magnitude is quite different in the three subregions. The share of informal employment in Northern, Southern and Western Europe is below the regional average accounting for 14.3 per cent. The share in Eastern Europe (31.5 per cent) and in Central and Western Asia (43.4 per cent) is substantially above the regional average. The share of non-agricultural employment in these three subregions represents 13.2 per cent, 28.3 per cent and 31.2 per cent, respectively (figure 16.G1 and table 5).

The share of informal employment in the emerging and developing subregion (36.8 per cent) is twice as high as in the developed subregion (15.6 per cent). The share of informal employment in Albania (61.0 per cent) and Armenia (52.1 per cent) is among the highest in the region and represents more than half of the employed population. The lowest rate in the region is in the single digits for developed countries (table 1 and Appendix B table B.1).

<sup>&</sup>lt;sup>33</sup> The category "not classified" in case of missing data on industrial sector is not displayed in this table.

Based on the number of "entrepreneurs" (own-account workers and employers), it is estimated that 54.7 per cent of all economic units in Europe and Central Asia are informal, 44.5 per cent are formal and the remainder less than 1 per cent are part of households. This estimated proportion of informal economic units is higher in Central and Western Asia (61.4 per cent), following the overall higher incidence of informality in the region (figure 16.65).

#### Composition

Informal employment represents 25.1 per cent of total employment in the region, with 19.4 per cent of employment in the informal sector, 5.3 per cent informal employment in the formal sector and 0.5 per cent in households.<sup>34</sup> The pattern in non-agricultural informal employment is similar. The share of informal employment in formal sector economic units is highest in Eastern Europe with 9.5 per cent (representing one-third of total informal employment and 40 per cent of informal employment among employees) in this subregion and up to 9.9 per cent excluding agriculture (table 5).

Employees represent the largest group among those in informal employment in the region and own-account workers follow. Employers and contributing family workers share more or less the same proportion. However, in Northern, Southern and Western Europe, own-account workers are the largest group in informal employment and the employers' group is also larger than the regional average. In Central and Western Asia, even though employees and own-account workers are still the major groups within informal employment, contributing family workers represent a considerable share at 17. 5 per cent, which is much higher than the 8.7 per cent regional average (figure 16.G2).

At the regional level, 60 per cent of own-account workers own informal economic units as in other regions. They are the most exposed to informality before employers (39.9 per cent) and employees (15.4 per cent). Northern, Southern and Western Europe present the lowest share of informal employment among employees (6.5 per cent). The difference between the proportion of employers (29.7 per cent) and employees (25.2 per cent) in informal employment is less in Central and Western Asia (figure 16.G3).

#### **Characteristics**

Informal employment represents a greater source of employment for men (26.4 per cent) than for women (23.6 per cent), and it is the same for non-agricultural informal employment, with 22.6 per cent for men and 18.8 per cent for women. In Central and Western Asia, the situation reverses when agriculture is included (47.3 per cent of women are in informal employment as compared to 41.1 per cent of men), but follows the regional pattern when excluding agriculture. Outside agriculture, the share of informal employment is higher for men (31.7 per cent) than for women (30.1 per cent) (table 5).

More than one-third of young workers in employment (35.7 per cent) are in informal employment compared to about only one-fifth (21.8 per cent) for adults. The share of informal employment drops steadily from the 25–29 age group to the 35–54 age group and rises again from the 55–64 age group to reach 40.8 per cent for the over 65s (table 5).

Informality falls in proportion with the rise in the level of education. The highest share of informal employment is found among those without education at 77.1 per cent, decreasing to 40.9 per cent for those with primary education, to 23.3 per cent for those with secondary education and to 15.2 per cent for those with tertiary education (figure 16.64 and table 5).

<sup>&</sup>lt;sup>34</sup> This proportion is most likely underestimated due to difficulties in identifying "households" as part of available micro datasets.

Informality is more prevalent in rural (33.2 per cent) than in urban (19.4 per cent) areas. The largest difference can be found in Central and Western Asia, where almost three quarters of the rural employed population (73.9 per cent) are in informal employment compared to 40.2 per cent in urban areas (table 5).

The incidence of informal employment is higher in the agricultural sector, with 71.6 per cent of the employed population in informal jobs. The industry (21.9 per cent) and service sectors (20.2 per cent) share similar proportions of informality. Among the three subregions, only in Northern, Southern and Western Europe does the service sector (14.2 per cent) have a higher level of informal employment than the industry sector (10.1 per cent) (table 5).

### Chapter 3

### Thematic issues in the informal economy

This chapter focuses on examining the relation between informal employment and key social and economic indicators of development. The objective is not to define causality between indicators — which would require an academic or research approach — but to provide the factual relation between them. An important feature of this chapter is the gender dimension, i.e. the difference between women and men in these key social and economic indicators.

### 3.1 Informal employment and level of economic and social development

Informality is related to the level of economic and social development. A measure of social development is the Human Development Index (HDI), which combines the indicators of long and healthy lives, knowledge and a decent standard of living.<sup>35</sup> Comparing national data on informal employment as a share of total employment with HDI values shows that countries with higher informality also have a lower HDI value (figure 17).

### Figure 17. Shares of informal employment in total employment and Human Development Index values (latest available year)



**Note:** The coefficient of determination R2=0.79. Developed countries are considered as high-income countries as defined by the World Bank; emerging countries as middle-income countries and developing countries as low-income countries. See income groupings in Appendix A.1.

Source: ILO harmonized estimates of informal employment and HDI from UNDP.

<sup>&</sup>lt;sup>35</sup> According to UNDP, "The HDI was created to emphasize that people and their capabilities should be the ultimate criteria for assessing the development of a country, not economic growth alone. The health dimension is assessed by life expectancy at birth, the education dimension is measured by mean of years of schooling for adults aged 25 years and more and expected years of schooling for children of school entering age. The standard of living dimension is measured by gross national income per capita" (source: http://hdr.undp.org/en/content/human-development-index-hdi).

In terms of income levels, the previous chapter showed that informal employment rates among developed countries are below 40 per cent, with an average of 18.3 per cent, while the share of informal employment among developing and emerging countries is on average 69.6 per cent. There is also a negative relation between the level of GDP per capita and the share of informal employment in total employment. Both the level of GDP per capita and its growth are potentially important elements for reducing informality, influencing employment generation and the economic capacity of economic units and workers. However, higher levels of GDP are not sufficient: figure 18 also shows significant levels of dispersion in each level of GDP.



Figure 18. Share of informal employment in total employment and GDP per capita (latest available year)

Note: The coefficient of determination R2=0.57.

Source: ILO harmonized data on the share of informal employment in total employment and World Development Indicators 2017 for GDP per capita.

Countries with the lowest level of GDP per capita tend to have the highest level of informality (figure 18). The gender gap in the share of informal employment<sup>36</sup> is also more likely to be positive in countries with the lowest level of GDP per capita, which means that women are more likely to be in informal employment than men (figure 19). The gender gap is actually positive in two out of three low- and lower-middle income countries.

<sup>&</sup>lt;sup>36</sup> The gender gap in the share of informal employment is the difference between the share of women in informal employment as a proportion of total women employment and the share of men in informal employment in total men employment. A positive gender gap means that women are more likely to be in informal employment than men.



Figure 19. The gender dimension of informality and development: gender gap in the share of informal employment in total employment (percentage points, latest available year)

**Note:** the gender gap in the share of informal employment in total employment is calculated as the difference between the share of informal employment among women and the men. This difference between two proportions is expressed in percentage points. A positive gap corresponds to a share of women in informal employment in total women's employment superior to the share of men in informal employment in total men's employment; it is represented in orange in the graph. A negative gender gap is represented in blue.

Source: ILO calculations for the share of informal employment for women and men and World Bank, 2017 (World development indicators) for GNI per capita.

Not all countries reaching the same level of GDP per capita share the same composition of industry sectors. The sectoral composition of GDP (i.e. the value added of respective main industry sectors to overall GDP), is correlated with estimates of informal employment in relation to the contribution of agricultural activities to GDP (figure 20, panel A) and of service activities (panel C). The relation is however not clear in the case of manufacturing/industry (panel B).

#### Figure 20. The sectoral composition of GDP (contribution of main industry sectors to GDP) and the share of informal employment in total employment (percentages, latest available year)



Developed
 Developing and emerging

Note: Panel A: R2= 0.59; Panel B: R2= 0; Panel C: R2= 0.57.

Source: ILO and WDI

Finally, there is a negative relation between the share of informal employment in total employment and the proportion of wage workers in total employment and conversely a positive relation between the share of informal employment and the proportion of own-account workers (figure 21). Own-account workers, as well as contributing family workers, for example are populations that are hard to reach by the usual tax, social security and labour administration policies to improve the situation of informal workers. This points to the need for innovative approaches or even non-conventional methods in order to facilitate their transition to formality (ILO, 2014b, 2017; ILO/GIZ, 2014).

### Figure 21. Share of status in employment and share of informal employment in total employment (percentages, latest available year)



Developed • Developing and emerging

**Note:** Panel A: R2= 0.64; Panel B: R2= 0.58. Orange dots for developing and emerging countries; blue dots for developed countries. **Source:** ILO harmonized data on the share of informal employment in total employment and ILO Trends for employment status as a percentage of total employment.

### 3.2 Informal employment and poverty

Recent estimates of informality reveal new insights into the relation between poverty and informality.<sup>37</sup> There is a clear positive relation between poverty and informality, with the poor facing higher rates of informal employment, but at the same time significant disparities are noted. These findings recognize the limitations of this joint analysis of employment (an individual based measure) and poverty (a household-based measure). This approach acknowledges the fact that poverty is strongly affected by household size and composition and that the working poor may benefit from decent working conditions and still be below the poverty line, not because they earn less than the poverty line, but because they share this labour income with many dependants (OECD, 2009; ILO, 2016b). With these limitations in mind, in developing and emerging countries the share of informal employment in total employment ranges from 50.4 per cent to more than 98 per cent in Cameroon and Rwanda among the poor (figure 22). Considering the non-poor in employ-

<sup>&</sup>lt;sup>37</sup> The analysis of informality and poverty presented in figures 22 to 24 refer for some countries to different datasets than those used for other indicators presented in the report. The same set of criteria to define informal employment and employment in the informal sector have been applied to household income and expenditure survey datasets to allow this combined analysis of poverty and informality. For countries concerned, alternative datasets used for this analysis are presented in Appendix A.2.

ment, 10 out of the 28 countries examined show average informal employment rates below 50 per cent. In developed countries, the share of informal employment in total employment ranges from 13.6 per cent to 62 per cent among the working poor compared to proportions among the non-poor, which are only half as much (from 3.7 per cent to 26.8 per cent).



Figure 22. Shares of informal employment among the poor (horizontal axis) and non-poor (vertical axis) (percentages, latest available year)

**Note:** The horizontal axis presents the share of informal employment among the poor and the vertical axis presents the share of informal employment among the non-poor. Blue dots cover selected developing and emerging countries with poverty being defined in reference to the absolute poverty line of US\$3.10PPP. Orange dots refer to developed countries with working poor, meaning workers with an income below 60 per cent of the national median household disposable income. Consumption and income are calculated on a per capita basis, including for developed countries. The analysis of informality and poverty is based for some countries on different datasets than those used for other indicators presented in the report (country datasets used for the joint poverty and informality analysis are presented in Appendix A.2).

Source: ILO calculations based on national household surveys.

It is often assumed that all informal workers are poor. However, these data show that this is not the case. Figure 23 presents a comparison of working poverty rates for workers in informal and in formal employment. Panel A covers selected developing and emerging countries using the common absolute international extreme and moderate poverty line of US\$3.10PPP per capita per day. Panel B focuses on developed countries and a relative poverty line of 60 per cent of national household disposable income. In both cases, while there is a strong correlation between poverty and informality, there are some workers in informal employment who are not poor, and others in formal employment who are poor (either because they earn lower incomes or because despite decent incomes, including decent labour incomes, they share their income with a high number of economic dependents within the household).

These data suggest that not all workers enter the informal economy for the same reason. It is worth noting that in 2015, in adopting Recommendation No. 204, ILO constituents from countries all over the world agreed that most people enter the informal economy not by choice, but as a consequence of a lack of opportunities in the formal economy and in the absence of other means of livelihood.<sup>38</sup>

<sup>&</sup>lt;sup>38</sup> See Kucera and Roncolato (2008) for a detailed discussion on this approach and alternative reading of empirical results. They note that most of the evidence comes from Latin America countries.



Figure 23. Poverty rates and the informal or formal nature of worker's main job (percentages, latest available year)

(at US3.10\$PPP per capita per day) in selected developing and emerging countries

Panel A. Poverty rate

Panel B. Relative poverty rate (60% median value of household disposable income) in selected developed countries



**Note:** Panel A covers selected developing and emerging countries. Proportions correspond to the share of those with per capita income or consumption below US\$3.10 PPP per day which corresponds to the latest international extreme and moderate poverty line. In panel B, the relative poverty rate for developed countries is defined as the share of those with an income below 60 per cent of the national median household disposable income. Consumption and income are calculated on a per capita basis, including for developed countries. The share of total informal employment (including agriculture) follows the common statistical definition as presented in Chapter 1. The analysis of informality and poverty is based for some countries on different datasets than those used for other indicators presented in the report (country datasets used for the joint poverty and informality analysis are presented in Appendix A.2)

Source: ILO calculations based on national household surveys

The analysis of the gender dimension of poverty among women and men whether in informal employment or in formal employment is not very conclusive, because poverty rates is heavily influenced by household size and structure in addition to personal features. Nevertheless, when comparing poverty rates between men (horizontal axis) and women (vertical axis) in informal employment (figure 24, panel A), most developing countries and more generally, most countries with the highest incidence of poverty show a higher poverty rate for women than for men, meaning that more women than men in informal employment live in households which are below the poverty line. The situation is reversed when comparing poverty rates for women and men in formal employment (figure 24, panel B).



## Figure 24. Working poverty rates among women and men in informal and in formal employment (percentages, latest available year)

**Note:** For emerging and developing countries (in blue), proportions correspond to the share of those with per capita income or consumption below US\$3.10 PPP per day. For developed countries, the relative poverty rate is defined as the share of those with an income below 60 per cent of the national median household disposable income. Consumption and income are calculated on a per capita basis, including for developed countries. The analysis of informality and poverty is based for some countries on different datasets than those used for other indicators presented in the report (country datasets used for the joint poverty and informality analysis are presented in Appendix A.2).

Source: ILO calculations based on national household surveys.

#### 3.3 Informal employment and education

While education is not a guarantee for access to formal and more productive employment, it is an important asset. This positive relation is observed in all regions in the world and for all categories of status in employment with, however, some variations. Training and skills development are essential contributing factors for the transition to the formal economy, as they can improve the competencies and employability of the workforce. While the data in this section focus solely on the relation between formal education and informality, it is important to note that most workers in the informal economy, who often do not have access to formal education and training, accumulate knowledge and skills through other means no matter where or how they were acquired: at the workplace, in the community, at home, through informal apprenticeships or "learning by doing" (ILO, 2008).

The global and regional estimates clearly highlight the link between the increase in the level of education of workers and the decrease in the share of informal employment everywhere and for

all statuses in employment.<sup>39</sup> The majority of workers with no education (93.8 per cent) are in informal employment (figure 25, panel A). The share of informal employment decreases to 84.6 per cent among workers with primary education, to 51.7 per cent for those with secondary education and 23.8 per cent for tertiary education. This pattern is largely influenced by developing and emerging countries where the largest share of workers in informal employment is concentrated. In developing and emerging countries, the share of informal employment in total employment drops from 93.9 per cent among workers with no education to 32.0 per cent among those with tertiary education (figure 25, panel B). The fall in the share of informal employment as the level of education rises seems to be true in developed countries too. However, in developed countries, where the level of informal employment is much lower, the differences in rates of informal employment between those with higher and lower education are much smaller (figure 25, panel C). In developed countries the share of informal employment ranges from 52.7 per cent among those with no education (who represent a minority) to 16.1 per cent among workers with tertiary education.





**Note:** Global and regional estimates based on data for 107 countries representing 86 per cent of the world employed population. Harmonized definition of informal employment and employment in the informal sector. Additional data available in Appendix C.1. **Source:** ILO calculation based on micro data. This figure is reproduced in figure 10 in Chapter 2.

Another way to look at education and informality is to compare the distribution of workers in informal employment by level of education to the one observed among workers in formal employment (figure 26). Half of the world population in informal employment has either no education or

<sup>&</sup>lt;sup>39</sup> Those results based on the analysis for 107 countries of the share of informal employment and of employment in the informal sector according the level education of workers (by employment status and by sex) confirm earlier findings from a large number of theoretical studies, mainly based on heterogeneous workers and/or firms and matching models, that show that more educated (and more productive) workers go to formal sector jobs, whereas less productive workers move into the informal sector (see, for example, Boeri, T., Garibaldi, P., 2005; Galiani, S., Weinschelbaum, F., 2007); they confirm and expand on more empirical studies focusing on the informal sector (for instance, Bernasek and Gallaway in 2002 which found that workers with the highest levels of education were more likely to be found in the formal sector, while the less educated people had higher probabilities of working in the informal sector (Bernasek, A; Gallaway, J.H. 2002).

a primary level. By contrast, just above 7.0 per cent of workers in informal employment worldwide have reached tertiary educational level. The corresponding proportions among workers in formal employment are 9.3 per cent (with no education or primary education at best) and 24.4 per cent with tertiary educational level respectively (figure 26, panel A). The lack of education among workers in informal employment is critical in Africa and the developing Arab States. In Africa, more than three-quarters of the workers in informal employment have primary education as the highest educational level (44.5 per cent have no education) and less than 2 per cent of them have reached tertiary educational level.

The situation in developed countries is slightly different (figure 26, panel C). Secondary and tertiary levels of education are the norm in these countries, but tertiary education does not represent a protection against informal employment. There are differences in the distribution of workers by level of education depending on the formal or informal nature of the main job, but these are smaller and less linear than in developing and emerging countries. The proportion of workers with either no or primary education is higher among those in informal employment but the proportions are below 10 per cent. In this group of countries, a significant proportion of workers in informal employment have a tertiary level of education: the share reaches 44.0 per cent in all developed countries with proportions that range from 18.5 per cent in Asia and the Pacific to 80.0 per cent in the developed countries of the Americas.



### Figure 26. Distribution of workers in informal and formal employment by level of education (percentages, latest available year)

Note: Global and regional estimates based on data for 107 countries representing 86 per cent of the world employed population. Harmonized definition of informal employment and employment in the informal sector. Additional data are available in Appendix C.2.

The positive effect of the increase in the level of education on access to formal employment is obvious among employees and employers, but far less among own-account workers, whose exposure to informal employment remains high (above 60 per cent) regardless of their level of education. This differentiated effect of the level of education on informality depending on employment status occurs for all levels of economic development and appears in all regions. The labour market structure in terms of employment status determined to a large extent the overall share of informal employment by level of education, i.e. the respective proportions of workers in categories of status in employment that are more exposed to informal employment than the others, namely contributing family workers (informal regardless of their level of education) and own-account workers.

The share of informal employment among employees is always lower than for other statuses (see table 1) and this is still true when the level of education is considered. Globally, informal employment rates among employees decrease sharply from 85.2 per cent for those with no education to 15.7 per cent among the most educated (tertiary education). A similar pattern is seen in developing and emerging countries with informality rates falling from 86.0 per cent to 21.5 per cent with the increase in the level of education (figure 27, panel B). The main difference regarding the situation of employees in developed countries lies in the lower share of informal employment among workers with secondary education (9.3 per cent) compared to those with a tertiary educational level (figure 27, panel C).





**Note:** Global and regional estimates based on data for 107 countries representing 86 per cent of the world's employed population. Harmonized definition of informal employment and employment in the informal sector. Contributing family workers are not displayed in figure 27 as they are considered as informal by statistical definition, independently of the formal or informal nature of the economic unit or any personal or employment related feature. Additional data available in Appendices C.1.

Source: ILO calculation based on micro data.

The category of employers shows the opportunity to access formal employment that comes with higher levels of education. Furthermore, the profile of informality among employers according to the level of education is surprisingly similar for all groups of countries defined according to their level of economic development. More than 90 per cent of employers with no education (in both developing, emerging and developed countries) own informal economic units. Holding a secondary level of education makes a major difference, since half of the employers are formal (i.e.

operating formal sector units). The rate of informal employment further decreases to just above 30 per cent among those with a tertiary educational level.

By contrast, the effect of an increase in the level of education on the share of informal employment among own-account workers is limited. The proportion of own-account workers operating informal economic units remains above 70 per cent including among those with secondary education. The lowest share of informal employment among own-account workers concerns ownaccount workers with tertiary education in developed countries (62.9 per cent).

Low productivity is often mentioned as a characteristic of informal economic units. Low level of education among workers in informal sector enterprises is certainly one important factor. Other factors include the lack of access to credit resulting in capital constraints and operations below the efficient scale of production for informal firms and the use of (low-skill) labour as a substitute; and the lack of access to markets and to key public goods and services. The comparative analysis of the levels of education of workers (entrepreneurs as well as employees) between informal and informal economic units can be seen as a contribution to the analysis of productivity differentials between the formal and informal sector economic units (box 4). In addition to these supply side

#### Box 4. Are educated workers more likely to go to formal sector jobs? Is labour in informal sector enterprises less qualified?

Are the most educated workers more likely to work in the formal sector? The answer is yes. Most educated workers are more likely to work in formal sector enterprises (figure 28). As the informal nature of employment among employers and own-account workers is defined according to the formal nature of their economic unit, the share of employment in the informal sector is similar for them to the share of informal employment. The case of employees is interesting as it shows that as the majority of highly educated employees join formal sector enterprises (more than 90 per cent in both developed, developing and emerging countries), the proportion of employees accessing formal jobs is slightly lower in developing and emerging countries (figure 26).



#### Figure 28. Share of employment in informal economic units and the level of education and status in employment (percentages, latest available year)

**Note:** Global and regional estimates based on data for 107 countries representing 86 per cent of the world employed population. Harmonized definition of informal employment and employment in the informal sector.

The answer to the question whether labour is less (formally) qualified in the informal sector, the response is also yes, but mainly in developing and emerging countries. In developing and emerging countries (figure 29, panel B), nearly 55 per cent of all workers in the informal sector have no education or primary education as the highest level of education compared to 15.5 per cent among workers in the formal sector. Nearly two-thirds of own-account workers operating informally are not educated or have primary level education; the proportion is almost halved among those operating formally (35.7 per cent). The overall level of education among employers (whether in the informal or in the formal sector) is higher and in this respect similar to the one observed among employees. Nonetheless, the proportion of low educated employers operating informally is more than double that of those operating a formal business (37.6 per cent compared to 14.0 per cent).

In developed countries (figure 29, panel C), levels of education are higher whatever the formal or informal nature of enterprises and categories of status in employment. The main difference lies in the respective shares of employers and own-account workers with secondary-level rather than tertiary-level education. Nevertheless, the level of education does not necessarily appear to be a key factor of differentiation between formal and informal economic units.



#### Figure 29. Educational composition of employment in the informal and formal sector (percentages, latest available year)

**Note:** Global and regional estimates based on data for 107 countries representing 86 per cent of the world employed population. Harmonized definition of informal employment and employment in the informal sector.

factors, the broader policy and regulatory environment often excludes or penalizes the informal self-employed and their income-earning activities. Many informal self-employed work in a public space or private homes without secure tenure and without basic infrastructure and transport services. Those who work in a public space often face harassment, bribes and confiscation of goods. Street vendors report that they do not invest in larger stock, digital scales or fancier display units for fear of confiscation or theft (Chen et al., 2016).

The gender dimension of the effect of the educational level on the share of informal employment highlights once again the need for universal and equal access to education, and not only to education but to a good level of education. While globally the percentage of women in informal employment is lower than that of men (see Chapter 2), the opposite situation occurs among loweducated workers. Among the less educated workers, women are more exposed to informality than men in developing, emerging and developed countries (figure 30). Nearly 91 per cent of

#### Figure 30. Share of informal employment (percentage of total employment) by level of education and sex (percentages, latest available year)



**Note:** Dashed lines represent the overall share of informal employment in total employment, independent of the level of education. Global and regional estimates based on data for 107 countries representing 86 per cent of the world employed population. Harmonized definition of informal employment and employment in the informal sector. Additional data available in Appendix C.1.

women with either no education or primary education are in informal employment compared to 87.2 per cent of men with similar levels of education. The situation is reversed when women reach secondary or higher levels of education.

At the global level, the situation by status in employment and sex shows similar trends. Three main observations can be drawn: the dramatic impact of the level of education among women employees; a delayed effect among the minority of women employers; and, despite higher shares of informal employment among own-account workers, a significant gender difference in favour of women own-account workers with secondary and tertiary education (figure 30, panels D to F).

#### 3.4 Informal employment and working conditions

This section seeks to assess the relationship between informality and certain working conditions. With a focus on employees, we start by quantifying the overlap between the different forms of employment and the incidence of informal employment. The scope of the analysis is then extended to all workers (employees as well as the self-employed) to compare the incidence of very short working hours and excessive working hours depending on the formal or informal nature of jobs. The goal of this statistical overview is primarily to highlight situations potentially critical either in terms of level of earnings and working poverty. These may result from very short hours of work, or from terms of health and/or safety and work–life balance issues associated with excessive working hours.

#### 3.4.1 Informal employment and non-standard forms of employment among employees

Non-standard forms of employment comprise four different employment arrangements that deviate from the "standard employment relationship", understood as work that is full time, indefinite, and part of a subordinate relationship between an employee and an employer (ILO, 2016a). The four types of employment are: (a) temporary waged employment; (b) part-time waged employment defined here as less than 35 weekly working hours; (c) temporary agency work and other forms of employment involving multiple parties; and (d) disguised employment relationships and dependent self-employment.<sup>40</sup>

This first section focuses on employees, specifically on the first two types of non-standard employment arrangements: temporary and part-time employment. It analyses the overlaps between standard employment, temporary and part-time employment, and the formal and informal nature of the main job among employees.<sup>41</sup>

<sup>&</sup>lt;sup>40</sup> Regional and global estimates are based on 96 countries representing 78 per cent of global employment. Some regions are not well represented and not displayed in tables or graphs.

<sup>&</sup>lt;sup>41</sup> In-depth analysis of the overlap between informal employment and non-standard employment can be found in ILO, 2016a and more recently for G20 countries in ILO, 2018.



Figure 31. Share of informal employment among employees in permanent full-time employment compared to non-standard forms of employment by sex (percentages, latest available year)

Note: Global and regional estimates based on data for 96 countries representing over 78 per cent of the world employed population. Harmonized definition of informal employment and informal sector.

Source: ILO calculation based on national labour force surveys or similar household survey data.

Compared to standard (permanent, full-time) employment, non-standard forms of employment are significantly more likely to be informal, for both women and men. Globally, 84.3 per cent of employees in full-time permanent employment are in formal employment without much variation among regions and levels of development. In other words, 15.7 per cent of employees in standard employment hold informal jobs (figure 31). The differences between regions and levels of development lie first in the share of employees in total employment and second in the proportion of employees in permanent full-time jobs (ILO, 2015c). The percentage of employees in informal employment significantly increases among part-time employees (44.0 per cent, analysed further in the next section); it increases even more among employees in temporary employment (56.7 per cent) and is the highest for employees in "temporary part-time jobs" (64.4 per cent), especially among men (68.1 per cent).

Women employees seem to have a relatively lower risk of being informal when working in nonstandard forms of employment and notably when working part-time. Just over one-third of women employees working less than 35 hours a week are in informal employment compared to 54.2 per cent among men employees in a similar situation. This is influenced by the situation in Europe and Central Asia, where a significant proportion of women employees working part-time might work in the public sector and in large formal enterprises that comply with national legislation on labour and social security protection that is relatively well developed and applied.

For employees, the informal nature of their main job means primarily the absence of social security coverage gained through their employment relationship with contributions paid by their employer on their behalf. It can also mean the absence of other employment benefits such as annual paid leave or paid sick leave in case of need. The reasons why employees in temporary and part-time employment are prone to informality are either the fact that they are outside the scope of current laws and regulations or they do not meet the qualifying thresholds in terms of length of employment or of the number of hours worked; or if legally covered, that those legal provisions are not effectively implemented.

Temporary workers are sometimes explicitly excluded from legal social security coverage. As far as social insurance is concerned, this is the case for instance in Egypt, Syrian Arab Republic, Thailand and Vanuatu.<sup>42</sup> As far as part-time work is concerned, some countries, such as Japan, the Republic of Korea and South Africa,<sup>43</sup> restrict eligibility among employees by fixing a minimum number of hours of work with a direct effect on the effective protection of workers in part-time employment. Indirectly, short and in particular very short hours of work resulting in low levels of income may also lead to an exclusion from the scope of current social security laws if conditioned by a minimum level of earnings. In Germany, Japan and the Republic of Korea for instance, workers with low earnings are excluded from coverage for unemployment benefits. Non-explicit indirect exclusions resulting from the inability to meet the minimum qualifying conditions (thresholds regarding the number of hours of work for a given period of reference, a qualifying period of contributions or a minimum level of earnings) are more numerous but not as much as the absence of effective implementation of legal coverage. The lack of effective implementation of laws and regulations is reflected by multiple circumstances that include the absence of employment contracts for temporary workers, financial constraints on the employer's side, heavy or inappropriate modalities for compliance, a lack of awareness or a deliberate choice not to comply (ILO, 2015c).

<sup>&</sup>lt;sup>42</sup> In Egypt, temporary and casual workers are excluded from the scope of social security law for maternity, sickness or unemployment benefits; in the Syrian Arab Republic, temporary workers are not covered by law for pensions; in Vanuatu, temporary workers in agriculture and forestry with employment contracts of less than two months are not covered for pensions; and in Thailand, both temporary and seasonal workers are not legally covered for pensions, maternity, sickness and unemployment benefits but they might be covered by other social insurance schemes (SSA/ISSA, 2017a, 2017b).

<sup>&</sup>lt;sup>43</sup> In Japan, in October 2016, coverage for part-time employees was initially required for large companies (500 or more employees) by reducing the monthly salary threshold for registration from 108,333 yen (US\$925) to 88,333 yen (US\$755) and by lowering the required weekly number of hours worked from 30 to 20. From April 2017, mandatory coverage of part-time employees was extended under the Employees' Pension Insurance system to companies with fewer than 500 employees. (SSA, 2017; National Institute of Population and Social Security Research, 2014). Still, there is a minimum of 20 scheduled working hours per week in Japan; at least 60 hours a month or 15 hours a week in the Republic of Korea; and more than 24 hours a month in South Africa.

#### 3.4.2 Informal employment and working time

This section provides some facts about the exposure of employees and more generally of all workers to informal employment in relation to working time. The objective is to answer two main questions: Are employees and more generally workers in part-time employment more exposed to informality than the others? Are workers in informal employment more likely to work either very short hours or on the contrary, excessive hours than those in formal employment?

The first question, directly linked to the previous section, concerns one of the main forms of nonstandard employment: part-time employment is defined here as less than 35 hours a week. The facts provided below assess the relative exposure to informality associated with part-time employment and to marginal employment whether based on legal grounds or resulting from a lack of compliance with laws and regulations. Figure 32 compares the proportion of informal employment among all workers (panel A); women (panel B) and employees (panel C) whether they are in marginal employment (working less than 20 hours a week), are in part-time employment (less than 35 hours a week) or are employed for 35 hours a week or more.

#### Figure 32. Share of informal employment among workers in marginal employment (less than 20 hours a week), in part-time employment (less than 35 hours of week) compared to those working 35 hours a week or more (percentages, latest available year)





Note: Global and regional estimates based on 106 countries representing 88 per cent of the world employed population.

Source: ILO calculations based on national household surveys.

First, the difference regarding the exposure to informal employment associated with working time lies primarily in whether the worker is in part-time employment, defined as less than 35 hours per week. The additional risk of being informally employed associated with marginal employment (less than 20 hours a week) is surprisingly limited.

Second, the effect of short hours of work on the incidence of informal employment is more limited when employees are considered rather than total employment. Worldwide, 48.3 per cent of employees in marginal employment and 44.0 per cent of employees working part-time hold informal jobs, compared to 41.8 per cent for employees working 35 hours or more (figure 32, panel C). The relatively higher exposure to informal employment for those on shorter working time becomes obvious when considering total employment (i.e. including own-account workers, employers and contributing family workers). The share of informal employment increases significantly from 56.5 per cent among workers in full-time employment to 75.1 per cent for workers in part-time employment and 78.5 per cent for marginal employment.

Workers in informal employment (employees as well as the self-employed) often work outside normal hours of work. They are at the same time not covered by labour laws and regulations or lack their effective implementation. They are relatively more exposed than workers in formal employment to very short hours of work, often as the sole available option rather than by choice, but also to excessive hours of work. Both situations have negative consequences: time related under-employment and a potentially higher risk of working poverty in the case of very short hours of work<sup>44</sup> and exposure to higher health and safety risks in addition to work–life balance issues without due financial compensation in the case of excessive hours.

Aiming to answer the second question ("Are workers in informal employment more likely to work outside normal working hours than those in formal employment?"), figure 33 presents the percentage of workers in informal employment working less than 20 hours a week as compared to the corresponding proportions among workers in formal employment. Panel A, B and C present the percentage of workers working for short hours in the world, in developing and emerging countries and in developed countries. Panels D to F consider the situation for different groups of workers: such as women (panel D), employees (panel E) and own-account workers (panel F).

The share of workers on very short working hours is systematically higher among workers in informal employment compared to those in formal employment whatever the region (figure 33, panel A) and the level of development (panels B and C) are. Globally, 10.1 per cent of workers in informal employment are working less than 20 hours a week as compared to 4.2 per cent of workers in formal employment. Africa shows the largest difference with a proportion more than five times higher among workers in informal employment (18.0 per cent) working less than 20 hours a week as compared to those in formal employment (3.2 per cent). The incidence of very short working hours among workers in informal employment is relatively higher in developed countries (12.3 per cent on average compared to 10.0 per cent in developing and emerging countries). However, in developed countries the difference between the proportion of workers in informal employment working very short hours compared to the proportion of workers in formal employment to the proportion of workers in formal employment to the proportion of workers in informal employment working very short hours compared to the proportion of workers in formal employment than in developing and emerging countries.

The situation of women is the most critical. More than 14 per cent of all women in informal employment work less than 20 hours a week as compared to 3.1 per cent of women in formal employment. This proportion reaches more than 20 per cent in Africa, the Americas and the Arab States. The most likely reason is the time spent on unpaid work.

As far as status in employment is concerned, the share of own-account workers working very short hours is significantly higher among those operating in informal economic units (11.0 per

<sup>&</sup>lt;sup>44</sup> A discussion and a quantitative assessment of the positive relation between short working hours and poverty can be found in Chapter 2, Addressing the income gap in ILO, 2016b.




Note: Global and regional estimates based on 106 countries representing 88 per cent of the world employed population. Additional data available in Appendices C.4 and C.5.

Source: ILO calculations based on national household surveys.

cent) as compared to the formal ones (6.7 per cent). This pattern is found in all regions, with the highest incidence of very short hours for those operating informally in Africa (18.3 per cent) and the Americas (21.6 per cent). The situation of employees is different. Both the overall incidence of short hours of work and the differences between employees in formal or informal employment are lower.



Figure 34. Incidence of time-related underemployment and the informal or formal nature of employment (percentages, latest available year)

**Note:** Time-related underemployment is defined here as persons who during the last week were willing to work additional hours, were available to do so, and had worked less hours than 35 hours a week (actual working hours in all jobs being considered as a basis). This definition allows international comparisons while being in line with the Resolution concerning the measurement of underemployment and inadequate employment situations. Results might be different from national ones as the reference period and in particular the minimum threshold relating to working time is chosen according to national circumstances.

Source: ILO calculations based on national household surveys.

The comparison of the incidence of time-related underemployment between workers in informal employment or in formal employment suggests an absence of other available options rather than a choice for those working informally. The proportion of workers in time-related underemployment is higher among workers in informal employment in most countries with available data (figure 34).

On the one hand, workers in informal employment are more likely to hold jobs associated with very short hours of work for pay or profit, than workers in formal employment but on the other hand workers in informal employment are also more likely to work excessive hours (more than 48 hours a week or even more than 60 hours a week as shown in figure C.2 in Appendix). This is particularly the case for men compared to women and for employees holding informal jobs in developing and emerging countries (figure 35).

Close to 45 per cent of all workers in informal employment are working more than 48 hours a week compared to 30.3 per cent among workers in formal employment (panel A). The higher exposure to long hours of work in informal employment concerns primarily developing and emerging countries (panel B). While nearly one out of two workers in informal employment work more than 48 hours a week in developing and emerging countries; the corresponding ratio is about one worker in four in developed countries.

Workers in informal employment in Asia and the Pacific are the most exposed to long hours of work (52.2 per cent), especially among employees (56.2 per cent). This phenomenon in Asia and the Pacific is extreme, but working longer hours when holding informal jobs seems to be the reality for half of all employees in the developing and emerging world (panel E). This reality is significantly different from the situation of employees in developed countries, as less than 16 per cent work long hours, without any difference between formal or informal employment.



Figure 35. Proportion of workers working more than 48 hours a week and the formal or informal nature of workers' main job (percentages, latest available year)

Note: Global and regional estimates based on 106 countries representing 88 per cent of the world employed population. Additional data available in Appendices C.4 and C.5.

Source: ILO calculations based on national household surveys.

Own-account workers show completely a different picture, not in terms of the overall incidence of long hours among those holding informal economic units (43.9 per cent worldwide), but regarding their relative situation compared to own-account workers operating in the formal sector. Everywhere, the relative minority of own-account workers owning formal economic units tend to work longer hours than their counterparts operating informally (panel F).

# Chapter 4

# Main findings

This publication aims to provide a statistical overview on informality by applying a harmonized definition of informal employment and employment in the informal sector to micro data for more than 100 countries representing more than 90 per cent of the world's employed population. The estimates are based on a common set of standardized criteria to determine informal employment and employment in the informal sector as the main job. The estimates are comparable across developed and developing countries and regions, but the ILO's country estimates may differ from national ones when they exist. This edition also provides global and regional estimates for 2016 based on percentages calculated from the latest available data.

The statistics show that informal employment plays a significant role in the global labour market. Two billion workers, representing 61.2 per cent of the world's employed population, are in informal employment. Half of the world's employed population work informally in non-agricultural activities. The level of socio-economic development is positively related to formality. Emerging and developing countries have substantially higher rates of informality than developed countries. The informal sector comprises the largest component of informal employment in all regions. When the share of informal employment is disaggregated by sex, men (63.0 per cent) have higher rates of informal employment than women (58.1 per cent) around the world, but there are actually more countries (55.5 per cent) where the share of women in informal employment exceeds the share of men. Women are more exposed to informal employment in sub-Saharan Africa, the Latin American countries and most low- and lower-middle income countries. They are more often found in the most vulnerable situations. Young people and older persons are found to be more affected by informality than persons aged between 25 and 64. The level of education is another key factor affecting the level of informality. Globally, increases in the level of education are related to decreases in the level of informality. People living in rural areas are almost twice as likely to be in informal employment (80.0 per cent) as those living in urban areas (43.7 per cent). The agricultural sector by nature is the sector with the highest level of informality (93.6 per cent) around the world. The industry (57.2 per cent) and service (47.2 per cent) sectors have relatively less informality.

# Informal employment by region

The level and composition of informality is different by region. Informal employment is the main source of employment for **Africa**, where its share reaches 85.8 per cent and 71.9 per cent of employment excluding agriculture. While there is little variation in the share of informal employment among countries in Northern Africa, there is substantial variation among the countries of sub-Saharan Africa. This is a major difference that characterizes countries in Southern Africa and explains to a large extent some of the differences observed for informality. Most workers (76.0 per cent) are in informal employment in the informal sector, with a relatively small proportion of informally employed in the formal sector (5.5 per cent) and in households (4.3 per cent). In this region, a higher proportion of women's employment (89.7 per cent) is informal than men's (82.7 per cent). Young people (94.9 per cent) and older persons (96.0 per cent) have very high levels of informal employment. The level of education is closely linked to informality in all subregions. Informal employment dominates the labour market in both rural (88.3 per cent) and urban areas (76.3 per

cent). There is a high concentration of informality in agriculture with 97.9 per cent of employment being informal. The rate of informality is relatively lower in the industry/manufacturing (77.4 per cent) and service sectors (70.2 per cent) but is still very high.

In **the Americas**, 40.0 per cent of total employment or 183 million persons are in informal employment. Informal employment ranges from 18.1 per cent of employment in Northern America to 53.1 per cent in Latin America and the Caribbean. Informal economic units provide the largest share of informal employment (29.3 per cent of the 40.0 per cent of total informal employment in the region). However, a significant proportion of employees in informal employment are employed in the formal sector (42.6 per cent of informally employed employees) or in households (15.2 per cent of all informally employed employees). The informality rate is higher for men than for women in the region, mainly due to Northern America. In Latin American countries, the informality rate is higher for women (54.3 per cent) than men (52.3 per cent). The informality rate is also higher for the youth population: 46.2 per cent compared to the adult's rate of 40.4 per cent. There is an inverse relation with the level of education. The informality rate decrease as education increases. The place of residence influences informality: 52.6 per cent of the employed population in rural areas and 35.8 per cent in urban areas are in informal employment. Informality accounts for 77.5 per cent of employment in agriculture, 38.4 per cent in industry and 35.5 per cent in services.

In **the Asia and Pacific** region, more than half (59.2 per cent) of the employed population engage in non-agricultural informal employment (64.8 per cent excluding China). If agricultural employment is included, the share of informal employment rises to 68.2 per cent (77.6 per cent excluding China). Since the region is so extensive and the development of countries varies, there is a wide range across countries in the share of informal employment. Informal employment in the informal sector dominates at 58.8 per cent (in Southern Asia up to 77.4 per cent), and informal employment in the formal sector and households accounts for 6.9 per cent and 2.5 per cent, respectively. More men (70.5 per cent) than women (64.1 per cent) are in informal employment. The two exceptions are South-Eastern Asia and Southern Asia. Informality decreases with higher levels of education. Informal employment predominates in rural areas (85.2 per cent) and is also almost half of employment in urban areas (47.4 per cent). Nearly all of agricultural employment (94.7 per cent) is informal and informal employment represents a greater share in industry (68.8 per cent) than in the service sector (54.1 per cent).

In **Europe and Central Asia**, a quarter (25.1 per cent) of the employed population engages in informal employment. The magnitude is quite different in the three subregions. The share of informal employment is more than twice as high in developing (36.8 per cent) than in developed (15.6 per cent) countries. The informal sector provides 19.4 per cent of total employment in the region. Another 5.3 per cent of employment is informal employment in the formal sector and only 0.5 per cent is informal employment in households. A slightly higher proportion of men's employment (26.4 per cent) is in informal employment than women's (23.6 per cent). More than one-third of the young population (35.7 per cent) and the older age group (40.8 per cent) are in informal employment compared to about only one-fifth (21.8 per cent) for the adult population. Informality declines with the increase in the level of education. Informality is more prevalent in rural areas (33.2 per cent) than in urban areas (19.4 per cent). Nearly three-quarters of the agricultural sector (71.6 per cent) is in informal employment. The industry (21.9 per cent) and service sectors (20.2 per cent) have similar proportions of informality.

## Informal employment and key social and economic indicators

Informal employment is related to the level of economic development. Informality rate among developed countries is usually well below 40 per cent with an average of 18.3 per cent, while informality rates among developing and emerging countries have a higher variation with an average of 69.6 per cent. Countries with high informality also have low HDIs. There is a negative correlation between the share of informal employment in total employment and the proportion of waged

workers and a positive correlation with the proportion of own-account workers. Women are more likely to be in informal employment than men in countries with the lowest level of GDP per capita. In sub-Saharan Africa, Latin America and most low- and lower-middle income countries, a larger proportion of women's employment than men's is informal employment.

There is a clear positive relation between informal employment and poverty, but the data also show that some informal workers are not poor and some formal workers are poor.

In all regions of the world and for all statuses in employment, people with higher level of education are less likely to be in informal employment. Half of all those engaged in informal employment globally have either no or primary level of education and just above 7 per cent of informal workers worldwide reach tertiary level. The positive effect of the increase in the level of education on access to formal employment is obvious among employees and employers, but far less among own-account workers, whose exposure to informal employment remains high (over 60 per cent) whatever their level of education. The share of informal employment among employees is always lower than for other statuses even when the level of education is considered. Nearly 91 per cent of women with either no education or primary education are in informal employment as compared to 87.2 per cent of men with similar level of education. Among the less educated workers, women have higher levels of informality than men in both developing, emerging and developed countries, but this pattern is reversed among women and men at secondary or higher levels of education.

Globally, 15.7 per cent of employees in permanent full-time employment hold informal jobs, i.e. having no employment related social and labour protections. The proportion of employees in informal employment increases significantly among part-time employees (44.0 per cent), and among employees in temporary employment (59.6 per cent) and is highest for employees in "temporary part-time jobs" (64.4 per cent), especially among men (68.1 per cent). Women part time employees are less likely than men to be informal. Just above one-third of women employees working less than 35 hours a week are in informal employment, as compared to 54.2 per cent among men.

Worldwide, the share of informal employment varies significantly from 56.5 per cent among workers in full-time employment to 75.1 per cent for workers in part-time employment and 78.5 per cent for marginal employment (less than 20 working hours a week). The incidence of informal employment is more limited when employees are considered alone: 48.3 per cent of employees in marginal employment hold informal jobs, compared to 41.8 per cent for employees working 35 hours or more. The percentage of workers with very short working hours is higher among workers in informal employment compared to those in formal employment. Globally, 10.1 per cent of workers in informal employment work less than 20 hours a week compared to 4.2 per cent of workers in formal employment. The difference is greater for women and for own-account workers. More than 14 per cent of all women in informal employment work less than 20 hours a week compared to 3.1 per cent among women in formal employment. The percentage of the own-account employed working very short hours is significantly higher among those operating in the informal sector (14.4 per cent) when compared to the formal (6.3 per cent). The proportion of workers in time-related underemployment is higher among workers in informal employment in most countries. Workers in informal employment are even more likely to work excessive hours (more than 48 hours a week or even more than 60 hours a week), especially employees. This phenomenon in Asia and the Pacific is extreme, but working longer hours when holding informal jobs seems to be the reality for half of all employees in the developing and emerging world. This reality is significantly different from the situation of employees in developed countries, as less than 16 per cent work long hours, without any difference between formal or informal employment. Own-account workers show a different picture, as own-account workers owning formal economic units tend to work longer hours than their counterparts operating informally.

# References

- Bernasek, A; Gallaway, J.H. 2002. Gender and informal sector employment in Indonesia, *Journal of Economic Issues*, Vol. 36 No. 2.
- Boeri, T.; Garibaldi, P. 2005. "Shadow sorting", in Frankel, J.A. and Pissarides, C. (eds) NBER International Seminar on Macroeconomics 2005 (2007). National Bureau of Economic Research. (p. 125 163) http://www.nber.org/chapters/c0339 [Feb. 2018]
- Chacaltana, J. 2016. Peru, 2002-2012: growth, structural change and formalization, CEPAL Review, No. 119, August 2016.
- Chen, M.; Alfers, L; Bali, N; Bird, M; Castellanos, T; Dobson, R; Hughes, K; Roever, S; Rogan, M. 2016. Technology at the base of the pyramid: Insights from Ahmedabad (India), Durban (South Africa) and Lima (Peru). Cambridge, MA: WIEGO. http://www.wiego.org/sites/default/files/ publications/files/WIEGO-Technology-Base-pyramid-2016.pdf [Feb. 2018].
- Galiani, S.; Weinschelbaum, F. 2007. Modelling informality formally: households and firms. https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=957329 [Feb. 2018].
- Hussmanns, R. 2004. *Measuring the informal economy: From employment in the informal sector to informal employment,* Working paper No. 53, Policy Integration Department, Bureau of Statistics (Geneva, ILO).
- International Labour Office (ILO). 1993. *Resolution concerning the measurement of employment in the informal sector,* Fifteenth International Conference of Labour Statisticians, (Geneva).
- -. 2002. Women and men in the informal economy: A statistical picture (Geneva).
- -. 2003. Guidelines concerning a statistical definition of informal employment, Seventeenth International Conference of Labour Statisticians (Geneva).
- -. 2008. Skills for improved productivity, employment growth and development, Report V, International Labour Conference, 97th Session (Geneva).
- —. 2013a. The informal economy and decent work: A policy resource guide supporting transitions to formality (Geneva).
- —. 2013b. Measuring informality: a statistical manual on the informal sector and informal employment (Geneva).
- 2014a. Employment policies for sustainable recovery and development, Recurrent discussion under the ILO Declaration on Social Justice for a Fair Globalization, Report VI, International Labour Conference, 103rd Session (Geneva).
- 2014b. Transitioning from the informal to the formal economy, Report V(1), International Labour Conference, 103rd Session (Geneva).
- 2015a. Resolution concerning efforts to facilitate the transition from the informal to the formal economy, adopted on 12 June 2015 (Geneva). http://www.ilo.ch/ilc/ILCSessions/104/textsadopted/WCMS\_380780/lang--en/index.htm [Feb. 2018].

- —. 2015b. Formalization of the informal economy: Follow-up to the resolution concerning efforts to facilitate the transition from the informal to the formal economy, 325th Session, Governing Body, Geneva, 29 October–12 November 2015, GB.325/POL/1/2. (Geneva)
- -. 2015c. World Employment and Social Outlook: The changing nature of jobs (Geneva).
- 2016a. Non-standard employment around the world: Understanding challenges, shaping prospects (Geneva).
- -. 2016b. World Employment and Social Outlook: Transforming jobs to end poverty (Geneva).
- 2017. World Social Protection Report 2017-19: Universal social protection to achieve the Sustainable Development Goals (Geneva).
- 2018. Informality and non-standard forms of employment, background paper prepared for the G20 Employment Working Group meeting, Buenos Aires, 20-22 February 2018.
- -. Forthcoming. The global transition to formality. New evidence and policy challenges (Geneva).
- -; Deutsche Gesellschaft für Internationale Zusammenarbeit (GIZ). 2014. Enterprise formalization: Fact or fiction? A quest for case studies (Geneva, Eschborn).
- Kucera, D; Roncolato, L. 2008. Informal employment: Two contested policy issues. International Labour Review. http://onlinelibrary.wiley.com/doi/10.1111/j.1564-913X.2008.00039.x/abstract [Feb. 2018].
- National Institute of Population and Social Security Research. 2014. Social Security in Japan. http://www.ipss.go.jp/s-info/e/ssj2014/PDF/ssj2014.pdf [Feb. 2018].
- Organisation for Economic Co-operation and Development (OECD). 2009. "Is work the best antidote to poverty?", in *Employment Outlook 2009* (Paris), pp. 166–210. Available at: http:// www.oecd.org/els/emp/45219514.pdf [Feb. 2018].
- Social Security Administration (SSA). 2017. International Update Recent Developments in Foreign Public and Private Pensions. https://www.ssa.gov/policy/docs/progdesc/intl\_update/2017-01/2017-01.pdf [Feb. 2018].
- Social Security Administration of the United States (SSA); International Social Security Association (ISSA). 2017a. Social security programs throughout the world: Asia and the Pacific, 2016 (Washington, DC, and Geneva).
- 2017b. Social security programs throughout the world: Africa. 2017 (Washington, DC, and Geneva).
- United Nations (UN). 2007. Sustainable Development Goals, http://www.un.org/sustainabledevelopment/ [Feb. 2018].

# **Appendices**

### Appendix A Country groupings, sources and method

Appendix A. – Country groupings, Sources & Country groupings, method Appendix A.1 Regional and income groupingsAppendix A.2 National sources: list of household surveysAppendix A.3 Methodological appendix

# Appendix B. Statistical appendix: Country data Key indicators on the size and composition of the informal economy

Appendix B. – Statistical appendix | Key indicators on the size and composition of the informal economy: **Country data** 

Table B.1	Share of informal employment in total employment and in non-agricultural employment by sex
Table B.2	Share of informal employment in total employment and in non-agricultural employment by urban/rural location
Table B.3	Share of informal employment in total employment by sta- tus in employment (including agriculture)
Table B.4	Share of informal employment in total employment by sta- tus in employment (excluding agriculture)
Table B.5	Distribution of workers in informal employment and in for- mal employment by employment status and sex (including agriculture)
Table B.6	Share of informal employment in agriculture, industry and services by sex
Table B.7	Distribution of workers in informal employment and in for- mal employment by sector
Table B.8	Share of employment in the informal sector, in the formal sector and in households by sex

# Appendix C. Statistical appendix: Global and regional estimates Micro determinants of informality and informality and working conditions

Appendix C. – Statistical Appendix   Micro	Table C.1	Share of informal employment by level of education, status in employment, sex and region
determinants of informality and informality and working conditions: <b>Global</b>	Table C.2	Distribution of informal employment and formal employ- ment by level of education, status in employment, sex and region
and regional estimates	Figure C.1	Share of informal employment by level of education, status in employment and region

Table C.3	Share of informal employment by age group, status in employment, sex and region
Table C.4	Informality and working time by sex for total employment and for employees

Figure C.2 Proportion of workers working more than 60 hours a week, depending of the formal or informal nature of main job

# Appendix A.1

# Regional and income groupings

#### Africa

Northern Africa Algeria Egypt Libya Morocco Sudan Tunisia Western Sahara

#### Sub-Saharan Africa

Central Africa Angola Cameroon **Central African Republic** Chad Congo Congo, Democratic Republic of the Equatorial Guinea Gabon Sao Tome and Principe Eastern Africa Burundi Comoros Djibouti Eritrea Ethiopia Kenya Madagascar Malawi Mauritius Mozambique Rwanda Somalia Tanzania, United Republic of

Uganda Zambia Zimbabwe

#### Southern Africa

Botswana Lesotho Namibia South Africa Swaziland

## Western Africa

Renin Burkina Faso Cabo Verde Côte d'Ivoire Gambia Ghana Guinea Guinea-Bissau Liberia Mali Mauritania Niger Nigeria Senegal Sierra Leone Togo

### Americas

Latin America and the Caribbean The Caribbean Bahamas Barbados Cuba Dominican Republic Haiti Jamaica Puerto Rico Saint Lucia Saint Vincent and the Grenadines Trinidad and Tobago United States Virgin Islands

Central America Belize Costa Rica El Salvador Guatemala Honduras Mexico Nicaragua Panama South America Argentina Bolivia, Plurinational State of Brazil Chile Colombia Ecuador Guyana Paraguay Peru Suriname Uruguay Venezuela, Bolivarian Republic of

#### Northern America Canada

United States

#### Arab States Bahrain

Iraq Jordan Kuwait Lebanon Occupied Palestinian Territory Oman Qatar Saudi Arabia Svrian Arab Republic United Arab Emirates Yemen

#### Asia and the Pacific

Eastern Asia China Hong Kong, China Japan Korea, Democratic People's Republic of Korea, Republic of Macau, China Mongolia Taiwan, China

### South-Eastern Asia and

the Pacific Pacific Islands Australia Fiji French Polynesia Guam New Caledonia New Zealand Papua New Guinea Samoa Solomon Islands Tonga Vanuatu South-Eastern Asia

### Brunei Darussalam

Cambodia Indonesia Lao People's Democratic Republic Malaysia Myanmar Philippines Singapore Thailand Timor-Leste Viet Nam

#### Southern Asia Afghanistan

Bangladesh Bhutan India Iran, Islamic Republic of Maldives Nepal Pakistan Sri Lanka

#### **Europe and Central Asia**

Central and Western Asia Central Asia Kazakhstan Kyrgyzstan Tajikistan Turkmenistan Uzbekistan

### Western Asia

Armenia Azerbaijan Cyprus Georgia Israel Turkey

#### Eastern Europe

Belarus Bulgaria Czech Republic Hungary Moldova, Republic of Poland Romania **Russian Federation** Slovakia Ukraine

#### Northern, Southern and Western Europe

Northern Europe Channel Islands Denmark Estonia Finland Iceland Ireland Latvia Lithuania Norway Sweden

# United Kingdom

Southern Europe Albania

Bosnia and Herzegovina Croatia Greece Italv Macedonia, the former Yugoslav Republic of Malta Montenegro Portugal Serbia Slovenia Spain Northern, Southern and Western Europe Austria Belgium France Germany Luxembourg Netherlands

Switzerland

#### Classification by income groups

# Developing countries (low income,

\$1,005 or less) Afghanistan Benin Burkina Faso Burundi

Burundi **Central African Republic** Chad Comoros Congo, Democratic Republic of the Eritrea Ethiopia Gambia Guinea Guinea-Bissau Haiti Korea, Democratic People's Republic of . Liberia Madagascar Malawi Mali Mozambique Nepal Niger Rwanda Senegal Sierra Leone Somalia Tanzania, United Republic of Togo Uganda Zimbabwe

Emerging countries (middle income) Lower-middle income

(\$1,006 to \$3,955) Angola Armenia Bangladesh Bhutan Bolivia, Plurinational State of Cabo Verde Cambodia Cameroon Congo Côte d'Ivoire Djibouti Egypt El Salvador Georgia Ghana Guatemala Honduras India Indonesia Jordan Kenya Kyrgyzstan Lao People's Democratic Republic Lesotho Mauritania Moldova, Republic of Mongolia Morocco Myanmar Nicaragua Nigeria Occupied Palestinian Territory Pakistan Papua New Guinea Philippines Sao Tome and Principe Solomon Islands Sri Lanka Sudan Swaziland Syrian Arab Republic Tajikistan Timor-Leste Tunisia Ukraine Uzbekistan Vanuatu Viet Nam Western Sahara Yemen Zambia

(\$3,956 to \$12,235) Albania Algeria Argentina Azerbaijan Belarus Belize Bosnia and Herzegovina Botswana Brazil Bulgaria China Colombia Costa Rica Croatia Cuba **Dominican Republic** Ecuador Equatorial Guinea Fiji Gabon Guyana Iran, Islamic Republic of Iraq . Jamaica Kazakhstan Lebanon Libya Macedonia, the former Yugoslav Republic of Malaysia Maldives Mauritius Mexico Montenegro Namibia Panama Paraguay Peru Romania **Russian Federation** Saint Lucia Saint Vincent and the Grenadines Samoa Serbia South Africa Suriname Thailand Tonga Turkey Turkmenistan Venezuela, Bolivarian Republic of

Upper-middle income

**Developed countries** (High income, \$12,236 or more) Australia Austria Bahamas Bahrain Barbados Belgium Brunei Darussalam Canada Channel Islands Chile Cyprus Czech Republic Denmark Estonia Finland France French Polynesia Germany Greece Guam Hong Kong, China Hungary Iceland Ireland Israel Italy Japan Korea, Republic of Kuwait Latvia Lithuania Luxembourg Macau, China Malta Netherlands New Caledonia New Zealand Norway Oman Poland Portugal Puerto Rico Qatar Saudi Arabia Singapore Slovakia Slovenia Spain Sweden Switzerland Taiwan, China Trinidad and Tobago United Arab Emirates United Kingdom United States United States Virgin Islands Uruguay

# Appendix A.2

# National sources: list of household surveys

Country	Year	Name of survey	Poverty analysis <sup>1</sup>
Albania	2013	Labour force survey	
Albania	2012	Living Standards Measurement Survey (LSMS)	•
Angola	2009	Inquérito Integrado sobre o Bem-Estar da População (IBEP)	•
Argentina	2016	Encuesta Permanente de Hogares (EPH)	
Armenia	2015	Labour force survey	
Armenia	2014	Integrated Living Conditions Survey	•
Austria	2012	EU statistics on income and living conditions (EU-SILC)	•
Bangladesh	2013	Labour force survey	
Belgium	2012	EU statistics on income and living conditions (EU-SILC)	•
Benin	2011	Enquête Modulaire Intégrée sur les Conditions de Vie des ménages (EMICOV)	•
Bolivia, Plurinational State of	2014	Encuesta de Hogares (EH)	•
Bosnia and Herzegovina	2005	Living Standards and Monitoring Survey (LSMS)	
Botswana	2009	Botswana Core Welfare Indicators Survey (BCWIS)	•
Brazil	2016	Pesquisa Nacional por Amostra de Domicílios (PNAD, Q3)	•
Brunei Darussalam	2014	Labour force survey	
Bulgaria	2012	EU statistics on income and living conditions (EU-SILC)	•
Burkina Faso	2014	Enquête multisectorielle continue	•
Cabo Verde	2015	Income survey	
Cambodia	2012	Labour force survey	
Cameroon	2007	Enquête Camerounaise Aupres des Menages (ECAM 3)	•
Chad	2003	the Comprehensive Household Survey (EIM)	
Chile	2016	Labour force survey (NENE, December)	
Chile	2013	CASEN	•
China	2013	China Household Income Project (CHIP)	•
Colombia	2015	Encuesta de Calidad de Vida	•
Comoros	2004	the Comprehensive Household Survey (EIM)	
Congo	2009	Enquête sur l'Emploi et le Secteur Informel au Congo (EESIC)	
Congo, Democratic Republic of the	2005	Labour force survey	
Costa Rica	2016	Encuesta Nacional de Hogares (ENAHO)	•
Côte d'Ivoire	2016	Labour force survey	
Croatia	2012	EU statistics on income and living conditions (EU-SILC)	•

<sup>&</sup>lt;sup>1</sup> Survey used for the joint analysis of poverty and informality. For a number of countries (such as Albania, Chile or India), a different dataset is used and indicated in this list, together with the main dataset used for the assessment of informality. For those countries, the same set of criteria to define informality was applied to the extent there were available in the survey.

Country	Year	Name of survey	Poverty analysis <sup>1</sup>
Cyprus	2012	EU statistics on income and living conditions (EU-SILC)	•
Czech Republic	2012	EU statistics on income and living conditions (EU-SILC)	•
Denmark	2012	EU statistics on income and living conditions (EU-SILC)	•
Dominican Republic	2014	Labour force survey (Encuesta de fuerza de trabajo) (ENFT)	
Ecuador	2015	Encuesta Nacional de Empleo, Desempleo y Subempleo (ENEMDU)	
Egypt	2013	Labour force survey	
El Salvador	2014	Encuesta de Hogares de Propósitos Múltiples (EHPM)	
Estonia	2012	EU statistics on income and living conditions (EU-SILC)	•
Finland	2012	EU statistics on income and living conditions (EU-SILC)	•
France	2012	EU statistics on income and living conditions (EU-SILC)	•
Gambia	2012	Labour force survey	
Germany	2013	German Socio-Economic Panel (GSOEP)	•
Ghana	2013	Ghana Living Standards Survey (GLSS)	•
Greece	2012	EU statistics on income and living conditions (EU-SILC)	•
Guatemala	2016	Encuesta Nacional de Empleo e Ingresos (ENEI, Q4)	
Honduras	2014	Encuesta de Hogares de Propositos Multiples (EHPM)	•
Hungary	2012	EU statistics on income and living conditions (EU-SILC)	•
Iceland	2012	EU statistics on income and living conditions (EU-SILC)	•
India	2012	Employment and unemployment survey	
India	2012	India Human Development Survey	•
Indonesia	2016	SAKERNAS (Q3)	
Iraq	2012	Household Socio-Economic Survey (IHSES)	
Ireland	2012	EU statistics on income and living conditions (EU-SILC)	•
Italy	2012	EU statistics on income and living conditions (EU-SILC)	•
Japan	2010	Japanese General Social Survey (JGSS)	•
Jordan	2010	Employment and Unemployment Survey (EUS)	
Korea, Republic of	2014	Korean Labor and Income Panel Study (KLIPS)	
Kyrgyzstan	2013	Life in Kyrgyzstan (LIK)	
Lao Peoples Democratic Republic	2010	Labour force survey	
Latvia	2012	EU statistics on income and living conditions (EU-SILC)	•
Liberia	2010	Labour force survey	
Lithuania	2012	EU statistics on income and living conditions (EU-SILC)	•
Luxembourg	2012	EU statistics on income and living conditions (EU-SILC)	•
Madagascar	2013	Enquête nationale sur le suivi des objectifs du millénaire pour le développement à Madagascar	٠
Malawi	2013	Labour force survey	
Mali	2015	Enquête Permanente emploi Auprès des Ménages (EPAM)	
Malta	2012	EU statistics on income and living conditions (EU-SILC)	•
Mexico	2015	Encuesta Nacional de Ocupación y Empleo (ENOE)	
Republic of Moldova	2010	Labour force survey	
Mongolia	2015	Labour force survey	
Morocco	2010	Household and youth survey	
Myanmar	2015	Labour force survey	
Namibia	2016	Labour force survey	

Country	Year	Name of survey	Poverty analysis
Nepal	2008	Labour force survey	
Netherlands	2012	EU statistics on income and living conditions (EU-SILC)	٠
Nicaragua	2014	Encuesta Nacional de Hogares para la Medición del Nivel de Vida (ENMV)	•
Niger	2011	Living Standard and Monitoring Survey (LSMS)	•
Nigeria	2013	Nigeria General Household Survey	•
Norway	2012	EU statistics on income and living conditions (EU-SILC)	٠
Occupied Palestinian Territory	2014	Labour force survey	
Pakistan	2015	Labour force survey	
Panama	2014	Labour force survey	
Paraguay	2015	Encuesta Permanente de Hogares (EPH)	
Peru	2015	Encuesta Nacional de Hogares (ENAHO)	•
Poland	2012	EU statistics on income and living conditions (EU-SILC)	•
Portugal	2012	EU statistics on income and living conditions (EU-SILC)	•
Republic of Moldova	2010	Labour force survey	
Romania	2012	EU statistics on income and living conditions (EU-SILC)	
Russian Federation	2014	Russia Longitudinal Monitoring Survey of HSE	•
Rwanda	2014	Integrated Household Living Conditions Survey (EICV4)	•
Samoa	2012	Labour force survey	
Senegal	2015	Enquête Nationale sur l'Emploi au Sénégal (ENES)	
Serbia	2016	Labour force survey (Q4)	
Sierra Leone	2014	Labour force survey	
Slovakia	2012	EU statistics on income and living conditions (EU-SILC)	•
Slovenia	2012	EU statistics on income and living conditions (EU-SILC)	٠
South Africa	2016	Labour force survey	
South Africa	2012	National Income Dynamics Study (NIDS)	٠
Spain	2012	EU statistics on income and living conditions (EU-SILC)	•
Sri Lanka	2013	Labour force survey	
Sweden	2012	EU statistics on income and living conditions (EU-SILC)	•
Switzerland	2012	EU statistics on income and living conditions (EU-SILC)	•
Syrian Arab Republic	2003	Household Income and Expenditure Survey (HIES)	
Tajikistan	2009	Living Standards Monitoring Survey (LSMS)	•
Tanzania, United Republic of	2014	Integrated labour force survey	
Timor-Leste	2013	Labour force survey	
Togo	2011	Questionnaire Unifié des Indicateurs de Base du Bien-être (QUIBB)	•
Tunisia	2014	Labour market panel survey	
Turkey	2015	Labour force survey	
Uganda	2012	Labour force and child labour survey	
United Kingdom	2012	EU statistics on income and living conditions (EU-SILC)	•
United States	2013	Consumer Expenditure Interview Survey	•
Uruguay	2016	Encuesta Continua de Hogares (ECH)	•
Venezuela, Bolivarian Republic of	2012	Encuesta de Hogares por Muestreo (EHM)	
Viet Nam	2015	Labour force survey	
Viet Nam	2008	Viet Nam Household Living Standards Survey (VHLSS)	•

Country	Year	Name of survey	Poverty analysis <sup>1</sup>
Yemen	2014	Labour force survey	
Zambia	2014	Labour force survey	
Zambia	2015	Zambia Living Conditions Monitoring Survey VII (LCMS)	•
Additional countries from ILOSTAT (for basic i	ndicator	on the size of informal employment only)	
Kazakhstan	2013	ILOSTAT	
Lesotho	2008	ILOSTAT	
Macedonia, the former Yugoslav Republic of	2015	ILOSTAT	
Philippines	2013	ILOSTAT	
Thailand	2013	ILOSTAT	
Ukraine	2015	ILOSTAT	
Zimbabwe	2014	ILOSTAT	

# Methodological appendix

# Countries with micro data used as a basis for estimate of missing values for the set of key indicators

The harmonized definition of informal employment and of employment in the informal sector (see box 2, Chapter 1) was applied to available micro datasets from 112 countries. An additional seven countries available from ILOSTAT are also taken into consideration in order to estimate missing values for key indicators on the size of the informal economy. Taken together, those 119 countries represent 63.3 per cent of the 188 ILO member States and over 90 per cent of the world's working population (table A3.1).

### Table A 3.1 Percentage of countries and total employment represented before estimating missing values for key indicators

		itries represented)		rcentage employ- resented)
	Nb	%	% of total employment	% non- agricultural employment
Africa	33	62.3	72.3	75.6
Northern Africa	3	42.9	64.9	61.8
Sub-Saharan Africa	30	65.2	73.6	79.9
Americas	19	57.6	93.0	93.1
Latin America and the Caribbean	18	58.1	95.3	95.7
Northern America	1	50.0	89.2	89.4
Arab States	5	41.7	44.2	41.9
Asia and the Pacific	19	48.7	94.7	94.1
Eastern Asia	4	50.0	96.6	96.7
South-Eastern Asia and the Pacific	10	45.5	89.2	85.8
Southern Asia	5	55.6	95.0	93.5
Europe and Central Asia	43	84.3	92.5	94.1
Northern, Southern and Western Europe	28	93.3	99.9	99.9
Eastern Europe	9	90.0	96.6	96.5
Central and Western Asia	6	54.5	63.0	66.1
World	119	63.3	90.5	91.5
Developing and emerging	85	64.9	91.6	93.3
Developed	34	59.6	85.1	85.3

## Missing values

Missing values have been estimated for key indicators on the size and composition of the informal economy for all ILO member States for which micro data were not available, with the exception of high-income countries in the Arab States region due to insufficient coverage from existing data (see list of key indicators in table A3.2 and country data in tables B.1 to B.8 under Appendix B). Estimated values are used for global and regional estimates only. The statistical Appendix B on key indicators on the size and composition of the informal economy focuses on countries with available datasets allowing for the application of the harmonized definition of informality. For countries with missing data (representing less than 10 per cent of global employment), estimated total value for a given indicator equals the mean value for this indicator in a given subregion<sup>2</sup> which is further divided into main income groups.<sup>3</sup> The same method was applied to estimate sub-indicators by status in employment, sex, sector or rural/urban location. Estimated mean values for ratios between sub-components and the total value of an indicator were then applied to the estimated total value for the indicator considered.

### Table A 3.2 Set of key indicators on the size and composition of the informal economy for which missing values have been estimated

Indicator	Disaggregation
About the size of the informal economy	
<ul> <li>Share of informal employment in total employment</li> </ul>	<ul><li>Sex</li><li>Urban and rural location</li></ul>
<ul> <li>Share of non-agricultural employment in non-agricultural employ- ment</li> </ul>	<ul><li>Sex</li><li>Urban and rural location</li></ul>
<ul> <li>Share of employment in the informal sector, in the formal sector, in households</li> </ul>	<ul><li>Sex</li><li>Urban and rural location</li></ul>
About the composition of the informal economy	
<ul> <li>Share of informal employment in the informal sector, in the formal sector, in households</li> </ul>	<ul> <li>Sex and status in employment</li> <li>Urban/rural location &amp; employment status</li> </ul>
<ul> <li>Share of informal employment in agriculture, industry, services</li> </ul>	<ul><li>Sex and employment status</li><li>Urban/rural location &amp; employment status</li></ul>
<ul> <li>Share of informal employment in total employment by status in employment (including and excluding agriculture)</li> </ul>	<ul><li>Sex</li><li>Urban and rural location</li></ul>
<ul> <li>Comparison of distributions of informal employment and of formal employment by status in employment</li> </ul>	<ul><li>Sex</li><li>Urban and rural location</li></ul>

# Global and regional estimates and year of reference

For each indicator, global and regional estimates of proportions (e.g. share of informal employment in total employment or share of employees in informal employment as a percentage of total employees) result from the weighted average of national proportions for the latest year available (as indicated in Appendix A.2). Those regional and global estimates are weighted by the denominator of the considered indicator using 2016 data from ILO's Trends Econometric Models<sup>4</sup> for total employment by sex, status in

<sup>&</sup>lt;sup>2</sup> Northern Africa, sub-Saharan Africa, Latin America and the Caribbean, Northern America, Arab States, Eastern Asia, South-Eastern Asia and the Pacific, Southern Asia, Northern, Southern and Western Europe, Eastern Europe, Central and Western Asia.

<sup>&</sup>lt;sup>3</sup> Low-income, lower-middle-income, upper-middle-income and high-income countries.

<sup>&</sup>lt;sup>4</sup> See ILO estimates and projections methodological note, available at: http://www.ilo.org/ilostat-files/Documents/ILO%20estimates%20and%20projections%20methodological%20note.pdf

employment<sup>5</sup> and sector<sup>6</sup> as relevant. When absolute numbers are presented in this report, they refer to 2016 by multiplying the estimated regional or global estimate by absolute numbers for 2016 from ILO's Trends Econometric Models as appropriate according to the denominator.

## Additional set of indicators

To assess some of the possible drivers of informality associated with personal features as well as decent work deficits faced by workers in informal employment, additional indicators on issues related to working conditions (working time, temporary or permananent nature of the employment relationship, timerelated under-employment), poverty or key socio-demographic features (such as age or level of education) have been processed systematically to cover a sufficient number of countries for global and regional estimates. No missing values are estimated for those additional indicators. For those additional indicators, the number of countries with available data and the proportion of global employment represented are indicated in the text or in notes under tables and figures. Global and regional estimates are calculated and presented only if countries with available data represent at least 75 per cent of the employed population. In practice, the lowest share of employment represented is for the analysis of informality and the temporary or permanent nature of employment where employment in countries with data available represent 78 per cent of global employment. Following the above principle, no global estimates are presented for the joint analysis of informality and poverty.

## Analysis of the availability of the questions in national survey questionnaires

The set of criteria used for the operational definition of employment in the informal sector and of informal employment supposes the inclusion of the relevant questions in national survey questionnaires, which is obviously not the case in all countries and also not the case in the 112 countries for which those indicators have been computed. The main gap concerns the criteria used to estimate employment in the informal sector (more likely to be asked in enterprise based surveys than in labour force surveys).

Concerning employment in the informal sector, the information about the institutional sector was available in all of the 112 countries. The main (and recommended) criteria related to the registration of the economic unit under national legislation is available in nearly half of the countries and whether the economic unit maintains a set of accounts in just above 26 per cent of the countries. Availability of either one or the other of those two criteria concerns 52.2 per cent of the countries (mainly developing and emerging countries). This means that the alternative path to estimate employment in the informal sector (contribution to social security or payment of income taxes by the employer, size and location of the activity) has been applied in 48 per cent of all countries for which data are presented in this report (see box 2, Chapter 1).

More countries provide the necessary information to estimate informal employment among employees. The totality of the countries covered here included at least one of the three recommended criteria (employer's contribution to social security, annual paid leave and paid sick leave). The question on the contribution to social security was available in all countries with two exceptions (Namibia and Timor-Leste), where the social security system is primarily based on non-contributory tax financed type of schemes. In those two countries, the information about annual paid leave and paid sick leave allowed a proper estimate of informal employment. The information on entitlement or benefit from annual paid leave is available in more than half of the countries (mainly developing and emerging countries) and the information about paid sick leave in less than 40 per cent of the 112 countries considered.

<sup>&</sup>lt;sup>5</sup> Employees, employers, own-account workers and contributing family workers.

<sup>&</sup>lt;sup>6</sup> Agriculture, industry and services.

# **Appendix B**

# Statistical appendix: Country data Key indicators on the size and composition of the informal economy

As mentioned previously, indicators presented in this statistical appendix follow for the first time a common definition of informal employment and of employment in the informal sector to enhance comparability between countries and regions. This may result in ILO country estimates that differ from national estimates. All indicators refer to informality in the main job.

Appendix B Key	r indicators on the size and composition of the informal economy: Country data
Table B.1	Share of informal employment in total employment and in non-agricultural employment by sex Table B.1 provides not only an indication about the size of the informal economy but also on its composition. The total share of informal employment is divided into informal employment in the informal sector (informal economic units), in the formal sector (including government units and non-profit institutions) and in households (as paid domestic workers or own-account workers producing goods exclusively for own final use by their households). Table B.1 includes notably SDG indicator 8.3.1 on the proportion of informal employment in non-agriculture employment, by sex.
Table B.2	Share of informal employment in total employment and in non-agricultural employment by urban or rural location Table B.2 presents the same set of indicators on the size and composition of the informal economy but focusing on the differ- ences between urban and rural areas.
Table B.3 Table B.4	Share of informal employment in total employment by status in employment: including agriculture in table B.3 and excluding agriculture in table B.4. The main objective of indicators presented in tables B.3 and B.4 is to help identify the workers who are most exposed to informal employment. The share of informal employment by employment status is of primary importance when thinking in terms of policy responses. Together with indicators about the distribution of informal employment, those indicators contribute to quantify the proportion of the employed population concerned by policies to formalize enterprises and/or to formalize jobs. For each country, tables B.3 and B.4 present i) the share of employees holding informal jobs either in the informal sector, in the formal sector or in households; ii) the share of own-account workers and employers operating in informal economic units; and iii) the share of contributing family workers (all in informal employment) but possibly working in the informal or formal sector or in households.
Table B.5	Distribution of workers in informal employment and in formal employment by employment status and sex (including agriculture) Table B.5 considers the composition by status in employment of informal employment compared to formal employment. The main objective is to identify status in employment most represented among workers in informal employment and workers in formal employment respectively.
Table B.6	Share of informal employment in agriculture, industry and services by sex Table B.6 focuses on the three aggregate sector categories (based on the International Standard Industrial Classification of All Economic Activities, ISIC). The assessment of the informal nature of jobs and economic units in the agricultural sector is based on the same set of criteria as those used in the other sectors.
Table B.7	<b>Distribution of workers in informal employment and in formal employment by broad sector of activity</b> Table B.7 is complementary to table B.6. It considers the composition by main sector of informal employment compared to formal employment. The main objective is to identify sectors most represented among workers in informal employment and workers in formal employment respectively. Workers in a given sector can show a higher risk of being informal (in table B.6), but do not necessarily represent the majority among those in informal employment (table B.7).
Table B.8	Share of employment in the informal sector, in the formal sector and in households by sex Table B.8 focuses on the characteristics of the economic units. The main difference with informal employment in the informal sector presented in table B.1 lies in the possible existence (even if it should be minor) of employees in formal employment in the informal sector.

Table B.1 – Share of informal employment in total employment and in non-agricultural employment by sex

-

			LS.	Share of informal employment in	formal ei	mploym		otal employment (%)	/ment (%					Share o	Share of non-agricultural informal employment in non-agriculture ( $\%$ ), SDG	ricultura	informa	ıl emplo	yment i	n non-ag	riculture	(%), SI	<b>JG 8.3.1</b>	
		Total	tal			W	Men			Women	nen			2	Total			Men	E			Mo	Women	
	letoT	In the informal sector	sector In the formal	spiouəsnou ui	Total	In the informal sector	In the formal sector	splodəsuod ni	letoT	In the informal sector	in the formal sector	spiouəsnou ui	letoT	In the informal sector	In the formal sector	splodəsuod ni	lstoT	In the informal sector	sector In the formal	splodesuod ni	letoT	In the informal sector	In the formal sector	splodəsuod ni
Africa																								
Northern Africa																								
Egypt	63.3	58.5	4.2	0.5	64.8	60.1	4.3	0.4	57.3	52.5	3.9	0.9	49.8	43.4	5.7	0.7	54.4	48.3	5.5	0.5	26.3	18.3	9.9	1.5
Morocco	79.9	58.7	19.8	1.4	81.1	61.6	19.1	0.4	73.9	43.8	23.5	9.9	75.6	53.3	20.4	1.9	77.2	56.6	20.1	0.5	67.6	36.8	22.0	8.8
Tunisia	58.8	53.3	5.1	0.4	60.0	55.8	4.2	0.0	54.7	44.7	8.2	1.8	53.5	47.6	5.5	0.5	55.5	51.2	4.4	0.0	46.9	35.6	9.1	2.2
Sub-Saharan Africa																								
Central Africa																								
Angola	94.1	77.1	16.5	0.5	92.8	68.6	24.1	0.1	95.4	86.1	8.5	0.8	89.5	59.2	29.5	0.9	88.2	49.0	39.0	0.2	91.2	72.8	16.7	1.7
Cameroon	90.9	83.7	6.4	0.8	86.6	77.1	9.2	0.3	95.2	90.4	3.6	1.2	79.0	62.5	14.7	1.8	73.6	54.7	18.2	0.7	86.7	73.5	9.7	3.5
Chad	94.0	92.0	2.0	0.0	91.7	89.4	2.3	0.0	97.3	95.7	1.6	0.0	75.8	68.0	7.8	0.0	68.5	60.2	8.3	0.0	87.9	80.8	7.1	0.0
Congo	85.3	56.9	25.9	2.5	88.3	59.5	26.4	2.5	82.9	54.9	25.5	2.5	85.0	55.6	26.8	2.6	88.5	58.9	27.1	2.6	82.1	52.9	26.6	2.7
Congo, Democratic Republic of the	91.9	88.2	3.7	0.0	86.6	81.2	5.3	0.0	97.2	95.0	2.2	0.0	76.2	67.0	9.3	0.0	68.0	55.6	12.4	0.0	90.0	86.1	4.0	0.0
Eastern Africa																								
Comoros	89.2	84.6	4.6	0.0	86.2	80.1	6.1	0.0	93.8	91.3	2.5	0.0	78.5	69.5	9.0	0.0	74.3	63.4	10.9	0.0	86.4	81.0	5.4	0.0
Madagascar	93.6	83.6	4.1	5.9	91.9	82.9	5.1	4.0	95.2	84.2	3.1	7.9	79.8	47.3	12.4	20.1	74.5	44.7	16.3	13.6	85.0	49.8	8.7	26.4
Malawi	83.0	74.1	4.6	4.3	83.7	74.7	4.8	4.3	82.4	73.7	4.5	4.3	67.2	49.4	10.8	7.0	67.1	49.0	10.8	7.3	67.3	49.7	10.9	6.8
Rwanda	94.3	90.9	1.5	1.9	92.0	88.0	2.1	1.9	96.2	93.4	0.9	1.9	78.6	67.0	4.6	7.1	78.9	69.3	4.7	5.0	78.0	62.5	4.3	11.2
Tanzania, United Republic of	90.6	83.5	4.9	2.3	88.2	79.9	6.7	1.6	93.1	87.2	3.0	2.9	69.3	48.1	15.2	6.0	64.7	42.1	18.9	3.8	74.9	55.5	10.7	8.7
Uganda	93.7	87.6	6.2	0.0	92.0	83.9	8.1	0.0	95.6	91.6	4.0	0.0	83.4	67.1	16.4	0.0	81.5	62.5	19.0	0.0	86.3	73.8	12.5	0.0
Zambia	87.9	79.0	2.9	6.1	81.6	71.2	4.4	6.0	93.6	86.0	1.4	6.2	71.5	52.3	6.3	12.9	65.5	48.1	6.7	9 F	80 1	58 G	0 0	17 G

Iotal	Share of informal employment in total employment (%)		S	hare of no	Share of non-agricultural informal employment in non-agriculture (%), SDG 8.3.1	ral inform	al employ	/ment in	non-agri	culture (%	6), SDG (	3.3.1	
Image: Construct of the formal construction of the f		Women		Total			Men				Women		
Ithern Africa         Ithern Africa         Botswana       65.6       52.1       5.2       8.3       63.5       51.2       4.2       8.2         Botswana       65.6       52.1       5.2       8.3       63.5       51.2       4.2       8.2         Namibia       67.0       46.8       5.6       14.6       66.1       50.4       4.7       11.0         South Africa       34.0       21.8       4.8       7.4       32.5       5.2       3.2       4.1       11.0         South Africa       94.5       90.6       5.2       1.6       90.7       87.1       4.2       0.3       20         Benin       94.5       91.6       65.5       9.6       19.5       91.3       65.3       97       16.2         Burkina Faso       94.6       65.5       9.6       19.3       65.3       97       16.2       63       64       97       85.3       66       0.4         Burkina Faso       94.6       65.5       7.1       7.1       70.4       53       66       0.4         Gambia       65.5       7.1       7.1       70.4       53       66       0.4 </th <th>In the informal sector In the formal sector In households Total In the informal</th> <th>In households sector sector</th> <th>lstoT Jerntotni edt al</th> <th>In the informal sector In the formal</th> <th>lu µonseµolds sector</th> <th>letoT</th> <th>In the informal sector</th> <th>in the formal sector</th> <th>splodesuod nl</th> <th>Total In the informal</th> <th>In the formal</th> <th>sector</th> <th>spjoyasnoy uj</th>	In the informal sector In the formal sector In households Total In the informal	In households sector sector	lstoT Jerntotni edt al	In the informal sector In the formal	lu µonseµolds sector	letoT	In the informal sector	in the formal sector	splodesuod nl	Total In the informal	In the formal	sector	spjoyasnoy uj
Botswana         65.6         5.1         5.2         8.3         63.5         51.2         4.2         8.2           Namibia         67.0         46.8         5.6         14.6         66.1         50.4         4.7         11.0           South Africa         34.0         21.8         4.8         7.4         32.5         3.2         4.1           South Africa         34.0         21.8         4.8         7.4         32.5         3.2         4.1           South Africa         34.0         51.8         4.8         1.1         91.6         87.1         4.7         11.0           Stern Africa         34.5         90.6         2.8         1.1         91.6         87.1         4.2         8.2           Benin         94.5         90.6         8.5         9.1         4.9         16.2         0.4           Cote of Voice         92.8         86.0         5.2         1.4         14.8         8.2         0.4           Benin         76.7         62.5         1.1         7.1         7.1         7.6         9.2         0.4           Cote of Voice         92.3         74.1         14.4         80.8         8.3         3.3<													
Namibia         67.0         46.8         5.6         14.6         66.1         50.4         4.7         11.0           South Africa         34.0         21.8         4.8         7.4         32.5         3.2         4.1           South Africa         34.0         21.8         4.8         7.4         32.5         5.5         3.2         4.1           Stern Africa         94.5         50.6         2.8         1.1         91.6         87.1         4.2         0.3           Benin         94.5         50.6         2.8         1.1         91.6         87.1         4.2         0.3           Burkina Faso         94.6         65.5         9.6         19.5         91.3         65.3         9.7         16.2           Coto Verde         46.5         71.1         71.1         70.4         53.4         8.8         2.0           Coto Verde         90.1         84.8         4.9         0.5         86.0         79.7         6.6         0.4           Gambia         76.7         61.1         71.4         90.8         88.2         61.1         16.2         61.1         16.2         61.4         16.2         51.4         16.2	51.2 4.2 8.2	6.3 8.4	55.1 4	41.2 6	6.8 7.2	50.0	39.6	5.8	4.6	60.5 4	42.8	7.9	9.8
South Africa         340         21.8         4.8         7.4         32.5         25.2         3.2         4.1           stern Africa         94.5         90.6         2.8         1.1         91.6         87.1         4.2         0.3           Burkina Faso         94.6         65.5         96         19.5         91.3         65.3         9.7         16.2           Burkina Faso         94.6         65.5         9.6         19.5         91.3         65.3         9.7         16.2           Burkina Faso         94.6         53.1.9         6.0         8.5         4.91         4.91         6.97         83.7         6.6         0.4           Cobe drivoire         92.8         86.0         5.2         1.1         70.4         5.3         8.2         0.4           Cobe drivoire         92.8         86.0         5.2         1.4         31.5         70.6         90.2         0.4           Ciberia         86.8         79.1         81.5         70.6         90.2         16.2         5.1         16.2           Mali         92.9         80.4         5.9         64.5         81.5         70.6         90.2         20.5         16.5	50.4 4.7 11.0	6.5 18.5	61.1 4	46.3 6.	.5 8.3	60.3	49.6	5.7	5.1	61.9 4	42.8	.4 1	11.6
stem Africa         Benin       94.5       90.6       2.8       1.1       91.6       87.1       4.2       0.3         Burkina Faso       94.6       55       9.6       19.5       91.3       65.3       97       16.2         Burkina Faso       94.6       65.5       9.6       19.5       91.3       65.3       97       16.2         Cabo Verde       46.5       31.9       6.0       8.5       16.1       71.1       71.1       70.4       53.4       8.8       8.2         Cofte d'Ivoire       92.8       86.0       5.2       1.6       90.7       83.7       6.6       0.4         Gambia       76.7       62.5       7.1       7.1       7.1       70.4       53.4       8.8       8.2         Giberia       86.8       79.1       6.0       17       81.5       70.6       90.4       8.3	25.2 3.2	6.8 11.7	34.0 2	21.4 4	4.9 7.7	32.4	24.9	3.3	4.2	35.9 1	17.0	6.8 1	12.2
Benin         94.5         90.6         2.8         1.1         91.6         87.1         4.2         0.3           Burkina Faso         94.6         65.5         9.6         19.5         91.3         65.3         9.7         16.2           Burkina Faso         94.6         65.5         9.6         19.5         91.3         65.3         9.7         16.2           Cabo Verde         46.5         31.9         6.0         8.5         1.1         7.1         7.1         8.3         6.6         0.4           Cabo Verde         46.5         7.1         7.1         7.0         83.7         6.6         0.4           Gambia         76.7         62.5         7.1         7.1         70.4         53.4         8.8         8.2           Gambia         76.7         62.5         7.1         7.1         70.4         53.4         8.8         8.2           Gambia         90.1         80.4         90.5         86.0         70.6         90.2         0.4           Mali         92.7         74.2         4.1         14.4         90.8         68.5         6.1         16.2         5.1           Miger         91.3         7	]									]			
Burkina Faso         94.6         65.5         9.6         19.5         91.3         65.3         9.7         16.2           Cabo Verde         46.5         31.9         6.0         8.5         49.1         41.8         5.3         2.0           Cabo Verde         46.5         31.9         6.0         8.5         49.1         41.8         5.3         2.0           Cabo Verde         92.8         86.0         5.2         1.6         90.7         83.7         6.6         0.4           Caba Verde         92.1         62.5         7.1         7.1         70.4         53.4         8.8         8.2           Gambia         90.1         84.8         4.9         0.5         86.0         79.7         5.9         0.4           Blai         92.7         74.2         4.1         14.4         90.8         85.5         6.1         16.2           Niger         91.3         77.3         96         44.5         87.4         69.8         3.7         8.3           Niger         91.3         77.3         96         44.5         89.5         68.1         16.2         5.5           Nigeria         92.5         88.9	87.1 4.2	1.5 1.9	90.6	83.7 4	4.9 2.0	83.4	74.3	8.5	0.7	95.8 9	90.5	2.4	2.9
Cabo Verde         46.5         31.9         6.0         8.5         49.1         41.8         5.3         2.0           Côte d'Ivoire         92.8         86.0         5.2         1.6         90.7         83.7         6.6         0.4           Gambia         76.7         62.5         7.1         7.1         70.4         53.4         8.8         8.2           Gambia         76.7         62.5         7.1         7.1         70.4         53.4         8.8         8.2           Ghana         90.1         84.8         4.9         0.5         86.0         7.97         5.9         0.4           Liberia         86.8         79.1         6.0         1.7         81.5         70.6         90.2         2.0           Mali         92.3         74.2         4.1         14.4         90.8         68.5         6.1         16.2           Niger         91.3         77.3         9.6         4.5         87.4         68.8         3.7         8.3           Niger         91.2         70.9         14.0         6.2         88.9         77.2         7.5         5.6           Niger         92.8         64.4         90.3	65.3 9.7	9.5 22.5	88.4 6	67.9 14	14.5 6.0	82.7	58.7	19.7	4.3	93.8 7	76.6	9.5	7.7
Côte d'Ivoire         928         86.0         5.2         1.6         90.7         83.7         6.6         0.4           Gambia         76.7         62.5         7.1         7.1         70.4         53.4         8.8         8.2           Gambia         76.7         62.5         7.1         7.1         70.4         53.4         8.8         8.2           Ghana         90.1         84.8         4.9         0.5         86.0         79.7         5.9         0.4           Liberia         86.8         79.1         6.0         1.7         81.5         70.6         90         2.0           Mali         92.7         74.2         4.1         14.4         90.8         68.5         6.1         16.2         5.1           Nigeria         92.9         80.4         2.9         96.8         73.8         3.7         8.3           Senegal         91.2         70.9         14.0         6.2         89.5         68.8         18.2         2.5           Senegal         91.2         70.9         90.3         71.2         7.5         7.5         5.6           Sitra Leone         92.8         10.1         17.7         7.5<	41.8 5.3 2.0	6.7 15.3	46.8 2	29.3 7	7.2 10.2	48.1	38.8	6.8	2.6	45.5 2	20.7	7.6 1	17.2
Gambia $76.7$ $62.5$ $7.1$ $7.1$ $70.4$ $53.4$ $8.8$ $8.2$ Ghana $90.1$ $84.8$ $4.9$ $0.5$ $86.0$ $79.7$ $5.9$ $0.4$ Liberia $86.8$ $79.1$ $6.0$ $1.7$ $81.5$ $70.6$ $90.$ $2.0$ Mali $92.7$ $74.2$ $4.1$ $14.4$ $90.8$ $88.5$ $6.1$ $16.2$ Mali $92.7$ $74.2$ $4.1$ $14.4$ $90.8$ $88.5$ $6.1$ $16.2$ Niger $91.3$ $77.3$ $9.6$ $4.5$ $87.4$ $69.8$ $12.5$ $5.1$ Nigeria $92.9$ $80.4$ $2.9$ $9.6$ $90.8$ $78.8$ $3.7$ $8.3$ Senegal $91.2$ $70.9$ $14.0$ $6.2$ $89.5$ $68.8$ $18.2$ $2.5$ Sitra Leone $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Gogo $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Interfactant Leone $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Gogo $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Interfactant Leone $92.8$ $80.6$ $5.8$ $6.1$ $10.7$ $7.6$ $7.6$ Gogo $62.3$ $37.4$ $4.9$ $6.9$ $30.4$ $5.2$ $1.0$ Interfactant Leone $56.3$ $30.4$ $5.2$	83.7 6.6 0.4 95.	3.1 3.3	87.7 7	75.8 9	9.0 2.9	82.4	68.7	12.8	0.8	93.8 8	83.9	4.7	5.3
Ghana         90.1         84.8         4.9         0.5         86.0         79.7         5.9         0.4           Liberia         86.8         79.1         6.0         1.7         81.5         70.6         9.0         2.0           Mali         92.7         74.2         4.1         14.4         90.8         68.5         6.1         16.2           Mali         92.7         74.2         4.1         14.4         90.8         68.5         6.1         16.2           Miger         91.3         77.3         9.6         4.5         87.4         69.8         12.5         5.1           Nigeria         92.9         96.4         2.9         96.6         70.8         3.7         8.3           Senegal         91.2         70.9         14.0         6.2         89.5         68.8         18.2         2.5           Sierra Leone         92.8         80.6         5.8         6.1         1.0         1.8         2.5         5.6           Sierra Leone         92.8         80.6         5.8         6.4         90.3         77.2         7.5         5.6           Gogo         92.8         6.4         90.3         77.2 <td>53.4 8.8 8.2</td> <td>4.9 5.8</td> <td>68.4</td> <td>52.1 9</td> <td>9.2 7.2</td> <td>62.4</td> <td>43.8</td> <td>11.1</td> <td>7.5</td> <td>77.7 6</td> <td>64.8 (</td> <td>6.2</td> <td>6.7</td>	53.4 8.8 8.2	4.9 5.8	68.4	52.1 9	9.2 7.2	62.4	43.8	11.1	7.5	77.7 6	64.8 (	6.2	6.7
Liberia         86.8         79.1         6.0         1.7         81.5         70.6         9.0         2.0           Mali         92.7         74.2         4.1         14.4         90.8         68.5         6.1         16.2           Mali         92.7         74.2         4.1         14.4         90.8         68.5         6.1         16.2           Nigeria         91.3         77.3         9.6         4.5         87.4         69.8         13.5         5.1           Nigeria         91.2         80.4         2.9         96.6         89.5         68.8         13.2         5.5         5.1           Stengal         91.2         70.9         14.0         6.2         89.5         68.8         18.2         2.5         5.6           Stenzalcone         92.8         10.0         1.7         92.9         90.1         1.0         1.8         7.2         7.5         5.6           Stenzal Loone         92.8         80.6         5.8         6.4         90.3         77.2         7.5         5.6           Stenzal Loone         92.8         80.6         5.8         6.4         90.3         77.2         7.5         5.6 <td>79.7 5.9 0.4</td> <td>3.9 0.7</td> <td>82.6 7</td> <td>73.2 8</td> <td>8.6 0.9</td> <td>73.7</td> <td>62.0</td> <td>11.1</td> <td>0.7</td> <td>89.9 8</td> <td>82.3 (</td> <td>6.5</td> <td>1.1</td>	79.7 5.9 0.4	3.9 0.7	82.6 7	73.2 8	8.6 0.9	73.7	62.0	11.1	0.7	89.9 8	82.3 (	6.5	1.1
Mali $92.7$ $74.2$ $4.1$ $14.4$ $90.8$ $68.5$ $6.1$ $16.2$ Niger $91.3$ $77.3$ $9.6$ $4.5$ $87.4$ $69.8$ $12.5$ $5.1$ Nigeria $92.9$ $80.4$ $2.9$ $9.6$ $90.8$ $78.8$ $3.7$ $8.3$ Senegal $91.2$ $70.9$ $14.0$ $6.2$ $89.5$ $68.8$ $18.2$ $2.5$ Senegal $91.2$ $70.9$ $14.0$ $6.2$ $89.5$ $68.8$ $18.2$ $2.5$ Sierra Leone $92.5$ $89.8$ $1.0$ $1.7$ $92.9$ $90.1$ $1.0$ $1.8$ Cogo $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Cogo $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Cogo $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Cogo $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Cogo $92.8$ $80.6$ $5.8$ $6.4$ $90.3$ $77.2$ $7.5$ $5.6$ Inerica and the $7.6$ $7.6$ $8.9$ $7.6$ $8.9$ $6.3$ $7.6$ $7.6$ Caribbean $56.3$ $37.4$ $4.9$ $6.9$ $56.3$ $30.4$ $5.2$ $1.0$ Caribbean $89.1$ $27.4$ $4.9$ $6.9$ $36.6$ $30.4$ $5.2$ $1.0$ Caribbean $39.1$ $27.4$ $4.9$ </td <td>70.6 9.0 2.0</td> <td>3.0 1.4</td> <td>77.5 6</td> <td>64.6 1(</td> <td>10.0 2.9</td> <td>68.6</td> <td>49.9</td> <td>15.3</td> <td>3.5</td> <td>86.5 7</td> <td>79.3</td> <td>4.8</td> <td>2.4</td>	70.6 9.0 2.0	3.0 1.4	77.5 6	64.6 1(	10.0 2.9	68.6	49.9	15.3	3.5	86.5 7	79.3	4.8	2.4
Niger     91.3     77.3     9.6     4.5     87.4     69.8     12.5     5.1       Nigeria     92.9     80.4     2.9     9.6     90.8     78.8     3.7     8.3       Senegal     91.2     70.9     14.0     6.2     89.5     68.8     18.2     2.5       Senegal     91.2     70.9     14.0     6.2     89.5     68.8     18.2     2.5       Sierra Leone     92.5     89.8     1.0     1.7     92.9     90.1     1.0     1.8       Iogo     92.8     80.6     5.8     6.4     90.3     77.2     7.5     5.6       Iogo     92.8     80.6     5.8     6.4     90.3     77.2     7.5     5.6       Iogo     92.8     80.6     5.8     6.4     90.3     77.2     7.5     5.6       Iogo     56.3     37.4     11.8     7.0     56.3     46.7     8.9     0.7       Inerica and the Caribbean     56.3     37.4     4.9     50.3     30.4     5.2     1.0       Caribbean     56.3     30.4     5.2     1.0     5.2     1.0	68.5 6.1 16.2	1.7 12.4	86.4 7	70.5 10	10.6 5.3	82.8	61.6	15.8	5.4	90.7 8	81.0 4	4.6	5.2
Nigeria     92.9     80.4     2.9     9.6     90.8     78.8     3.7     8.3       Senegal     91.2     70.9     14.0     6.2     89.5     68.8     18.2     2.5       Senegal     91.2     70.9     14.0     6.2     89.5     68.8     18.2     2.5       Sierra Leone     92.5     80.6     5.8     1.0     1.7     92.9     90.1     1.0     1.8       Ogo     92.8     80.6     5.8     6.4     90.3     77.2     7.5     5.6       Inerica and the Caribbean      5.8     5.4     90.3     77.2     7.5     5.6       Caribbean      56.3     37.4     11.8     7.0     56.3     46.7     8.9     0.7       Republic     56.3     37.4     4.9     6.9     30.6     30.4     5.2     1.0	69.8 12.5	6.2 3.7	85.1 7	72.6 10	10.3 2.2	74.7	56.8	16.2	1.6	94.2 8	86.3	5.2	2.7
Senegal         91.2         70.9         14.0         6.2         89.5         68.8         18.2         2.5           Sierra Leone         92.5         89.8         1.0         1.7         92.9         90.1         1.0         1.8           Iogo         92.8         80.6         5.8         6.4         90.3         77.2         7.5         5.6           Iogo         92.8         80.6         5.8         6.4         90.3         77.2         7.5         5.6           Inerica and the Caribbean         Antican	78.8 3.7 8.3	2.0 11.0	89.0	82.9 4	4.2 1.8	83.5	75.9	6.2	1.5	93.2 8	88.5	2.7	2.1
Sierra Leone     92.5     89.8     1.0     1.7     92.9     90.1     1.0     1.8       logo     92.8     80.6     5.8     6.4     90.3     77.2     7.5     5.6       merica and the Caribbean        90.3     77.2     7.5     5.6       Caribbean         90.3     77.2     7.5     5.6       Dominican      56.3     37.4     11.8     7.0     56.3     46.7     8.9     0.7       Republic      56.3     37.4     11.8     7.0     56.3     46.7     8.9     0.7       Costa Rica       39.1     27.4     4.9     6.9     36.6     30.4     5.2     1.0	68.8 18.2 2.5	7.9 11.8	87.0	59.1 19	19.1 8.7	83.9	55.2	25.9	2.8	91.3 6	64.5 1	10.1 1	16.6
logo         92.8         80.6         5.8         6.4         90.3         77.2         7.5         5.6           Imerica and the Caribbean         Earlibbean         Ear	90.1 1.0 1.8	1.0 1.5	86.0	80.8 2	2.6 2.6	86.4	80.9	2.6	2.9	85.7 8	80.8	2.6	2.3
Imerica and the Caribbean         Caribbean           Caribbean         56.3         37.4         11.8         7.0         56.3         46.7         8.9         0.7           Republic         56.3         37.4         11.8         7.0         56.3         46.7         8.9         0.7           Republic         56.3         37.4         11.8         7.0         56.3         46.7         8.9         0.7           Republic         56.3         37.4         4.9         6.9         36.6         30.4         5.2         1.0           Costa Rica         39.1         27.4         4.9         6.9         36.6         30.4         5.2         1.0	77.2 7.5	4.2 7.2	90.0	73.6 6	6.7 9.8	85.8	67.4	9.4	9.0	93.6 7	79.0	4.3 1	10.4
7.4 11.8 7.0 56.3 46.7 8.9 0.7 7.4 4.9 6.9 36.6 30.4 5.2 1.0													
56.3     37.4     11.8     7.0     56.3     46.7     8.9     0.7       39.1     27.4     4.9     6.9     36.6     30.4     5.2     1.0													
56.3         37.4         11.8         7.0         56.3         46.7         8.9         0.7           39.1         27.4         4.9         6.9         36.6         30.4         5.2         1.0													
39.1 27.4 4.9 6.9 36.6 30.4 5.2 1.0	46.7 8.9 0.7 56	16.5 17.0	52.6	31.3 13	13.1 8.1	49.8	38.4	10.4	6.0	56.2 2	21.9 1	16.7 1	17.6
39.1 27.4 4.9 6.9 36.6 30.4 5.2 1.0													
	36.6 30.4 5.2 1.0 42.9 22.7	4.5 15.7	37.4 2	25.5 4	4.2 7.8	33.5	28.0	4.3	1.3	42.5 2	22.1 ,	4.0 1	16.3
El Salvador 69.6 53.9 11.2 4.5 68.4 54.0 13.9 0.4 71.3	54.0 13.9 0.4	7.5 10.1	63.1 4	48.1 9	9.4 5.6	55.9	42.7	12.7	0.6	70.2 5	53.5 (	6.1 1	10.6
Guatemala 79.7 63.3 8.8 7.7 79.1 64.5 9.2 5.4 80.8	64.5 9.2	8.0 12.0	72.7	54.5 12	12.0 6.1	67.9	52.1	14.7	1.2	78.7 5	57.5	8.7 1	12.5
Honduras 79.9 70.0 1.8 8.1 81.4 72.5 1.1 7.8 77.6	72.5 1.1	3.0 8.7	72.7	65.7 2	2.6 4.3	69.9	67.3	1.9	0.7	75.6 6	64.1	3.3	8.1

			45	Share of informal employment in	ormal er	nnlovm	-	otal emnlovment (%)	ment (%					hare o	100-201	Stare of non-acricultural informal employment in non-acriculture (%) SN6 8 3 1	informa	emnlo	vment in	ทกท-ลฮา	iculture	US (%)	G 8.3.1	
		Total				Men				Women	nen			Total				Men	_	0		Women	nen	
	Total	In the informal sector	In the formal sector	spiodəsuod ni	Total	In the informal sector	In the formal sector	spiodəsuod ni	Total	In the informal sector	In the formal sector	spiodəsuod ni	lstoT	In the informal sector	sector sector	spiodəsuod ni	letoT	In the informal sector	In the formal sector	spiodəsuod ni	letoT	In the informal sector	In the formal sector	splodəsuod nl
Mexico	53.4	29.1	19.8	4.5	50.1	27.9	21.6	9.0	58.8	31.0	17.0	10.9	53.2	33.1	14.9	5.2	49.6	33.8	15.1	0.7	58.2	32.2	14.7	11.3
Nicaragua	77.4	71.5	1.4	4.4	80.1	77.8	1.0	1.3	73.1	61.3	2.2	9.6	68.6	60.1	2.1	6.5	66.5	62.4	1.8	2.4	70.7	57.8	2.4	10.5
Panama	52.3	34.3	14.0	3.9	53.5	40.1	12.9	9.0	50.6	26.4	15.6	8.7	46.0	26.2	15.2	4.6	45.3	30.4	14.3	0.7	46.9	21.3	16.3	9.4
South America																								
Argentina	47.2	32.8	8.5	5.8	47.1	37.6	9.3	0.2	47.2	26.6	7.4	13.3	47.0	32.7	8.5	5.9	47.0	37.5	9.3	0.2	47.2	26.5	7.4	13.3
Bolivia, Plurinational State of	83.1	68.7	12.4	1.9	82.1	68.6	13.4	0.1	84.2	68.9	11.1	4.2	75.6	55.3	17.5	2.8	74.6	55.9	18.5	0.2	76.9	54.6	16.1	6.2
Brazil	46.0	30.3	11.9	3.9	46.1	36.9	8.8	0.4	46.0	21.5	16.1	8.4	42.5	26.4	11.8	4.3	41.1	31.4	9.2	0.5	44.1	20.3	14.9	8.9
Chile	40.5	32.2	6.3	2.0	39.8	33.9	5.6	0.2	41.7	29.7	7.3	4.6	39.3	30.2	6.8	2.2	37.9	31.4	6.3	0.3	41.1	28.6	7.6	4.9
Colombia	60.6	55.9	3.0	1.8	61.9	59.4	2.3	0.2	58.8	50.7	3.9	4.2	55.4	50.0	3.4	2.0	54.1	51.1	2.9	0.1	56.9	48.5	4.1	4.3
Ecuador	59.0	46.4	11.1	1.6	57.8	46.8	10.9	0.1	60.9	45.8	11.3	3.8	50.4	35.9	12.4	2.1	48.4	36.1	12.1	0.2	53.1	35.6	12.7	4.8
Paraguay	70.6	46.0	17.6	7.0	70.1	48.8	20.7	0.7	71.3	42.1	13.3	15.9	64.4	34.9	20.8	8.7	62.7	36.5	25.3	0.9	66.7	32.9	15.2	18.5
Peru	69.2	55.5	11.8	1.9	65.5	53.5	11.9	0.2	73.5	57.8	11.7	4.0	59.1	40.5	15.9	2.7	54.2	37.9	16.1	0.2	64.6	43.5	15.7	5.4
Uruguay	24.5	19.6	1.9	3.0	25.2	22.7	2.2	0.2	23.8	15.8	1.5	6.4	24.1	18.8	2.0	3.3	24.5	21.9	2.4	0.3	23.6	15.3	1.6	6.7
Venezuela, Bolivarian Republic of	39.7	26.3	6.3	7.2	39.5	29.0	5.8	4.7	40.0	22.3	7.0	10.7	38.3	24.2	6.4	7.7	37.2	26.0	5.9	5.3	39.6	21.8	6.9	10.9
Northern America																								
United States	18.6	16.6	1.5	0.5	19.4	18.1	0.7	9.0	17.8	14.8	2.5	0.5	18.3	16.3	1.5	0.5	18.8	17.6	0.7	9.0	17.7	14.8	2.5	0.5
Arab States																								
Iraq	66.9	59.7	7.2	0.0	66.69	62.2	7.7	0.0	49.0	44.8	4.2	0.1	66.7	59.5	7.2	0.0	69.7	62.0	7.7	0.0	48.3	44.1	4.1	0.1
Jordan	44.9	43.9	1.0	0.0	48.5	47.8	0.7	0.0	26.9	24.8	2.1	0.0	43.0	42.0	1.0	0.0	47.1	46.3	0.8	0.0	23.5	21.3	2.2	0.0
Occupied Palestinian Territory	64.3	31.2	30.8	2.3	63.4	32.9	29.9	0.5	68.3	24.0	34.5	9.8	60.1	25.8	34.2	0.1	60.3	28.1	32.2	0.0	59.1	14.5	44.4	0.2
Syrian Arab Republic	70.1	66.0	4.1	0.0	71.9	67.8	4.1	0.0	61.6	57.6	4.0	0.0	61.0	55.6	5.4	0.0	66.0	60.9	5.1	0.0	24.1	16.2	7.8	0.0
Yemen	77.8	68.5	9.0	0.3	77.5	68.2	8.9	0.3	82.9	72.6	9.7	0.5	69.0	56.3	12.2	0.5	69.2	56.9	11.9	0.5	64.6	44.8	18.6	1.1

		sbionasuon ni			0.0	0.1	0.0	4.8			3.4		12.5	2.3	10.1	0.3	0.9	3.2	1.4		23.0	6.1	1.6	7.5	3 G
G 8.3.1	nen	ln the formal sector			6.0	5.8	9.8	3.6			10.2		13.4	24.2	11.2	18.7	17.4	9.6	17.3		29.1	14.7	7.4	6.3	10.0
(%), SD	Women	In the informal sector			44.4	10.5	26.7	23.3			18.2		3.9	67.0	60.7	65.0	63.9	44.7	35.6		35.3	54.8	77.6	58.9	C UV
iculture		Total			50.4	16.3	36.5	31.7			31.9		29.8	93.5	81.9	84.0	82.3	57.4	54.3		87.4	75.5	86.6	72.7	Б. А. 1
non-agr		splonesuon ni			0.0	0.0	0.0	9.8			0.8		0.9	0.5	8.5	0.1	0.2	7.9	0.1		5.3	0.7	0.5	0.6	2 E
/ment in	_	in the formal sector			6.3	3.2	2.8	2.2			11.3		30.0	18.4	6.5	27.1	16.4	21.9	22.1		21.2	11.7	12.7	6.9	6 3
l emplo	Men	In the informal sector			49.2	13.2	21.0	27.3			23.0		4.4	67.6	64.1	46.7	65.7	22.6	38.9		53.7	66.3	64.1	63.1	53 F
informa		letoT			55.5	16.4	23.8	39.3			35.1		35.3	86.6	79.0	73.9	82.3	52.4	61.1		80.2	78.6	77.3	70.6	65.2
Share of non-agricultural informal employment in non-agriculture (%), SDG 8.3.1		splodesuod nl			0.0	0.0	0.0	7.4			1.8	1	6.0	1.4	9.1	0.2	0.5	6.4	0.7	1	9.7	1.6	0.9	1.3	2.8
non-agi	-	In the formal sector			6.2	4.3	5.6	2.9			10.9		22.7	21.2	8.3	23.2	16.9	18.1	19.8		23.1	12.2	11.0	6.8	10.2
share of	Total	In the informal sector			47.3	12.0	23.3	25.3			21.2		4.2	67.3	62.7	55.2	64.9	29.5	37.4		49.2	64.3	68.5	62.7	491
		letoT			53.5	16.3	28.8	35.6			33.8		32.9	89.8	80.2	78.5	82.3	54.0	57.9		82.0	78.1	80.4	70.8	62.2
L		sblodəsuod nl			0.0	0.1	0.0	3.8			5.2	1	12.1	1.7	13.6	0.1	0.5	1.7	4.9		36.7	2.5	0.3	2.1	2.4
	en	in the formal sector			5.9	5.8	9.6	2.7			9.9		13.2	16.6	8.1	5.1	16.8	5.2	9.5		14.5	5.9	1.7	1.8	9.2
_	Women	In the informal sector			45.6	12.5	30.3	43.4			18.8		3.8	77.2	65.4	90.4	71.3	69.6	60.9		42.4	81.6	92.6	88.2	55.0
ment (%		lstoT			51.5	18.4	39.9	49.9			33.9		29.0	95.5	87.1	95.5	88.6	76.6	75.2		93.5	90.0	97.5	92.1	66.6
total employment (%)		splodəsuon ni			0.0	0.0	0.0	7.4			2.8	1	0.9	0.4	11.4	0.0	0.1	5.0	2.6		22.6	0.4	0.2	0.4	1.8
	_	In the formal sector			6.2	3.2	2.7	1.6			10.6		29.0	12.6	4.4	8.5	11.9	14.0	13.4		13.0	6.7	5.3	4.7	7.2
ployme	Men	In the informal sector			50.2	15.7	23.2	47.7			23.3		4.2	77.8	68.9	83.1	71.6	50.3	61.2		51.6	80.7	85.2	74.6	63.5
ormal en		letoT			56.4	18.9	25.9	56.7			36.7		34.0	90.9	84.7	91.7	83.5	69.3	77.3		87.2	87.7	90.7	79.6	72.5
Share of informal employment in		splouəsnou ul			0.0	0.0	0.0	5.7			3.7		5.7	1.0	12.2	0.1	0.3	3.9	3.7		26.7	0.8	0.2	0.8	2.0
Sha	-	In the formal sector			6.1	4.3	5.5	2.1			10.4		22.3	14.5	5.8	6.8	14.0	11.0	11.5		13.5	6.5	3.4	4.0	7.9
	Total	In the informal sector			48.4	14.3	26.1	45.7	ific		21.7		4.0	77.5	67.5	86.7	71.5	56.9	61.0		48.9	80.9	90.7	77.6	60.6
		letoT			54.4	18.7	31.5	53.5	the Pac		35.7		31.9	93.1	85.6	93.6	85.7	71.8	76.2		89.0	88.2	94.3	82.4	70.4
			Asia and the Pacific	Eastern Asia	China	Japan	Korea, Republic of	Mongolia	South-Eastern Asia and the Pacific	Pacific Islands	Samoa	South-Eastern Asia	Brunei Darussalam	Cambodia	Indonesia	Lao People's Democratic Republic	Myanmar	Timor-Leste	Viet Nam	Southern Asia	Bangladesh	India	Nepal	Pakistan	Sri Lanka

			-S	Share of informal employment in	formal e	molovm		total employment (%)	ment (%					Share o	f non-ag	Share of non-agricultural informal employment in non-agriculture (%). SDG 8.3.1	informa	l emplo	vment ir	1 non-ag	riculture	(%). SI	06 8.3.1	
		Total	al			ž	Men			Women	nen			Total	al			Men	=			Mo	Women	
	letal	In the informal sector	sector sector	spiouəsnou ul	letal	In the informal sector	sector In the formal	spiouəsnou ui	lstoT	In the informal sector	in the formal sector	spiouəsnou ul	letoT	In the informal sector	sector sector	spiouəsuon ni	lstoT	In the informal sector	sector sector	spiouəsnou ui	Total	In the informal sector	In the formal sector	splodəsuod nl
<b>Europe and Central Asia</b>																								
<b>Central and Western Asia</b>	sia																							
Central Asia																								
Kyrgyzstan	48.6	37.2	9.6	1.8	50.6	40.9	7.9	1.8	45.3	31.4	12.2	1.7	42.1	32.7	7.1	2.3	47.4	38.2	6.7	2.5	34.9	25.0	7.7	2.2
Tajikistan	74.8	54.4	14.4	5.9	75.9	52.8	15.4	7.8	72.8	57.2	12.9	2.8	70.5	39.8	20.5	10.2	74.9	44.5	19.3	11.1	56.8	25.1	24.4	7.3
Western Asia																								
Armenia	52.1	39.2	1.4	11.6	52.8	44.2	1.2	7.5	51.4	33.8	1.6	16.1	26.5	24.0	2.1	0.4	32.4	30.4	1.7	0.3	19.1	16.0	2.6	0.5
Cyprus	15.1	15.0	0.1	0.0	17.0	16.9	0.1	0.0	13.1	13.0	0.1	0.0	14.1	14.0	0.1	0.0	15.4	15.3	0.1	0.0	12.6	12.6	0.1	0.0
Turkey	34.8	32.0	2.4	0.4	28.9	27.7	1.2	0.0	48.3	42.0	5.0	1.2	21.9	19.3	2.2	0.5	20.6	19.6	1.0	0.0	25.6	18.3	5.5	1.8
Eastern Europe																								
Bulgaria	15.9	15.0	0.9	0.0	17.6	16.9	0.7	0.0	13.9	12.9	1.0	0.0	13.6	12.7	0.9	0.0	15.0	14.2	0.8	0.0	12.1	11.1	1.0	0.0
Czech Republic	9.2	8.6	0.6	0.0	11.0	10.6	0.3	0.0	7.0	6.1	6.0	0.0	9.1	8.6	0.5	0.0	10.8	10.5	0.3	0.0	7.0	6.1	0.8	0.0
Hungary	12.2	11.8	0.4	0.0	14.5	14.2	0.3	0.0	9.6	9.1	0.5	0.0	10.5	10.2	0.4	0.0	12.7	12.4	0.3	0.0	8.1	7.7	0.5	0.0
Moldova, Republic of	28.9	21.9	6.6	0.4	32.0	25.4	6.5	0.1	25.9	18.4	6.7	0.8	13.7	7.6	5.6	0.6	18.9	13.4	5.5	0.1	9.0	2.3	5.7	1.0
Poland	38.0	20.1	17.8	0.0	42.2	21.1	21.1	0.0	32.7	19.0	13.7	0.0	31.6	12.1	19.5	0.0	36.8	13.7	23.1	0.0	25.1	10.0	15.1	0.0
Romania	28.9	27.4	1.5	0.0	29.1	27.5	1.7	0.0	28.6	27.3	1.3	0.0	11.1	9.3	1.9	0.0	12.5	10.5	2.0	0.0	9.4	7.8	1.7	0.0
Russian Federation	35.9	24.4	11.5	0.0	39.1	27.9	11.3	0.0	32.7	20.9	11.8	0.0	35.6	24.3	11.3	0.0	38.7	27.9	10.8	0.0	32.5	20.8	11.7	0.0
Slovakia	16.7	16.4	0.3	0.0	20.7	20.5	0.2	0.0	12.1	11.7	0.5	0.0	16.5	16.2	0.3	0.0	20.4	20.3	0.2	0.0	12.0	11.5	0.5	0.0
Northern, Southern and Western Europe	d Wester	'n Europ	e																					
Northern Europe																								
Denmark	11.2	4.1	7.2	0.0	14.4	4.7	9.7	0.0	7.5	3.4	4.1	0.0	10.8	4.0	6.8	0.0	13.7	4.6	9.1	0.0	7.5	3.4	4.1	0.0
Estonia	6.9	5.4	1.5	0.0	7.6	6.2	1.5	0.0	6.1	4.7	1.4	0.0	6.3	4.8	1.5	0.0	6.7	5.2	1.5	0.0	5.9	4.4	1.5	0.0
Finland	6.3	6.2	0.1	0.0	8.3	8.2	0.1	0.0	4.2	4.1	0.1	0.0	5.1	5.1	0.0	0.0	6.6	6.5	0.0	0.0	3.6	3.6	0.0	0.0
Iceland	4.9	4.6	0.3	0.0	5.9	5.6	0.3	0.0	3.9	3.5	0.4	0.0	4.8	4.5	0.3	0.0	5.9	5.6	0.3	0.0	3.8	3.4	0.4	0.0

			Sh	Share of informal employment in t	ormal en	nploym		otal employment (%)	yment (%	(9			0,	Share of	i non-ag	Share of non-agricultural informal employment in non-agriculture (%), SDG 8.3.1	informa	l employ	/ment in	i non-agr	iculture	(%), SD(	3 8.3.1	
		To	Total			×	Men			Women	len			Total	a			Men				Women	en	
	letoT	In the informal sector	In the formal sector	splodesuod nl	lstoT	In the informal sector	In the formal sector	spiouəsuon ni	letoT	In the informal sector	in the formal sector	splodesuod nl	lstoT	In the informal sector	In the formal sector	splodesuod nl	letoT	In the informal sector	in the formal sector	splodesuod nl	letal	In the informal sector	sector sector	splodəsuod nl
Ireland	13.5	11.8	1.7	0.0	18.4	16.4	2.0	0.0	7.9	9.9	1.3	0.0	11.1	9.4	1.8	0.0	14.8	12.7	2.2	0.0	7.2	5.9	1.3	0.0
Latvia	13.2	11.2	2.0	0.0	15.1	12.8	2.3	0.0	11.5	9.8	1.7	0.0	11.2	9.1	2.0	0.0	12.1	9.6	2.5	0.0	10.4	8.7	1.7	0.0
Lithuania	12.6	8.2	4.4	0.0	11.4	6.7	4.6	0.0	13.8	9.7	4.2	0.0	9.8	5.6	4.2	0.0	8.2	4.0	4.2	0.0	11.2	7.1	4.1	0.0
Norway	7.4	7.1	0.2	0.0	9.0	8.8	0.2	0.0	5.5	5.3	0.2	0.0	6.9	6.7	0.2	0.0	8.3	8.1	0.2	0.0	5.3	5.1	0.2	0.0
Sweden	8.2	2.6	5.6	0.0	8.4	3.0	5.5	0.0	7.9	2.2	5.7	0.0	8.3	2.6	5.7	0.0	8.6	3.0	5.7	0.0	7.9	2.1	5.8	0.0
United Kingdom	13.6	13.5	0.1	0.0	16.1	16.0	0.1	0.0	10.9	10.8	0.1	0.0	13.4	13.3	0.1	0.0	15.8	15.7	0.1	0.0	10.8	10.7	0.1	0.0
Southern Europe																								
Albania	61.0	29.1	0.5	31.4	59.0	31.1	0.6	27.3	63.5	26.6	0.5	36.4	33.1	24.9	0.9	7.3	38.6	29.0	0.9	8.8	24.1	18.3	1.0	4.8
Bosnia and Herzegovina	30.1	30.0	0.1	0.0	26.5	26.5	0.0	0.0	36.6	36.3	0.4	0.0	13.2	13.1	0.2	0.0	12.8	12.8	0.0	0.0	14.0	13.5	0.5	0.0
Croatia	13.0	12.8	0.2	0.0	14.0	13.8	0.2	0.0	11.8	11.6	0.2	0.0	10.8	10.6	0.2	0.0	11.4	11.2	0.2	0.0	10.1	9.8	0.2	0.0
Greece	32.8	32.2	0.5	0.0	34.4	33.8	0.6	0.0	30.4	30.0	0.4	0.0	25.5	25.0	0.6	0.0	28.2	27.6	0.6	0.0	21.6	21.1	0.5	0.0
Italy	19.0	16.0	3.0	0.0	21.2	18.2	3.0	0.0	15.9	12.8	3.1	0.0	18.3	15.2	3.1	0.0	20.3	17.2	3.1	0.0	15.4	12.3	3.1	0.0
Malta	8.1	7.9	0.2	0.0	9.0	8.9	0.1	0.0	9.9	6.2	0.5	0.0	7.7	7.5	0.2	0.0	8.4	8.2	0.1	0.0	6.6	6.1	0.5	0.0
Portugal	12.1	11.9	0.3	0.0	13.5	13.2	0.3	0.0	10.7	10.4	0.2	0.0	10.5	10.2	0.3	0.0	11.8	11.5	0.3	0.0	9.0	8.8	0.2	0.0
Serbia	22.1	6.4	10.7	5.0	21.3	8.2	8.8	4.4	23.0	4.2	13.1	5.7	15.3	3.6	5.7	6.0	17.0	5.1	6.5	5.5	13.3	1.9	4.7	6.7
Slovenia	5.0	4.8	0.3	0.0	6.3	6.0	0.3	0.0	3.5	3.3	0.2	0.0	4.1	3.9	0.2	0.0	5.0	4.8	0.2	0.0	3.0	2.9	0.2	0.0
Spain	27.3	17.3	9.9	0.0	26.2	17.7	8.5	0.0	28.7	16.9	11.8	0.0	26.8	16.4	10.4	0.0	25.5	16.5	9.0	0.0	28.4	16.3	12.1	0.0
Western Europe																								
Austria	10.0	8.8	1.2	0.0	9.2	8.4	0.8	0.0	11.1	9.3	1.7	0.0	8.9	7.7	1.2	0.0	8.2	7.4	0.8	0.0	9.9	8.1	1.8	0.0
Belgium	13.5	10.8	2.6	0.0	15.3	12.7	2.6	0.0	11.4	8.8	2.7	0.0	12.8	10.1	2.7	0.0	14.4	11.7	2.7	0.0	11.0	8.3	2.7	0.0
France	9.8	5.4	4.4	0.0	9.6	6.5	3.1	0.0	9.9	4.2	5.7	0.0	8.9	4.5	4.4	0.0	8.6	5.4	3.2	0.0	9.2	3.5	5.7	0.0
Germany	10.2	9.7	0.5	0.0	10.3	9.9	0.4	0.0	10.2	9.6	0.6	0.0	10.1	9.6	0.5	0.0	10.1	9.8	0.4	0.0	10.1	9.5	0.6	0.0
Luxembourg	1.2	0.9	0.3	0.0	0.8	0.6	0.2	0.0	1.7	1.3	0.3	0.0	1.0	0.9	0.1	0.0	0.8	0.6	0.2	0.0	1.2	1.2	0.0	0.0
Netherlands	9.4	8.4	0.9	0.0	9.9	9.1	0.8	0.0	8.8	7.6	1.1	0.0	9.2	8.3	1.0	0.0	9.7	8.9	0.8	0.0	8.6	7.5	1.1	0.0
Switzerland	10.4	4.2	6.2	0.0	11.1	3.9	7.2	0.0	9.5	4.6	4.9	0.0	10.1	3.9	6.2	0.0	10.9	3.6	7.4	0.0	9.1	4.2	4.8	0.0
Source: II O calculations based on national labour force or similar bousehold surve	no pase	nationa	Inhour	force or s	imilar ho	Johoan	2	data 200	vibuouc	v Δ O for	locizio -		and woon											

Source: ILO calculations based on national labour force or similar household survey data. See appendix A.2 for original sources and years.

ition
loca
rural
an/
/ urb
þ
ployment
empl
tural
gricul
1-ag
1001
-
ani
yment
nplo
e
ota
nt
Ħ
ЯШ
임
em
mal
for
f in
0 a.
– Shar
2 -
Table B.2

					Total informal employ	ormal e	mploym	ment (%)								Non-a	gricultur	al infor	mal emp	Non-agricultural informal employment (%)	(%)			
		D	Total			Ľ	Urban			Rural	al			Total	-			Urban	an			Rural	'al	
	letoT	In the informal sector	In the formal sector	sblodəsuod ni	letoT	In the informal sector	In the formal sector	splodesuod ni	letoT	In the informal sector	In the formal sector	spiodəsuod ni	Total	In the informal sector	In the formal sector	splodesuod ni	letoT	In the informal sector	In the formal sector	spiouəsuou ni	Total	In the informal sector	sector In the formal	spionasuon ni
Africa																								
Northern Africa																								
Egypt	63.3	58.5	4.2	0.5	51.0	44.7	5.4	0.9	72.1	68.5	3.4	0.2	49.8	43.4	5.7	0.7	47.4	40.7	5.8	1.0	52.5	46.5	5.6	0.4
Morocco	79.9	58.7	19.8	1.4	72.5	50.7	19.8	2.0	89.1	68.8	19.8	0.5	75.6	53.3	20.4	1.9	71.7	50.6	18.9	2.2	85.5	60.1	24.2	1.2
Tunisia	58.8	53.3	5.1	0.4	53.2	47.2	5.5	0.5	71.8	67.4	4.3	0.2	53.5	47.6	5.5	0.5	51.5	45.5	5.5	0.6	61.0	55.2	5.6	0.2
Sub-Saharan Africa																								
Central Africa																								
Angola	94.1	77.1	16.5	0.5	89.4	61.1	27.5	0.8	98.4	91.6	9.9	0.2	89.5	59.2	29.5	0.9	88.5	57.6	30.0	0.9	93.6	65.5	27.4	0.7
Cameroon	90.9	83.7	6.4	0.8	81.5	66.5	13.2	1.8	97.2	95.4	1.8	0.0	79.0	62.5	14.7	1.8	77.3	58.7	16.4	2.3	84.6	75.3	9.1	0.2
Chad	94.0	92.0	2.0	0.0	68.2	62.1	6.1	0.0	96.7	95.2	1.6	0.0	75.8	68.0	7.8	0.0	66.4	60.0	6.4	0.0	81.4	72.7	8.7	0.0
Congo	85.3	56.9	25.9	2.5	85.3	56.9	25.9	2.5	95.8	85.0	2.5	8.3	85.0	55.6	26.8	2.6	85.0	55.6	26.8	2.6	89.3	81.3	5.9	2.1
Congo, Democratic Republic of the	91.9	88.2	3.7	0.0	80.9	74.8	6.2	0.0	95.2	92.2	3.0	0.0	76.2	67.0	9.3	0.0	76.2	68.6	7.6	0.0	76.3	64.3	12.0	0.0
Eastern Africa																								
Comoros	89.2	84.6	4.6	0.0	77.8	69.8	8.1	0.0	93.4	90.0	3.4	0.0	78.5	69.5	9.0	0.0	72.4	62.7	9.7	0.0	83.2	74.8	8.4	0.0
Madagascar	93.6	83.6	4.1	5.9	79.7	52.3	11.7	15.8	96.4	89.9	2.5	4.0	79.8	47.3	12.4	20.1	77.2	45.8	13.2	18.1	82.4	48.7	11.7	22.0
Malawi	83.0	74.1	4.6	4.3	66.9	45.5	11.9	9.4	85.1	77.8	3.7	3.6	67.2	49.4	10.8	7.0	62.3	37.7	13.9	10.6	69.0	53.6	9.7	5.7
Rwanda	94.3	90.9	1.5	1.9	97.2	95.4	1.0	0.8	79.4	68.2	4.0	7.3	78.6	67.0	4.6	7.1	83.9	75.0	4.1	4.8	72.6	57.9	5.1	9.7
Tanzania, United Republic of	90.6	83.5	4.9	2.3	77.6	62.6	11.8	3.2	96.9	93.5	1.5	1.8	69.3	48.1	15.2	6.0	67.4	45.5	17.3	4.6	74.4	55.3	9.3	9.8
Uganda	93.7	87.6	6.2	0.0	79.2	59.8	13.4	6.0	93.9	83.8	5.3	4.8	83.4	67.1	16.4	0.0	76.2	54.5	15.6	6.1	87.0	64.2	15.0	7.8
Zambia	87.9	79.0	2.9	6.1	79.0	68.1	5.2	5.8	94.4	86.9	1.2	6.3	71.5	52.3	6.3	12.9	68.1	51.8	7.7	8.6	79.6	53.6	2.9	23.1

					Total in	Total informal employ	molovme	(%)								Non-ag	ricultura	al inform	nal empl	Non-agricultural informal employment (%)	(%			
		Total	al			Urban	an			Rural	al			Total	_			Urban				Rural	a	
	lstoT	In the informal sector	In the formal sector	spiodəsuod ni	latoT	In the informal sector	In the formal sector	spiodəsuod ni	lstoT	In the informal sector	In the formal sector	spiodəsuod ni	lstoT	In the informal sector	In the formal sector	spiouəsnou ui	Total	In the informal sector	In the formal sector	spiodəsuod ni	Total	In the informal sector	In the formal sector	splodesuod ni
Southern Africa																								
Botswana	65.6	52.1	5.2	8.3	54.0	42.4	5.1	6.5	82.7	66.5	5.3	10.9	55.1	41.2	6.8	7.2	50.6	38.4	5.5	6.7	67.8	48.9	10.4	8.5
Namibia	67.0	46.8	5.6	14.6	58.0	45.7	6.1	6.2	81.4	48.6	4.7	28.1	61.1	46.3	6.5	8.3	57.5	45.4	6.2	5.9	71.3	48.6	7.4	15.2
South Africa	34.0	21.8	4.8	7.4	29.3	18.8	3.6	7.0	48.1	30.8	8.5	8.8	34.0	21.4	4.9	7.7	29.2	18.5	3.6	7.1	50.5	31.4	9.4	9.8
Western Africa																								
Benin	94.5	90.6	2.8	1.1	89.3	82.1	5.3	1.9	98.3	96.6	1.0	0.6	90.6	83.7	4.9	2.0	86.8	9.77	9.9	2.3	96.1	92.2	2.5	1.5
Burkina Faso	94.6	65.5	9.6	19.5	86.3	57.1	18.1	11.1	96.7	67.8	7.3	21.7	88.4	67.9	14.5	6.0	83.7	56.9	20.5	6.3	94.9	84.5	4.8	5.6
Cabo Verde	46.5	31.9	6.0	8.5	45.1	30.8	5.5	8.7	49.2	34.1	6.9	8.2	46.8	29.3	7.2	10.2	44.9	29.7	5.9	9.3	52.5	28.2	11.1	13.1
Côte d'Ivoire	92.8	86.0	5.2	1.6	88.4	76.6	8.7	3.1	96.8	94.8	1.9	0.2	87.7	75.8	9.0	2.9	86.3	72.6	10.0	3.7	92.2	86.1	5.7	0.4
Gambia	76.7	62.5	7.1	7.1	63.7	48.2	10.6	4.9	89.7	76.7	3.6	9.4	68.4	52.1	9.2	7.2	62.3	46.5	10.8	4.9	81.2	63.6	5.8	11.8
Ghana	90.1	84.8	4.9	0.5	83.6	75.0	7.8	0.8	96.4	94.0	2.1	0.3	82.6	73.2	8.6	0.9	80.6	70.4	9.3	0.9	88.2	80.6	6.7	0.9
Liberia	86.8	79.1	6.0	1.7	86.2	72.7	8.5	5.0	90.4	80.9	4.3	5.3	77.5	64.6	10.0	2.9	76.9	57.7	13.7	5.4	81.7	67.9	7.8	5.9
Mali	92.7	74.2	4.1	14.4	83.5	63.5	12.3	7.6	95.4	77.3	1.7	16.5	86.4	70.5	10.6	5.3	81.6	63.6	14.1	3.9	91.7	78.1	6.8	6.8
Niger	91.3	77.3	9.6	4.5	80.2	61.6	15.6	3.0	93.1	79.9	8.6	4.7	85.1	72.6	10.3	2.2	76.7	57.3	16.3	3.0	93.5	87.8	4.3	1.4
Nigeria	92.9	80.4	2.9	9.6	89.3	82.6	4.1	2.6	95.2	79.1	2.1	14.0	89.0	82.9	4.2	1.8	88.3	82.1	4.4	1.8	89.8	83.9	4.0	1.8
Senegal	91.2	70.9	14.0	6.2	86.5	58.7	9.5	18.3	97.9	65.0	1.2	31.7	87.0	59.1	19.1	8.7	85.3	58.3	10.0	17.1	95.5	69.2	3.1	23.2
Sierra Leone	92.5	89.8	1.0	1.7	91.0	87.9	1.3	1.8	93.2	90.8	0.9	1.6	86.0	80.8	2.6	2.6	85.5	79.7	3.1	2.7	86.3	81.5	2.3	2.5
Togo	92.8	80.6	5.8	6.4	89.4	74.9	9.5	5.1	96.1	86.2	2.2	7.7	90.0	73.6	6.7	9.8	87.7	72.3	9.3	6.1	92.8	69.9	2.7	20.3
Americas																								
Latin America and the Caribbean	9 Caribbe	an																						
The Caribbean																								
Dominican Republic	56.3	37.4	11.8	7.0	67.3	48.5	10.4	8.4	53.5	34.6	12.2	6.7	52.6	31.3	13.1	8.1	60.1	35.2	13.4	11.6	51.0	30.5	13.1	7.4
Central America																								
Costa Rica	39.1	27.4	4.9	6.9	35.0	24.0	4.4	6.6	51.9	37.8	6.3	7.8	37.4	25.5	4.2	7.8	34.7	23.7	4.1	6.9	49.1	33.2	4.4	11.6
El Salvador	69.69	53.9	11.2	4.5	61.0	46.9	10.0	4.2	86.5	67.7	13.6	5.2	63.1	48.1	9.4	5.6	58.9	45.3	9.1	4.5	76.9	57.4	10.2	9.3
Guatemala	79.7	63.3	8.8	7.7	71.4	54.5	11.7	5.2	89.0	73.0	5.6	10.4	72.7	54.5	12.0	6.1	68.7	50.7	13.0	4.9	80.6	61.9	10.1	8.6

					Total informal employ	ırmal em	Iploymer	yment (%)								Non-ag	gricultur	al inforn	nal emp	Non-agricultural informal employment (%)	(%			
Total	Total					Urban	=			Rural	al			Total	al			Urban	u			Rural	al	
Total In the informal sector sector sector In households	sector In the formal sector	sector	splodəsuod ni		Total	In the informal sector	In the formal sector	splodəsuod nl	letoT	In the informal sector	In the formal sector	spiouəsnou ui	lstoT	In the informal sector	In the formal sector	spiouəsuou ui	letoT	In the informal sector	In the formal sector	spiouəsuou ui	letoT	In the informal sector	in the formal sector	sployəsnoy ul
79.9 70.0 1.8 8.1	1.8 8.1	8.1			70.1	63.1	2.3	4.6	91.4	9.77	1.3	12.2	72.7	65.7	2.6	4.3	68.0	61.6	2.5	3.9	83.2	75.1	2.7	5.4
53.4 29.1 19.8 4.5	19.8 4.5	4.5		-	46.9	28.0	14.3	4.6	58.6	29.9	24.2	4.5	53.2	33.1	14.9	5.2	46.9	28.2	14.0	4.6	59.8	38.1	15.8	5.8
77.4 71.5 1.4 4.4 6	1.4 4.4	4.4		9	67.9	61.4	1.5	5.0	91.5	86.7	1.3	3.6	68.6	60.1	2.1	6.5	66.3	59.2	1.7	5.4	78.5	63.8	3.8	10.8
52.3 34.3 14.0 3.9 52	14.0 3.9	3.9		52	52.3	34.3	14.0	3.9	:	÷	÷	:	46.0	26.2	15.2	4.6	46.0	26.2	15.2	4.6	:	:	÷	÷
47.2 32.8 8.5 5.8 47.2	8.5 5.8	5.8		47		32.8	8.5	5.8	:	÷	÷	:	47.0	32.7	8.5	5.9	47.0	32.7	8.5	5.9	:	:	:	÷
83.1 68.7 12.4 1.9 75.3	12.4 1.9	1.9		75.		55.4	17.1	2.8	96.6	92.0	4.1	0.4	75.6	55.3	17.5	2.8	74.3	53.6	17.7	3.0	85.4	67.6	16.0	1.8
46.0 30.3 11.9 3.9 40.1	11.9 3.9	3.9	_	40.	-	25.9	10.7	3.5	50.5	33.5	12.8	4.1	42.5	26.4	11.8	4.3	39.6	25.4	10.7	3.5	45.0	27.3	12.8	4.9
40.5 32.2 6.3 2.0 38.8	6.3 2.0	2.0	_	38.	~	30.0	6.8	2.0	51.7	46.5	3.3	1.9	39.3	30.2	6.8	2.2	38.6	29.6	6.9	2.1	46.8	37.5	5.7	3.6
60.6 55.9 3.0 1.8 56.0	3.0 1.8	1.8		56.	0	51.0	3.3	1.8	89.2	86.3	1.3	1.6	55.4	50.0	3.4	2.0	54.4	49.1	3.4	1.9	76.3	67.5	3.9	4.9
59.0 46.4 11.1 1.6 50.8	11.1 1.6	1.6		50.	∞	36.6	12.4	1.9	76.8	67.5	8.2	1.1	50.4	35.9	12.4	2.1	48.9	34.6	12.3	2.0	57.4	42.0	12.8	2.7
70.6 46.0 17.6 7.0 60.6	17.6 7.0	7.0	_	60.	9	31.7	21.1	7.7	86.4	68.6	12.0	5.9	64.4	34.9	20.8	8.7	60.0	30.7	21.3	7.9	77.2	46.8	19.5	11.0
69.2 55.5 11.8 1.9 61.1	11.8 1.9	1.9	_	61.		44.5	14.2	2.3	95.7	91.1	3.9	0.6	59.1	40.5	15.9	2.7	57.2	38.7	15.9	2.7	85.9	66.5	16.5	2.9
24.5 19.6 1.9 3.0 24.4	1.9 3.0	3.0		24.	4	19.4	2.0	3.1	27.1	24.2	0.8	2.1	24.1	18.8	2.0	3.3	23.9	18.7	2.0	3.2	28.8	21.9	1.9	5.0
39.7 26.3 6.3 7.2 39.7	6.3 7.2	7.2		39.		26.3	6.3	7.2	:	÷	:	÷	38.3	24.2	6.4	7.7	38.3	24.2	6.4	7.7	÷	÷	÷	:
																I								
18.6 16.6 1.5 0.5 18	1.5 0.5	0.5		18	18.2	16.3	1.4	0.5	24.3	20.3	2.8	1.1	18.3	16.3	1.5	0.5	18.0	16.1	1.4	0.5	22.2	18.4	2.6	1.2
66.9 59.7 7.2 0.0 62	7.2 0.0	0.0		62	62.7	55.5	7.2	0.0	76.8	69.5	7.2	0.0	66.7	59.5	7.2	0.0	61.9	54.6	7.2	0.0	66.5	56.4	10.1	0.0
44.9 43.9 1.0 0.0 29	1.0 0.0	0.0	_	29	29.4	28.4	1.0	0.0	48.1	47.1	0.9	0.0	43.0	42.0	1.0	0.0	24.8	23.7	1.2	0.0	46.4	45.4	1.0	0.0
64.3 31.2 30.8 2.3 6:	30.8 2.3	2.3		63	63.1	30.3	30.8	2.0	67.2	33.5	30.7	3.1	60.1	25.8	34.2	0.1	59.4	25.6	33.8	0.1	61.9	26.6	35.2	0.1
70.1 66.0 4.1 0.0 7	4.1 0.0	0.0		~	76.1	72.9	3.3	0.0	64.0	59.2	4.9	0.0	61.0	55.6	5.4	0.0	58.0	52.1	5.9	0.0	62.7	57.6	5.1	0.0
77.8 68.5 9.0 0.3 6	9.0 0.3	0.3		39	68.5	50.6	17.3	9.0	80.6	73.9	6.4	0.3	69.0	56.3	12.2	0.5	66.8	48.0	18.2	0.6	69.9	60.0	9.5	0.4

					Total in	Total informal emplo	mplovme	ovment (%)								Non-a	Non-agricultural informal employment (%)	al infor	mal emp	olovment	(%)			
		Total	al			Urban				Rural	ral			Total	al			Urban	an			Rural	a	
	letoT	In the informal sector	in the formal sector	spiouəsnou ui	lstoT	In the informal sector	sector In the formal	spiodesuod ni	Total	In the informal sector	ln the formal sector	splodesuod nl	lstoT	In the informal sector	In the formal sector	splodesuod nl	lstoT	In the informal sector	in the formal sector	spiodəsuod ni	lstoT	In the informal sector	In the formal sector	spionasuon ni
Asia and the Pacific																								
Eastern Asia																								
China	54.4	48.4	6.1	0.0	36.2	31.3	4.9	0.0	82.0	74.0	8.0	0.0	53.5	47.3	6.2	0.0	35.3	30.4	5.0	0.0	81.4	73.3	8.1	0.0
Japan	18.7	14.3	4.3	0.0	17.2	13.4	3.7	0.0	23.0	16.9	6.0	0.0	16.3	12.0	4.3	0.0	16.2	12.5	3.7	0.0	16.7	10.5	6.2	0.0
Korea, Republic of	31.5	26.1	5.5	0.0	:	÷	÷	÷	:	÷	÷	:	28.8	23.3	5.6	0.0	÷	÷	÷	÷	:	÷	÷	÷
Mongolia	53.5	45.7	2.1	5.7	39.7	29.9	2.7	7.2	93.4	91.6	0.2	1.6	35.6	25.3	2.9	7.4	35.4	25.2	2.9	7.3	41.6	29.9	2.0	9.7
South-Eastern Asia and the Pacific	d the Pa	cific																						
Pacific Islands																								
Samoa	35.7	21.7	10.4	3.7	38.4	21.7	14.0	2.8	34.2	21.6	8.4	4.1	33.8	21.2	10.9	1.8	37.6	21.3	14.2	2.0	31.7	21.1	8.9	1.6
South-Eastern Asia																								
Brunei Darussalam	31.9	4.0	22.3	5.7	33.8	4.0	23.8	6.0	25.2	3.9	16.8	4.5	32.9	4.2	22.7	6.0	34.9	4.3	24.3	6.3	25.5	3.9	16.8	4.8
Cambodia	93.1	77.5	14.5	1.0	84.7	64.2	19.2	1.2	95.8	81.9	13.0	1.0	89.8	67.3	21.2	1.4	84.2	63.1	19.9	1.2	93.0	69.6	21.9	1.4
Indonesia	85.6	67.5	5.8	12.2	80.3	64.1	6.7	9.5	91.3	71.2	4.9	15.2	80.2	62.7	8.3	9.1	78.1	62.1	7.6	8.4	84.2	64.0	9.7	10.5
Lao People's Democratic Republic	93.6	86.7	6.8	0.1	85.9	70.3	15.5	0.2	96.7	93.4	3.3	0.0	78.5	55.2	23.2	0.2	78.5	54.8	23.5	0.2	78.5	55.8	22.7	0.1
Myanmar	85.7	71.5	14.0	0.3	79.3	58.6	20.1	0.6	88.3	76.5	11.6	0.1	82.3	64.9	16.9	0.5	78.2	56.1	21.5	0.6	86.8	74.6	11.8	0.4
Timor-Leste	71.8	56.9	11.0	3.9	52.6	32.1	17.0	3.4	84.6	73.5	6.9	4.2	54.0	29.5	18.1	6.4	46.4	23.2	19.3	3.9	64.7	38.5	16.3	9.9
Viet Nam	76.2	61.0	11.5	3.7	55.0	36.7	17.1	1.3	85.3	71.5	9.1	4.8	57.9	37.4	19.8	0.7	48.9	29.1	19.0	0.7	65.9	44.8	20.5	0.7
Southern Asia																								
Bangladesh	89.0	48.9	13.5	26.7	77.9	39.5	23.2	15.2	93.4	52.5	9.7	31.1	82.0	49.2	23.1	9.7	74.5	37.9	27.5	9.1	87.5	57.5	19.9	10.1
India	88.2	80.9	6.5	0.8	75.9	58.6	15.2	2.2	93.0	89.5	3.1	0.3	78.1	64.3	12.2	1.6	74.5	56.0	16.1	2.3	81.6	72.4	8.4	0.8
Nepal	94.3	90.7	3.4	0.2	85.1	74.5	9.4	1.1	95.8	93.3	2.4	0.1	80.4	68.5	11.0	0.9	79.0	64.0	13.4	1.7	81.1	70.9	9.7	0.5
Pakistan	82.4	77.6	4.0	0.8	66.5	57.8	7.3	1.4	91.9	89.4	2.1	0.4	70.8	62.7	6.8	1.3	65.7	56.8	7.5	1.4	78.9	72.1	5.8	1.1
Sri Lanka	70.4	9.09	7.9	2.0	58.6	44.7	10.0	3.9	72.6	63.4	7.5	1.6	62.2	49.1	10.2	2.8	57.1	42.7	10.2	4.1	63.5	50.8	10.2	2.5

					Total in	formal	Total informal employm	nent (%)								Non-a	gricultur	al infor	mal em	Non-agricultural informal employment (%)	: (%)			
		2	Total			Ŀ	Urban			Rural	al			Total	al			Urban	an			Rural	al	
	Total	In the informal sector	In the formal sector	splodəsuod nl	letoT	In the informal sector	In the formal sector	splodəsuod ni	løtoT	In the informal sector	ln the formal sector	splodesuod ni	latoT	In the informal sector	In the formal sector	splodəsuod ni	letoT	In the informal sector	In the formal sector	splodəsuod ni	letoT	In the informal sector	In the formal sector	splodəsuod nl
<b>Europe and Central Asia</b>																								
<b>Central and Western Asia</b>	sia																							
Central Asia																								
Kyrgyzstan	48.6	37.2	9.6	1.8	41.0	32.6	7.6	0.8	53.4	40.1	10.9	2.4	42.1	32.7	7.1	2.3	39.8	31.4	7.6	0.8	44.7	34.1	6.7	3.9
Tajikistan	74.8	54.4	14.4	5.9	68.6	47.8	15.1	5.7	76.8	56.6	14.2	6.0	70.5	39.8	20.5	10.2	64.8	41.5	16.7	9.9	74.0	38.8	22.8	12.4
Western Asia																								
Armenia	52.1	39.2	1.4	11.6	29.3	24.7	1.6	3.0	77.0	55.0	1.1	20.9	26.5	24.0	2.1	0.4	25.4	23.3	1.6	0.4	30.2	26.3	3.4	0.5
Cyprus	15.1	15.0	0.1	0.0	13.7	13.6	0.1	0.0	19.0	18.9	0.1	0.0	14.1	14.0	0.1	0.0	13.5	13.4	0.1	0.0	15.7	15.7	0.1	0.0
Turkey	34.8	32.0	2.4	0.4	:	:	÷	:	:	÷	:	:	21.9	19.3	2.2	0.5	:	:	÷	:	:	:	÷	:
Eastern Europe																								
Bulgaria	15.9	15.0	0.9	0.0	12.8	11.9	0.9	0.0	24.2	23.3	0.9	0.0	13.6	12.7	0.9	0.0	12.4	11.5	0.9	0.0	17.7	16.8	0.9	0.0
Czech Republic	9.2	8.6	0.6	0.0	9.5	8.9	0.6	0.0	8.6	8.2	0.5	0.0	9.1	8.6	0.5	0.0	9.5	8.9	0.6	0.0	8.4	8.0	0.4	0.0
Hungary	12.2	11.8	0.4	0.0	12.6	12.4	0.2	0.0	11.5	10.8	0.7	0.0	10.5	10.2	0.4	0.0	12.1	11.9	0.2	0.0	7.4	6.7	0.7	0.0
Moldova, Republic of	28.9	21.9	9.9	0.4	12.5	6.5	5.6	0.5	43.5	35.6	7.6	0.4	13.7	7.6	5.6	9.0	11.4	5.7	5.2	0.5	17.7	10.8	6.2	0.7
Poland	38.0	20.1	17.8	0.0	31.3	12.9	18.4	0.0	47.7	30.8	17.0	0.0	31.6	12.1	19.5	0.0	29.9	11.2	18.7	0.0	34.8	13.7	21.1	0.0
Romania	28.9	27.4	1.5	0.0	12.8	11.0	1.8	0.0	54.0	52.9	1.1	0.0	11.1	9.3	1.9	0.0	8.5	9.9	1.8	0.0	18.7	16.8	1.9	0.0
Russian Federation	35.9	24.4	11.5	0.0	35.2	25.4	9.8	0.0	38.4	20.4	18.0	0.0	35.6	24.3	11.3	0.0	35.3	25.5	9.8	0.0	37.0	19.3	17.7	0.0
Slovakia	16.7	16.4	0.3	0.0	16.2	15.8	0.4	0.0	17.5	17.3	0.2	0.0	16.5	16.2	0.3	0.0	16.0	15.6	0.4	0.0	17.2	17.0	0.2	0.0
Northern, Southern and Western Europe	d Weste	rm Euro	be																					
Northern Europe																								
Denmark	11.2	4.1	7.2	0.0	10.8	4.3	6.5	0.0	11.8	3.9	8.0	0.0	10.8	4.0	6.8	0.0	10.7	4.3	6.4	0.0	11.0	3.7	7.3	0.0
Estonia	6.9	5.4	1.5	0.0	6.3	4.4	1.8	0.0	7.9	7.1	0.8	0.0	6.3	4.8	1.5	0.0	6.3	4.4	1.9	0.0	6.3	5.4	0.9	0.0
Finland	6.3	6.2	0.1	0.0	4.3	4.3	0.0	0.0	7.8	7.7	0.1	0.0	5.1	5.1	0.0	0.0	4.2	4.2	0.0	0.0	5.9	5.9	0.1	0.0
Iceland	4.9	4.6	0.3	0.0	5.1	4.9	0.2	0.0	4.5	4.0	0.5	0.0	4.8	4.5	0.3	0.0	5.1	4.8	0.3	0.0	4.4	3.8	0.5	0.0
Ireland	13.5	11.8	1.7	0.0	11.5	9.2	2.3	0.0	16.7	16.1	0.6	0.0	11.1	9.4	1.8	0.0	11.4	9.1	2.4	0.0	10.6	9.9	0.7	0.0
Latvia	13.2	11.2	2.0	0.0	11.1	9.1	1.9	0.0	16.0	14.0	2.0	0.0	11.2	9.1	2.0	0.0	11.0	9.1	1.9	0.0	11.4	9.2	2.2	0.0

					Total inf	Total informal employ	mploym	ment (%)								Non-a	ricultur	al inform	mal emp	Non-agricultural informal employment (%)	(%)			
		2	Total			Urban	an			Rural	ral			Total	al			Urban	an			Ru	Rural	
	letoT	In the informal sector	sector sector	spiouəsuon ni	letoT	In the informal sector	sector sector	splodesuod nl	letoT	In the informal sector	In the formal sector	spiouəsnou ui	Total	In the informal sector	sector sector	splodesuod nl	letoT	In the informal sector	sector sector	splodesuod nl	letoT	In the informal sector	sector In the formal	spionasuon ni
Lithuania	12.6	8.2	4.4	0.0	10.8	5.4	5.3	0.0	15.4	12.3	3.0	0.0	9.8	5.6	4.2	0.0	10.4	5.3	5.1	0.0	8.7	6.2	2.5	0.0
Norway	7.4	7.1	0.2	0.0	5.4	5.3	0.1	0.0	12.0	11.6	0.4	0.0	6.9	6.7	0.2	0.0	5.3	5.2	0.1	0.0	10.8	10.4	0.4	0.0
Sweden	8.2	2.6	5.6	0.0	8.8	2.5	6.3	0.0	7.8	2.6	5.2	0.0	8.3	2.6	5.7	0.0	8.8	2.5	6.3	0.0	7.9	2.6	5.4	0.0
United Kingdom	13.6	13.5	0.1	0.0	12.9	12.8	0.1	0.0	18.1	17.9	0.2	0.0	13.4	13.3	0.1	0.0	12.9	12.8	0.1	0.0	16.7	16.5	0.2	0.0
Southern Europe																								
Albania	61.0	29.1	0.5	31.4	:	÷	÷	:	:	÷	÷	:	33.1	24.9	0.9	7.3	:	÷	:	:	:	÷	÷	:
Bosnia and Herzegovina	30.1	30.0	0.1	0.0	26.9	26.9	0.0	0.0	54.5	53.4	1.1	0.0	13.2	13.1	0.2	0.0	12.2	12.2	0.0	0.0	23.5	21.7	1.9	0.0
Croatia	13.0	12.8	0.2	0.0	10.4	10.4	0.0	0.0	16.0	15.6	0.4	0.0	10.8	10.6	0.2	0.0	9.9	9.8	0.0	0.0	11.9	11.5	0.4	0.0
Greece	32.8	32.2	0.5	0.0	24.1	23.7	0.4	0.0	45.9	45.2	0.8	0.0	25.5	25.0	0.6	0.0	22.8	22.4	0.4	0.0	30.9	30.0	0.9	0.0
Italy	19.0	16.0	3.0	0.0	18.2	15.2	3.0	0.0	24.0	20.6	3.4	0.0	18.3	15.2	3.1	0.0	17.7	14.7	3.0	0.0	22.2	18.6	3.6	0.0
Malta	8.1	7.9	0.2	0.0	8.1	7.9	0.2	0.0	÷	:	÷	÷	7.7	7.5	0.2	0.0	7.7	7.5	0.2	0.0	÷	÷	÷	÷
Portugal	12.1	11.9	0.3	0.0	10.4	10.1	0.3	0.0	16.9	16.7	0.2	0.0	10.5	10.2	0.3	0.0	9.6	9.3	0.3	0.0	13.2	13.0	0.2	0.0
Serbia	22.1	6.4	10.7	5.0	12.4	4.2	6.3	1.9	35.2	9.4	16.7	9.1	15.3	3.6	5.7	6.0	10.8	3.5	5.4	2.0	24.5	4.0	6.3	14.3
Slovenia	5.0	4.8	0.3	0.0	÷	÷	÷	:	:	÷	÷	:	4.1	3.9	0.2	0.0	÷	÷	:	:	:	÷	÷	÷
Spain	27.3	17.3	9.9	0.0	25.8	15.2	10.6	0.0	31.9	24.0	7.9	0.0	26.8	16.4	10.4	0.0	25.9	15.2	10.8	0.0	29.9	20.8	9.1	0.0
Western Europe																								
Austria	10.0	8.8	1.2	0.0	10.0	8.7	1.3	0.0	10.1	8.9	1.2	0.0	8.9	7.7	1.2	0.0	9.8	8.5	1.2	0.0	7.8	6.5	1.3	0.0
Belgium	13.5	10.8	2.6	0.0	13.4	10.7	2.6	0.0	15.7	13.0	2.7	0.0	12.8	10.1	2.7	0.0	12.8	10.1	2.7	0.0	13.4	10.6	2.7	0.0
France	9.8	5.4	4.4	0.0	8.6	4.1	4.5	0.0	11.8	7.8	4.1	0.0	8.9	4.5	4.4	0.0	8.5	3.9	4.6	0.0	9.7	5.5	4.2	0.0
Germany	10.2	9.7	0.5	0.0	10.7	10.3	0.4	0.0	8.7	8.1	0.6	0.0	10.1	9.6	0.5	0.0	10.7	10.3	0.4	0.0	8.3	7.8	0.5	0.0
Luxembourg	1.2	0.9	0.3	0.0	1.0	0.9	0.0	0.0	1.4	0.9	0.5	0.0	1.0	0.9	0.1	0.0	1.0	1.0	0.0	0.0	1.0	0.8	0.2	0.0
Netherlands	9.4	8.4	0.9	0.0	:	÷	:	:	:	÷	:	:	9.2	8.3	1.0	0.0	:	÷	:	:	:	÷	÷	:
Switzerland	10.4	4.2	6.2	0.0	9.7	4.1	5.7	0.0	12.2	4.6	7.7	0.0	10.1	3.9	6.2	0.0	9.6	3.9	5.7	0.0	11.6	3.8	7.8	0.0
Note: "" no data available by urban / rural location	ole by ui	ban / ru	ral locat	ion																				

Source: ILO calculations based on national labour force or similar household survey data. See appendix A.2 for original sources and years.

(e)
ultu
agric
ding
inclu
ent (
loym
emp
us in
' stat
nt by
yme
emplo
total
it in 1
ymer
emplo
rmal
f info
are o
– Shar
e B.3 -
Table B.3

		To	Total			Employees	yees			Employers		0wn-a	Own-account workers	rkers	Con	<b>Contributing family workers</b>	mily work	ers
	Inf	ormal emp	Informal employment (%)	(%	Inf	ormal emp	Informal employment (%)	(%)	Informal	Informal employment (%)	ent (%)	Informal	Informal employment (%)	ent (%)	Inf	Informal employment (%)	loyment (9	(%)
	lstoT	In the informal sector	In the formal sector	spiodəsuod ni	lstoT	In the informal sector	ln the formal sector	spiodəsuod ni	lstoT	In the informal sector	spiodəsuod ni	lstoT	In the informal sector	sblodəsuod ni	lstoT	In the informal sector	ln the formal sector	spionasuon ni
Africa																		
Northern Africa																		
Egypt	63.3	58.5	4.2	0.5	40.5	32.7	6.9	0.8	97.5	97.5	0.0	99.2	99.2	0.0	100	99.7	0.3	0.0
Morocco	79.9	58.7	19.8	1.4	69.4	32.4	34.5	2.5	9.99	66.6	0.0	94.0	94.0	0.0	100	88.0	12.0	0.0
Tunisia	58.8	53.3	5.1	0.4	43.7	36.1	7.0	0.6	98.9	98.9	0.0	99.7	99.7	0.0	100	99.8	0.2	0.0
Sub-Saharan Africa																		
Central Africa																		
Angola	94.1	77.1	16.5	0.5	82.1	29.0	51.6	1.5	96.4	96.4	0.0	99.6	9.66	0.0	100	91.5	8.5	0.0
Cameroon	90.9	83.7	6.4	0.8	61.8	26.2	31.1	4.6	78.8	78.8	0.0	95.6	92.6	0.0	100	95.1	4.9	0.0
Chad	94.0	92.0	2.0	0.0	35.1	24.6	10.5	0.0	91.7	91.7	0.0	98.7	98.7	0.0	100	92.6	4.4	0.0
Congo	85.3	56.9	25.9	2.5	82.6	40.1	38.9	3.6	80.4	80.4	0.0	90.1	90.0	0.1	100	80.2	16.2	3.7
Congo, Democratic Republic of the	91.9	88.2	3.7	0.0	40.9	16.8	24.1	0.0	84.0	84.0	0.0	98.1	98.1	0.0	100	96.4	3.6	0.0
Eastern Africa																		
Comoros	89.2	84.6	4.6	0.0	66.7	51.8	14.9	0.0	87.8	87.8	0.0	0.66	99.0	0.0	100	97.7	2.3	0.0
Madagascar	93.6	83.6	4.1	5.9	64.9	29.5	26.1	9.4	84.6	84.6	0.0	97.1	95.7	1.4	100	87.3	1.4	11.3
Malawi	83.0	74.1	4.6	4.3	78.7	56.7	10.6	11.4	58.5	58.5	0.0	84.4	84.4	0.0	100	90.3	9.7	0.0
Rwanda	94.3	90.9	1.5	1.9	84.2	72.1	5.4	6.8	71.0	71.0	0.0	98.0	98.0	0.0	100	100.0	0.0	0.0
Tanzania, United Republic of	90.6	83.5	4.9	2.3	63.3	25.9	36.9	0.5	53.7	51.9	1.8	92.5	91.0	1.4	100	95.2	0.7	4.1
Uganda	93.7	87.6	6.2	0.0	87.1	57.9	29.2	0.0	79.9	79.9	0.0	95.0	95.0	0.0	100	98.3	1.7	0.0
Zambia	87.9	79.0	2.9	6.1	58.3	31.7	10.4	16.2	73.0	9.99	6.5	95.6	90.4	5.3	100	98.7	0.4	0.9

		Total	<u>a</u>			Emplo	Employees			Employers		0wn-ad	Own-account workers	rkers	Conti	ibuting fa	Contributing family workers	ers
	Infe	ırmal emp	Informal employment (%)	(%	Ξţ	ormal emp	Informal employment (%)		Informal	Informal employment (%)	ant (%)	Informal	Informal employment (%)	ant (%)	Info	rmal emp	Informal employment (%)	()
	lstoT	In the informal sector	ln the formal sector	spiodəsuod ni	lstoT	In the informal sector	sector In the formal	splodəsuod al	Total	In the informal sector	spiotasuot ni	lstoT	In the informal sector	spiodəsuod ni	lstoT	In the informal sector	In the formal sector	splodəsuod ni
Southern Africa																		
Botswana	65.6	52.1	5.2	8.3	51.3	32.2	7.4	11.7	9.66	9.66	0.0	100.0	100.0	0.0	100	100.0	0.0	0.0
Namibia	67.0	46.8	5.6	14.6	62.0	45.2	8.0	8.8	46.4	34.1	12.4	81.3	48.9	32.4	100	78.1	0.0	21.9
South Africa	34.0	21.8	4.8	7.4	27.8	14.3	5.5	8.0	37.8	36.4	1.4	88.1	82.3	5.7	100	69.6	30.4	0.0
Western Africa																		
Benin	94.5	90.6	2.8	1.1	66.5	33.3	23.0	10.2	74.0	74.0	0.0	97.9	97.9	0.0	100	97.6	2.4	0.0
Burkina Faso	94.6	65.5	9.6	19.5	78.4	34.6	36.8	6.9	69.1	63.1	6.0	92.4	86.0	6.4	100	54.9	12.0	33.2
Cabo Verde	46.5	31.9	6.0	8.5	57.0	35.8	8.8	12.5	43.9	43.9	0.0	16.0	16.0	0.0	100	100.0	0.0	0.0
Côte d'Ivoire	92.8	86.0	5.2	1.6	78.6	53.5	20.0	5.1	89.5	89.0	0.5	97.1	96.7	0.4	100	97.4	2.0	0.6
Gambia	76.7	62.5	7.1	7.1	59.5	20.4	25.1	14.1	56.3	53.0	3.3	79.2	75.0	4.3	100	82.6	8.8	8.6
Ghana	90.1	84.8	4.9	0.5	63.9	43.6	18.3	2.0	93.9	93.9	0.0	99.2	99.2	0.0	100	98.8	1.2	0.0
Liberia	86.8	79.1	6.0	1.7	65.5	37.2	23.2	5.1	63.9	62.1	1.8	91.8	91.0	0.8	100	94.5	5.3	0.2
Mali	92.7	74.2	4.1	14.4	89.4	38.6	42.1	8.8	86.6	82.3	4.3	91.7	85.0	6.7	100	54.3	0.5	45.3
Niger	91.3	77.3	9.6	4.5	81.6	44.9	31.3	5.5	50.4	48.9	1.6	89.1	87.1	2.0	100	69.3	21.6	9.1
Nigeria	92.9	80.4	2.9	9.6	58.3	38.6	15.5	4.2	0.0	0.0	0.0	99.7	99.7	0.0	100	21.6	2.8	75.7
Senegal	91.2	70.9	14.0	6.2	76.4	18.2	46.3	11.9	76.9	72.8	4.1	96.9	93.0	2.8	100	93.0	0.4	6.6
Sierra Leone	92.5	89.8	1.0	1.7	52.0	38.8	10.8	2.4	98.9	90.9	8.0	95.8	94.4	1.4	100	97.9	1.1	1.0
Togo	92.8	80.6	5.8	6.4	94.4	36.8	23.3	34.4	43.1	43.1	0.0	90.3	90.3	0.0	100	93.9	6.1	0.0
Americas																		
Latin America and the Caribbean	ean																	
The Caribbean																		
Dominican Republic	56.3	37.4	11.8	7.0	38.6	5.9	20.3	12.4	38.0	38.0	0.0	82.6	82.6	0.0	100	76.5	23.5	0.0
Central America																		
Costa Rica	39.1	27.4	4.9	6.9	32.8	17.6	6.3	8.9	17.3	17.3	0.0	67.0	67.0	0.0	100	94.2	5.8	0.0
El Salvador	69.6	53.9	11.2	4.5	52.1	26.5	18.0	7.5	80.8	80.8	0.0	97.3	97.3	0.0	100	92.6	4.4	0.0
Guatemala	79.7	63.3	8.8	7.7	67.8	46.6	13.8	7.3	93.0	87.8	5.2	99.3	90.7	8.6	100	88.8	3.2	8.0

		Total	al			Employees	yees			Employers		0wn-a	Own-account workers	kers	Cont	ributing fa	<b>Contributing family workers</b>	irs
	Infe	ormal emp	Informal employment (%)	(%)	Info	ırmal emp	Informal employment (%)	(%)	Informal	Informal employment (%)	ent (%)	Informal	Informal employment (%)	nt (%)	Info	rmal emp	Informal employment (%)	()
	lstoT	In the informal sector	In the formal sector	spioqəsnoq uj	lstoT	In the informal sector	In the formal sector	splotasuod nl	lstoT	In the informal sector	splodəsuod al	lstoT	In the informal sector	sblodesuod nl	lstoT	In the informal sector	In the formal sector	spiouəsnou ui
Honduras	79.9	70.0	1.8	8.1	61.3	51.4	3.7	6.2	94.1	83.2	10.9	99.2	86.8	12.4	100	100.0	0.0	0.0
Mexico	53.4	29.1	19.8	4.5	46.3	15.2	24.4	6.7	41.2	41.2	0.0	9.99	9.99	0.0	100	39.9	60.1	0.0
Nicaragua	77.4	71.5	1.4	4.4	59.2	48.6	2.7	8.0	96.8	96.8	0.0	98.8	98.3	0.5	100	100.0	0.0	0.0
Panama	52.3	34.3	14.0	3.9	31.9	7.1	19.0	5.7	83.9	83.9	0.0	98.9	98.9	0.0	100	79.9	20.1	0.0
South America																		
Argentina	47.2	32.8	8.5	5.8	32.3	13.4	11.2	7.7	73.8	73.8	0.0	96.2	96.2	0.0	100	91.5	8.5	0.0
Bolivia, Pluri- national State of	83.1	68.7	12.4	1.9	64.8	28.0	31.5	5.3	80.5	80.5	0.0	93.3	93.3	0.0	100	95.1	4.9	0.0
Brazil	46.0	30.3	11.9	3.9	33.8	14.1	14.1	5.6	17.6	17.6	0.0	81.1	81.1	0.0	100	0.0	100.0	0.0
Chile	40.5	32.2	6.3	2.0	18.8	7.9	8.1	2.8	73.8	73.8	0.0	99.3	99.3	0.0	100	85.5	14.5	0.0
Colombia	60.6	55.9	3.0	1.8	26.8	17.4	5.8	3.6	89.3	89.3	0.0	93.7	93.7	0.0	100	95.4	4.6	0.0
Ecuador	59.0	46.4	11.1	1.6	44.6	26.5	15.3	2.8	31.7	31.7	0.0	77.2	77.2	0.0	100	74.4	25.6	0.0
Paraguay	70.6	46.0	17.6	7.0	61.4	17.7	31.3	12.5	33.4	33.4	0.0	85.7	85.7	0.0	100	100.0	0.0	0.0
Peru	69.2	55.5	11.8	1.9	45.2	18.5	22.5	4.2	50.3	50.3	0.0	89.6	89.6	0.0	100	89.6	10.4	0.0
Uruguay	24.5	19.6	1.9	3.0	12.6	5.9	2.6	4.2	6.2	6.2	0.0	63.1	63.1	0.0	100	96.9	3.2	0.0
Venezuela, Bolivarian Republic of	39.7	26.3	6.3	7.2	11.6	0.3	10.3	6.0	51.4	40.3	11.1	81.5	64.8	16.7	100	74.5	20.2	5.4
Northern America																		
United States	18.6	16.6	1.5	0.5	13.4	11.5	1.7	0.2	:	÷	:	69.1	64.7	4.4	100	97.6	0.0	2.4
Arab States																		
Iraq	66.9	59.7	7.2	0.0	53.7	43.6	10.0	0.0	72.4	72.4	0.0	÷	:	÷	:	:	:	:
Jordan	44.9	43.9	1.0	0.0	32.6	31.5	1.2	0.0	94.3	94.3	0.0	98.7	98.7	0.0	100	98.6	1.4	0.0
Occupied Palestinian Territory	64.3	31.2	30.8	2.3	64.4	22.7	41.7	0.0	37.6	37.6	0.1	57.6	51.4	6.2	100	59.7	24.2	16.1
Syrian Arab Republic	70.1	66.0	4.1	0.0	52.0	44.9	7.1	0.0	86.3	86.3	0.0	88.9	88.9	0.0	100	98.1	1.9	0.0
Yemen	77.8	68.5	9.0	0.3	64.0	47.6	15.7	0.7	70.8	70.8	0.0	93.7	93.7	0.0	100	91.2	8.8	0.0

		Total	a			Employees	yees			Employers		0wn-a	Own-account workers	rkers	Cont	<b>Contributing family workers</b>	imily work	ers
	-	ormal emp	Informal employment (%)	(%	Inf	ormal emp	Informal employment (%)	(9	Informa	Informal employment (%)	ent (%)	Informa	Informal employment (%)	ent (%)	Infe	Informal employment (%)	loyment (9	()
	Total	In the informal sector	In the formal sector	spiodəsuod ni	lstoT	In the informal sector	In the formal sector	spiodəsuod ni	lstoT	In the informal sector	splouəsnou ul	letoT	In the informal sector	spiouəsnou uj	lstoT	In the informal sector	In the formal sector	spiouəsnou ui
Asia and the Pacific																		
Eastern Asia																		
China	54.4	48.4	6.1	0.0	50.5	43.4	7.0	0.0	53.2	53.2	0.0	78.0	78.0	0.0	100	82.1	17.9	0.0
Japan	18.7	14.3	4.3	0.0	4.2	0.3	3.9	0.0	42.2	42.2	0.0	91.3	91.3	0.0	100	70.9	29.1	0.0
Korea, Republic of	31.5	26.1	5.5	0.0	28.3	25.4	2.9	0.0	5.9	5.9	0.0	35.2	35.2	0.0	100	33.0	67.0	0.0
Mongolia	53.5	45.7	2.1	5.7	43.6	40.4	2.2	0.9	50.7	49.5	1.2	85.1	62.3	22.8	100	67.3	22.7	10.1
South-Eastern Asia and the Pacific	acific																	
Pacific Islands																		
Samoa	35.7	21.7	10.4	3.7	25.4	10.8	12.2	2.4	85.3	77.2	8.1	95.9	86.9	0.0	100	71.7	3.9	24.4
South-Eastern Asia																		
Brunei Darussalam	31.9	4.0	22.3	5.7	31.4	1.2	24.0	6.2	6.4	6.4	0.0	57.7	57.7	0.0	100	56.4	43.6	0.0
Cambodia	93.1	77.5	14.5	1.0	87.1	55.2	29.6	2.2	86.4	86.4	0.0	97.2	97.2	0.0	100	95.5	4.5	0.0
Indonesia	85.6	67.5	5.8	12.2	80.7	60.7	9.3	10.8	56.2	49.6	6.7	90.1	76.9	13.2	100	74.3	9.2	16.5
Lao People's Demo- cratic Republic	93.6	86.7	6.8	0.1	62.8	20.2	42.3	0.3	76.8	76.8	0.0	0.66	99.0	0.0	100	99.5	0.5	0.0
Myanmar	85.7	71.5	14.0	0.3	87.9	66.1	21.1	0.6	61.1	61.1	0.0	76.6	76.6	0.0	100	74.3	25.7	0.0
Timor-Leste	71.8	56.9	11.0	3.9	40.4	5.9	25.3	9.2	63.8	63.8	0.0	94.9	94.9	0.0	100	98.7	1.3	0.0
Viet Nam	76.2	61.0	11.5	3.7	51.2	25.6	24.7	1.0	44.3	44.1	0.2	92.0	86.7	5.3	100	82.9	10.4	6.7
Southern Asia																		
Bangladesh	89.0	48.9	13.5	26.7	85.3	36.4	31.8	17.1	68.6	54.2	14.4	88.3	59.4	28.9	100	52.7	3.7	43.7
India	88.2	80.9	6.5	0.8	59.1	22.5	32.4	4.3	89.0	89.0	0.0	94.3	94.3	0.0	100	98.8	1.2	0.0
Nepal	94.3	90.7	3.4	0.2	67.9	47.3	19.3	1.4	92.4	92.4	0.0	100.0	100.0	0.0	100	100.0	0.0	0.0
Pakistan	82.4	77.6	4.0	0.8	62.7	52.3	8.5	2.0	61.4	61.4	0.0	93.7	93.7	0.0	100	96.9	3.1	0.0
Sri Lanka	70.4	60.6	7.9	2.0	52.9	36.4	12.9	3.6	61.6	61.6	0.0	93.5	93.5	0.0	100	92.3	7.7	0.0
		Total	a			Employees	yees			Employers		0wn-ad	Own-account workers	rkers	Cont	tributing fa	<b>Contributing family workers</b>	ers
---------------------------------------	------------	---------------------------	-------------------------	---------------	-------	---------------------------	-------------------------	---------------	----------	---------------------------	---------------	----------	---------------------------	---------------	-------	---------------------------	------------------------------------	---------------
	Infe	Informal employment (%)	loyment (9	()	Infe	ormal emp	Informal employment (%)	(9	Informal	Informal employment (%)	ent (%)	Informal	Informal employment (%)	ent (%)	Infe	ormal emp	Informal employment (%)	(ه
	Total	In the informal sector	In the formal sector	spiodəsuod ni	lstoT	In the informal sector	In the formal sector	spiotesuod ni	lstoT	In the informal sector	spiodəsuod ni	lstoT	In the informal sector	spioyəsnoy uj	Total	In the informal sector	In the formal sector	spiodəsuod ni
<b>Europe and Central Asia</b>																		
<b>Central and Western Asia</b>																		
Central Asia																		
Kyrgyzstan	48.6	37.2	9.6	1.8	39.0	27.6	8.3	3.1	11.8	11.8	0.0	49.6	49.3	0.3	100	53.2	46.8	0.0
Tajikistan	74.8	54.4	14.4	5.9	62.7	32.4	21.5	8.8	94.1	94.1	0.0	98.9	98.9	0.0	100	99.4	0.6	0.0
Western Asia																		
Armenia	52.1	39.2	1.4	11.6	23.4	21.2	1.9	0.4	0.9	0.9	0.0	90.8	62.3	28.5	100	76.1	3.8	20.1
Cyprus	15.1	15.0	0.1	0.0	5.9	5.8	0.1	0.0	90.8	90.8	0.0	90.2	90.2	0.0	100	93.8	6.2	0.0
Turkey	34.8	32.0	2.4	0.4	18.3	16.0	1.8	0.6	12.9	12.9	0.0	60.1	60.1	0.0	100	89.9	10.1	0.0
Eastern Europe																		
Bulgaria	15.9	15.0	0.9	0.0	7.9	7.0	0.9	0.0	61.4	61.4	0.0	93.6	93.6	0.0	100	62.1	37.9	0.0
Czech Republic	9.2	8.6	9.0	0.0	1.6	1.1	9.0	0.0	35.8	35.8	0.0	43.7	43.7	0.0	100	56.1	43.9	0.0
Hungary	12.2	11.8	0.4	0.0	1.7	1.4	0.3	0.0	61.9	61.9	0.0	84.2	84.2	0.0	100	80.0	20.0	0.0
Moldova, Republic of	28.9	21.9	9.9	0.4	9.6	1.3	7.7	0.6	0.0	0.0	0.0	74.8	74.8	0.0	100	57.3	42.7	0.0
Poland	38.0	20.1	17.8	0.0	26.7	4.7	22.1	0.0	54.7	54.7	0.0	90.0	0.06	0.0	100	97.3	2.7	0.0
Romania	28.9	27.4	1.5	0.0	2.8	0.7	2.1	0.0	48.6	48.6	0.0	96.9	96.9	0.0	100	100.0	0.0	0.0
Russian Federation	35.9	24.4	11.5	0.0	34.2	23.4	10.8	0.0	40.1	40.1	0.0	:	:	:	100	60.1	39.9	0.0
Slovakia	16.7	16.4	0.3	0.0	4.2	3.8	0.4	0.0	76.2	76.2	0.0	85.1	85.1	0.0	100	100.0	0.0	0.0
Northern, Southern and Western Europe	ern Europi	81																
Northern Europe																		
Denmark	11.2	4.1	7.2	0.0	3.1	2.2	0.9	0.0	0.0	0.0	0.0	100.0	100.0	0.0	100	21.2	78.8	0.0
Estonia	6.9	5.4	1.5	0.0	4.1	2.5	1.6	0.0	19.8	19.8	0.0	50.5	50.5	0.0	100	100.0	0.0	0.0
Finland	6.3	6.2	0.1	0.0	0.4	0.4	0.0	0.0	38.4	38.4	0.0	47.5	47.5	0.0	100	37.6	62.4	0.0
Iceland	4.9	4.6	0.3	0.0	3.5	3.3	0.2	0.0	8.6	8.6	0.0	15.6	15.6	0.0	100	13.9	86.1	0.0
Ireland	13.5	11.8	1.7	0.0	3.4	1.4	1.9	0.0	44.5	44.5	0.0	75.0	75.0	0.0	100	91.8	8.2	0.0
Latvia	13.2	11.2	2.0	0.0	9.0	6.9	2.1	0.0	26.1	26.1	0.0	66.4	66.4	0.0	100	85.7	14.3	0.0

		Total	8			Employees	yees			Employers		0wn-a	<b>Own-account workers</b>	'kers	Cont	ributing fa	<b>Contributing family workers</b>	ers
	Inf	Informal employment (%)	loyment ()	(%	Infe	ormal emp	Informal employment (%)	()	Informal	Informal employment (%)	ent (%)	Informal	Informal employment (%)	int (%)	Info	ırmal emp	Informal employment (%)	()
	lstoT	In the informal sector	In the formal sector	spiodəsuod ni	lstoT	In the informal sector	In the formal sector	splodəsuod al	Total	In the informal sector	splotasuot nl	Total	In the informal sector	splodəsuod al	lstoT	In the informal sector	ln the formal sector	spiodesuod ni
Lithuania	12.6	8.2	4.4	0.0	8.2	3.7	4.5	0.0	20.6	20.6	0.0	52.1	52.1	0.0	100	80.2	19.8	0.0
Norway	7.4	7.1	0.2	0.0	3.7	3.6	0.1	0.0	40.7	40.7	0.0	52.7	52.7	0.0	100	56.1	43.9	0.0
Sweden	8.2	2.6	5.6	0.0	8.1	2.0	6.1	0.0	6.1	6.1	0.0	10.2	10.2	0.0	100	0.0	100.0	0.0
United Kingdom	13.6	13.5	0.1	0.0	3.8	3.7	0.1	0.0	75.4	75.4	0.0	84.2	84.2	0.0	:	÷	:	:
Southern Europe																		
Albania	61.0	29.1	0.5	31.4	18.8	6.5	1.3	10.9	57.1	57.1	0.0	77.6	46.9	30.7	100	41.9	0.1	58.0
Bosnia and Herzegovina	30.1	30.0	0.1	0.0	0.0	0.0	0.0	0.0	100.0	100.0	0.0	100.0	100.0	0.0	100	98.8	1.2	0.0
Croatia	13.0	12.8	0.2	0.0	4.7	4.5	0.2	0.0	67.8	67.8	0.0	93.6	93.6	0.0	100	100.0	0.0	0.0
Greece	32.8	32.2	0.5	0.0	4.1	3.5	0.7	0.0	59.9	59.9	0.0	91.0	91.0	0.0	100	94.9	5.1	0.0
Italy	19.0	16.0	3.0	0.0	3.7	1.6	2.1	0.0	54.5	54.5	0.0	66.0	66.0	0.0	100	42.4	57.6	0.0
Malta	8.1	7.9	0.2	0.0	2.5	2.3	0.1	0.0	31.2	31.2	0.0	56.2	56.2	0.0	100	54.2	45.8	0.0
Portugal	12.1	11.9	0.3	0.0	2.7	2.5	0.2	0.0	57.8	57.8	0.0	78.0	78.0	0.0	100	82.2	17.9	0.0
Serbia	22.1	6.4	10.7	5.0	8.8	0.9	7.0	0.8	0.9	0.9	0.0	45.5	23.7	21.8	100	14.0	86.0	0.0
Slovenia	5.0	4.8	0.3	0.0	0.9	0.7	0.2	0.0	22.4	22.4	0.0	44.8	44.8	0.0	100	59.3	40.7	0.0
Spain	27.3	17.3	9.9	0.0	18.0	0.9	12.0	0.0	56.6	56.6	0.0	78.3	78.3	0.0	100	84.3	15.8	0.0
Western Europe																		
Austria	10.0	8.8	1.2	0.0	5.2	4.0	1.2	0.0	28.4	28.4	0.0	49.0	49.0	0.0	100	55.1	44.9	0.0
Belgium	13.5	10.8	2.6	0.0	6.0	3.1	2.8	0.0	62.7	62.7	0.0	82.5	82.5	0.0	100	80.8	19.2	0.0
France	9.8	5.4	4.4	0.0	5.6	0.9	4.7	0.0	35.8	35.8	0.0	47.2	47.2	0.0	100	58.0	42.0	0.0
Germany	10.2	9.7	0.5	0.0	7.6	7.1	0.5	0.0	21.4	21.4	0.0	48.1	48.1	0.0	100	44.9	55.2	0.0
Luxembourg	1.2	0.9	0.3	0.0	0.9	0.9	0.1	0.0	0.0	0.0	0.0	0.4	0.4	0.0	100	33.9	66.1	0.0
Netherlands	9.4	8.4	0.9	0.0	2.9	2.0	0.9	0.0	39.4	39.4	0.0	46.7	46.7	0.0	100	56.1	43.9	0.0
Switzerland	10.4	4.2	6.2	0.0	7.1	0.5	6.7	0.0	16.5	16.5	0.0	44.3	44.3	0.0	100	35.6	64.4	0.0
Note: "…" no data available																		

culture)
iding agri
nt (exclu
nployme
itus in ei
nt by sta
mployment
tale
in to
oyment
ial emplo
f inform
- Share of
Table B.4 -

		P	Total			Emplo	Employees			Employers		0wn-a	<b>Own-account workers</b>	kers	Cont	<b>Contributing family workers</b>	mily work	ers
	Ē	ormal em	Informal employment (%)	%)	Ē	ormal emp	nformal employment (%)	(%)	Informa	Informal employment (%)	ent (%)	Informal	Informal employment (%)	int (%)	Info	Informal employment (%)	loyment (9	(%)
	lstoT	In the informal sector	In the formal sector	splodøsuod al	lstoT	In the informal sector	In the formal sector	spiotasuot ni	lstoT	In the informal sector	sbiodəsuod ni	lstoT	In the informal sector	sblodesuod ni	lstoT	In the informal sector	In the formal sector	splodəsuod ni
Africa																		
Northern Africa																		
Egypt	49.8	43.4	5.7	0.7	35.0	26.7	7.4	0.9	92.9	92.9	0.0	99.0	0.66	0.0	100	98.1	1.9	0.0
Morocco	75.6	53.3	20.4	1.9	66.5	32.6	31.0	2.9	63.2	63.2	0.0	96.4	96.4	0.0	100	100.0	0.0	0.0
Tunisia	53.5	47.6	5.5	0.5	41.3	33.7	6.9	0.6	98.7	98.7	0.0	99.5	99.5	0.0	100	100.0	0.0	0.0
Sub-Saharan Africa																		
Central Africa																		
Angola	89.5	59.2	29.5	0.9	81.8	26.8	53.5	1.6	94.8	94.8	0.0	99.1	99.1	0.0	100	95.0	5.0	0.0
Cameroon	79.0	62.5	14.7	1.8	60.5	23.4	32.1	5.0	50.1	50.1	0.0	89.1	89.1	0.0	100	71.4	28.6	0.0
Chad	75.8	68.0	7.8	0.0	34.5	23.8	10.7	0.0	88.9	88.9	0.0	93.9	93.9	0.0	100	64.1	36.0	0.0
Congo	85.0	55.6	26.8	2.6	82.4	39.7	39.1	3.7	78.3	78.3	0.0	90.5	90.4	0.1	100	73.9	21.3	4.8
Congo, Democratic Republic of the	76.2	67.0	9.3	0.0	39.1	14.7	24.5	0.0	82.8	82.8	0.0	97.9	97.9	0.0	100	90.9	9.1	0.0
Eastern Africa																		
Comoros	78.5	69.5	9.0	0.0	64.1	48.6	15.4	0.0	84.8	84.8	0.0	98.1	98.1	0.0	100	91.8	8.2	0.0
Madagascar	79.8	47.3	12.4	20.1	60.9	21.8	28.5	10.7	62.0	62.0	0.0	89.6	83.3	6.3	100	33.4	4.3	62.3
Malawi	67.2	49.4	10.8	7.0	56.8	17.0	23.0	16.8	41.6	41.6	0.0	74.2	74.2	0.0	100	63.0	37.0	0.0
Rwanda	78.6	67.0	4.6	7.1	76.0	58.7	6.8	10.5	70.9	70.9	0.0	87.0	87.0	0.0	100	100.0	0.0	0.0
Tanzania, United Republic of	69.3	48.1	15.2	6.0	60.5	21.9	38.1	0.5	53.7	52.0	1.8	74.8	70.2	4.6	100	28.7	14.8	56.6
Uganda	83.4	67.1	16.4	0.0	83.1	46.0	37.1	0.0	62.3	62.3	0.0	84.9	84.9	0.0	100	87.8	12.2	0.0
Zambia	71.5	52.3	6.3	12.9	56.3	29.1	10.1	17.1	56.2	43.9	12.3	92.8	86.1	6.7	100	82.4	8.3	9.3
Southern Africa																		
Botswana	55.1	41.2	6.8	7.2	47.6	31.3	8.0	8.4	9.66	9.66	0.0	100.0	100.0	0.0	100	100.0	0.0	0.0
Namibia	61.1	46.3	6.5	8.3	59.8	43.1	8.4	8.3	30.8	26.2	4.6	68.9	60.8	8.1	100	83.6	0.0	16.4
South Africa	34.0	21.4	4.9	7.7	27.3	13.4	5.6	8.4	39.3	37.8	1.5	88.1	82.3	5.8	100	70.2	29.8	0.0

		Ţġ	Total			Employees	yees			Employers		0wn-ac	Own-account workers	kers	Cont	ributing fa	<b>Contributing family workers</b>	irs
	Ţ	ormal emp	Informal employment (%)	(%	Infe	ormal emp	Informal employment (%)		Informal	Informal employment (%)	ent (%)	Informal	Informal employment (%)	int (%)	Info	ırmal empl	Informal employment (%)	()
	letoT	In the informal sector	In the formal sector	sblodəsuod ni	lstoT	In the informal sector	In the formal sector	sblodesuod ni	Total	In the informal sector	spiodəsuod ni	lstoT	In the informal sector	spiodəsuod ni	lstoT	In the informal sector	In the formal sector	sblodesuod ni
Western Africa																		
Benin	90.6	83.7	4.9	2.0	65.6	31.2	23.7	10.7	71.7	71.7	0.0	96.4	96.4	0.0	100	94.2	5.8	0.0
Burkina Faso	88.4	67.9	14.5	6.0	77.8	33.8	37.3	6.7	64.0	61.1	2.9	94.6	93.5	1.1	100	54.2	5.4	40.4
Cabo Verde	46.8	29.3	7.2	10.2	52.7	29.1	9.7	13.8	49.4	49.4	0.0	22.9	22.9	0.0	100	100.0	0.0	0.0
Côte d'Ivoire	87.7	75.8	9.0	2.9	76.9	49.0	21.8	6.1	80.7	79.8	6.0	94.9	94.2	0.7	100	92.1	6.1	1.9
Gambia	68.4	52.1	9.2	7.2	57.8	19.1	25.1	13.7	54.5	52.5	2.0	71.9	67.6	4.3	100	73.8	23.7	2.5
Ghana	82.6	73.2	8.6	0.9	63.2	42.4	18.7	2.0	91.6	91.6	0.0	98.3	98.3	0.0	100	94.8	5.2	0.0
Liberia	77.5	64.6	10.0	2.9	62.0	31.1	25.1	5.7	53.2	50.7	2.5	86.1	84.8	1.3	100	83.4	15.6	1.0
Mali	86.4	70.5	10.6	5.3	89.0	36.8	43.4	8.8	78.0	76.8	1.3	85.6	82.4	3.2	100	64.4	5.6	30.0
Niger	85.1	72.6	10.3	2.2	68.8	21.7	38.6	8.5	49.2	49.2	0.0	91.2	91.0	0.1	100	80.3	19.3	0.5
Nigeria	89.0	82.9	4.2	1.8	56.8	38.2	16.1	2.5	0.0	0.0	0.0	99.7	99.7	0.0	100	64.2	4.7	31.1
Senegal	87.0	59.1	19.1	8.7	74.9	17.8	45.1	12.1	74.0	69.2	4.8	95.8	90.4	3.6	100	73.6	1.6	24.8
Sierra Leone	86.0	80.8	2.6	2.6	48.8	34.6	11.6	2.6	96.2	70.7	25.5	96.0	94.0	2.0	100	96.2	3.0	0.7
Togo	90.0	73.6	6.7	9.8	94.3	36.2	21.7	36.4	38.2	38.2	0.0	86.7	86.7	0.0	100	94.7	5.3	0.0
Americas																		
Latin America and the Caribbean	ean																	
The Caribbean																		
Dominican Republic	52.6	31.3	13.1	8.1	38.9	5.4	20.5	13.0	32.3	32.3	0.0	79.3	79.3	0.0	100	67.3	32.7	0.0
Central America																		
Costa Rica	37.4	25.5	4.2	7.8	30.9	15.6	5.3	10.0	16.0	16.0	0.0	67.7	67.7	0.0	100	92.6	7.5	0.0
El Salvador	63.1	48.1	9.4	5.6	45.1	22.1	14.3	8.8	75.0	75.0	0.0	96.6	96.6	0.0	100	95.4	4.7	0.0
Guatemala	72.7	54.5	12.0	6.1	59.4	32.8	17.6	9.0	91.3	90.7	0.6	99.0	98.6	0.5	100	91.3	7.8	0.9
Honduras	72.7	65.7	2.6	4.3	54.0	42.0	4.5	7.5	87.6	87.6	0.0	99.1	98.9	0.1	100	100.0	0.0	0.0
Mexico	53.2	33.1	14.9	5.2	42.5	16.5	18.8	7.3	48.5	48.5	0.0	84.0	84.0	0.0	100	63.5	36.6	0.0
Nicaragua	68.6	60.1	2.1	6.5	49.4	35.6	3.5	10.4	93.3	93.3	0.0	98.3	97.6	0.7	100	100.0	0.0	0.0
Panama	46.0	26.2	15.2	4.6	29.9	4.7	19.1	6.1	82.1	82.1	0.0	98.6	98.6	0.0	100	29.4	70.6	0.0

		Total	al			Employees	lees			Employers		Own-ad	Own-account workers	kers	Conti	ibuting fa	Contributing family workers	irs
	Inf	ormal emp	Informal employment (%)	(%)	Info	Informal employment (%)	oyment (%		Informal	Informal employment (%)	ent (%)	Informal	Informal employment (%)	nt (%)	Info	rmal empl	Informal employment (%)	()
	lstoT	In the informal sector	In the formal sector	spiotasuot ni	lstoT	In the informal sector	in the formal sector	sblodesuod ni	lstoT	In the informal sector	spiouəsnou ui	lstoT	In the informal sector	spiodesuod ni	letoT	In the informal sector	In the formal sector	splodəsuod nl
South America																		
Argentina	47.0	32.7	8.5	5.9	32.1	13.3	11.1	7.8	73.8	73.8	0.0	96.3	96.3	0.0	100	92.1	7.9	0.0
Bolivia, Plurinational State of	75.6	55.3	17.5	2.8	63.1	25.3	32.3	5.6	67.6	67.6	0.0	89.9	6.68	0.0	100	81.8	18.2	0.0
Brazil	42.5	26.4	11.8	4.3	32.8	11.9	14.9	5.9	14.5	14.5	0.0	77.8	77.8	0.0	100	0.0	100.0	0.0
Chile	39.3	30.2	6.8	2.2	17.8	0.9	8.7	3.1	72.0	72.0	0.0	99.3	99.3	0.0	100	83.2	16.8	0.0
Colom bia	55.4	50.0	3.4	2.0	25.9	16.2	6.0	3.6	86.3	86.3	0.0	92.4	92.4	0.0	100	93.4	6.7	0.0
Ecuador	50.4	35.9	12.4	2.1	38.4	19.4	15.5	3.4	20.0	20.0	0.0	70.5	70.5	0.0	100	46.5	53.5	0.0
Paraguay	64.4	34.9	20.8	8.7	59.9	15.4	31.4	13.1	30.7	30.7	0.0	79.2	79.2	0.0	100	100.0	0.0	0.0
Peru	59.1	40.5	15.9	2.7	41.5	12.3	24.5	4.8	34.7	34.7	0.0	84.6	84.6	0.0	100	67.9	32.1	0.0
Uruguay	24.1	18.8	2.0	3.3	11.9	4.8	2.7	4.4	6.5	6.5	0.0	65.7	65.7	0.0	100	95.1	4.9	0.0
Venezuela, Bolivarian Republic of	38.3	24.2	6.4	7.7	11.4	0.2	10.3	1.0	47.7	34.5	13.2	80.6	62.1	18.5	100	71.1	22.5	6.4
Northern America																		
United States	18.3	16.3	1.5	0.5	13.3	11.4	1.7	0.2	÷	÷	÷	68.4	63.8	4.6	100	97.3	0.0	2.7
Arab States																		
Iraq	66.7	59.5	7.2	0.0	53.1	43.0	10.1	0.0	72.4	72.4	0.0	:	÷	:	:	÷	:	:
Jordan	43.0	42.0	1.0	0.0	30.7	29.5	1.2	0.0	94.1	94.1	0.0	98.5	98.5	0.0	100	97.8	2.2	0.0
Occupied Palestinian Territory	60.1	25.8	34.2	0.1	62.6	19.3	43.3	0.0	35.6	35.6	0.0	48.7	48.4	0.3	100	45.2	54.5	0.2
Syrian Arab Republic	61.0	55.6	5.4	0.0	49.3	41.7	7.6	0.0	81.1	81.1	0.0	84.5	84.5	0.0	100	91.6	8.4	0.0
Yemen	69.0	56.3	12.2	0.5	58.2	39.6	17.9	0.8	56.6	56.6	0.0	90.6	90.6	0.0	100	75.5	24.6	0.0

		Total	al			Empl	Employees			Employers		0wn-a	Own-account workers	rkers	Cont	tributing f	<b>Contributing family workers</b>	ers
	Inf	ormal emp	Informal employment (%)	(%	Ē	ormal em	Informal employment (%)	(%)	Informa	Informal employment (%)	ent (%)	Informa	Informal employment (%)	ent (%)	Inf	ormal emp	Informal employment (%)	%)
	leto	n the informal ector	n the formal ector	splodəsuod n	leto	n the informal ector	n the formal ector	splodəsuod n	leto	n the informal ector	spiodəsuod n	leto	n the informal ector	spiodəsuod n	leto	n the informal ector	n the formal ector	spionasuon n
Europe and Central Asia																		
Central and Western Asia																		
Central Asia																		
Kyrgyzstan	42.1	32.7	7.1	2.3	38.6	27.0	8.4	3.2	10.7	10.7	0.0	50.0	49.5	0.5	100	53.2	46.8	0.0
Tajikistan	70.5	39.8	20.5	10.2	62.1	22.0	26.7	13.3	91.0	91.0	0.0	98.6	98.6	0.0	100	99.2	0.8	0.0
Western Asia																		
Armenia	26.5	24.0	2.1	0.4	22.4	20.1	1.9	0.4	0.0	0.0	0.0	57.4	57.2	0.3	100	19.0	79.3	1.7
Cyprus	14.1	14.0	0.1	0.0	6.0	5.9	0.1	0.0	90.2	90.2	0.0	89.5	89.5	0.0	100	100.0	0.0	0.0
Turkey	21.9	19.3	2.2	0.5	16.4	14.0	1.8	9.0	11.6	11.6	0.0	51.0	51.0	0.0	100	72.4	27.6	0.0
Eastern Europe																		
Bulgaria	13.6	12.7	0.9	0.0	7.0	6.1	0.9	0.0	60.8	60.8	0.0	92.2	92.2	0.0	100	61.0	39.0	0.0
Czech Republic	9.1	8.6	0.5	0.0	1.7	1.1	9.0	0.0	36.4	36.4	0.0	44.0	44.0	0.0	100	64.5	35.5	0.0
Hungary	10.5	10.2	0.4	0.0	1.7	1.3	0.4	0.0	61.8	61.8	0.0	81.6	81.6	0.0	100	75.7	24.3	0.0
Moldova, Republic of	13.7	7.6	5.6	0.6	7.0	0.5	5.9	9.0	0.0	0.0	0.0	68.4	68.4	0.0	100	20.3	79.7	0.0
Poland	31.6	12.1	19.5	0.0	26.4	4.4	22.0	0.0	51.6	51.6	0.0	81.3	81.3	0.0	100	87.8	12.2	0.0
Romania	11.1	9.3	1.9	0.0	2.7	0.7	2.1	0.0	48.1	48.1	0.0	89.6	89.6	0.0	100	100.0	0.0	0.0
Russian Federation	35.6	24.3	11.3	0.0	33.9	23.4	10.6	0.0	40.6	40.6	0.0	:	÷	:	100	59.3	40.7	0.0
Slovakia	16.5	16.2	0.3	0.0	4.2	3.9	0.4	0.0	75.1	75.1	0.0	84.7	84.7	0.0	100	100.0	0.0	0.0
Northern, Southern and Western Europe	ern Europu	0																
Northern Europe																		
Denmark	10.8	4.0	6.8	0.0	3.1	2.2	0.9	0.0	0.0	0.0	0.0	100.0	100.0	0.0	100	21.6	78.4	0.0
Estonia	6.3	4.8	1.5	0.0	4.0	2.4	1.6	0.0	19.2	19.2	0.0	45.5	45.5	0.0	100	100.0	0.0	0.0
Finland	5.1	5.1	0.0	0.0	0.4	0.4	0.0	0.0	38.4	38.4	0.0	45.1	45.1	0.0	100	45.3	54.7	0.0
Iceland	4.8	4.5	0.3	0.0	3.5	3.3	0.2	0.0	8.6	8.6	0.0	16.6	16.6	0.0	100	13.9	86.1	0.0
Ireland	11.1	9.4	1.8	0.0	3.4	1.4	2.0	0.0	42.0	42.0	0.0	76.7	76.7	0.0	100	88.1	11.9	0.0
Latvia	11.2	9.1	2.0	0.0	8.7	6.5	2.2	0.0	20.7	20.7	0.0	57.7	57.7	0.0	100	100.0	0.0	0.0
Lithuania	9.8	5.6	4.2	0.0	7.7	3.5	4.3	0.0	21.4	21.4	0.0	47.1	47.1	0.0	100	35.0	65.0	0.0

		2	Total			Employees	yees			Employers		0wn-a	Own-account workers	rkers	Cont	ributing fa	<b>Contributing family workers</b>	ers
	Ē	formal em	Informal employment (%)	%)	Info	ırmal emp	Informal employment (%)	()	Informal	Informal employment (%)	ent (%)	Informal	Informal employment (%)	ent (%)	Info	ırmal emp	Informal employment (%)	()
	letoT	In the informal sector	In the formal sector	splodasuod nl	lstoT	In the informal sector	In the formal sector	spiodesuod ni	lstoT	In the informal sector	spiodəsuod ni	lstoT	In the informal sector	spiodəsuod ni	lstoT	In the informal sector	in the formal sector	spiodəsuod ni
Norway	6.9	6.7	0.2	0.0	3.7	3.6	0.1	0.0	37.4	37.4	0.0	53.2	53.2	0.0	100	56.9	43.1	0.0
Sweden	8.3	2.6	5.7	0.0	8.2	2.0	6.2	0.0	5.3	5.3	0.0	10.5	10.5	0.0	100	0.0	100.0	0.0
United Kingdom	13.4	13.3	0.1	0.0	3.8	3.7	0.1	0.0	76.6	76.6	0.0	84.9	84.9	0.0	:	÷	:	÷
Southern Europe																		
Albania	33.1	24.9	0.9	7.3	17.8	6.4	1.2	10.2	55.4	55.4	0.0	62.0	61.9	0.1	100	99.4	0.6	0.0
Bosnia and Herzegovina	13.2	13.1	0.2	0.0	0.0	0.0	0.0	0.0	100.0	100.0	0.0	100.0	100.0	0.0	100	92.2	7.8	0.0
Croatia	10.8	10.6	0.2	0.0	4.7	4.5	0.2	0.0	68.6	68.6	0.0	92.3	92.3	0.0	:	÷	:	÷
Greece	25.5	25.0	0.6	0.0	4.0	3.4	0.6	0.0	58.6	58.6	0.0	87.6	87.6	0.0	100	88.0	12.0	0.0
Italy	18.3	15.2	3.1	0.0	3.7	1.6	2.1	0.0	53.5	53.5	0.0	65.3	65.3	0.0	100	40.1	59.9	0.0
Malta	7.7	7.5	0.2	0.0	2.5	2.4	0.1	0.0	31.6	31.6	0.0	55.9	55.9	0.0	100	54.2	45.8	0.0
Portugal	10.5	10.2	0.3	0.0	2.6	2.3	0.2	0.0	57.5	57.5	0.0	75.8	75.8	0.0	100	74.1	25.9	0.0
Serbia	15.3	3.6	5.7	6.0	7.7	0.6	6.2	0.9	0.0	0.0	0.0	62.4	22.9	39.5	100	5.3	94.7	0.0
Slovenia	4.1	3.9	0.2	0.0	0.9	0.7	0.2	0.0	22.4	22.4	0.0	42.2	42.2	0.0	100	59.1	40.9	0.0
Spain	26.8	16.4	10.4	0.0	18.4	6.0	12.4	0.0	55.5	55.5	0.0	77.1	77.1	0.0	100	81.3	18.7	0.0
Western Europe																		
Austria	8.9	7.7	1.2	0.0	5.2	4.0	1.2	0.0	27.0	27.0	0.0	52.9	52.9	0.0	100	34.2	65.8	0.0
Belgium	12.8	10.1	2.7	0.0	6.0	3.1	2.9	0.0	62.8	62.8	0.0	81.6	81.6	0.0	100	75.4	24.6	0.0
France	8.9	4.5	4.4	0.0	5.5	0.8	4.8	0.0	31.3	31.3	0.0	46.7	46.7	0.0	100	65.0	35.0	0.0
Germany	10.1	9.6	0.5	0.0	7.6	7.1	0.5	0.0	18.1	18.1	0.0	48.5	48.5	0.0	100	67.9	32.1	0.0
Luxem bourg	1.0	0.9	0.1	0.0	0.9	0.9	0.1	0.0	0.0	0.0	0.0	0.5	0.5	0.0	100	23.1	76.9	0.0
Netherlands	9.2	8.3	1.0	0.0	3.0	2.0	0.9	0.0	39.4	39.4	0.0	46.8	46.8	0.0	100	53.5	46.5	0.0
Switzerland	10.1	3.9	6.2	0.0	7.2	0.5	6.7	0.0	15.1	15.1	0.0	46.3	46.3	0.0	100	22.7	77.3	0.0
<b>ote:</b> "" no data available																		

Note: "..." no data available

Table B.5 – Distribution of workers in informal employment and in formal employment by employment status and sex (including agriculture)

					Total										Men									3	Women				
	Info	ırmal e	mployi	Informal employment (%)	(%	For	Formal employment (9	ployme	ent (%)		Inform	Informal employment (%)	ployme	ent (%)		Forma	al emp	Formal employment (%)	nt (%)	Ξ	aforma	Informal employment (%)	ymen	t (%)		Forma	l empl	Formal employment (%)	(%)
	letoT	səəyolqm3	Employers	own-account Workers	Contributing family workers	letoT	Employees	Gwn-account Employers	Own-account workers Contributing	family workers	letoT	Employees	Own-account Employers	workers Contributing	family workers	Total Profess	Employees	Gwn-account Employers	workers Contributing	family workers Total	Employees	Employers	tnuosse.nw0	workers Contributing	tamily workers	letoT 	Employees	Own-account Employers	workers Contributing
Africa																													
Northern Africa																													
Egypt	100	41.2	18.8	20.5	19.5	100	99.0	0.8	0.3 0	0.0	100 4	46.9 22	22.2 22	22.4 8.	8.5 10	100 98.7		1.0 0.4	4 0.0	100	16.	5 3.4	11.3	3 68.8	8 100	00 99.	9 0.1	1 0.1	0.0
Morocco	100	38.1	2.3	32.9	26.7	100	83.7	5.7 1	10.6 0	0.0	100 4	42.4 2	2.9 39	39.1 15	15.5 10	100 82.	2	6.3 11.2	.2 0.0	100	26.4	4 0.5	16.0	0 57.2	2 100	00 87.1	ŝ	5 9.4	0.0
Tunisia	100	53.1	11.6	29.5	5.8	100	99.7	0.2	0.2 0	0.0	100 4	49.7 13	13.4 32	32.3 4.	4.6 10	100 99.	2	0.3 0.2	2 0.0	100	64.	6 5.1	20.1	1 10.2	2 100	00 100.	0.0	0.0 0.0	0.0
Sub-Saharan Africa																													
Central Africa																													
Angola	100	42.1	1.7	48.0	8.2	100	97.5	0.7	1.8 0	0.0	100 5	51.1 2	2.0 41	41.1 5.	5.8 10	100 95.	95.9 1.	1.0 3.1	1 0.0	100	30.7	7 1.4	56.7	7 11.3	3 100	.66 00	.3 0.1	1 0.5	0.0
Cameroon	100	16.1	3.3	51.5	29.2	100	75.4	6.7 1	17.9 0	0.0 1	100 2	23.1 4	4.9 50	50.3 21	21.8 10	100 76.2		6.6 17.	17.2 0.0	100	<u>∞</u>	8 1.5	52.8	8 36.9	9 100	00 72.	5 7.2	2 20.3	3 0.0
Chad	100	3.2	0.3	65.7	30.8	100	86.8	0.4 ]	12.8 0	0.0 1	100	5.6 0	0.4 81	81.8 12	12.2 10	100 91.	91.6 0.	0.4 8.1	1 0.0	100	0.4	t 0.2	47.0	0 52.4	1 100	00 57.8	.8 0.7	7 41.5	5 0.0
Congo	100	24.7	1.0	71.9	2.3	100	39.0	1.8	59.2 0	0.0 1	100 3	39.4 1	1.5 57	57.1 2.	2.1 10	100 52	52.3 2.	2.8 45.	45.0 0.0	100	9.4	t 0.5	87.6	6 2.6	100	00 18.1	0	9 81.1	1 0.0
Congo, Democratic Republic of the	100	11.0	4.0	69.2	15.8	100	88.3	4.3	7.4 0	0.0	100 1	13.1 6	6.2 71	71.0 9.	9.8 10	100 86.	86.6 5.	5.7 7.7	7 0.0	100	0 10.0	0 2.1	66.	9 21.1	1 100	91.7	.7 2.0	0 6.4	0.0
Eastern Africa																													
Comoros	100	29.0	3.0	66.0	2.0	100	93.2	2.7	4.2 0	0.0 1	100 3	32.4 3	3.8 61	61.7 2.	2.1 10	100 93.	93.0 3.	3.6 3.4	4 0.0	100	21.4	4 0.9	75.9	9 1.9	100	00 92.	5 0.7	7 6.8	0.0
Madagascar	100	7.4	3.5	46.4	42.7	100	66.3	10.5 2	23.2 0	0.0 1	100	9.1 5	5.3 61	61.0 24	24.7 10	100 65.	65.8 11	11.3 23.	23.0 0.0	100	5.7	7 1.7	31.5	5 61.1	1 100	00 67.7	∞.	8 23.	.5 0.0
Malawi	100	36.2	0.8	55.4	7.6	100	47.4	2.8 4	49.8 0	0.0 1	100 4	42.1 1	1.0 50	50.6 6.	6.3 10	100 52.7	.7 4.1	.1 43.2	.2 0.0	100	30.	8 0.6	59.8	8 8.9	100	00 41.	9 1.	8 56.4	4 0.0
Rwanda	100	18.4	0.3	72.6	8.7	100	67.9	2.8 2	29.4 0	0.0 1	100 2	25.2 0	0.4 68	68.8 5.	5.5 10	100 72.4		3.2 24.4	.4 0.0	100	0 12.2	2 0.3	76.0	0 11.5	5 100	00 62.	.5 1.9	9 35.7	7 0.0
Tanzania, United Republic of	100	12.2	1.4	42.7	43.7	100	60.3	10.1 2	29.6 0	0.0	100 1	16.6 2	2.0 46.	6	34.6 10	100 62.7		11.7 25.6	.6 0.0	100	7.8	8 0.8	38.3	3 53.2	2 100	00 55.	.8 7.4	4 36.8	8 0.0
Uganda	100	19.4	1.1	54.2	25.3	100	48.0	4.7 4	47.3 0	0.0 1	100 2	25.5 1	1.5 50	50.7 22	22.3 10	100 51.5		5.2 43.3	.3 0.0	100	13.4	4 0.7	57.7	7 28.2	2 100	00 42.	0 3.	9 54.1	1 0.0
Zambia	100	14.7	0.3	48.7	36.3	100	82.0	0.7 1	17.3 0	0.0	100 2	20.4 0	0.3 60	60.7 18	18.6 10	100 84.9		0.5 14.6	.6 0.0	100	8.8	3 0.2	37.3	3 53.8	3 100	00 73.0	.0 1.3	3 25.7	7 0.0

					Total										Men									Won	Women				
	Info	ırmal e	mploy	Informal employment (%)	(%	Forn	Formal employmen	ployme	ent (%)		Inform	Informal employment (%)	loymen	ıt (%)		Formal	Formal employment (%)	yment	(%)	Info	ırmal e	mploy	Informal employment (%)	(%	Ē	Formal employment (%)	nployn	nent ('	(%)
	letoT	Employees	Employers	workers Own-account	Contributing family workers	letoT	Employees	Employers	Own-account workers Contributing	family workers	letoT	Employees	Own-account Employers	workers Contributing	tamily workers Total	Employees	Employers	Own-account workers	Contributing family workers	letoT	Employees	Employers	workers workers	Contributing family workers	letoT	Employees	Employers	workers Workers	Contributing family workers
Southern Africa																													
Botswana	100	70.8	4.8	17.3	7.1	100	100.0	0.0	0.0 0.0		100 68	68.4 6.7	7 18.1	.1 6.8	100	0 99.9	9 0.1	0.0	0.0	100	73.5	2.6	16.4	7.5	100	100.0	0.0	0.0	0.0
Namibia	100	61.1	2.5	28.8	1.7	100	79.8	6.1	14.1 0.0		100 70	70.3 2.8	8 20.4	.4 6.6	100	0 80.9	9 8.1	11.0	0.0	100	51.5	2.2	37.5	8.8	100	78.7	3.7	17.6	0.0
South Africa	100	70.2	5.8	23.2	0.8	100	93.5	4.9	1.6 0.0		100 66	66.9 9.0	23	.4 0.7	100	0 91.6	g. 6.6	1.8	0.0	100	74.2	2.0	22.9	1.0	100	96.3	2.4	1.3	0.0
Western Africa																													
Benin	100	6.4	1.1	78.4	14.2	100	61.1	7.2	31.8 0.0		100 8.	8.9 1.7	7 81.0	.0 8.3	100	0 68.2	2 6.9	24.9	0.0	100	3.8	0.5	76.0	19.8	100	39.5	7.2	53.3	0.0
Burkina Faso	100	6.3	0.7	42.5	50.6	100	31.5	5.4 (	63.1 0.0	_	100 8.	8.7 0.9	9 62.2	.2 28.2	2 100	0 26.3	3 5.5	68.2	0.0	100	3.8	0.4	21.7	74.1	100	47.7	2.5	49.8	0.0
Cabo Verde	100	70.3	3.2	11.3	15.2	100	45.5	3.5	51.0 0.0		100 76	76.7 2.1	1 11.7	.7 9.5	100	0 43.8	3 5.5	50.7	0.0	100	61.0	4.4	10.7	23.9	100	47.4	0.7	52.0	0.0
Côte d'Ivoire	100	19.7	1.1	62.0	17.3	100	72.7	1.7	25.6 0.0		100 25	25.8 1.4	60	.2 12.6	3 100	0 76.3	3 1.7	22.0	0.0	100	10.0	0.5	64.8	24.8	100	58.3	1.6	40.1	0.0
Gambia	100	17.2	1.3	59.2	22.3	100	41.5	3.6	54.9 0.0		100 23	23.0 1.4	4 55.3	.3 20.3	3 100	0 43.8	3.5	52.8	0.0	100	10.7	1.3	63.2	24.8	100	33.3	3.8	62.9	0.0
Ghana	100	17.2	6.9	52.5	23.4	100	91.5	4.3	4.2 0.0		100 24	24.5 7.8	8 50.0	.0 17.7	7 100	0 93.5	5 4.1	2.4	0.0	100	10.3	6.2	54.9	28.6	100	86.7	4.8	8.6	0.0
Liberia	100	16.3	1.5	69.0	13.2	100	55.2	5.4	39.4 0.0		100 23	23.9 1.5	5 63.1	.1 11.5	5 100	0 64.1	1 5.1	30.8	0.0	100	8.6	1.5	75.1	14.9	100	33.4	6.2	60.4	0.0
Mali	100	12.6	1.2	70.5	15.8	100	18.4	2.2	79.4 0.0		100 17	17.1 1.4	4 68.1	.1 13.4	t 100	0 21.6	3.1	75.2	0.0	100	5.2	0.7	74.2	19.9	100	8.1	0.4	91.6	0.0
Niger	100	7.6	0.3	91.2	1.0	100	13.0	1.9	85.1 0.0		100 9.	9.0 0.3	3 89.4	4 1.3	100	12.	9 2.1	85.0	0.0	100	4.7	0.2	94.7	0.4	100	11.6	0.5	87.9	0.0
Nigeria	100	54.3	0.0	44.7	1.0	100	99.7	0.0	0.3 0.0	_	100 59	59.3 0.0	0 39.5	5 1.2	100	0 99.7	7 0.0	0.3	0.0	100	48.2	0.0	51.0	0.8	100	99.7	0.0	0.3	0.0
Senegal	100	26.2	0.4	48.8	24.6	100	83.0	1.1	15.8 0.0		100 33	33.5 0.6	6 45.4	4 20.5	5 100	0 83.6	5 1.4	15.1	0.0	100	15.7	0.1	53.6	30.6	100	82.3	0.6	17.1	0.0
Sierra Leone	100	5.9	0.5	83.1	10.5	100	60.0	0.1	40.0 0.0		100 9.	9.0 0.7	7 80.9	.9 9.5	100	0 71.1	1 0.1	28.9	0.0	100	3.0	0.4	85.1	11.6	100	41.7	0.1	58.2	0.0
Togo	100	35.2	0.3	60.6	3.9	100	23.2	4.7	72.1 0.0		100 47	47.8 0.4	47	.9 3.9	100	0 26.6	5 6.5	60.9	0.0	100	23.1	0.2	72.9	3.8	100	19.8	2.5	77.7	0.0
Americas																													
Latin America and the Caribbean	Caribi	hean																											
The Caribbean																[													
Dominican Republic	100	37.9	2.3	57.6	2.2	100	79.3	4.9	15.9 0.0		100 24.	1.4 2.7	7 71.4	.4 1.6	100	0 75.4	t 5.9	18.6	0.0	100	58.9	1.7	36.2	3.2	100	86.5	2.9	10.7	0.0
Central America																]													
Costa Rica	100	70.4	3.7	25.4	0.4	100	82.7	10.1	7.2 0.0		100 67	67.1 4.9	9 27.5	5 0.4	100	0 79.1	1 12.8	8.1	0.0	100	75.4	1.8	22.3	0.5	100	89.2	5.1	5.7	0.0
El Salvador	100	41.8	5.2	43.1	9.9	100	94.0	3.0	3.0 0.0		100 50	50.2 6.5	5 34.6	.6 8.8	100	0 93.9	9 3.6	2.5	0.0	100	31.9	3.8	53.1	11.3	100	94.2	2.2	3.6	0.0
Guatemala	100	53.1	3.4	32.9	10.6	100	98.2	1.0	0.8 0.0	_	100 59	59.4 3.8	8 28.2	.2 8.7	100	0 97.5	5 1.4	1.2	0.0	100	41.6	2.7	41.5	14.2	100	99.7	0.1	0.2	0.0
Honduras	100	36.0	3.2	45.6	15.2	100	97.6	0.9	1.6 0.0		100 40	40.0 3.7	7 42.2	2 14.2	2 100	0 97.2	2 0.9	2.0	0.0	100	28.9	2.2	51.7	17.2	100	97.8	1.0	1.1	0.0

					Total										Men									Women	en				
	Info	Informal employment (%)	mployn	nent (9	(%)	Forn	nal em	Formal employmen	nt (%)		nform	Informal employment (%)	oymen	t (%)		Formal employment (%)	employ	/ment (	(%	Info	Informal employment (%)	mployr	nent (9	(%	For	Formal employment (%)	ploym	ent (%	
	letoT	Employees	Employers	Own-account Dwn-account	Contributing family workers	letoT	Employees	Own-account Employers	vorkers workers Contributing	family workers	letoT	Employees Employees	0wn-account	workers Contributing tamily workers	tamily workers Total	Employees	Employers	workers Own-account	Contributing family workers	Total	Employees	Employers	workers Own-account	Contributing family workers	letoT	Employees	Employers	workers Dwn-account	Contributing family workers
Mexico	100	59.5	3.5	27.9	9.1	100	78.4	5.6 1	16.0 0.0		100 64	64.6 4.6	23	.8 6.9	100	73.1	6.9	20.0	0:0	100	52.4	1.8	33.4	12.4	100	88.7	3.0	8.4	0.0
Nicaragua	100	36.2	10.6	36.3	16.9	100	96.8	1.4 ]	1.8 0.0		100 41	41.0 14.7	7 30.8	8 13.6	100	96.3	2.0	1.8	0:0	100	27.8	3.9	45.6	22.7	100	97.6	0.6	1.8	0.0
Panama	100	42.2	4.8	46.8	6.2	100	98.4	1.0 0	0.6 0.0	0.0 10	100 38	38.6 6.4	50	.9 4.1	100	98.1	1.3	0.6	0.0	100	48.0	2.3	40.0	9.7	100	99.0	0.5	0.5	0.0
South America																													
Argentina	100	52.9	5.6	40.3	1.3	100	96.9	1.7 1	1.4 0.0	0.0 10	100 47.	7.6 6.6	3 45.1	1 0.7	100	95.9	2.4	1.6	0:0	100	60.5	4.1	33.4	2.1	100	98.2	0.8	1.0	0.0
Bolivia, Plurinational State of	100	55.6	3.8	32.8	7.8	100	90.3	2.7 7	7.0 0.1	0.0 10	100 54	54.4 4.4	34	.4 6.8	100	88.9	3.2	7.9	0.0	100	57.1	3.0	30.8	9.2	100	91.7	2.4	5.9	0.0
Brazil	100	49.5	1.7	43.9	4.9	100	84.3	6.8 8	8.9 0.0	0.0 10	100 42	42.4 2.3	3 52.3	3 3.0	100	81.9	8.1	10.0	0.0	100	59.2	0.8	32.4	7.6	100	87.8	4.9	7.3	0.0
Chile	100	33.8	7.5	50.5	8.3	100	98.0	1.8 (	0.2 0.0		100 32	32.4 10.0	0 51.0	0 6.6	100	97.5	2.2	0.3	0.0	100	35.7	4.1	49.8	10.5	100	98.7	1.1	0.2	0.0
Colombia	100	22.0	6.2	65.7	6.1	100	92.1	1.2 6	6.8 0.0		100 19	19.2 8.0	0 69.4	4 3.5	100	92.3	1.7	6.1	0.0	100	25.6	3.9	60.8	9.7	100	91.6	0.5	7.9	0.0
Ecuador	100	37.9	1.7	41.7	18.7	100	74.6	5.8 1	19.6 0.0	0.0 10	100 46	46.7 2.3	3 40.0	0 11.0	100	74.1	9.9	19.3	0.0	100	25.3	0.8	43.7	30.2	100	75.0	4.3	20.6	0.0
Paraguay	100	48.1	2.3	37.2	12.4	100	73.6	11.4 1	15.1 0.0	0.0 10	100 49	49.5 3.5	5 37.7	7 9.4	100	71.1	14.0	14.9	0.0	100	46.1	0.7	36.3	17.0	100	77.4	7.3	15.3	0.0
Peru	100	33.4	3.2	47.8	15.7	100	82.3	6.4 1	11.3 0.0	0.0 10	100 37	37.9 4.6	3 49.5	5 8.0	100	82.8	7.3	10.0	0.0	100	28.3	1.6	45.9	24.3	100	81.5	5.0	13.5	0.0
Uruguay	100	37.6	1.2	57.0	4.3	100	83.7	5.6 1	10.7 0.0	0.0 10	100 34	34.2 1.7	7 61.4	4 2.8	100	81.8	7.3	10.9	0.0	100	41.8	0.5	51.3	6.4	100	86.0	3.6	10.5	0.0
Venezuela, Bolivarian Republic of	100	16.9	9.3	62.1	11.7	100	84.9	5.8	9.3 0.0	0.0 10	100 16	16.6 11.5	62	.6 9.4	100	83.1	6.2	10.7	0.0	100	17.3	5.8	62.0	14.9	100	87.6	5.6	6.9	0.0
Northern America																													
United States	100	73.7	0.0	25.9	0.4	100	97.6	0.0 2	2.4 0.0	0.0 10	100 69.	9.8 0.0	29	.8 0.3	100	97.2	0.0	2.8	0.0	100	78.1	0.0	21.4	0.5	100	98.2	0.0	1.8	0.0
Arab States																													
Iraq	100	57.4	3.6	39.0	0.0	100	97.3	2.7 0		0.0 10	100 58	58.0 3.6	38	.4 0.0	100	96.8	3.2	0.0	0.0	100	45.0	0.0	55.0	0.0	100	100.0	0.0	0.0	0.0
Jordan	100	64.6	13.5	21.3	0.7	100	99.2	0.6 (	0.2 0.0	0.0 10	100 62	62.9 14.2	22	.3 0.7	100	99.2	0.6	0.2	0.0	100	85.2	5.2	8.7	1.0	100	99.4	0.5	0.1	0.0
Occupied Palestinian Territory	100	69.7	3.6	16.4	10.2	100	68.0	10.6 2	21.3 0.0	0.0 10	100 72	72.9 4.4	4 15.8	8 7.0	100	63.5	11.8	24.6	0.0	100	57.2	0.5	18.8	23.6	100	88.1	4.9	7.0	0.0
Syrian Arab Republic	100	50.5	5.7	40.3	3.5	100	88.7	1.7 9	9.6 0.0	0.0 10	100 49	49.7 5.9	9 41.8	8 2.7	100	86.5	2.1	11.5	0.0	100	65.6	2.0	17.0	15.4	100	97.8	0.4	1.9	0.0
Yemen	100	51.3	4.2	44.3	0.3	100	86.0	5.1 8	8.9 0.0	0.0 10	100 54.	1.6 4.7	7 40.4	4 0.3	100	85.9	5.8	8.3	0.0	100	33.4	3.0	63.3	0.3	100	98.4	0.5	1.1	0.0

					Total	뫼									Men									>	Women					
	Info	Informal employment (%)	mploy	ment	(%)	For	rmal ei	Formal employmer	nent (%)	<u>_</u>	Infor	mal en	nploym	Informal employment (%)		Forma	Formal employment (%)	oymer	nt (%)		Informal employment (%)	al emp	loymer	nt (%)		Formal employment (%)	emplo	ymen	t (%)	
	letoT	Employees	Employers	workers Own-account	Contributing family workers	letoT	Employees	Employers	workers Own-account	Contributing family workers	letoT	səəyolqm3	Gwn-account Employers	Own-account workers Contributing	tamily workers	letoT 299volum3	Employees Employees	Own-account	workers Contributing	tamily workers	letoT	Employees	Own-account Employers	workers Contributing	family workers Total	Employees	Employers	Own-account	workers Contributing	family workers
Asia and the Pacific																														
Eastern Asia																														
China	100	52.0	3.5	28.8	15.7	100	82.0	5.0	13.1	0.0	100	56.2	4.8	34.0 5.1		100 77.	77.5 7.3	3 15.2	.2 0.0		100 46.	4	1.8 22.6	.6 29.2	2 100	0 87.7	7 2.0	0 10.3		0.0
Japan	100	29.6	6.8	44.9	18.7	100	98.0	1.4	9.0	0.0	100	25.0	9.6	58.6 6.	6.8 1	100 97.2	.2 2.0	0 0.8	8 0.0		100 35	35.1 3.2	2 27.8	.8 34.0	0 100	0 99.1	1 0.5	0.4		0.0
Korea, Republic of	100	67.7	1.2	17.4	13.7	100	77.1	8.5	14.4	0.0	100	71.7	2.0	22.4 3.	3.9 1	100 72.8	.8 10.2	.2 17.0	0.0 0.0		100 64.	.3 0.4	4 12.6	.6 22.6	6 100	0 84.	5 5.5	10.0		0.0
Mongolia	100	37.8	1.7	56.9	3.6	100	80.8	2.8	16.5	0.0	100	36.9	2.2	59.2 1.	1.7 1	100 80.7	ŝ	.8 15.6	.6 0.0		100 38.	1.1 6.1	1 53.	.9 6.1	1 100	0 80.	8 1.8	3 17.4		0.0
South-Eastern Asia and the Pacific	nd the	Pacific																												
Pacific Islands																														
Samoa	100	31.5	7.4	58.5	2.7	100	96.1	1.3	2.6	0.0	100	33.2	7.2 5	58.0 1.	1.6 1	100 96.1	1.1 1.1	1 2.8	8 0.0		100 28.	1.1 7.9	58	.4 5.6	5 100	0 95.	9 1.7	2.4		0.0
South-Eastern Asia	E																													
Brunei Darussalam	100	90.2	9.0	7.6	1.6	100	93.3	4.1	2.6	0.0	100	90.9	0.7	7.3 1.	1.1	100 92.1	.1 5.2	2 2.7	7 0.0		100 89.	1.2 0.4	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	.0 2.5	5 100	0 95.0	0 2.6	3 2.5	5 0.0	0
Cambodia	100	44.4	0.2	49.8	5.6	100	82.0	0.3	17.6	0.0	100	48.4	0.2 4	46.4 5.	5.0 1	100 85.6	.6 0.2	2 14.1	.1 0.0		100 40.	.6 0.1	1 53.1	.1 6.3	3 100	75.	.3 0.5	5 24.2		0.0
Indonesia	100	37.7	2.6	42.1	17.6	100	57.7	12.8	29.6	0.0	100	40.0	3.5	50.3 6.	6.2 1	100 53.	5 16.0	.0 30.6	.6 0.0		100 34.	0	1.0 29.0	.0 36.0	0 100	0 67.1	1 6.2	26.7		0.0
Lao People's Democratic Republic	100	12.6	0.5	54.6	32.4	100	91.7	1.7	9.9	0.0	100	16.6	0.7 (	60.7 22	22.0 1	100 93.	1.2 1.8	8 5.0	0 0.0		8.	.9 0.2	48	.9 42.1	1 100	68	.0 1.6	9.4		0.0
Myanmar	100	41.0	2.7	29.3	27.0	100	34.7	10.6	54.7	0.0	100	44.8	4.2	35.3 15	15.8 1	100 23.7	.7 13.6	.6 62.8	.8 0.0		100 37	37.1 1.4	4 23.9	.9 37.6	6 100	0 53.4	4 6.4	40.2	.2 0.0	0
Timor-Leste	100	17.1	0.9	49.2	32.8	100	88.9	1.8	9.3	0.0	100	19.5	1.2	59.3 20	20.1 1	100 85.	.5 2.3	3 12.2	.2 0.0		100 11.	0	0.3 27.7	.7 61.0	0 100	0 95.1	1 0.6	6.4.3	3 0.0	0
Viet Nam	100	27.7	1.8	48.6	21.9	100	80.2	6.9	12.9	0.0	100	35.6	2.4 4	47.8 14.	2	100 77.4	.4 9.7	12	.9 0.0		100 19	19.1 1.	1.2 49.4	.4 30.4	4 100	0 82.	9 4.3	3 12.9		0.0
Southern Asia																														
Bangladesh	100	41.6	0.7	36.4	21.3	100	58.2	2.8	39.1	0.0	100	43.8	1.1	50.0 5.	5.2 1	100 50.	50.6 3.3	3 46.1	.1 0.0		100 37	.6 0.1	1 12.0	.0 50.3	3 100	0 86.3	3 0.8	3 12.9	9 0.0	0
India	100	13.7	1.2	70.3	14.9	100	68.4	1.0	30.6	0.0	100	14.5	1.4	74.7 9.	5	100 69.7	.7 1.2	2 29.1	.1 0.0		100 11.	4 0	5 56	.6 31.4	4 100	62	.4 0.2	37.4	4 0.0	0
Nepal	100	16.9	1.4	37.8	43.9	100	98.6	1.4	0.0	0.0	100	26.0	2.4 4	49.1 22.	5	100 98.	3 1	.8 0.0	0 0.0		100 8.	6	0.5 27.	8 62	.8 100	0 99.7	7 0.3	0.0	0.0 0.0	0
Pakistan	100	29.7	1.0	41.3	28.1	100	83.8	3.1	13.1	0.0	100	33.8	1.4 4	46.6 18.	2	100 82.	.8 3.2	13	.9 0.0		100 16.	0	0.0 24.	5 59	5 100	0 96.4	4 1.1	2.4	4 0.0	0
Sri Lanka	100	42.8	2.4	43.9	10.9	100	89.3	3.5	7.2	0.0	100	46.6	2.9 4	46.7 3.	3.8 1	100 85.5	5 5.0	0 9.5	5 0.0		100 34	3 1.1	1 35.5	.5 29.1	1 100	0 95.8	8 0.8	3.4		0.0

					Total									×	Men									Women	ua				
	Info	rmal e	mployn	Informal employment (%)	(	Form	Formal employment	loymei	nt (%)	5	forma	Informal employment (%)	yment	(%)	Ë	ırmal e	mployn	Formal employment (%)	()	Infor	mal en	nployn	Informal employment (%)	()	Forr	Formal employment (%)	ployme	ent (%	
	letoT	Employees	Employers	Own-account workers	Contributing family workers	letoT	Employees	Own-account Employers	Contributing	tamily workers Total	Employees	Employers	Own-account Own-account	Contributing family workers	letoT	Employees	Employers	Own-account Vorkers	Contributing family workers	letoT	Employees	Employers	Own-account workers	Contributing family workers	letoT	Employees	Employers	Own-account Workers	Contributing family workers
<b>Europe and Central Asia</b>																													
<b>Central and Western Asia</b>	lsia																												
Central Asia																													
Kyrgyzstan	100	44.9	0.3	33.2	21.6	100 6	66.3 1	1.8 31	31.9 0.0	100	44.8	3 0.1	39.7	15.5	100	54.5	2.6	42.9	0.0	100	45.1	0.5	22.2	32.1	100	82.7	0.7	16.6	0.0
Tajikistan	100	43.7	0.8	53.3	2.2	100 5	97.5 0	0.2 2	2.3 0.0	100	43.5	5 1.0	52.9	2.6	100	92.6	0.4	4.0	0.0	100	42.9	0.4	54.8	1.8	100	100.0	0.0	0.0	0.0
Western Asia																													
Armenia	100	26.2	0.0	59.7	14.2	100	91.7 1	1.8 6	6.5 0.0	100	30.0	0.0	61.4	8.5	100	88.4	2.9	8.7	0.0	100	21.8	0.0	57.5	20.7	100	95.2	0.7	4.1	0.0
Cyprus	100	28.0	9.3	55.9	6.8	100 9	98.4 0	0.2 1	1.4 0.0	100	21.4	t 12.6	60.3	5.8	100	97.9	0.3	1.8	0.0	100	38.2	4.2	49.3	8.3	100	99.1	0.1	0.8	0.0
Turkey	100	36.3	1.7	29.4	32.6	100 8	83.8 6	6.1 1(	10.1 0.0	100	42.6	3 2.7	38.8	15.9	100	80.2	7.3	12.6	0.0	100	27.3	0.3	16.0	56.4	100	95.2	2.3	2.6	0.0
Eastern Europe																													
Bulgaria	100	41.1	13.0	42.0	3.9	100 5	97.8 1	1.7 0	0.6 0.0	100	38.2	2 14.9	44.6	2.3	100	96.8	2.5	0.7	0.0	100	45.2	10.5	38.2	6.2	100	98.9	0.7	0.4	0.0
Czech Republic	100	15.1	12.7	66.2	6.0	100 8	89.4 2	2.2 8	8.4 0.0	100	10.4	t 17.4	70.2	2.1	100	87.7	2.9	9.5	0.0	100	23.4	5.4	58.1	13.1	100	91.7	1.3	7.0	0.0
Hungary	100	15.9	32.2	48.7	3.3	100 5	97.0 2	2.1 1	1.0 0.0	100	12.3	3 39.4	46.3	2.1	100	96.7	2.2	1.1	0.0	100	22.1	20.0	52.5	5.3	100	97.4	1.8	0.8	0.0
Moldova, Republic of	100	17.7	0.0	81.7	0.5	100 8	84.1 2	2.0 1/	14.0 0.0	100	14.5	0.0	85.1	0.5	100	78.3	3.7	18.0	0.0	100	22.1	0.0	77.2	0.7	100	89.3	0.4	10.3	0.0
Poland	100	54.8	5.6	32.8	6.9	100 5	94.8 2	2.9 2	2.3 0.0	100	54.3	3 6.4	35.5	3.9	100	93.0	4.1	2.9	0.0	100	55.1	4.3	28.8	11.8	100	96.6	1.8	1.7	0.0
Romania	100	7.3	1.8	59.6	31.3	100 5	98.6 0	0.7 0	0.7 0.0	100	7.3	2.3	73.4	17.0	100	97.9	1.0	1.2	0.0	100	7.4	1.3	41.1	50.3	100	99.3	0.5	0.3	0.0
Russian Federation	100	97.2	1.7	0.0	11	100	90.3 1	1.2 8	8.6 0.0	100	97.1	l 2.0	0.0	1.0	100	88.5	1.7	9.9	0.0	100	97.5	1.2	0.0	1.3	100	91.9	0.8	7.4	0.0
Slovakia	100	21.4 14.9	14.9	62.7	1.0	100 5	96.9	0.9 2	2.2 0.0	100	17.3	3 16.3	66.1	0.4	100	96.2	1.3	2.5	0.0	100	29.6	12.0	56.1	2.3	100	97.7	0.5	1.8	0.0
Northern, Southern and Western Europe	id Wesi	tern Eu	rope																										
Northern Europe																													
Denmark	100	33.5	0.0	57.3	9.2	100 1	100.0 0	0.0 0	0.0 0.0	100	27.0	0.0 (	65.6	7.4	100	100.0	0.0	0.0	0.0	100	43.1	0.0	45.0	11.9	100	100.0	0.0	0.0	0.0
Estonia	100	50.0	10.5	37.8	1.7	100 5	93.7 3	3.4 2	2.9 0.0	100	43.6	3 13.0	42.5	0.8	100	91.8	4.8	3.4	0.0	100	58.3	7.1	31.8	2.8	100	95.6	1.9	2.5	0.0
Finland	100	5.0	22.5	64.8	7.7	100 5	92.0 2	2.7 5	5.3 0.0	100	4.9	24.4	64.3	6.5	100	89.3	4.1	9.9	0.0	100	5.1	18.2	66.2	10.5	100	94.7	1.3	4.1	0.0
Iceland	100	62.9	7.4	26.4	0.3	100 8	89.2 3	3.8 7	7.0 0.0	100	63.8	3 8.7	27.4	0.1	100	85.3	5.5	9.2	0.0	100	69.8	5.2	24.4	0.7	100	93.4	2.0	4.6	0.0
Ireland	100	19.6	14.6	9.09	5.2	100 5	93.6 3	3.0 3	3.4 0.0	100	15.9	) 15.3	65.3	3.5	100	89.9	4.5	5.6	0.0	100	29.9	12.5	47.8	9.8	100	97.5	1.5	1.0	0.0

					Total	_								×	Men									Women				
	Infe	Informal employment (%)	mploy	ment (	(%)	Forr	nal em	Formal employment	ent (%)	=	forma	empla	Informal employment (%)	(%)	Fo	rmal en	nployn	Formal employment (%)		Inforn	nal em	Informal employment (%)	int (%)		Forma	Formal employment (%)	oymen	t (%)
	letoT	Employees	Employers	workers Own-account	Contributing family workers	letoT	Employees	Umu-scconut Employers	Own-account workers Contributing	tamily workers Total	Employees	Employers	Own-account workers	Contributing family workers	letoT	Employees	Employers	Own-account workers Contributing	Contributing family workers	letoT	Employees	Own-account Employers	workers Contributing	family workers	letoT	Fmployees	Own-account Employers	workers Contributing
Latvia	100	52.2	7.4	34.7	5.6	100	93.2	3.7	3.1 0.0	100	50.8	8.6	34.4	6.2	100	90.7	5.6	3.7	0.0	100 5	54.3 5	5.8 35	35.2 4.	4.8 1	100 95	95.5 2.	2.0 2.	.5 0.0
Lithuania	100	52.6	3.6	34.6	9.2	100	92.8	2.2	5.0 0.0	100	44.4	1 3.8	44.1	7.6	100	90.1	3.3	9.9	0.0	100 5	58.8 3	3.7 27.	0	10.4 1	100 95.	5	1.0 3.	.5 0.0
Norway	100	49.4	10.5	37.3	2.8	100	96.3	1.2	2.5 0.0	100	41.9	) 12.4	43.3	2.5	100	95.3	1.7	2.9	0.0	100 6	63.3 7	7.4 25.	6	3.4 1	100 97.	7.3 0.	5 2	.2 0.0
Sweden	100	86.8	2.6	7.5	3.2	100	90.3	3.8	6.0 0.0	100	84.9	9 4.1	8.2	2.9	100	86.6	5.6	7.9	0.0	100 8	88.8	0.9 6.0	8	5	100 94.	1.3 1.	8 3.	9 0.0
United Kingdom	n 100	20.5	11.7	67.8	0.0	100	96.9	0.7	2.4 0.0	100	16.3	13.5	70.2	0.0	100	95.2	1.1	3.8	0.0	100 2	27.2 8	8.9 63.	6	0.0	100 98.	8.8 0.	3 0	.9 0.0
Southern Europe																			]									
Albania	100	12.7	2.4	35.1	49.8	100	82.2	2.6 1	15.2 0.0	100	15.7	, 3.1	43.1	38.0	100	75.6	3.8	20.5	0.0	100	8.2 1	1.1 22.	9	68.2 1	100 91.	4	0.9 7.7	7 0.0
Bosnia and Herzegovina	100	0.0	17.3	65.6	17.1	100	100.0	0.0	0.0 0.0	100	0.0	19.1	74.1	6.8	100	100.0	0.0	0.0	0.0	100	0.0 1	14.4 51.	6	33.7 1	100 10	100.0 0.	0.0 0.0	0.0
Croatia	100	24.0	21.1	40.8	14.2	100	97.5	2.0	0.6 0.0	100	15.0	) 27.0	50.0	8.0	100	96.6	2.6	0.8	0.0	100 3	34.8 1	13.8 29	29.7 21	21.7 1	100 98.	4	1.3 0.3	3 0.0
Greece	100	8.5	14.0	65.3	12.2	100	92.6	4.4	3.0 0.0	100	7.1	16.7	68.5	7.7	100	90.3	5.4	4.4	0.0	100 1	10.9 9	9.6 6(	60.0 19.	5	100 95	95.7 3.	0 1.3	3 0.0
Italy	100	15.2	19.3	58.1	7.5	100	89.7	3.6	6.7 0.0	100	11.0	) 21.8	61.9	5.3	100	87.3	4.8	7.9	0.0	100 2	22.8 1	15.0 50.	6	11.4 1	100 92.	2.9 2.	.0 5.1	1 0.0
Malta	100	24.6	16.9	58.3	0.3	100	92.1	3.6	4.3 0.0	100	15.2	19.7	64.9	0.2	100	88.9	5.2	5.9	0.0	100 4	46.0 1	12.2 41.	5	0.3 1	100 97.	7.6 0.	8 1.7	7 0.0
Portugal	100	14.5	18.1	63.2	4.2	100	94.4	2.4	3.2 0.0	100	9.0	20.8	66.7	3.6	100	92.4	3.4	4.2	0.0	100 2	22.1 1.	14.8 58	58.1 5.	5.0 1	100 96.	5.5 1.	3 2.	.3 0.0
Serbia	100	26.7	0.2	37.3	35.9	100	81.7	5.1 1	13.2 0.0	100	30.2	2 0.3	49.4	20.1	100	74.7	6.5	18.8	0.0	100 2	22.9 0	0.0 22	22.1 55	55.0 1	100 91	91.0 3.	2 5	.8 0.0
Slovenia	100	10.1	10.8	48.0	31.0	100	92.0	3.1	4.9 0.0	100	9.0	12.9	56.8	21.3	100	89.5	4.4	6.1	0.0	100 1	11.8 8	8.1 34.	∞	45.4 1	100 94.	1.9 1.	5 3.	.6 0.0
Spain	100	54.8	10.3	33.1	1.8	100	93.6	3.0	3.4 0.0	100	43.4	13.5	41.6	1.6	100	92.3	3.7	4.0	0.0	100 6	67.6 6	6.8 23.	9	2.0 1	100 95.	5.2 2.	0 2	.8 0.0
Western Europe																												
Austria	100	42.8	12.2	30.7	14.3	100	92.5	3.7	3.8 0.0	100	38.0	15.1	33.2	13.7	100	90.2	5.4	4.5	0.0	100 4	47.8 9	9.9 27	27.7 14	14.7 1	100 95.	ŝ	1.6 3.1	1 0.0
Belgium	100	30.5	15.7	49.0	4.8	100	96.1	1.9	2.1 0.0	100	26.0	) 18.5	53.1	2.5	100	94.5	2.7	2.7	0.0	100 3	37.8 1	11.0 42.	6 8	9	100 97	97.7 0.	6	1.3 0.0
France	100	48.1	15.0	33.0	3.9	100	92.7	3.1	4.2 0.0	100	36.5	5 20.3	41.5	1.7	100	90.5	4.5	5.1	0.0	100 6	60.5 5	9.4 23	23.9 6.	6.3 1	100 95.	5.2 1.	5 3.	.3 0.0
Germany	100	62.7	8.8	24.9	3.6	100	92.8	3.9	3.3 0.0	100	58.6	3 10.4	28.8	2.2	100	90.6	5.7	3.7	0.0	100 6	67.6 6	6.6 20.	5 5	2	100 95.	5.3 1.	9 2.	.8 0.0
Luxembourg	100	50.4	0.0	1.2	48.4	100	90.6	4.0	5.4 0.0	100	51.1	0.0	0.0	48.9	100	89.3	5.1	5.6	0.0	100 5	50.0 0	0.0 1	9 48	48.1 1	100 92.	2.3 2.	6 5.1	1 0.0
Netherlands	100	23.8	15.6	56.2	4.4	100	89.9	2.7	7.4 0.0	100	19.7	, 19.2	59.0	2.1	100	87.4	3.9	8.7	0.0	100 2	29.1 1	.0.9 52.	9	7.4 1	100 92.	2.9 1.	4 5.	.8 0.0
Switzerland	100	51.5	8.0	24.4	16.1	100	90.4	5.5	4.1 0.0	100	57.4	1 8.7	21.7	12.2	100	87.8	7.9	4.4	0.0	100 4	43.5 6	6.6 28.	9	21.3 1	100 93.	3.3 2.	80	9 0.0
Source: ILO calculations based on national labour force or similar household survey	based	on natic	onal la	abour fi	orce or s	imilar h	vuseh	old surv		See app	endix	A.2 for	origina	data. See appendix A.2 for original sources and years.	s and y(	ears.												

Maires		Total (%)		Ä	Agriculture (%)			Industry (%)			Services (%)	
Africa	Total	Men	Women	Total	Men	Women	Total	Men	Women	Total	Men	Women
AIFICA												
Northern Africa												
Egypt	63.3	64.8	57.3	98.2	97.7	99.3	67.5	68.3	48.7	41.0	45.9	24.3
Morocco	79.9	81.1	73.9	90.9	90.6	92.6	80.3	81.4	73.0	72.5	74.3	65.0
Tunisia	58.8	60.0	54.7	88.1	85.7	95.5	60.7	65.2	41.0	50.1	50.5	49.1
Sub-Saharan Africa												
Central Africa												
Angola	94.1	92.8	95.4	99.4	9.66	99.2	85.1	86.5	68.2	90.2	88.7	91.8
Cameroon	90.9	86.6	95.2	99.3	98.7	99.7	86.8	81.3	93.1	70.3	66.3	77.3
Chad	94.0	91.7	97.3	99.9	99.8	100.0	94.1	89.7	98.7	65.1	59.2	78.1
Congo	85.3	88.3	82.9	91.7	83.1	95.4	90.5	93.0	88.2	83.6	87.3	80.7
Congo, Democratic Republic of the	91.9	86.6	97.2	98.3	97.2	99.1	85.1	81.6	95.8	73.5	62.6	89.0
Eastern Africa												
Comoros	89.2	86.2	93.8	99.0	98.9	99.1	92.6	89.9	98.3	72.8	67.6	82.1
Madagascar	93.6	91.9	95.2	99.3	99.2	99.5	80.0	77.8	83.0	79.8	73.1	85.6
Malawi	83.0	83.7	82.4	91.6	92.7	90.7	76.2	77.2	75.2	64.8	64.4	65.2
Rwanda	94.3	92.0	96.2	99.9	99.7	9.99	93.1	93.2	92.5	73.9	72.3	76.3
Tanzania, United Republic of	90.6	88.2	93.1	99.8	9.66	100.0	T.TT	75.9	79.9	68.5	63.7	74.4
Uganda	93.7	92.0	95.6	0.06	98.6	99.4	87.9	86.4	92.0	82.1	79.7	85.2
Zambia	87.9	81.6	93.6	98.2	96.5	99.3	69.0	65.5	86.2	72.2	65.5	79.5
Southern Africa												
Botswana	65.6	63.5	68.0	96.7	96.2	97.5	61.5	59.7	66.4	53.1	44.9	59.6
Namibia	67.0	66.1	67.9	90.4	86.6	95.4	71.7	72.8	66.4	58.4	54.3	61.5
South Africa	34.0	32.5	35.9	34.8	34.3	36.0	29.8	31.3	24.5	35.3	33.0	37.5
Western Africa												
Benin	94.5	91.6	97.2	9.66	99.5	99.7	95.3	92.4	98.4	88.2	77.2	94.8
Burkina Faso	94.6	91.3	97.6	96.9	94.5	99.0	92.2	87.9	97.9	86.8	79.9	92.4

Table B.6 – Share of informal employment in agriculture, industry and services by sex

		Total (%)		A	Agriculture (%)			Industry (%)			Services (%)	
	Total	Men	Women	Total	Men	Women	Total	Men	Women	Total	Men	Women
Cabo Verde	46.5	49.1	43.8	45.1	52.7	29.8	50.3	59.8	24.6	45.8	42.6	48.1
Côte d'Ivoire	92.8	90.7	95.9	98.2	97.9	98.9	88.0	86.7	92.2	87.7	81.5	93.9
Gambia	76.7	70.4	84.7	96.3	92.6	96.9	78.5	77.4	86.5	65.6	55.4	77.0
Ghana	90.1	86.0	94.0	99.5	99.1	99.8	88.7	84.5	94.7	80.5	68.1	88.7
Liberia	86.8	81.5	92.0	97.2	96.2	98.2	79.3	75.9	88.9	77.1	65.4	86.2
Mali	92.7	90.8	95.0	96.5	92.6	97.6	90.3	87.2	96.5	85.4	81.1	89.7
Niger	91.3	87.4	95.9	93.2	90.5	90.6	92.4	81.5	97.5	81.7	72.8	91.9
Nigeria	92.9	90.8	95.1	99.3	99.1	99.7	96.3	94.2	98.9	86.9	79.6	92.0
Senegal	91.2	89.5	93.7	99.1	98.9	<u> 99.5</u>	90.8	89.7	96.0	86.1	81.6	90.9
Sierra Leone	92.5	92.9	92.1	96.3	96.5	96.0	88.7	90.6	87.1	85.6	85.8	85.4
Togo	92.8	90.3	95.2	98.3	97.8	98.9	84.6	80.4	89.6	90.6	86.5	94.0
Americas												
Latin America and the Caribbean												
The Caribbean												
Dominican Republic	56.3	56.3	56.2	80.0	82.0	58.0	53.2	56.9	41.5	52.4	47.1	58.0
Central America												
Costa Rica	39.1	36.6	42.9	52.3	52.2	52.8	39.9	39.7	40.8	36.7	31.0	42.7
El Salvador	69.6	68.4	71.3	97.4	97.4	98.0	62.6	63.3	61.3	63.2	52.3	72.2
Guatemala	79.7	79.1	80.8	95.8	95.6	96.8	78.1	75.8	82.9	70.4	63.6	77.5
Honduras	79.9	81.4	77.6	98.0	98.1	97.1	70.2	69.4	71.3	73.8	70.2	77.0
Mexico	53.4	50.1	58.8	54.7	52.3	72.8	50.4	51.9	46.1	54.4	48.2	60.8
Nicaragua	77.4	80.1	73.1	90.6	96.7	95.9	64.1	67.6	56.1	70.2	66.0	73.4
Panama	52.3	53.5	50.6	88.3	86.3	95.6	53.1	51.8	57.9	43.9	42.0	45.6
South America												
Argentina	47.2	47.1	47.2	67.3	64.0	82.8	55.6	56.1	53.3	44.4	42.2	46.5
Bolivia, Plurinational State of	83.1	82.1	84.2	99.5	99.2	100.0	83.5	82.4	87.3	72.3	69.0	75.2
Brazil	46.0	46.1	46.0	78.0	76.6	83.6	40.7	42.5	34.2	43.0	40.4	45.4
Chile	40.5	39.8	41.7	52.0	53.0	49.4	35.1	32.5	45.6	40.7	40.8	40.5
Colombia	60.6	61.9	58.8	90.5	90.3	91.5	55.2	54.4	57.4	55.4	54.0	56.9

		Total (%)		A	Agriculture (%)			Industry (%)			Services (%)	
	Total	Men	Women	Total	Men	Women	Total	Men	Women	Total	Men	Women
Ecuador	59.0	57.8	60.9	86.0	83.2	91.3	59.3	59.3	59.2	47.2	42.4	52.1
Paraguay	70.6	70.1	71.3	95.7	94.3	98.9	72.2	73.4	67.4	61.9	56.9	66.6
Peru	69.2	65.5	73.5	94.9	93.1	97.2	59.9	56.6	69.4	58.9	53.0	63.9
Uruguay	24.5	25.2	23.8	29.6	29.9	28.3	32.6	33.5	28.9	21.7	20.2	23.0
Venezuela, Bolivarian Republic of	39.7	39.5	40.0	57.4	57.0	62.4	37.7	36.9	40.8	38.4	37.4	39.5
Northern America												
United States	18.6	19.4	17.8	52.3	62.8	30.7	17.6	19.3	15.0	18.4	18.7	18.1
Arab States												
Iraq	6.99	69.9	49.0	86.0	83.9	95.6	78.2	79.8	31.6	63.1	66.1	49.2
Jordan	44.9	48.5	26.9	68.9	66.3	87.6	56.8	58.4	37.3	38.2	42.5	21.7
Occupied Palestinian Territory	64.3	63.4	68.3	97.3	95.9	99.5	85.9	85.4	92.1	46.4	44.3	53.7
Syrian Arab Republic	70.1	71.9	61.6	97.3	96.1	9.66	83.4	84.2	65.5	48.8	54.7	18.0
Yemen	77.8	77.5	82.9	99.3	99.1	100.0	91.5	91.0	99.2	63.2	63.7	52.1
Asia and the Pacific												
Eastern Asia												
China	54.4	56.4	51.5	81.7	80.7	83.2	60.2	62.1	56.0	49.3	50.4	48.1
Japan	18.7	18.9	18.4	79.4	77.1	83.2	19.5	19.9	18.2	15.0	14.1	15.9
Korea, Republic of	31.5	25.9	39.9	81.6	68.5	0.66	24.3	22.0	32.4	30.7	24.8	37.3
Mongolia	53.5	56.7	49.9	98.3	97.8	0.66	39.1	38.5	40.6	34.2	39.8	29.7
South-Eastern Asia and the Pacific												
Pacific Islands												
Samoa	35.7	36.7	33.9	60.1	54.7	77.4	34.9	33.2	42.4	33.6	35.8	30.6
South-Eastern Asia												
Brunei Darussalam	31.9	34.0	29.0	15.8	18.4	8.0	47.5	55.2	31.2	27.7	26.2	29.4
Cambodia	93.1	90.9	95.5	9.66	99.4	99.8	97.1	96.9	97.4	85.4	80.5	91.0
Indonesia	85.6	84.7	87.1	97.2	96.2	0.66	87.2	86.2	89.7	76.9	74.7	79.7
Lao People's Democratic Republic	93.6	91.7	95.5	9.66	99.5	99.7	84.9	82.4	88.7	75.9	69.8	82.4
Myanmar	85.7	83.5	88.6	88.9	84.7	94.9	90.0	90.0	90.1	78.2	77.4	79.1
Timor-Leste	71.8	69.3	76.6	97.8	97.0	0.66	74.4	72.6	81.4	48.5	45.8	53.6
Viet Nam	76.2	77.3	75.2	0.06	98.7	99.3	59.6	67.9	47.3	56.7	55.5	57.9

		Total (%)		A	Agriculture (%)			Industry (%)			Services (%)	
	Total	Men	Women	Total	Men	Women	Total	Men	Women	Total	Men	Women
Southern Asia												
Bangladesh	89.0	87.2	93.5	97.7	97.1	98.9	90.5	88.7	94.2	76.7	75.9	80.3
India	88.2	87.7	0.06	99.7	99.7	99.7	81.4	77.5	88.8	77.6	78.7	71.0
Nepal	94.3	90.7	97.5	99.5	99.2	99.7	90.8	88.3	95.8	73.7	70.3	80.6
Pakistan	82.4	79.6	92.1	99.2	98.8	99.7	78.7	75.9	96.5	65.3	67.1	46.9
Sri Lanka	70.4	72.5	9.99	89.9	91.4	87.5	72.5	75.9	65.8	56.1	58.9	50.2
Europe and Central Asia												
Central and Western Asia												
Central Asia												
Kyrgyzstan	48.5	50.6	45.3	67.6	59.3	83.8	54.3	54.3	54.0	38.8	44.5	32.4
Tajikistan	74.8	75.9	72.8	80.6	78.1	82.6	71.9	73.3	51.5	70.0	75.8	57.2
Western Asia												
Armenia	52.1	52.8	51.4	0.66	98.2	99.7	44.0	46.8	34.4	20.8	25.0	16.9
Cyprus	15.1	17.0	13.1	67.8	73.1	54.6	16.7	17.7	12.3	13.4	14.4	12.7
Turkey	34.8	28.9	48.3	84.3	72.7	98.0	24.0	22.7	29.8	20.9	19.5	24.3
Eastern Europe												
Bulgaria	15.9	17.6	13.9	48.1	45.9	53.0	8.7	11.0	4.8	15.8	17.5	14.4
Czech Republic	9.2	11.0	7.0	12.0	13.8	7.5	7.9	9.6	3.0	9.9	12.1	8.1
Hungary	12.2	14.5	9.6	37.9	32.7	54.4	9.1	10.6	5.6	11.2	14.3	8.8
Moldova, Republic of	28.9	32.0	25.9	68.9	61.6	78.1	24.8	32.9	8.3	9.9	11.1	9.1
Poland	38.0	42.2	32.7	91.0	88.6	93.9	24.6	26.9	17.5	35.0	44.6	26.7
Romania	28.9	29.1	28.6	90.5	85.8	96.5	8.9	11.6	2.8	12.6	13.4	12.0
Russian Federation	35.9	39.1	32.7	43.7	46.2	38.5	41.2	44.2	34.3	33.6	35.5	32.2
Slovakia	16.7	20.7	12.1	25.4	27.3	20.0	18.6	22.4	6.0	15.4	18.6	13.2
Northern, Southern and Western Europe												
Northern Europe												
Denmark	11.2	14.4	7.5	45.4	47.8	30.2	9.2	10.1	5.7	11.0	14.4	7.5
Estonia	6.9	7.6	6.1	19.3	22.0	13.9	3.3	4.0	1.8	7.7	8.9	6.9
Finland	6.3	8.3	4.2	34.6	37.4	27.5	5.9	6.3	4.3	4.9	6.8	3.6
Iceland	4.9	5.9	3.9	6.4	6.0	8.0	4.9	5.2	3.8	4.8	6.1	3.8

		Total (%)		A	Agriculture (%)			Industry (%)			Services (%)	
	Total	Men	Women	Total	Men	Women	Total	Men	Women	Total	Men	Women
Ireland	13.5	18.4	7.9	61.2	61.1	61.7	12.7	14.5	6.1	10.8	15.0	7.3
Latvia	13.2	15.1	11.5	40.8	39.6	44.8	8.3	8.9	7.1	12.1	14.0	10.9
Lithuania	12.6	11.4	13.8	44.4	36.3	0.09	8.9	8.3	10.3	10.1	8.2	11.4
Norway	7.4	9.0	5.5	44.5	46.7	34.7	7.9	9.2	0.8	6.7	8.1	5.5
Sweden	8.2	8.4	7.9	3.4	2.1	8.2	1.8	1.5	3.1	9.1	10.1	8.1
United Kingdom	13.6	16.1	10.9	35.5	36.6	31.5	16.4	18.3	8.8	12.6	14.6	11.0
Southern Europe												
Albania	61.0	59.0	63.5	96.4	93.7	98.7	41.4	49.4	20.4	29.5	32.8	25.2
Bosnia and Herzegovina	30.1	26.5	36.6	89.1	82.8	97.1	5.8	6.1	5.0	16.4	16.2	16.9
Croatia	13.0	14.0	11.8	52.4	49.1	60.1	8.8	9.6	9.9	11.6	12.7	10.8
Greece	32.8	34.4	30.4	90.5	87.9	93.7	26.7	27.8	21.3	25.2	28.3	21.6
Italy	19.0	21.2	15.9	40.8	43.7	34.2	15.4	16.8	10.1	19.4	22.5	16.3
Malta	8.1	9.0	9.9	36.4	39.8	14.2	7.5	8.1	4.6	7.8	8.5	6.8
Portugal	12.1	13.5	10.7	51.0	43.7	65.3	8.3	10.3	3.7	11.3	12.8	10.2
Serbia	22.1	21.3	23.0	53.1	38.8	76.9	12.8	15.3	6.8	16.5	18.2	14.9
Slovenia	5.0	6.3	3.5	41.0	41.7	38.5	3.1	3.9	1.3	4.6	5.9	3.5
Spain	27.3	26.2	28.7	37.8	37.0	40.0	12.4	13.5	7.5	31.0	31.4	30.5
Western Europe												
Austria	10.0	9.2	11.1	37.2	32.6	43.5	4.2	4.5	3.0	10.6	10.3	10.8
Belgium	13.5	15.3	11.4	78.7	80.3	74.8	9.2	10.4	4.5	13.7	16.1	11.7
France	9.8	9.6	9.9	33.9	31.5	39.0	5.4	5.6	4.9	10.2	10.5	9.9
Germany	10.2	10.3	10.2	19.3	17.2	23.4	7.2	7.0	8.0	11.1	12.1	10.4
Luxembourg	1.2	0.8	1.7	10.2	0.8	33.2	0.3	0.2	0.7	1.1	1.0	1.2
Netherlands	9.4	9.9	8.8	24.7	21.2	37.4	8.1	8.0	8.9	9.3	10.0	8.6
Switzerland	10.4	11.1	9.5	21.3	15.9	31.1	2.2	1.6	4.2	12.0	14.7	9.5
Source: 11.0 calculations based on national labour force or similar household survey data.	Ir force or simi	ar household	survev data. See	See appendix A 2 for original solutces and vears	for original so	irces and vears						

				Total	_							Men								Women	en			
	Infor	Informal employment (%)	loyment	(%)	Form	al empl	Formal employment (%	(%	Inforn	nal empl	Informal employment (%)	(%)	Forma	il emplo	Formal employment (%)		Inform	Informal employment (%)	yment (	(%	Form	Formal employment (%)	oyment	(%)
	latoT	Agriculture	lndustry	Services	letoT	Agriculture	Y12200	Services	lstoT	Agriculture	lndustry	Services	lstoT	Agriculture	ر Linuatry	Services	lstoT	Agriculture	lndustry	Services	letoT	Agriculture	lndustry	Services
Africa																								
Northern Africa																								
Egypt	100	44.8	24.6	30.6	100	1.4	20.9	<i>T.T</i>	100	35.1	29.8	35.1	100	1.5	24.6 7	73.9	100	81.1	3.3	15.6	100	1.0	9.9	92.4
Morocco	100	42.3	20.1	37.6	100	18.1	21.1	60.8	100	34.6	23.5	41.9	100	15.3	22.9 6	61.8	100	64.3	10.4	25.2	100	22.8	17.1	60.1
Tunisia	100	18.3	31.7	50.0	100	3.4	28.2	68.4	100	19.0	39.6	41.3	100	4.9	32.6 6	62.5	100	14.3	8.1	77.6	100	0.7	12.6	86.7
Sub-Saharan Africa	e c																							
Central Africa																								
Angola	100	45.6	11.6	42.8	100	4.2	29.3	66.5	100	46.9	16.7	36.4	100	2.9	35.0 6	62.2	100	44.3	4.6	51.1	100	4.9	30.5	64.6
Cameroon	100	64.6	11.7	23.8	100	3.8	14.5	81.7	100	64.4	12.9	22.7	100	5.5	19.4 7	75.1	100	64.4	10.0	25.7	100	2.1	8.7	89.2
Chad	100	85.5	2.4	12.1	100	1.7	2.2	96.1	100	84.9	3.7	11.4	100	2.3	4.9 9	92.8	100	85.5	0.7	13.8	100	0.0	0.2	99.8
Congo	100	32.4	22.8	44.9	100	20.7	17.0	62.4	100	27.2	23.1	49.7	100	38.1	12.0 4	49.8	100	36.6	22.7	40.7	100	12.3	20.8	66.9
Congo, Democratic Republic of the	100	64.3	10.4	25.3	100	9.5	15.1	75.4	100	65.9	12.4	21.7	100	10.7	15.9 7	73.4	100	61.4	8.8	29.8	100	12.4	8.3	79.3
Eastern Africa												1				J								
Comoros	100	70.1	9.8	20.1	100	8.2	8.6	83.2	100	69.3	11.6	19.2	100	7.1	11.5 8	81.4	100	73.0	5.3	21.7	100	12.2	1.7	86.1
Madagascar	100	76.5	8.0	15.5	100	7.9	31.1	61.0	100	79.9	7.2	12.9	100	9.1	27.7 6	63.2	100	72.8	9.0	18.2	100	6.5	34.9	58.5
Malawi	100	70.5	6.8	22.7	100	30.8	10.2	59.0	100	65.5	9.2	25.3	100	23.8	12.4 6	63.9	100	75.3	4.6	20.2	100	38.6	7.5	53.9
Rwanda	100	78.6	7.6	13.8	100	2.1	10.1	87.8	100	68.0	13.4	18.6	100	2.1	11.8 8	86.2	100	87.2	3.0	9.9	100	1.9	7.1	91.0
Tanzania, United Republic of	100	72.4	6.3	21.4	100	1.3	15.3	83.4	100	70.8	9.0	20.2	100	2.0	19.6 7	78.4	100	74.0	3.2	22.8	100	0.2	9.3	90.5
Uganda	100	78.6	5.7	15.8	100	15.6	15.6	68.7	100	75.7	8.7	15.6	100	16.9	21.3 6	61.8	100	81.4	2.6	16.0	100	13.5	6.6	79.9
Zambia	100	77.8	6.2	16.0	100	13.6	27.0	59.4	100	74.0	9.6	16.5	100	16.3	30.8 5	53.0	100	81.7	2.9	15.3	100	11.0	9.5	79.5

				Total								Men								Women	en			
	Infor	Informal employment (%)	loyment	(%)	Form	al emple	Formal employment (%	%)	Inforn	nal empl	Informal employment (%)	(%)	Forma	l emplo	Formal employment (%)		Inform	Informal employment (%)	oyment (	(%)	Form	Formal employment (%)	yment (	%)
	letoT	Agriculture	lndustry	Services	lstoT	Agriculture	lndustry	Services	letoT	Agriculture	lndustry	Services	lstoT	Agriculture	lndustry	Services	letoT	Agriculture	Judustry	Services	lstoT	Agriculture	lndustry	Services
Southern Africa																								
Botswana	100	32.7	19.6	47.7	100	2.0	22.1	75.8	100	38.9	28.6	32.5	100	2.5	31.7 (	65.8	100	25.8	9.6	64.6	100	1.4	9.9	88.8
Namibia	100	47.2	15.8	37.0	100	13.4	16.6	70.0	100	44.9	26.2	29.0	100	16.9	23.8	59.3	100	50.0	4.9	45.1	100	7.3	7.5	85.2
South Africa	100	4.3	20.3	75.4	100	4.2	24.7	71.1	100	5.5	31.0	63.5	100	5.1	32.8 (	62.1	100	2.9	7.7	89.5	100	2.9	13.3	83.8
Western Africa																								
Benin	100	49.0	11.5	39.5	100	3.4	9.4	87.2	100	57.4	16.2	26.4	100	3.1	14.0 8	82.9	100 4	41.1	6.7	52.2	100	4.0	3.5	92.6
Burkina Faso	100	54.0	12.0	34.0	100	22.0	12.8	65.2	100	58.4	13.8	27.8	100	27.7	15.5	56.8	100 4	49.3	10.1	40.6	100	12.8	5.4	81.8
Cabo Verde	100	26.8	19.1	54.1	100	28.3	16.4	55.3	100	32.9	26.7	40.4	100	29.0	17.6	53.4	100	16.2	6.3	77.5	100	26.9	13.8	59.3
Côte d'Ivoire	100	42.1	14.5	43.4	100	8.6	22.3	69.1	100	45.4	18.1	36.5	100	8.2	22.9 (	68.9	100	37.4	9.1	53.6	100	8.6	16.5	74.9
Gambia	100	36.8	15.4	47.8	100	4.6	13.7	81.7	100	33.5	25.0	41.5	100	3.7	17.4	79.0	100 4	40.9	4.3	54.8	100	7.1	3.7	89.2
Ghana	100	46.4	14.8	38.9	100	2.2	16.3	81.5	100	52.6	18.8	28.7	100	2.6	19.9	77.5	100 4	40.9	11.0	48.1	100	1.3	9.0	89.8
Liberia	100	50.4	10.4	39.1	100	9.2	17.3	73.6	100	53.5	15.4	31.1	100	8.9	20.9	70.2	100 4	47.7	5.6	46.7	100	9.9	7.7	82.5
Mali	100	63.1	7.0	29.9	100	28.1	9.2	62.7	100	9.66	8.9	24.6	100	30.4	12.9	56.7	100	57.6	3.7	38.8	100	23.8	2.2	74.0
Niger	100	56.2	12.5	31.3	100	33.9	8.4	57.7	100	67.0	8.6	24.4	100	38.8	10.8	50.4	100	35.4	19.3	45.3	100	22.0	8.5	69.5
Nigeria	100	53.7	6.9	39.4	100	5.4	4.0	90.6	100	61.8	7.5	30.8	100	6.3	5.1 8	88.6	100	43.6	6.1	50.3	100	2.5	1.6	95.9
Senegal	100	58.1	22.0	20.0	100	9.1	37.1	53.8	100	54.4	26.9	18.6	100	8.0	38.9	53.2	100	63.4	15.1	21.5	100	10.7	20.0	69.3
Sierra Leone	100	64.8	6.0	29.1	100	30.8	9.4	59.8	100	62.3	11.2	26.5	100	28.6	15.0 5	56.4	100 (	67.2	1.0	31.7	100	33.6	1.8	64.6
Togo	100	52.5	7.1	40.4	100	14.5	20.0	65.5	100	61.1	9.3	29.6	100	17.0	27.4	55.7	100 4	44.4	4.8	50.9	100	11.9	12.9	75.2
Americas																								
Latin America and the Caribbean	the Car.	ibbean																						
The Caribbean																								
Dominican Republic	100	20.2	14.3	65.5	100	6.5	16.3	77.1	100	29.9	20.1	50.0	100	8.4	19.5 7	72.1	100	3.6	5.3	91.1	100	3.5	9.8	86.7
Central America																								
Costa Rica	100	13.9	17.3	68.9	100	8.1	16.5	75.5	100	20.2	24.0	55.7	100	10.3	20.4 (	69.2	100	5.3	8.1	86.7	100	3.6	8.8	87.7
El Salvador	100	24.2	17.8	58.1	100	1.4	23.6	75.0	100	39.9	21.0	39.1	100	2.2	24.9 7	72.9	100	5.0	13.8	81.2	100	0.3	21.8	78.0
Guatemala	100	38.5	20.6	40.8	100	6.9	23.5	69.6	100	49.8	20.7	29.5	100	8.8	25.6 (	65.6	100	18.2	20.6	61.3	100	2.7	18.8	78.6

				Total								Men								Women	en			
	Infor	nal emp	Informal employment (%)	(%)	Form	al emplo	Formal employment (%)	(%	Inform	al emplo	Informal employment (%)	(%)	Forma	Formal employment (%)	ment (%		Inform	al emplo	Informal employment (%)	(%)	Forn	Formal employment (%)	oyment (	%)
	lstoT	Agriculture	Yıtzubnî	Services	letoT	Agriculture	lndustry	Services	lstoT	Agriculture	lndustry	Services	lstoT	Agriculture	lndustry	Services	lstoT	Agriculture	findustry	Services	Total	Agriculture	lndustry	Services
Honduras	100	40.2	15.6	44.2	100	3.6	28.7	67.8	100	55.3	15.4	29.2	100	5.3 3	33.5 6	61.2	100	10.1	15.9	73.9	100	1.0	22.2	76.7
Mexico	100	14.1	24.5	61.4	100	13.3	27.6	59.0	100	20.5	32.4 /	47.2	100 1	8.8 3	30.2 5	51.0	100	4.6	13.3	82.1	100	2.4	22.1	75.5
Nicaragua	100	36.3	13.3	50.3	100	4.3	24.8	70.9	100	48.0	15.2	36.8	100	5.9 2	26.1 6	68.0	100	17.7	10.3	72.0	100	2.2	23.1	74.7
Panama	100	25.4	21.8	52.9	100	3.7	21.3	74.9	100	31.3	27.1 /	41.6	100	5.7 2	28.8 6	65.5	100	15.4	12.7	71.9	100	0.8	9.7	89.6
South America																								
Argentina	100	0.8	29.2	70.1	100	0.3	20.9	78.8	100	1.0	43.0	56.0	100 (	0.5 3	30.3 6	69.2	100	0.4	9.0	90.6	100	0.1	7.1	92.9
Bolivia, Plurinational State of	100	32.3	22.7	45.1	100	0.7	20.5	78.8	100	32.5	30.7	36.7	100	1.2 2	28.0 7	70.8	100	32.0	12.6	55.4	100	0.0	9.1	6.06
Brazil	100	23.7	19.4	57.0	100	6.1	25.6	68.4	100	28.2	27.8	43.9	100	7.8 3	33.9 5	58.4	100	17.8	7.8	74.4	100	3.2	14.0	82.8
Chile	100	12.3	20.2	67.5	100	7.7	25.4	6.99	100	17.6	26.1	56.3	100 1	10.3 3	35.8 5	53.9	100	5.2	11.9	82.9	100	3.8	10.1	86.1
Colombia	100	23.1	17.9	59.1	100	3.8	22.6	73.7	100	32.1	21.1	46.8	100	5.7 2	29.0 6	65.3	100	9.8	13.2	77.0	100	1.3	14.2	84.5
Ecuador	100	34.8	21.1	44.1	100	8.2	20.8	71.0	100	37.4	28.0	34.6	100 1	0.2 2	26.1 6	63.7	100	31.0	10.5	58.6	100	4.6	11.3	84.1
Paraguay	100	29.4	20.5	50.1	100	3.3	19.8	76.9	100	33.9	28.1	38.0	100	5.0 2	24.8 7	70.2	100	22.8	9.4	67.9	100	0.6	11.7	87.7
Peru	100	33.5	16.7	49.8	100	3.8	23.4	72.9	100	37.4	22.1	40.5	100	5.0 3	30.5 6	64.6	100	29.2	10.7	60.1	100	2.2	12.0	85.9
Uruguay	100	10.9	26.8	62.3	100	8.5	18.1	73.4	100	15.2	38.3	46.5	100 1	12.0 2	25.7 6	62.3	100	4.9	10.9	84.3	100	3.8	8.3	87.9
Venezuela, Bolivarian Republic of	100	10.2	20.7	69.1	100	5.0	22.5	72.6	100	15.9	27.0	57.1	100	7.8 3	30.0	62.2	100	1.8	11.5	86.7	100	0.7	11.0	88.2
Northern America																								
United States	100	4.5	17.0	78.5	100	0.9	18.4	80.6	100	7.0	26.0 (	6.99	100	1.0 2	26.9 7	72.0	100	1.5	6.7	91.8	100	0.7	8.3	90.9
Arab States																								
Iraq	100	10.7	35.8	53.5	100	4.1	23.2	72.7	100	10.2	39.5	50.3	100	5.2 2	26.5 6	68.3	100	15.4	6.5	78.1	100	0.7	14.8	84.5
Jordan	100	2.8	25.5	71.8	100	0.9	14.2	84.9	100	2.8	26.6	70.6	100	1.2 1	16.4 8	82.4	100	1.8	11.1	87.1	100	0.1	5.6	94.3
Occupied Palestinian Territory	100	15.0	41.0	44.1	100	0.7	11.6	87.7	100	12.1	46.8	41.1	100 (	0.9 1	13.3 8	85.9	100	25.9	18.7	55.4	100	0.3	3.2	96.5
Syrian Arab Republic	100	19.3	35.0	45.8	100	1.0	12.6	86.5	100	18.7	35.6	45.7	100	1.7 1	14.8 8	83.5	100	26.7	22.9	50.4	100	0.1	5.0	95.0
Yemen	100	37.4	21.4	41.3	100	1.1	7.6	91.4	100	41.3	19.2	39.5	100	1.4	7.7 9	90.9	100	24.6	32.8	42.6	100	0.0	0.7	99.3

				Total	-							Men								Women	len			
	Infor	rmal emp	Informal employment (%)	(%)	Form	ial empl	Formal employment (%)	(%)	Inforn	nal empl	Informal employment (%)	(%)	Forma	l emplo	Formal employment (%)	(9	Inform	Informal employment (%)	pyment (	(%)	Forn	Formal employment (%)	oyment	(%)
	Total	Agriculture	۲itsubnl	Services	Total	Agriculture	findustry.	Services	letoT	Agriculture	lndustry	Services	letal	Agriculture	futur (19	Services	Total	Agriculture	findustry.	Services	letoT	Agriculture	Industry	Services
Asia and the Pacific																								
Eastern Asia																								
China	100	14.0	11.3	74.7	100	3.6	8.6	87.8	100	5.7	18.1	76.2	100	1.6	12.6 8	85.8	100	24.5	3.2	72.3	100	5.8	2.9	91.3
Japan	100	15.2	27.6	57.2	100	0.9	25.9	73.3	100	16.0	37.3	46.8	100	1.1	34.1 (	64.8	100	14.2	14.8	71.0	100	0.6	14.9	84.4
Korea, Republic of	100	13.5	19.1	67.4	100	1.4	27.6	71.0	100	13.7	27.7	58.6	100	2.2	34.8 (	63.0	100	13.2	11.2	75.7	100	0.1	15.4	84.5
Mongolia	100	65.7	12.4	21.8	100	1.8	31.0	67.2	100	65.9	16.2	17.9	100	2.7	47.6 4	49.7	100	66.1	7.5	26.4	100	0.9	14.9	84.2
South-Eastern Asia and the Pacific	a and th	e Pacific										I												
Pacific Islands																								
Samoa	100	49.8	8.8	41.5	100	25.1	12.4	62.5	100	56.9	8.9	34.2	100	37.3	14.1 4	48.6	100	17.6	9.9	72.5	100	2.8	7.3	89.9
South-Eastern Asia	sia											l												
Brunei Darussalam	100	0.3	28.4	71.3	100	0.7	14.4	85.0	100	0.3	39.7	0.09	100	0.7	15.9 8	83.4	100	0.1	11.5	88.4	100	9.0	10.7	88.8
Cambodia	100	46.9	20.9	32.2	100	3.4	9.7	86.9	100	45.2	22.3	32.6	100	3.2	8.1 8	88.7	100	48.7	19.5	31.8	100	3.2	13.6	83.2
Indonesia	100	38.8	22.1	39.1	100	6.9	20.2	72.9	100	38.4	23.5	38.1	100	8.3	20.7	71.0	100	39.6	20.0	40.4	100	3.2	17.7	79.1
Lao People's Democratic Republic	100	71.8	9.7	18.5	100	3.6	21.9	74.5	100	71.1	11.2	17.7	100	3.2	23.1	73.7	100	72.3	8.3	19.3	100	4.5	19.5	76.0
Myanmar	100	61.0	14.4	24.5	100	47.4	10.0	42.6	100	55.7	17.6	26.7	100	50.9	9.9	39.3	100	66.7	11.2	22.1	100	33.7	11.6	54.7
Timor-Leste	100	62.8	10.1	27.2	100	4.2	10.2	85.6	100	63.7	10.9	25.4	100	5.5	11.3 8	83.2	100	9.09	8.6	30.8	100	2.0	6.7	91.3
Viet Nam	100	57.8	17.1	25.1	100	1.9	37.1	61.0	100	54.8	22.6	22.6	100	2.4	36.2 (	61.4	100	61.0	11.3	27.7	100	1.4	37.8	60.8
Southern Asia																								
Bangladesh	100	50.2	19.9	29.9	100	9.7	16.9	73.4	100	42.2	22.7	35.1	100	8.3	18.9	72.8	100	64.6	14.6	20.8	100	11.0	13.3	75.7
India	100	48.6	25.1	26.4	100	1.1	42.6	56.3	100	45.1	25.3	29.6	100	0.9	47.5	51.6	100	60.1	23.0	17.0	100	2.0	29.0	69.0
Nepal	100	72.5	11.2	16.3	100	4.8	15.5	79.7	100	65.1	16.0	18.9	100	4.9	19.9	75.2	100	78.9	6.8	14.4	100	5.6	7.5	87.0
Pakistan	100	49.7	22.9	27.5	100	2.0	29.1	68.9	100	41.1	24.4	34.5	100	2.0	30.8 (	67.3	100	76.4	18.1	5.5	100	3.3	9.3	87.4
Sri Lanka	100	37.0	26.3	36.7	100	9.7	23.3	67.0	100	34.4	27.6	38.0	100	8.4	22.8 (	68.8	100	43.3	23.3	33.3	100	12.1	23.6	64.3

				Total	_							Men								Women	Ę			
	Inforn	nal emp	Informal employment (%)	(%)	Form	ial empl	Formal employment (%)	(%	Inforn	nal empl	Informal employment (%)	(%	Formal	employ	Formal employment (%)		Informa	l emplo	Informal employment (%)	(%	Forma	Formal employment (%)	ment (%	()
	Total	Agriculture	lndustry	Services	lstoT	Agriculture	lndustry	Services	lstoT	Agriculture	lndustry	Services	lstoT	Agriculture	, rudustry	Services	Total	Agriculture	lndustry	Services	lstoT	Agriculture	lndustry	Services
<b>Europe and Central Asia</b>	Asia																							
<b>Central and Western Asia</b>	ern Asia																							
Central Asia																								
Kyrgyzstan	100	34.9	18.1	47.0	100	15.8	14.4	69.8	100	32.1	22.6	45.3	100 2	22.7 1	19.5 5	57.9	100 3	39.7 1	10.4 4	49.9	100	6.4	7.4	86.2
Tajikistan	100	54.1	14.2	31.8	100	40.5	17.2	42.3	100	41.1	21.0	37.9	100 3	36.8 2	24.5 38	38.7	100 7	72.8	4.2 2	23.0	100 4	41.9	10.9	47.2
Western Asia																								
Armenia	100	67.2	13.3	19.5	100	0.7	18.6	80.7	100	52.2	23.9	24.0	100 1	1.0 2	27.1 7	71.9	100 8	81.9	3.0 1	15.1	100	0.3	7.2	92.6
Cyprus	100	20.2	16.9	63.0	100	1.9	16.9	81.2	100	26.1	22.9	51.0	100 2	2.3 2	25.4 7:	72.4	100 1	10.0	6.0 8	84.0	100	1.3	6.8	91.9
Turkey	100	45.8	21.3	32.9	100	4.3	33.6	62.1	100	35.0	28.0	36.9	100	5.0 3	36.5 58	58.4	100 6	60.8 1	12.0 2	27.2	100	1.1 2	24.8	74.2
Eastern Europe																								
Bulgaria	100	21.4	16.4	62.2	100	4.4	32.7	63.0	100	24.1	22.8	53.2	100 6	6.1 3	39.8 54	54.1	100 1	17.6	T. T.	74.8	100	2.6 2	24.8	72.6
Czech Republic	100	3.4	33.3	63.4	100	2.5	39.3	58.2	100	4.3	43.9	51.8	100	3.3 5	50.6 4	46.1	100	1.6 1	10.8 8	87.7	100	1.4	25.4	73.1
Hungary	100	14.4	24.1	61.5	100	3.1	32.1	64.7	100	14.2	31.4	54.5	100 4	4.7 4	42.8 5	52.5	100 1	15.1 ]	11.5 7	73.4	100	1.3	20.1	78.6
Moldova, Republic of	100	70.0	14.4	15.6	100	14.6	20.2	65.3	100	64.4	22.9	12.7	100 2	21.3 2	24.7 5.	54.1	100 7	77.1	3.6 1	19.2	100	8.5	15.8	75.7
Poland	100	25.6	20.1	54.4	100	1.5	37.2	61.2	100	24.2	26.8	49.0	100 2	2.3 5	53.2 4.	44.6	100 2	28.1	9.4 6	62.5	100	0.8	20.5	78.7
Romania	100	71.9	9.1	19.0	100	3.3	40.1	56.7	100	67.9	14.7	17.4	100 4	4.8 4	47.4 4	47.8	100 7	77.0	2.0 2	21.0	100	1.2	31.1	67.7
Russian Federation	100	7.8	30.7	61.5	100	5.7	25.0	69.3	100	9.2	42.0	48.8	100	7.0 3	34.8 5	58.2	100	5.8 ]	16.7 7	77.5	100	4.5	15.6	79.9
Slovakia	100	5.4	38.6	56.0	100	3.3	34.4	62.4	100	6.7	50.0	43.3	100 4	4.7 4	45.5 4	49.8	100	3.0 ]	10.6 8	86.5	100	1.6	22.3	76.2
Northern, Southern and Western Europe	n and We.	stern Eu	irope																					
Northern Europe																								
Denmark	100	9.3	15.3	75.4	100	1.4	19.5	79.0	100	12.0	19.4	68.6	100 2	2.2 2	29.1 6	68.7	100	3.7	7.1 8	89.2	100	0.7	9.6	89.7
Estonia	100	10.2	14.7	75.1	100	3.1	31.3	65.6	100	14.1	22.3	63.6	100 4	4.1 4	42.9 5;	53.0	100	5.3	5.4 8	89.3	100	2.1 ]	19.1	78.8
Finland	100	21.9	20.1	58.1	100	2.8	21.4	75.8	100	25.2	25.5	49.4	100	3.8 3	34.5 6	61.7	100 1	14.7	8.4 7	76.9	100	1.7	8.1	90.2
Iceland	100	5.4	18.2	76.5	100	4.0	18.1	77.9	100	6.4	24.2	69.4	100 6	6.3 2	27.3 61	66.4	100	3.5	7.9 8	88.6	100	1.6	8.1	90.2
Ireland	100	26.0	16.5	57.6	100	2.7	18.8	78.5	100	30.0	19.9	50.1	100 4	4.5 2	27.9 6	67.6	100 1	13.8	6.5 7	79.7	100	0.8	8.9	90.4
Latvia	100	20.8	15.2	64.1	100	4.5	25.2	70.3	100	25.6	21.6	52.9	100 6	6.7 3	37.8 5	55.6	100 1	14.9	7.4 7	<i>T.T</i>	100	2.5 ]	13.0	84.5

				Total	_							Men								Women	ua			
	Infor	Informal employment (%)	loyment	(%)	Forn	nal empl	Formal employment (%)	(%)	Infor	Informal employment (%)	oyment	(%)	Forma	ıl emplo	Formal employment (%)	()	Inform	Informal employment (%)	pyment (	(%)	Form	Formal employment (%)	yment ('	(%
	Total	Agriculture	lndustry	Services	Total	Agriculture	lndustry	Services	Total	Agriculture	lndustry	Services	Total	Agriculture	lndustry	Services	letoT	Agriculture	lndustry	Services	lstoT	Agriculture	Judustry	Services
Lithuania	100	32.4	16.4	51.2	100	6.1	25.3	68.6	100	37.7	23.0	39.3	100	8.7	33.3	58.0	100	29.3	11.4	59.3	100	3.4	17.0	79.6
Norway	100	12.8	21.1	66.1	100	1.4	20.8	77.9	100	15.6	31.2	53.1	100	1.9	33.0 6	65.1	100	7.2	1.0	91.8	100	0.8	7.7	91.5
Sweden	100	0.7	4.4	94.9	100	1.7	19.4	78.9	100	0.7	5.7	93.6	100	2.5	30.6 6	6.9	100	0.8	2.7	96.5	100	0.8	7.1	92.2
United Kingdom	100	3.3	23.1	73.6	100	1.0	18.6	80.4	100	4.0	32.7	63.4	100	1.3	28.0 7	70.7	100	2.2	6.7	91.2	100	0.6	8.5	6.06
Southern Europe												1				J				1				
Albania	100	66.2	12.8	21.0	100	3.5	25.6	70.9	100	57.7	18.3	24.0	100	5.4	26.2 6	68.5	100	78.2	4.5	17.3	100	1.5	25.4	73.2
Bosnia and Herzegovina	100	56.6	6.8	36.6	100	2.3	36.5	61.2	100	56.8	10.2	33.0	100	3.5	46.3 5	50.2	100	55.6	2.8	41.6	100	0.7	20.4	79.0
Croatia	100	29.7	16.2	54.1	100	4.4	27.8	67.8	100	31.0	22.6	46.4	100	5.7	37.8 5	56.5	100	28.5	7.6	63.9	100	2.9	16.3	80.8
Greece	100	34.2	11.5	54.3	100	1.8	16.2	82.0	100	32.9	15.0	52.1	100	2.6	22.2 7	75.2	100	36.3	5.4	58.3	100	1.0	8.5	90.4
Italy	100	7.5	21.7	70.8	100	2.6	28.2	69.2	100	9.0	29.1	61.9	100	3.1	39.1	57.8	100	4.9	8.2	86.9	100	1.8	13.8	84.4
Malta	100	5.3	20.4	74.3	100	0.8	22.0	77.2	100	7.5	26.3	66.2	100	1.1	29.3 (	69.6	100	0.8	6.9	92.4	100	0.3	10.1	89.6
Portugal	100	19.8	16.5	63.8	100	2.7	25.9	71.4	100	19.2	24.9	55.9	100	4.0	34.9 (	61.2	100	20.7	5.2	74.2	100	1.4	16.7	81.9
Serbia	100	47.2	13.0	39.8	100	12.6	26.7	60.8	100	39.8	20.1	40.2	100	17.7	31.4	50.9	100	55.8	4.5	39.7	100	5.5	20.1	74.4
Slovenia	100	47.4	14.0	38.6	100	5.2	33.1	61.7	100	42.1	21.3	36.6	100	5.0	45.3 4	49.7	100	54.3	3.8	41.9	100	5.6	18.7	75.6
Spain	100	5.6	8.6	85.8	100	3.5	23.3	73.2	100	7.8	14.2	78.1	100	4.8	33.1 (	62.1	100	3.1	2.2	94.7	100	1.9	11.0	87.2
Western Europe																								
Austria	100	16.6	10.8	72.7	100	3.2	27.7	69.1	100	16.4	18.7	64.9	100	3.4	39.9 5	56.8	100	17.1	3.3	79.6	100	2.8	13.5	83.7
Belgium	100	6.3	14.4	79.3	100	0.3	22.3	77.5	100	6.8	22.6	70.6	100	0.3	34.5 6	65.2	100	5.3	3.0	91.6	100	0.2	8.5	91.2
France	100	8.8	11.1	80.1	100	1.9	21.0	77.1	100	11.4	17.1	71.5	100	2.7	31.3 6	0.99	100	5.9	4.6	89.5	100	1.0	9.8	89.2
Germany	100	2.5	20.5	77.0	100	1.2	29.6	69.2	100	2.9	28.3	68.9	100	1.5	42.4 5	56.0	100	2.1	11.4	86.5	100	0.8	14.8	84.4
Luxem bourg	100	10.8	2.3	86.9	100	1.1	10.6	88.3	100	1.6	3.2	95.2	100	1.7	15.4 8	83.0	100	14.3	2.2	83.5	100	0.4	4.5	95.1
Netherlands	100	5.7	13.7	80.6	100	1.8	16.2	81.9	100	6.2	19.9	73.9	100	2.5	25.0 7	72.6	100	5.8	5.8	88.4	100	1.0	5.9	93.1
Switzerland	100	6.8	4.2	89.0	100	2.9	22.0	75.1	100	5.9	4.2	89.9	100	3.9	31.4 6	64.7	100	7.9	4.7	87.4	100	1.8	11.2	87.0
		-		for the second sec		_	-	-				-	-											

					Tota	l employ	Total employment (%)	(%								Noi	Non-agricultural employment (%)	iltural ei	mployme	ent (%)				
		Total	_			Men	5			Women	en			Total				Men				Women	en	
	IstoT	In the informal sector	in the formal sector	spionasuon ni	lstoT	In the informal sector	In the formal sector	spionasuon ni	letoT	In the informal sector	In the formal sector	spionasuon ni	lstoT	In the informal sector	In the formal sector	spiodəsuod ni	IstoT	In the informal sector	in the formal sector	spionasuon ni	lstoT	In the informal sector	In the formal sector	spioyəsnoy ul
Africa																								
Northern Africa																								
Føvnt	100	59.9	39.6	0.6	100	61.6	37.9	0.5	100	53.2	45.9	6.0	100	45.3	54.0 (	0.8	100	50.3	49.1	0.6	100	19.5	79.0	1.5
Morocco	_		38.3	1.6	100	62.9	36.7		100	45.6	46.6	7.8	100			2.2			41.1	0.6	100	39.2	50.4	10.4
Tunisia	100	53.4	46.2	0.5	100	56.0	44.0	0.0	100	44.8	53.3	2.0	100	47.7	51.8 (	0.5	100	51.3 /	48.7	0.0	100	35.7	62.0	2.4
Sub-Saharan Africa												]								1				
Central Africa																								
Angola	100	79.5	20.0	0.6	100	71.3	28.6	0.2	100	88.2	10.8	1.0	100	63.2	35.7 ]	1.0	100	53.3 4	46.4	0.3	100	76.5	21.4	2.1
Cameroon	100	83.8	15.4	0.8	100	77.3	22.4	0.3	100	90.5	8.3	1.2	100	62.9	35.3 ]	1.8	100	55.1 4	44.2	0.7	100	73.7	22.8	3.5
Chad	100	92.0	8.0	0.0	100	89.4	10.6	0.0	100	95.7	4.3	0.0	100	68.0	32.0 (	0.0	100	60.2	39.8	0.0	100	80.8	19.2	0.0
Congo	100	57.3	40.2	2.5	100	59.8	37.7	2.5	100	55.3	42.1	2.5	100	56.0	41.4 2	2.6	100	59.2	38.2	2.6	100	53.4	43.9	2.7
Congo, Democratic Republic of the	100	88.8	11.2	0.0	100	82.2	17.8	0.0	100	95.2	4.8	0.0	100	68.6	31.4 (	0.0	100	57.8	42.2	0.0	100	86.6	13.4	0.0
Eastern Africa																								
Comoros	100	84.6	15.4	0.0	100	80.1	19.9	0.0	100	91.3	8.7	0.0	100	69.5	30.5 (	0.0	100	63.4	36.6	0.0	100	81.0	19.0	0.0
Madagascar	100	83.8	10.2	6.0	100	83.3	12.6	4.1	100	84.3	7.7	8.0	100	48.1	31.5 2	20.4	100	45.9	40.2	13.8	100	50.1	23.1	26.8
Malawi	100	75.0	20.5	4.6	100	75.5	20.0	4.6	100	74.5	21.0	4.5	100	50.2	42.0 7	7.7	100	50.2	41.7	8.1	100	50.3	42.3	7.4
Rwanda	100	91.1	6.9	2.0	100	88.3	9.7	2.0	100	93.6	4.4	2.0	100	67.6	24.7	7.7	100	69.8	24.8	5.4	100	63.3	24.6	12.1
Tanzania, United Republic of	100	83.5	14.2	2.3	100	79.9	18.5	1.6	100	87.2	9.9	2.9	100	48.2	45.8 (	0.9	100	42.1	54.1	3.8	100	55.5	35.7	8.7
Uganda	100	87.7	12.3	0.0	100	84.0	16.0	0.0	100	91.7	8.3	0.0	100	67.2	32.8 (	0.0	100	62.8	37.2	0.0	100	73.8	26.2	0.0
Zambia	100	80.1	13.7	6.2	100	73.0	20.8	6.3	100	86.5	7.3	6.2	100	54.7	32.1 1	13.2	100	51.1	38.9	10.0	100	60.1	22.2	17.8
Southern Africa																								
Botswana	100	52.2	39.1	8.7	100	51.2	40.5	8.4	100	53.4	37.5	9.1	100	41.3	51.1 7	7.7	100	39.7	55.6	4.8	100	43.0	46.4	10.7
Namibia	100	46.8	37.9	15.3	100	50.4	38.1	11.5	100	42.9	37.6	19.5	100	46.3	44.7 9	9.1	100	49.6	44.9	5.5	100	42.8	44.4	12.7
South Africa	100	35.1	55.7	9.2	100	39.2	56.1	4.8	100	29.7	55.3	15.0	100	33.0	57.4 9	9.6	100	36.9	58.2	4.9	100	28.1	56.4	15.6

Table B.8 – Share of employment in the informal sector, in the formal sector and in households by sex

					Tot	al emplo	Total employment (%)	(%								Ż	on-agric	ultural	employn	Non-agricultural employment (%)				
		2	Total			×	Men			Women	nen			To	Total			Men	E.			Women	len	
	lstoT	In the informal sector	In the formal sector	splodesuod ni	Total	In the informal sector	In the formal sector	splodesuod nl	lstoT	In the informal sector	In the formal sector	sblodesuod nl	Total	In the informal sector	In the formal sector	splodesuod nl	Total	In the informal sector	In the formal sector	splodesuod ni	lstoT	In the informal sector	In the formal sector	splodesuod ni
Western Africa																								
Benin	100	90.6	8.3	1.1	100	87.1	12.6	0.3	100	93.8	4.3	1.9	100	83.7	14.3	2.0	100	74.3	25.1	0.7	100	90.5	9.9	2.9
Burkina Faso	100	65.6	14.9	19.5	100	65.4	18.3	16.3	100	65.7	11.8	22.5	100	68.2	25.7	6.1	100	59.1	36.5	4.4	100	76.7	15.6	7.7
Cabo Verde	100	31.9	59.5	8.5	100	41.8	56.2	2.0	100	21.7	63.0	15.3	100	29.3	60.4	10.2	100	38.8	58.7	2.6	100	20.7	62.1	17.2
Côte d'Ivoire	100	86.0	12.4	1.6	100	83.7	15.9	0.4	100	89.5	7.3	3.3	100	75.8	21.3	2.9	100	68.7	30.5	0.8	100	83.9	10.9	5.3
Gambia	100	62.7	30.1	7.3	100	53.6	38.0	8.4	100	74.3	19.9	5.9	100	52.3	40.3	7.4	100	44.0	48.2	7.8	100	65.0	28.2	6.9
Ghana	100	86.1	13.2	0.7	100	82.1	17.4	0.5	100	89.9	9.4	0.7	100	75.4	23.4	1.1	100	66.3	32.7	1.0	100	82.9	15.8	1.3
Liberia	100	79.7	17.7	2.6	100	71.5	25.2	3.3	100	87.9	10.2	1.9	100	65.6	29.8	4.6	100	51.5	42.7	5.8	100	79.7	16.9	3.4
Mali	100	74.2	11.4	14.4	100	68.6	15.2	16.2	100	80.9	6.8	12.4	100	70.6	24.1	5.3	100	61.7	32.8	5.4	100	81.0	13.8	5.2
Niger	100	77.3	18.2	4.5	100	69.8	25.1	5.1	100	86.1	10.2	3.7	100	72.6	25.2	2.2	100	56.9	41.5	1.6	100	86.3	11.0	2.7
Nigeria	100	80.4	10.0	9.6	100	78.8	12.9	8.3	100	82.1	6.9	11.0	100	82.9	15.2	1.8	100	75.9	22.7	1.5	100	88.5	9.5	2.1
Senegal	100	70.9	22.8	6.2	100	68.8	28.7	2.5	100	74.1	14.1	11.9	100	59.2	32.1	8.8	100	55.2	42.0	2.8	100	64.5	18.8	16.7
Sierra Leone	100	89.9	8.5	1.7	100	90.1	8.1	1.8	100	89.6	8.8	1.5	100	81.0	16.4	2.6	100	80.9	16.2	2.9	100	81.1	16.7	2.3
Togo	100	81.1	12.3	9.9	100	77.9	16.4	5.7	100	84.3	8.4	7.4	100	74.4	15.7	10.0	100	68.3	22.5	9.2	100	79.6	9.8	10.7
Americas																								
Latin America and the Caribbean	aribbean																							
The Caribbean																								
Dominican Republic	100	38.2	54.8	7.0	100	47.6	51.7	0.7	100	23.3	59.7	17.0	100	32.0	59.9	8.1	100	39.3	59.8	0.9	100	22.4	0.09	17.6
Central America																								
Costa Rica	100	27.4	64.7	7.9	100	30.4	68.1	1.5	100	22.7	59.6	17.7	100	25.5	65.6	8.9	100	28.0	70.3	1.7	100	22.1	59.5	18.4
El Salvador	100	53.9	41.4	4.7	100	54.0	45.4	0.5	100	53.7	36.0	10.3	100	48.1	46.1	5.8	100	42.7	56.6	0.8	100	53.5	35.8	10.8
Guatemala	100	63.3	29.0	7.7	100	64.5	30.0	5.4	100	60.9	27.1	12.0	100	54.5	39.3	6.2	100	52.1	46.7	1.2	100	57.5	29.9	12.5
Honduras	100	70.3	21.5	8.2	100	72.8	19.4	7.8	100	66.1	25.0	8.9	100	66.1	29.4	4.5	100	67.8	31.5	0.8	100	64.4	27.3	8.3
Mexico	100	29.1	66.3	4.7	100	27.9	71.5	0.7	100	31.0	57.9	11.1	100	33.1	61.5	5.4	100	33.8	65.4	0.8	100	32.2	56.3	11.6
Nicaragua	100	71.6	23.8	4.6	100	77.8	20.8	1.4	100	61.4	28.7	9.9	100	60.1	33.1	6.8	100	62.4	35.0	2.5	100	57.9	31.2	10.9
Panama	100	36.3	58.8	5.0	100	42.4	56.6	1.0	100	27.7	61.9	10.5	100	28.2	66.0	5.8	100	32.8	66.0	1.3	100	22.7	66.0	11.4

					Tota	Total employme	oyment (%)	(%								2	Non-agricultural employment (%)	cultural	employn	nent (%)				
		Total	tal			W	Men			Wo	Women			Total	al			Men	u			Women	nen	
	Total	In the informal sector	In the formal sector	spiodəsuod ni	letoT	In the informal sector	In the formal sector	spiodəsuod ni	letoT	In the informal sector	In the formal sector	spionasuon ni	Total	In the informal sector	In the formal sector	spiodəsuod ni	Total	In the informal sector	In the formal sector	spiodəsuod ni	letoT	In the informal sector	In the formal sector	spiouəsnou ul
South America																								
Argentina	100	32.8	59.5	7.7	100	37.6	62.2	0.3	100	26.6	55.9	17.5	100	32.7	59.5	7.8	100	37.5	62.3	0.3	100	26.5	55.9	17.6
Bolivia, Plurinational State of	100	68.9	29.1	2.0	100	68.7	31.1	0.2	100	69.0	26.6	4.4	100	55.5	41.6	2.9	100	56.2	43.6	0.2	100	54.7	38.9	6.4
Brazil	100	32.0	61.2	6.8	100	38.6	60.5	0.9	100	23.1	62.2	14.8	100	28.1	64.3	7.6	100	33.2	65.8	1.0	100	22.0	62.6	15.5
Chile	100	32.2	63.8	4.0	100	33.9	65.6	0.4	100	29.7	61.2	9.1	100	30.2	65.4	4.4	100	31.4	68.1	0.5	100	28.6	61.6	9.7
Colombia	100	56.5	41.3	2.2	100	60.09	39.8	0.2	100	51.2	43.5	5.2	100	50.5	46.9	2.6	100	51.8	48.0	0.2	100	49.0	45.5	5.5
Ecuador	100	47.0	50.3	2.7	100	47.7	52.1	0.3	100	46.0	47.6	6.4	100	36.3	60.1	3.6	100	36.7	63.0	0.4	100	35.9	56.1	8.0
Paraguay	100	46.2	46.7	7.1	100	49.0	50.3	0.7	100	42.3	41.7	16.0	100	35.1	56.1	8.8	100	36.7	62.4	0.9	100	33.1	48.2	18.7
Peru	100	57.0	40.6	2.4	100	55.4	44.3	0.3	100	58.7	36.4	4.9	100	42.0	54.6	3.4	100	39.9	59.8	0.4	100	44.4	48.8	6.8
Uruguay	100	29.9	63.8	6.4	100	37.0	62.3	0.7	100	21.1	65.6	13.3	100	26.9	66.2	6.9	100	33.1	66.1	0.8	100	20.0	66.2	13.8
Venezuela, Bolivarian Republic of	100	30.8	59.4	9.8	100	35.6	57.5	7.0	100	23.9	62.1	14.0	100	27.6	61.8	10.6	100	30.9	61.3	7.9	100	23.3	62.5	14.2
Northern America																								
United States	100	16.6	81.8	1.6	100	18.2	80.4	1.5	100	14.9	83.5	1.7	100	16.3	82.1	1.6	100	17.6	80.9	1.5	100	14.8	83.5	1.7
Arab States																								
Iraq	100	59.7	40.3	0.0	100	62.2	37.8	0.0	100	44.8	55.2	0.1	100	59.5	40.5	0.0	100	62.0	38.0	0.0	100	44.1	55.8	0.1
Jordan	100	43.9	56.1	0.0	100	47.8	52.2	0.0	100	24.8	75.2	0.0	100	42.0	58.0	0.0	100	46.3	53.7	0.0	100	21.3	78.7	0.0
Occupied Palestinian Territory	100	31.4	66.4	2.3	100	33.1	66.4	0.5	100	24.0	66.2	9.8	100	26.0	74.0	0.1	100	28.2	71.7	0.0	100	14.5	85.3	0.2
Syrian Arab Republic	100	66.0	34.0	0.0	100	67.8	32.2	0.0	100	57.6	42.4	0.0	100	55.6	44.4	0.0	100	60.9	39.1	0.0	100	16.2	83.8	0.0
Yemen	100	68.8	30.9	0.4	100	68.5	31.2	0.4	100	72.7	26.8	0.5	100	56.6	42.9	0.5	100	57.1	42.4	0.5	100	45.0	53.9	1.1

					Tota	l emplo	Total employment (%)	(%								No	n-agricı	iltural e	Non-agricultural employment (%)	ent (%)				
		Total				Men	=			Women	en			Total				Men				Women	en	
	Total	In the informal sector	In the formal sector	splodesuod ni	Total	In the informal sector	In the formal sector	splodesuod ni	letoT	In the informal sector	In the formal sector	splodesuod ni	letal	In the informal sector	In the formal sector	splodesuod nl	Total	In the informal sector	in the formal sector	splodəsuod ni	letoT	In the informal sector	in the formal sector	spionasuon ni
Asia and the Pacific																								
Eastern Asia																								
China	100	48.4	51.7	0.0	100	50.2	49.8	0.0	100	45.6	54.4	0.0	100	47.3	52.7	0.0	100	49.2	50.8	0.0	100	44.4	55.6	0.0
Japan	100	17.0	82.5	0.4	100	18.1	81.8	0.1	100	15.7	83.5	0.8	100	14.7	84.9	0.5	100	15.5	84.4	0.1	100	13.7	85.5	0.9
Korea, Republic of	100	26.1	73.7	0.2	100	23.2	76.8	0.0	100	30.3	69.2	0.5	100	23.3	76.5	0.2	100	21.0	79.0	0.0	100	26.7	72.8	0.5
Mongolia	100	45.7	48.5	5.8	100	47.8	44.8	7.5	100	43.5	52.6	4.0	100	25.4	67.1	7.5	100	27.4	62.8	9.9	100	23.4	71.7	5.0
South-Eastern Asia and the Pacific	e Pacifi	5										1				1				1				
Pacific Islands																								
Samoa	100	22.1	72.8	5.1	100	23.8	72.3	3.9	100	19.2	73.8	7.0	100	21.7	75.6	2.7	100	23.6	75.2	1.2	100	18.6	76.3	5.0
South-Eastern Asia																								
Brunei Darussalam	100	4.6	89.7	5.8	100	4.6	94.5	6.0	100	4.5	83.3	12.2	100	4.7	89.1	6.1	100	4.8	94.2	1.0	100	4.6	82.7	12.7
Cambodia	100	<i>T.T.</i>	21.2	1.1	100	78.1	21.4	0.5	100	77.3	21.0	1.7	100	67.6	31.0	1.4	100	68.0	31.4	0.6	100	67.1	30.6	2.3
Indonesia	100	67.5	20.2	12.3	100	68.9	19.7	11.5	100	65.4	21.0	13.7	100	62.7	28.1	9.2	100	64.1	27.4	8.6	100	60.7	29.2	10.2
Lao People's Democratic Republic	100	86.7	13.2	0.1	100	83.1	16.9	0.0	100	90.4	9.6	0.1	100	55.2	44.7	0.2	100	46.7	53.2	0.1	100	65.0	34.7	0.3
Myanmar	100	71.8	28.0	0.3	100	71.9	28.0	0.1	100	71.6	27.9	0.5	100	65.4	34.1	0.5	100	66.2	33.6	0.2	100	64.4	34.7	0.9
Timor-Leste	100	57.1	38.1	4.9	100	50.4	43.5	6.1	100	69.8	27.7	2.5	100	29.8	62.3	7.9	100	22.8	67.7	9.4	100	45.1	50.5	4.5
Viet Nam	100	61.1	35.2	3.7	100	61.3	36.1	2.6	100	60.9	34.3	4.9	100	37.5	61.9	0.7	100	39.1	60.9	0.1	100	35.7	63.0	1.4
Southern Asia																								
Bangladesh	100	49.1	24.2	26.7	100	51.7	25.7	22.6	100	42.7	20.6	36.7	100	49.4	40.9	9.7	100	53.9	40.8	5.3	100	35.8	41.2	23.0
India	100	81.1	18.0	0.8	100	81.0	18.7	0.4	100	81.8	15.7	2.5	100	64.7	33.7	1.6	100	66.7	32.7	0.7	100	55.1	38.7	6.2
Nepal	100	91.1	8.6	0.3	100	85.8	14.0	0.2	100	95.8	3.9	0.3	100	69.4	29.6	1.0	100	65.1	34.3	0.6	100	78.2	20.0	1.8
Pakistan	100	80.4	18.6	1.0	100	77.8	21.7	0.5	100	89.1	8.3	2.7	100	6.99	31.4	1.8	100	67.5	31.7	0.8	100	61.5	28.9	9.5
Sri Lanka	100	9.09	37.3	2.1	100	63.5	34.5	1.9	100	55.0	42.4	2.5	100	49.2	47.8	3.0	100	53.5	43.8	2.7	100	40.4	55.9	3.8

					Tota	l emplo	Total employment (%)	(%								No	n-agricu	Non-agricultural employment (%)	mployme	ent (%)				
		Total	al			Men	Ę			Women	nen			Total				Men				Women	nen	
	Total	In the informal sector	In the formal sector	splodəsuod ni	letoT	In the informal sector	ln the formal sector	splodəsuod ni	letoT	In the informal sector	In the formal sector	splodəsuod ni	letoT	In the informal sector	In the formal sector	spionasuon ni	letal	In the informal sector	In the formal sector	splodəsuod ni	Total	In the informal sector	In the formal sector	spionasuon ni
<b>Europe and Central Asia</b>																								
<b>Central and Western Asia</b>																								
Central Asia																								
Kyrgyzstan	100	60.9	37.2	1.8	100	57.2	40.9	1.8	100	9.99	31.6	1.9	100	64.8	32.8	2.5	100	59.3	38.2	2.5	100	72.5	25.2	2.4
Tajikistan	100	62.5	31.6	6.0	100	59.4	32.7	7.8	100	67.6	29.5	2.9	100	42.5	47.2	10.4	100	47.3	41.4	11.2	100	27.2	65.2	7.6
Western Asia												,				1				1				
Armenia	100	39.2	49.2	11.6	100	44.2	48.3	7.5	100	33.8	50.2	16.1	100	24.0	75.6	0.4	100	30.4	69.3	0.3	100	16.0	83.5	0.5
Cyprus	100	15.0	85.0	0.0	100	16.9	83.2	0.0	100	13.0	87.0	0.0	100	14.0	86.0	0.0	100	15.3	84.7	0.0	100	12.6	87.4	0.0
Turkey	100	32.0	67.4	0.5	100	27.7	72.3	0.1	100	42.0	56.3	1.7	100	19.3	80.1	0.7	100	19.6	80.3	0.1	100	18.3	79.3	2.4
Eastern Europe																								
Bulgaria	100	15.0	85.0	0.0	100	16.9	83.1	0.0	100	12.9	87.1	0.0	100	12.7	87.3	0.0	100	14.2	85.8	0.0	100	11.1	88.9	0.0
Czech Republic	100	8.6	91.4	0.0	100	10.6	89.4	0.0	100	6.1	93.9	0.0	100	8.6	91.5	0.0	100	10.5	89.5	0.0	100	6.1	93.9	0.0
Hungary	100	11.8	88.2	0.0	100	14.2	85.8	0.0	100	9.1	90.9	0.0	100	10.2	89.9	0.0	100	12.4	87.6	0.0	100	7.7	92.3	0.0
Moldova, Republic of	100	21.9	<i>T.T.T</i>	0.4	100	25.4	74.5	0.1	100	18.4	80.9	0.8	100	7.6	91.8	9.0	100	13.4	86.5	0.1	100	2.3	96.7	1.0
Poland	100	20.1	79.9	0.0	100	21.1	78.9	0.0	100	19.0	81.0	0.0	100	12.1	88.0	0.0	100	13.7	86.3	0.0	100	10.0	90.0	0.0
Romania	100	35.4	64.6	0.0	100	33.6	66.4	0.0	100	37.7	62.3	0.0	100	19.3	80.7	0.0	100	17.9	82.1	0.0	100	21.1	78.9	0.0
Russian Federation	100	24.4	75.6	0.0	100	27.9	72.1	0.0	100	20.9	79.1	0.0	100	24.3	75.7	0.0	100	27.9	72.1	0.0	100	20.8	79.2	0.0
Slovakia	100	16.4	83.6	0.0	100	20.5	79.5	0.0	100	11.7	88.3	0.0	100	16.2	83.8	0.0	100	20.3	79.8	0.0	100	11.5	88.5	0.0
Northern, Southern and Western Europe	lestern E	urope																						
Northern Europe																								
Denmark	100	4.1	95.9	0.0	100	4.7	95.3	0.0	100	3.4	96.6	0.0	100	4.0	96.0	0.0	100	4.6	95.4	0.0	100	3.4	90.6	0.0
Estonia	100	5.4	94.6	0.0	100	6.2	93.8	0.0	100	4.7	95.3	0.0	100	4.8	95.2	0.0	100	5.2	94.8	0.0	100	4.4	92.6	0.0
Finland	100	6.2	93.8	0.0	100	8.2	91.8	0.0	100	4.1	95.9	0.0	100	5.1	94.9	0.0	100	6.5	93.5	0.0	100	3.6	96.4	0.0
Iceland	100	4.6	95.4	0.0	100	5.6	94.4	0.0	100	3.5	96.5	0.0	100	4.5	95.5	0.0	100	5.6	94.4	0.0	100	3.4	90.6	0.0
Ireland	100	11.8	88.3	0.0	100	16.4	83.6	0.0	100	6.6	93.4	0.0	100	9.4	90.6	0.0	100	12.7	87.4	0.0	100	5.9	94.1	0.0

					Tot	al emplo	Total employment (%)	(%								ž	Non-agricultural employment (%)	ultural	employm	nent (%)				
		To	Total			Men	ua			Women	nen			Total	al			Men	ч			Women	ien	
	lstoT	In the informal sector	In the formal sector	spionasuon ni	letoT	In the informal sector	In the formal sector	splodəsuod ni	letoT	In the informal sector	In the formal sector	splodəsuod ni	letal	In the informal sector	In the formal sector	splodəsuod ni	latoT	In the informal sector	In the formal sector	splodəsuod ni	Total	In the informal sector	In the formal sector	splodəsuod ni
Latvia	100	11.2	88.8	0.0	100	12.8	87.2	0.0	100	9.8	90.2	0.0	100	9.1	90.9	0.0	100	9.6	90.4	0.0	100	8.7	91.3	0.0
Lithuania	100	8.2	91.8	0.0	100	6.7	93.3	0.0	100	9.7	90.4	0.0	100	5.6	94.4	0.0	100	4.0	96.0	0.0	100	7.1	92.9	0.0
Norway	100	7.1	92.9	0.0	100	8.8	91.2	0.0	100	5.3	94.7	0.0	100	6.7	93.3	0.0	100	8.1	91.9	0.0	100	5.1	94.9	0.0
Sweden	100	2.6	97.4	0.0	100	3.0	97.1	0.0	100	2.2	97.8	0.0	100	2.6	97.5	0.0	100	3.0	97.0	0.0	100	2.1	97.9	0.0
United Kingdom	100	26.0	74.1	0.0	100	23.7	76.3	0.0	100	28.5	71.5	0.0	100	25.7	74.3	0.0	100	23.3	76.7	0.0	100	28.3	71.7	0.0
Southern Europe																								
Albania	100	30.8	35.5	33.7	100	33.2	36.8	30.0	100	27.8	33.9	38.2	100	28.0	60.8	11.3	100	32.3	54.8	12.9	100	21.0	70.4	8.6
Bosnia and Herzegovina	100	45.0	55.0	0.0	100	43.5	56.5	0.0	100	47.5	52.4	0.1	100	31.0	69.0	0.0	100	32.3	67.7	0.0	100	28.5	71.5	0.1
Croatia	100	12.8	87.2	0.0	100	13.8	86.2	0.0	100	11.6	88.4	0.0	100	10.6	89.4	0.0	100	11.2	88.8	0.0	100	9.8	90.2	0.0
Greece	100	32.2	67.8	0.0	100	33.8	66.2	0.0	100	30.0	70.0	0.0	100	25.0	75.1	0.0	100	27.6	72.5	0.0	100	21.1	78.9	0.0
Italy	100	16.0	84.1	0.0	100	18.2	81.8	0.0	100	12.8	87.2	0.0	100	15.2	84.8	0.0	100	17.2	82.8	0.0	100	12.3	87.7	0.0
Malta	100	7.9	92.1	0.0	100	8.9	91.2	0.0	100	6.2	93.9	0.0	100	7.5	92.5	0.0	100	8.2	91.8	0.0	100	6.1	93.9	0.0
Portugal	100	11.9	88.2	0.0	100	13.2	86.8	0.0	100	10.4	89.6	0.0	100	10.2	89.8	0.0	100	11.5	88.5	0.0	100	8.8	91.2	0.0
Serbia	100	6.4	88.6	5.0	100	8.2	87.5	4.4	100	4.2	90.1	5.7	100	3.6	90.3	6.1	100	5.1	89.5	5.5	100	1.9	91.3	6.8
Slovenia	100	4.8	95.2	0.0	100	6.0	94.0	0.0	100	3.3	96.8	0.0	100	3.9	96.1	0.0	100	4.8	95.2	0.0	100	2.9	97.1	0.0
Spain	100	17.3	82.7	0.0	100	17.7	82.3	0.0	100	16.9	83.1	0.0	100	16.4	83.6	0.0	100	16.5	83.5	0.0	100	16.3	83.7	0.0
Western Europe																								
Austria	100	8.8	91.2	0.0	100	8.4	91.6	0.0	100	9.3	90.7	0.0	100	7.7	92.3	0.0	100	7.4	92.6	0.0	100	8.1	91.9	0.0
Belgium	100	10.8	89.2	0.0	100	12.7	87.3	0.0	100	8.8	91.3	0.0	100	10.1	89.9	0.0	100	11.7	88.3	0.0	100	8.3	91.7	0.0
France	100	6.1	93.9	0.0	100	7.1	92.9	0.0	100	5.1	94.9	0.0	100	5.2	94.8	0.0	100	5.9	94.1	0.0	100	4.5	95.5	0.0
Germany	100	28.0	72.1	0.0	100	26.9	73.1	0.0	100	29.0	71.0	0.0	100	27.9	72.1	0.0	100	26.9	73.1	0.0	100	29.0	71.0	0.0
Luxembourg	100	0.9	99.1	0.0	100	0.6	99.4	0.0	100	1.3	98.7	0.0	100	0.9	99.1	0.0	100	0.6	99.4	0.0	100	1.2	98.8	0.0
Netherlands	100	9.1	90.9	0.0	100	9.6	90.4	0.0	100	8.4	91.6	0.0	100	8.9	91.1	0.0	100	9.4	90.6	0.0	100	8.3	91.7	0.0
Switzerland	100	4.2	95.8	0.0	100	3.9	96.1	0.0	100	4.6	95.4	0.0	100	3.9	96.1	0.0	100	3.6	96.4	0.0	100	4.2	95.8	0.0
			notional labour facos as aimilas ha			o la			1.1															

## Appendix C

## Statistical appendix: Global and regional estimates

## Micro determinants of informality and informality and working conditions

Appendix C presents global and regional estimates for some of the drivers of informality associated with personal features (age and level of education) as well as indicators related to working conditions in the informal economy (working time and permanent or temporary nature of employment). Appendix C complements statistical summaries presented in Chapter 2 by adding the gender dimension to the analysis of profiles of informality (or formality) by age and level of education and Chapter 3 on thematic issues.

	Micro level determinants of informality
Table C.1	Share of informal employment by level of education, status in employment, sex and region Indicators presented in table C.1 assess whether the level of education is an important driver of informality by comparing the share of informal employment according to increasing levels of education for different groups of workers (women and men in total employment or as employees, employers and own-account workers).
Table C.2	<b>Distribution of informal employment and formal employment by level of education, status in employment, sex and region</b> Table C.2 should be analysed jointly with table C.1. The distribution of informal employment and formal employment by level of education is influenced by the overall educational profile of the employed population in the country. This is through the comparison of the respective educational profiles of workers in informal and in formal employment that conclusions might be drawn on the over-representation of low or high educated workers. Table C.2 presents the distribution of informal employ- ment and formal employment by level of education for different groups of workers: total (informal and formal) employment in table C.2.1; employees in table C.2.2; employers in table C.2.3; own-account workers in table C.2.4; and contributing family workers (all being in informal employment) in table C.2.5.
Figure C.1	Share of informal employment by level of education, status in employment and region Figure C.1 complements section 3.3 in chapter 3 by adding a regional perspective. It corresponds to indicators presented in table C.1.
Table C.3	Share of informal employment by age group, status in employment, sex and region Table C.3 presents the share of informal employment along the life course. The overall share of informal employment and the shape over the life course is determined to a large extent by the labour market structure in terms of employment. Indicators presented in table C.3 cover the gender dimension across main statuses in employment.
Informality and w	orking conditions
Table C.4	Incidence of marginal and part-time employment among workers in informal employment compared to workers in formal employment

Workers in informal employment are not covered – either legally or in practice – by formal arrangements, including protections regarding normal working hours. The objective of this first indicator is to assess if workers in informal employment are more likely to work very short hours for pay or profit (defined here as less than 15 hours and less than 20 hours a week) with the underlying risk of working poverty; or more generally if they are more likely to be in part-time employment (defined as less than 35 hours a week).

	Incidence of excessive hours of work among workers in informal employment compared to workers in formal employment Two indicators related to the incidence of excessive hours are presented in table C.4: the incidence of excessive hours defined as a) more than 48 working hours a week; and b) more than 60 working hours a week among workers in informal employment compared to workers in formal employment. Table C.4.1 considers total employment and table C.4.2 focuses on employees, comparing the respective situation of work- ers in informal employment to the situation of workers in formal employment.
Figure C.2	<b>Proportion of workers working more than 60 hours a week depending on the formal or informal nature of main job</b> This figure includes global and regional estimates for the world, developing and emerging countries, and developed coun- tries for total employment and by status in employment to complement section 3.4 in chapter 3. Figure C.2 refers to total formal and informal employment, including agriculture.

Table C.1 – Share of informal employment by level of education, status in employment, sex and region

			Men	_			Wor	Women			Gender gap	Gender gap (women-men)	
		High	Highest level of (	rel of education (%)		Ξ	ghest level of	Highest level of education (%)		Highe	est level of e	Highest level of education (% points)	nts)
		None	Primary	Secondary	Tertiary	None	Primary	Secondary	Tertiary	None	Primary	Secondary	Tertiary
C.1.1 Tot	Total employment												
	Africa	90.9	85.5	67.0	30.1	97.0	92.4	70.3	21.6	6.1	6.9	3.3	-8.5
eintr	Americas	76.5	70.3	50.0	32.5	92.3	79.2	53.4	35.0	15.8	9.0	3.5	2.6
coni	Arab States	87.4	76.4	56.1	27.4	94.9	83.6	28.3	10.9	7.5	7.3	-27.8	-16.5
	Asia and the Pacific	94.3	89.8	64.5	39.6	96.0	89.8	57.5	24.6	1.7	-0.0	-7.1	-15.0
mer	Europe and Central Asia	70.1	41.2	37.2	23.0	92.3	59.3	34.8	22.8	22.3	18.1	-2.4	-0.2
<b>e</b> Total	al	92.5	85.1	61.1	35.5	96.1	88.0	55.5	27.0	3.6	2.9	-5.6	-8.5
:	Americas	44.2	44.6	22.8	19.2	54.1	45.1	20.2	17.9	9.9	0.5	-2.6	-1.2
loper tries	Asia and the Pacific	87.1	70.8	24.0	12.8	94.2	76.7	23.5	17.3	7.1	6.0	-0.4	4.4
unoc	Europe and Central Asia	29.3	29.3	18.0	12.1	28.1	28.9	14.9	11.3	-1.2	-0.4	-3.2	-0.8
	al	44.1	38.9	20.2	16.4	63.2	42.7	18.0	15.8	19.1	3.8	-2.2	-0.5
Afr	Africa	90.9	85.5	67.0	30.1	97.0	92.4	70.3	21.6	6.1	6.9	3.3	-8.5
Am	Americas	74.9	68.5	45.9	22.8	90.6	76.4	48.2	22.6	15.7	7.9	2.3	-0.1
רום Ara	Arab States	87.4	76.4	56.1	27.4	94.9	83.6	28.3	10.9	7.5	7.3	-27.8	-16.5
	Asia and the Pacific	94.3	89.7	61.8	34.6	96.0	89.6	53.9	23.6	1.7	-0.1	-7.8	-11.0
Eui	Europe and Central Asia	63.0	37.8	24.6	15.5	87.5	47.4	21.6	15.0	24.5	9.6	-3.0	-0.5
Total	al	92.3	83.9	54.4	26.1	95.9	86.2	47.4	20.9	3.6	2.3	-7.0	-5.2

			×	Men			Wo	Women			Gender gap	Gender gap (women-men)	
		Ŧ	Highest level o	rel of education (%)		÷	ghest level o	Highest level of education (%)		Highe	est level of ec	Highest level of education (% points)	ints)
		None	Primary	Secondary	Tertiary	None	Primary	Secondary	Tertiary	None	Primary	Secondary	Tertiary
C.1.2	Employees												
2	Africa	76.7	63.8	42.5	18.3	86.7	6.99	37.9	13.4	10.0	3.0	-4.6	-4.9
	Americas	77.5	54.5	33.5	25.4	84.5	62.0	38.1	27.6	6.9	7.5	4.6	2.2
s gni nuoc	Arab States	80.6	67.2	42.4	16.2	71.1	59.5	20.7	9.4	-9.5	-7.7	-21.7	-6.8
	Asia and the Pacific	89.4	87.4	57.7	22.2	91.5	84.8	49.6	19.0	2.1	-2.6	-8.0	-3.2
	Europe and Central Asia	56.8	34.6	31.8	20.2	74.4	37.9	27.6	17.6	17.6	3.3	-4.2	-2.6
Ð	Total	84.5	76.5	52.2	22.3	89.5	76.7	45.5	20.6	5.0	0.2	-6.7	-1.7
	Americas	22.2	26.4	12.0	12.9	31.5	26.3	11.5	14.2	9.3	-0.1	-0.5	1.3
lopei tries	Asia and the Pacific	58.7	67.4	10.6	6.8	94.4	71.2	11.1	9.6	35.8	3.8	0.5	2.8
	Europe and Central Asia	18.8	11.2	8.0	5.1	19.5	14.6	8.3	6.9	0.7	3.4	0.3	1.8
	Total	22.9	19.6	9.1	10.0	42.3	25.4	9.4	11.6	19.4	5.8	0.3	1.6
	Africa	76.7	63.8	42.5	18.3	86.7	6.99	37.9	13.4	10.0	3.0	-4.6	-4.9
	Americas	73.3	51.7	29.4	15.8	79.2	57.4	32.9	17.4	5.9	5.7	3.5	1.7
rld	Arab States	80.6	67.2	42.4	16.2	71.1	59.5	20.7	9.4	-9.5	-7.7	-21.7	-6.8
οW	Asia and the Pacific	89.4	87.3	54.3	19.0	91.5	84.5	45.3	17.7	2.2	-2.8	-9.0	-1.3
	Europe and Central Asia	50.8	27.7	16.2	9.8	63.3	27.0	14.6	10.3	12.5	-0.8	-1.5	0.5
	Total	83.8	74.0	44.1	15.7	88.5	72.6	36.5	15.6	4.7	-1.3	-7.6	-0.2

Highest level of education (%)     Highest level of education (%)     Highest level of education (%)     None     Primary     Secondary     Tertiary     None     Prinary     Secondary     Teri				Σ	Men			Wo	Women			Gender gap	Gender gap (women-men)	
Number Introduction     None     Frimary Entany     None     Frimary Entany     Recondary Intro     None     Frimary Entany     Recondary Intro     None     Primary Entany     Recondary Entany     None     Primary Entany     Recondary Entany     None     Primary Entany     None			Ŧ	ighest level o	f education (%		Ξ	ghest level o	f education (%	()	High	est level of ec	Highest level of education (% points)	ints)
Implores     Employes     Endologed			None	Primary	Secondary	Tertiary	None	Primary	Secondary	Tertiary	None	Primary	Secondary	Tertiary
Africa     94.4     80.5     77.4     52.6     95.6     86.0     61.7     37.4       Americas     71.5     69.5     42.6     20.2     82.8     69.9     39.7     21.7       Americas     71.5     69.5     42.6     20.2     82.8     69.9     39.7     21.7       Arab States     83.5     79.8     68.8     64.1     97.4     78.7     60.7     51.2       Arab States     83.5     79.8     68.8     64.1     97.4     78.7     60.7     51.2       Anot be and the Pacific     90.5     70.4     52.3     32.8     93.2     74.1     30.1       Memicas     96.1     90.4     72.6     62.9     93.3     69.5     59.2     69.5     59.2       Asia and the Pacific     100.0     61.6     53.6     54.0     51.9     24.6       Mericas     101.0     61.6     53.6     53.2     54.0     53.2       Intal     Mericas     101.0     51.4 <t< th=""><th>C.1.3</th><th>Employers</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th></t<>	C.1.3	Employers												
Americas     71.5     69.5     4.2.6     20.2     82.8     6.9.9     39.7     21.7       Arab States     83.5     79.8     6.8.8     6.4.7     97.4     78.7     6.0.7     51.2       Arab States     83.5     79.8     6.8.8     6.4.7     97.4     78.7     6.0.7     51.2       Furnope and Central Asia     41.6     21.6     36.6     20.9     38.2     40.3     44.1     30.1       Total     90.5     70.4     52.3     32.8     93.2     75.5     51.9     24.6       Americas     96.1     90.4     72.6     62.9     93.3     69.5     59.5     59.2       Americas     96.1     90.4     72.6     62.9     93.3     69.5     59.2     59.2       Americas     96.1     90.4     72.6     51.9     24.6     40.3       Intal     100.0     51.6     36.9     24.4     97.8     19.1     42.2     46.4       Intal     104.1     50.6 <th>S</th> <th>Africa</th> <th>94.4</th> <th>80.5</th> <th>77.4</th> <th>52.6</th> <th>92.6</th> <th>86.0</th> <th>61.7</th> <th>37.4</th> <th>1.2</th> <th>5.6</th> <th>-15.7</th> <th>-15.2</th>	S	Africa	94.4	80.5	77.4	52.6	92.6	86.0	61.7	37.4	1.2	5.6	-15.7	-15.2
Mark States     83.5     79.8     68.8     64.7     97.4     78.7     60.7     51.2       Rain and the Pacific     90.8     70.7     52.4     40.4     91.2     73.5     54.0     25.1       Europe and Central Asia     41.6     21.6     36.6     20.9     38.2     40.3     44.1     30.1       Total     90.5     70.4     52.3     32.8     93.2     75.5     51.9     24.6       Americas     96.1     90.4     72.6     62.9     93.3     69.5     59.2       Americas     96.1     90.4     72.6     62.9     93.3     64.1     30.1       Total     90.6     51.9     37.5      76.0     53.6     43.3       Atrica     91.4     80.5     71.4     52.6     95.6     64.7     37.4       Mericas     71.9     71.4     52.6     95.6     65.7     47.3     47.3       Matricas     71.9     72.6     95.6     83.1     70.8		Americas	71.5	69.5	42.6	20.2	82.8	6.69	39.7	21.7	11.3	0.4	-3.0	1.5
Asia and the Pacific     90.8     70.7     52.4     40.4     91.2     73.5     54.0     25.1       Europe and Central Asia     41.6     21.6     36.6     20.9     38.2     40.3     44.1     30.1       Total     90.5     70.4     52.3     32.8     93.2     75.5     51.9     24.6       Americas     96.1     90.4     72.6     62.9     93.2     75.5     51.9     24.6       Americas     96.1     90.4     72.6     62.9     93.2     75.5     51.9     24.6       Luope and Central Asia     100.0     51.6     37.5      76.0     53.6     42.3       Total     97.6     63.1     44.9     31.1     96.2     66.1     47.4     43.4       Africa     94.4     80.5     77.4     52.6     95.6     86.0     61.7     37.4       Africa     94.4     90.5     77.4     52.6     95.6     86.0     67.7     57.4       Africa     7		Arab States	83.5	79.8	68.8	64.7	97.4	78.7	60.7	51.2	13.9	-1.1	-8.1	-13.5
Europe and Central Asia     41.6     21.6     36.6     20.9     38.2     40.3     44.1     30.1       Total     90.5     70.4     52.3     32.8     93.2     75.5     51.9     24.6       Americas     90.5     70.4     52.3     32.8     93.2     75.5     51.9     24.6       Americas     90.6     70.4     72.6     62.9     93.3     69.5     59.2     59.2       Americas     100.0     51.6     36.9     24.4     97.8     19.1     42.2     46.4       Intel     97.6     63.1     44.9     31.1     96.2     66.1     47.4     43.4       Africa     71.9     70.2     43.1     70.6     53.6     53.6     23.6       Mericas     71.9     70.2     43.4     70.6     53.6     60.7     53.6     53.6       Mericas     71.9     72.6     83.1     70.8     83.1     70.8     70.9     73.4       Markicas     71.9     71		Asia and the Pacific	90.8	70.7	52.4	40.4	91.2	73.5	54.0	25.1	0.4	2.8	1.5	-15.3
Total     90.5     70.4     52.3     32.8     93.2     75.5     51.9     24.6       Americas     96.1     90.4     72.6     62.9     93.9     92.3     69.5     59.2       Americas     96.1     90.4     72.6     62.9     93.9     92.3     69.5     59.2       Americas     100.0     51.6     36.9     24.4     97.8     19.1     42.2     46.4       Total     100.0     60.9     51.9     37.5      76.0     53.6     42.3       Total     97.6     63.1     44.9     31.1     96.2     66.1     47.4     43.4       Africa     94.4     80.5     77.4     52.6     95.6     86.0     61.7     37.4       Americas     71.9     70.2     43.7     21.6     83.1     70.8     40.9     22.4       Arab States     83.5     79.4     97.4     97.4     77.4     52.6     51.9     51.4       Arab States     70.4		Europe and Central Asia	41.6	21.6	36.6	20.9	38.2	40.3	44.1	30.1	-3.4	18.8	7.5	9.2
Americas     96.1     90.4     72.6     62.9     93.9     92.3     69.5     59.2       Avia and the Pacific     100.0     51.6     36.9     24.4     97.8     19.1     42.2     46.4       Europe and Central Asia     100.0     60.9     51.9     37.5      76.0     53.6     42.3       Ital     97.6     63.1     44.9     31.1     96.2     66.1     47.4     43.4       Africa     94.4     80.5     77.4     52.6     95.6     66.1     47.4     37.4       Americas     71.9     70.2     43.7     21.6     83.1     70.8     40.9     22.4       Arab States     83.5     79.8     68.8     64.7     97.4     73.4       Arab States     83.5     79.8     68.7     97.4     78.7     60.7     51.2       Arab States     83.5     79.8     68.7     97.4     78.7     60.7     51.2       Asia and the Pacific     90.8     70.4     50.1	Ð	Total	90.5	70.4	52.3	32.8	93.2	75.5	51.9	24.6	2.7	5.0	-0.4	-8.2
Asia and the Pacific     100.0     51.6     36.9     24.4     97.8     19.1     42.2     46.4       Europe and Central Asia     100.0     60.9     51.9     37.5      76.0     53.6     42.3       Iotal     97.6     63.1     44.9     31.1     96.2     66.1     47.4     43.4       Africa     94.4     80.5     77.4     52.6     95.6     66.1     47.4     43.4       Africa     71.9     70.2     43.7     21.6     83.1     70.8     40.9     22.4       Arab States     83.5     79.8     68.8     64.7     97.4     79.9     51.2       Asia and the Pacific     90.8     70.4     50.1     34.1     91.3     72.5     51.9     30.8       Luope and Central Asia     48.7     37.0     48.2     32.5     51.4     30.6     51.6     51.6     51.9     51.6		Americas	96.1	90.4	72.6	62.9	93.9	92.3	69.5	59.2	-2.2	1.9	-3.1	-3.8
Europe and Central Asia     100.0     60.9     51.9     37.5      76.0     53.6     42.3       Total     97.6     63.1     44.9     31.1     96.2     66.1     47.4     43.4       Africa     94.4     80.5     77.4     52.6     95.6     86.0     61.7     37.4       Americas     71.9     70.2     43.7     21.6     83.1     70.8     40.9     22.4       Americas     71.9     70.2     43.7     21.6     83.1     70.8     40.9     22.4       Arab States     83.5     79.8     68.8     64.7     97.4     73.2     51.2     30.8       Arab States     83.5     79.8     68.8     64.7     97.4     73.2     51.9     30.8       Arab States     83.5     70.4     50.1     34.1     91.3     72.5     51.9     30.8       Arab States     48.7     37.0     48.2     32.5     51.4     30.6     51.4     51.2       Arab States<		Asia and the Pacific	100.0	51.6	36.9	24.4	97.8	19.1	42.2	46.4	-2.2	-32.5	5.3	22.0
Total     97.6     63.1     44.9     31.1     96.2     66.1     47.4     43.4       Africa     94.4     80.5     77.4     52.6     95.6     86.0     61.7     37.4       Americas     71.9     70.2     43.7     21.6     83.1     70.8     40.9     22.4       Americas     71.9     70.2     43.7     21.6     83.1     70.8     40.9     22.4       Arab States     83.5     79.8     68.8     64.7     97.4     78.7     60.7     51.2       Asia and the Pacific     90.8     70.4     50.1     34.1     91.3     72.5     51.9     30.8       Europe and Central Asia     48.7     37.0     48.2     32.5     32.5     51.4     39.6		Europe and Central Asia	100.0	60.9	51.9	37.5	:	76.0	53.6	42.3	:	15.1	1.7	4.8
Africa 94.4 80.5 77.4 52.6 95.6 86.0 61.7 37.4   Americas 71.9 70.2 43.7 21.6 83.1 70.8 40.9 22.4   Americas 71.9 70.2 43.7 21.6 83.1 70.8 40.9 22.4   Arab States 83.5 79.8 68.8 64.7 97.4 78.7 60.7 51.2   Asia and the Pacific 90.8 70.4 50.1 34.1 91.3 72.5 51.9 30.8   Europe and Central Asia 48.7 37.0 48.2 32.5 38.2 67.2 51.4 39.6   Total 90.5 70.2 50.8 75.0 50.9 50.9 50.9		Total	97.6	63.1	44.9	31.1	96.2	66.1	47.4	43.4	-1.4	3.0	2.5	12.2
Americas 71.9 70.2 43.7 21.6 83.1 70.8 40.9 22.4   Arab States 83.5 79.8 68.8 64.7 97.4 78.7 60.7 51.2   Arab States 83.5 79.8 68.8 64.7 97.4 78.7 60.7 51.2   Arab States 83.5 79.8 50.1 34.1 91.3 72.5 51.9 30.8   Asia and the Pacific 90.8 70.4 50.1 34.1 91.3 72.5 51.9 30.8   Europe and Central Asia 48.7 37.0 48.2 32.5 38.2 67.2 51.4 39.6   Total 90.5 70.2 50.8 32.5 93.2 50.9 50.9 50.9		Africa	94.4	80.5	77.4	52.6	95.6	86.0	61.7	37.4	1.2	5.6	-15.7	-15.2
Arab States     83.5     79.8     68.8     64.7     97.4     78.7     60.7     51.2       Asia and the Pacific     90.8     70.4     50.1     34.1     91.3     72.5     51.9     30.8       Europe and Central Asia     48.7     37.0     48.2     32.5     38.2     67.2     51.4     39.6     .       Total     91.5     70.5     51.8     30.8     30.8     .     .     30.6     .     .		Americas	71.9	70.2	43.7	21.6	83.1	70.8	40.9	22.4	11.2	0.6	-2.9	0.9
Asia and the Pacific     90.8     70.4     50.1     34.1     91.3     72.5     51.9     30.8       Europe and Central Asia     48.7     37.0     48.2     32.5     38.2     67.2     51.4     39.6     -       Total     50.5     50.8     32.5     58.2     67.2     51.4     39.6     -	rld	Arab States	83.5	79.8	68.8	64.7	97.4	78.7	60.7	51.2	13.9	-1.1	-8.1	-13.5
48.7 37.0 48.2 32.5 38.2 67.2 51.4 39.6 . 90.5 70.2 50.8 32.2 93.2 75.0 50.9 30.9	οW	Asia and the Pacific	90.8	70.4	50.1	34.1	91.3	72.5	51.9	30.8	0.5	2.1	1.8	-3.3
QUE 70.2 E0.8 32.2 G3.2 75.0 E0.G 30.0		Europe and Central Asia	48.7	37.0	48.2	32.5	38.2	67.2	51.4	39.6	-10.5	30.2	3.1	7.1
		Total	90.5	70.2	50.8	32.2	93.2	75.0	50.9	30.9	2.7	4.8	0.2	-1.3
			Z	Men			Moi	Women			Gender gap (	Gender gap (women-men)		
-----------------	-------------------------	------	----------------	--------------------------------	----------	------	-----------------	--------------------------------	----------	------	-----------------	---------------------------------------	----------	
		-	ighest level o	Highest level of education (%)		Ť	ighest level of	Highest level of education (%)	()	High	est level of ed	Highest level of education (% points)	ints)	
		None	Primary	Secondary	Tertiary	None	Primary	Secondary	Tertiary	None	Primary	Secondary	Tertiary	
C.1.4	Own-account workers													
S	Africa	94.9	95.2	93.6	83.3	96.7	95.8	92.5	80.0	1.8	0.6	-1.1	-3.3	
bnd eintrie:	Americas	74.4	86.6	84.3	60.8	94.2	93.0	87.0	67.7	19.9	6.4	2.7	6.9	
	Arab States	96.8	91.9	93.0	81.4	99.5	96.1	92.2	42.2	2.7	4.2	-0.9	-39.3	
qolə grig	Asia and the Pacific	95.0	90.9	81.9	88.4	94.3	87.9	75.1	76.7	-0.6	-2.9	-6.7	-11.7	
	Europe and Central Asia	82.7	59.9	76.4	46.3	96.4	87.1	89.0	70.6	13.7	27.2	12.7	24.3	
Ð	Total	94.2	90.5	83.1	80.5	95.1	90.5	79.4	71.4	0.9	0.0	-3.7	-9.1	
	Americas	99.9	9.66	90.6	66.4	99.8	99.8	89.0	68.9	-0.1	0.2	-1.6	2.5	
opec tries	Asia and the Pacific	99.3	80.4	69.2	48.8	90.6	80.9	70.4	65.4	-8.7	0.5	1.2	16.6	
	Europe and Central Asia	83.1	82.5	67.1	55.6	82.3	84.2	68.8	56.2	-0.8	1.7	1.7	0.6	
	Total	97.6	86.1	70.1	62.0	94.9	88.2	72.0	64.5	-2.7	2.1	1.9	2.6	
	Africa	94.9	95.2	93.6	83.3	96.7	95.8	92.5	80.0	1.8	0.6	-1.1	-3.3	
	Americas	75.1	87.1	84.7	64.3	94.4	93.4	87.1	68.4	19.3	6.2	2.4	4.0	
rld	Arab States	96.8	91.9	93.0	81.4	99.5	96.1	92.2	42.2	2.7	4.2	-0.9	-39.3	
oW	Asia and the Pacific	95.0	90.8	81.3	85.5	94.3	87.9	74.9	74.7	-0.7	-2.9	-6.4	-10.7	
	Europe and Central Asia	82.7	66.0	69.3	54.0	95.8	85.8	74.0	57.0	13.0	19.8	4.6	3.1	
	Total	94.2	90.5	81.8	73.5	95.1	90.5	78.6	67.8	0.9	0.0	-3.2	-5.7	

Table C.2 – Distribution of informal employment and formal employment by level of education, status in employment, sex and region

Table C.2.1 Distribution of workers by level of education | Total informal and total formal employment

				Total					Men					Women		
		м %	% workers with a		given level of education	tion	0M %	orkers with	% workers with a given level of education	el of educa	ition	ж %	orkers with	% workers with a given level of education	el of educa	tion
		None	Primary	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other
Percentage	Percentage of people with a given level of education by sex I Informal employment	sex I Informa	il employm	ent												
S	Africa	44.5	32.9	17.4	1.9	3.3	40.6	33.5	21.0	2.5	2.4	48.8	32.3	13.2	1.3	4.4
	Americas	10.6	25.9	45.6	14.0	3.9	10.4	28.0	45.8	11.9	3.9	10.7	23.0	45.5	16.8	4
s gni Iuoc	Arab States	38.8	37.5	20.5	3.1	0.0	35.7	40.2	20.9	3.2	0.0	63.0	19.0	15.2	2.8	0
	Asia and the Pacific	23.1	27.7	42.5	4.3	2.5	21.7	29.2	41.2	4.8	3.2	26.3	24.8	44.4	3.3	1
	Europe and Central Asia	4.0	19.7	57.5	18.3	0.6	2.4	22.0	59.3	15.8	0.5	6.0	17.0	55.2	21.1	0.6
Ð	Total	24.3	28.1	39.8	5.2	2.6	22.5	29.5	39.6	5.3	3.0	27.6	25.5	40.1	4.9	1.9
	Americas	1.3	5.1	13.6	80.0	0.0	1.4	5.4	15.0	78.1	0.0	1.3	4.6	11.9	82.2	0.0
	Asia and the Pacific	1.3	8.6	71.4	18.5	0.2	0.5	7.1	70.3	21.9	0.2	2.3	10.5	72.4	14.6	0.2
unoc  əxə(	Europe and Central Asia	0.2	7.0	58.6	25.4	8.8	0.2	7.1	60.5	22.0	10.3	0.2	6.8	55.9	30.5	9
	Total	0.9	6.7	45.0	44.0	3.6	0.7	6.5	46.0	42.5	4.3	1.1	6.9	43.6	45.9	2.6
	Africa	44.5	32.9	17.4	1.9	3.3	40.6	33.5	21.0	2.5	2.4	48.8	32.3	13.2	1.3	4.4
	Americas	8.9	22.0	39.7	26.2	3.2	8.8	24.0	40.3	23.8	3.2	8.9	19.4	39.0	29.6	3.2
hd	Arab States	38.8	37.5	20.5	3.1	0.0	35.7	40.2	20.9	3.2	0.0	63.0	19.0	15.2	2.8	0.0
oW	Asia and the Pacific	22.7	27.4	42.9	4.5	2.5	21.4	28.9	41.6	5.0	3.1	25.8	24.5	45.0	3.6	-i
	Europe and Central Asia	2.3	14.0	58.0	21.5	4.2	1.4	15.2	59.9	18.6	5.0	3.5	12.6	55.5	25.1	З.
	Total	23.1	27.0	40.1	7.2	2.7	21.5	28.5	39.9	7.1	3.1	26.0	24.4	40.3	7.4	2.0

				Total					Men					Women		
		м %	% workers with a	a given leve	given level of education	tion	0M %	orkers with	% workers with a given level of education	el of educa	tion	w %	orkers with	% workers with a given level of education	el of educa	tion
		None	Primary	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other
Percentage	Percentage of people with a given level of education by sex I Formal employment	sex I Formal	employmen	Ħ												
S	Africa	14.3	20.8	40.0	24.4	0.6	16.0	21.6	40.1	21.7	9.0	11.1	18.6	39.4	30.4	0.5
and artie	Americas	2.7	11.1	50.3	31.8	4.2	3.7	13.6	51.6	27.5	3.6	1.2	7.5	48.3	38.1	4.9
s gnii Iuoc	Arab States	10.8	25.4	41.1	22.8	0.1	11.9	30.2	38.8	19.1	0.0	3.4	6.1	55.3	35.1	0.1
	Asia and the Pacific	3.1	7.4	63.8	20.8	5.0	3.6	8.6	61.4	19.6	6.8	2.3	5.5	67.2	22.5	2.5
	Europe and Central Asia	0.4	12.3	54.0	33.1	0.2	9.0	17.1	53.6	28.6	0.1	0.3	6.3	54.6	38.7	0.2
Ð	Total	3.4	9.4	59.0	24.1	4.1	4.2	11.4	57.3	22.0	5.1	2.3	6.5	61.4	27.3	2.5
	Americas	0.3	1.5	12.0	86.1	0.0	0.4	1.7	13.1	84.7	0.0	0.2	1.3	10.7	87.7	0.0
	Asia and the Pacific	0.0	0.9	66.1	32.6	0.4	0.0	0.8	59.3	39.6	0.3	0.0	1.0	75.7	22.6	0.6
unoc Iəvə(	Europe and Central Asia	0.1	3.4	59.0	35.8	1.7	0.1	3.8	60.8	33.4	1.9	0.1	3.0	56.9	38.6	1.5
	Total	0.2	2.3	44.4	52.2	0.9	0.2	2.5	44.6	51.7	0.9	0.1	2.1	44.2	52.8	0.8
	Africa	14.3	20.8	40.0	24.4	0.6	16.0	21.6	40.1	21.7	0.6	11.1	18.6	39.4	30.4	0.5
	Americas	1.5	6.2	30.5	59.8	2.0	2.1	7.8	32.9	55.3	1.9	0.6	4.1	27.4	65.7	2.2
rlq	Arab States	10.8	25.4	41.1	22.8	0.1	11.9	30.2	38.8	19.1	0.0	3.4	6.1	55.3	35.1	0.1
οW	Asia and the Pacific	2.7	9.9	64.1	22.3	4.5	3.2	7.6	61.1	22.2	6.0	2.0	5.0	68.2	22.5	2.3
	Europe and Central Asia	0.2	6.0	57.5	35.0	1.2	0.3	7.9	58.6	31.9	1.4	0.1	3.9	56.2	38.6	1.1
	Total	2.3	7.0	53.9	33.8	3.0	2.9	8.5	53.2	31.7	3.7	1.5	4.9	55.1	36.7	1.9

		ļ	(	1		

	☱
	-
	Ð
	Ξ
	=
	6
	=
	0
	Ē
	H
	9
	-
	ā
	Ε
	-
	õ
	e
	_
	-
	2
	=
	(0
	_
	3
	Ξ
	5
	₽
	-
	ŝ
	ã
	yee
	ະ
	1
	믈
	Ē
	-
	-
	_
	_
	_
	E
:	Ition
:	ation
:	ICation
:	ucation
:	ducation
:	education
:	r education
:	ot education
:	of education
:	education
	vel of education
:	evel of education
-	level of education 1
•	y level of education 1
-	by level of education 1
	by level of education
	's by level of education
	ers by level of education l
	ers by level of education 1
	Kers by level of education 1
	irkers by level of education 1
	vorkers by level of education 1
	workers by level of education 1
	t workers by level of education 1
	of workers by level of education
	i of workers by level of education
	in of workers by level of education
	on of workers by level of education 1
	tion of workers by level of education
	ution of workers by level of education
	bution of workers by level of education
	libution of workers by level of education 1
	tribution of workers by level of education
	stribution of workers by level of education
	listribution of workers by level of education
	Distribution of workers by level of education
	Distribution of workers by level of education
	Z Distribution of workers by level of education 1
	2.2 Distribution of workers by level of education
	C.Z.Z Distribution of workers by level of education 1

Table C.2	Table C.2.2 Distribution of workers by level of education   Employees	of education	I Employ		ormal and	l in formal	in informal and in formal employment	art								
				Total					Men					Women		
		м %	% workers with a giv	a given leve	ven level of education	tion	DM %	% workers with a given level of education	a given lev	el of educa	tion	0M %	% workers with a given level of education	a given leve	l of educat	ion
		None	Primary	Secondary	Tertiary	<b>Other</b>	None	Primary	Secondary	Tertiary	Other	None	Primary \$	Secondary	Tertiary	Other
Percentage	Percentage of employees with a given level of education by sex I Informal wage employment	ion by sex I Infor	rmal wage e	employment												
S	Africa	33.2	31.9	28.5	5.0	1.5	33.5	31.4	28.9	4.8	1.4	32.0	33.6	27.5	5.2	1.7
bné sintr	Americas	8.7	21.9	45.8	18.2	5.3	10.5	24.9	44.8	15.0	4.8	6.4	18.0	47.2	22.4	6.0
s gni Inoc	Arab States	34.1	40.6	22.1	3.2	0.0	33.8	42.4	21.1	2.7	0.0	30.2	21.2	39.6	8.9	0.1
	Asia and the Pacific	9.6	23.8	59.0	5.0	2.6	9.6	25.3	57.4	4.8	2.9	9.7	21.0	61.9	5.5	2.0
	Europe and Central Asia	1.8	15.7	61.9	20.4	0.2	1.5	19.4	61.7	17.3	0.2	2.3	10.7	62.3	24.6	0.2
Ð	Total	11.1	24.0	55.1	7.2	2.7	11.6	25.7	53.4	6.3	2.8	10.1	20.7	58.0	8.8	2.4
	Americas	0.8	3.6	10.4	85.1	0.0	0.9	4.4	12.2	82.5	0.0	0.7	2.9	8.7	87.7	0.0
lopeo L'ries	Asia and the Pacific	1.2	9.0	67.4	22.3	0.1	0.2	6.7	65.3	27.7	0.0	2.1	11.5	68.9	17.3	0.1
	Europe and Central Asia	0.1	4.7	53.9	25.3	15.9	0.1	4.5	54.8	19.7	20.8	0.2	4.9	52.9	31.3	10.8
	Total	9.0	5.0	36.1	52.8	5.5	0.5	4.9	36.6	50.9	7.1	0.8	5.3	35.5	54.8	3.7
	Africa	33.2	31.9	28.5	5.0	1.5	33.5	31.4	28.9	4.8	1.4	32.0	33.6	27.5	5.2	1.7
	Americas	6.8	17.6	37.4	34.1	4.1	8.4	20.4	37.7	29.6	3.8	4.9	14.0	37.0	39.7	4.4
rlq	Arab States	34.1	40.6	22.1	3.2	0.0	33.8	42.4	21.1	2.7	0.0	30.2	21.2	39.6	8.9	0.1
DW	Asia and the Pacific	9.5	23.5	59.2	5.3	2.5	9.5	25.0	57.5	5.1	2.9	9.5	20.7	62.1	5.8	1.9
	Europe and Central Asia	1.2	12.0	59.2	22.1	5.5	1.1	14.8	59.6	18.1	6.5	1.4	8.5	58.7	27.2	4.2
	Total	10.4	22.7	53.8	10.3	2.9	11.0	24.6	52.5	8.8	3.1	9.3	19.3	55.9	13.0	2.5

% workers with a given level of clucation         % w					Total					Men					Women		
Tertiany         Other         Note         Friany         Secondary         Tertiany         Other         Friandy         Secondary         Tertiany         Tertiany         Secondary         Tertiany         Secondary         Tertiany         Secondary         Tertiany         Secondary         Tertiany         Secondary         Secondary         Secondary         Secondary			M %	orkers with	a given lev	el of educa	tion	ом %	orkers with	a given lev	el of educa	ition	M %	orkers wit	ı a given lev	el of educ	atio
27.7         0.5         11.3         19.7         44.0         24.5         0.6         5.3         16.2         44.0           31.6         4.5         1.9         13.1         54.1         26.9         4.0         0.8         7.2         48.9           31.6         4.5         1.9         13.1         54.1         26.9         4.0         0.8         7.2         48.9           23.7         0.1         10.6         29.2         40.6         19.5         0.0         2.8         6.5         55.7         55.7         55.7         55.7         55.7         55.7         55.7         54.7         55.6         61.1         67.5         55.7         54.7         55.6         61.1         67.5         55.7         54.7         55.7         54.7         55.7         54.7         55.7         54.7         55.7         54.7         55.7         54.7         55.7         54.7         55.7         54.7         55.7         54.7         55.7         55.7         54.7         55.7         54.7         55.7         54.7         55.3         54.7         55.3         54.7         55.7         54.7         55.7         55.7         54.7         55.7         55.			None		Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other
Afficiation         9.2         18.5         4.1         2.7         0.5         11.3         19.7         4.0         2.45         0.6         5.3         16.2         440           Americas         1.4         10.6         51.9         31.6         4.5         19.3         54.1         26.9         4.0         28         6.2         55.7         48.9           Americas         1.3         5.1         6.43         2.24         6.9         1.7         56.7         0.0         28         6.2         55.7         48.9           Loope and the Pacific         1.3         5.1         6.3         5.2         58.7         0.1         10.6         5.3         5.2         58.7         0.1         0.6         5.9         5.1         5.4           Loope and the Pacific         1.4         10.9         5.4         5.1         24         5.7         5.7         5.8         5.1         5.9         5.4         5.3         5.4         5.4         5.4         5.4         5.4         5.4         5.4         5.4         5.4         5.5         5.4         5.7         5.4         5.4         5.4         5.4         5.4         5.4         5.4         5.4<	Percentage	of employees with a given level of education	by sex I Forr	nal wage er	nployment												
Americase         14         106         51.9         31.6         13.1         54.1         26.9         4.0         0.8         7.2         4.9         37.9           Americase         13         5.1         6.3         2.3         0.1         10.6         2.92         40.6         19.5         0.0         2.8         5.7         3.5.7         3.5.7           Momericase         0.4         10.3         5.1         6.3         2.3         0.1         0.6         2.8         6.0         7.2         4.9         3.7.9           Momericase         0.4         10.3         5.3         2.3         0.1         10.6         5.7	S	Africa	9.2	18.5	44.1	27.7	0.5	11.3	19.7	44.0	24.5	0.6	5.3	16.2	44.0	34.1	
And States         94         245         4.3         237         01         106         292         406         195         00         28         62         557           Europe and the Pacific         1.3         5.1         64.3         22.4         6.9         1.7         5.7         61.9         2.13         9.5         0.9         4.1         67.5           Europe and the Pacific         1.3         5.1         64.3         2.24         6.9         1.7         5.7         61.9         2.3         5.9         6.4         6.5           Iteration         0.4         1.9         54.9         5.1         2.4         6.9         1.7         5.7         61.9         2.3         5.9         5.4           Americas         0.4         1.0         5.4         1.7         0.1         2.4         6.9         1.0         6.5         6.1         6.5         6.1         6.5         6.1         6.5         6.1         6.5         6.1         6.5         6.1         6.5         6.1         6.5         6.1         6.5         6.5         6.5         6.5         6.5         6.5         6.1         6.5         6.1         6.5         6.1		Americas	1.4	10.6	51.9	31.6	4.5	1.9	13.1	54.1	26.9	4.0	0.8	7.2	48.9	37.9	
Asia and the Pacific         13         51         64.3         224         6.9         1.7         5.7         61.9         21.3         9.5         9.9         4.1         67.5           Intal         11         13         59         54.9         336         0.2         0.5         15.5         55.2         28.7         0.1         0.3         5.9         5.1         5.1           Intal         1         7.9         59         51.1         2.4         9.6         58.0         2.3         6.1         0.3         6.3         6.1           Mericas         0         1         1         2.4         9.6         58.0         53.0		Arab States	9.4	24.5	42.3	23.7	0.1	10.6	29.2	40.6	19.5	0.0	2.8	6.2	55.7	35.2	
Endpend Central Asia         04         109         549         336         02         155         552         287         01         03         59         547           Intal         13         79         593         533         559         51         24         96         10         03         59         61           Intal         1         1         1         2         4         96         580         234         66         10         55         61           Americas         04         16         122         858         00         04         18         135         842         00         85         51         13         13         13         13         13         13         13         13         13         13         13         13         13         14         13         13         14         13         14         16         10         16         13         13         13         14         13         13         13         13         13         13         13         13         13         13         13         13         13         13         13         14         13         13         14 <th></th> <td>Asia and the Pacific</td> <td>1.3</td> <td>5.1</td> <td>64.3</td> <td>22.4</td> <td>6.9</td> <td>1.7</td> <td>5.7</td> <td>61.9</td> <td>21.3</td> <td>9.5</td> <td>0.9</td> <td>4.1</td> <td>67.5</td> <td>24.2</td> <td></td>		Asia and the Pacific	1.3	5.1	64.3	22.4	6.9	1.7	5.7	61.9	21.3	9.5	0.9	4.1	67.5	24.2	
Total         1         2         5         5         5         5         6         1         5         6         1           Total         1         1         1         2         5         5         5         5         6         1           Americas         0.4         1.6         1.2         85.8         0.0         0.4         1.8         1.35         84.2         0.0         0.3         1.3 </td <th></th> <td>Europe and Central Asia</td> <td>0.4</td> <td>10.9</td> <td>54.9</td> <td>33.6</td> <td>0.2</td> <td>0.5</td> <td>15.5</td> <td>55.2</td> <td>28.7</td> <td>0.1</td> <td>0.3</td> <td>5.9</td> <td>54.7</td> <td>38.9</td> <td></td>		Europe and Central Asia	0.4	10.9	54.9	33.6	0.2	0.5	15.5	55.2	28.7	0.1	0.3	5.9	54.7	38.9	
Americas         0.4         1.6         1.2         8.5         0.0         0.4         1.8         1.3         8.42         0.0         0.3         1.3         10.3           Americas         0.0         0.6         66.9         32.1         0.4         0.0         0.5         59.6         39.6         0.2         0.0         0.8         76.2           Kai and the Pacific         0.0         0.6         66.9         32.1         0.4         0.0         0.5         59.6         39.6         0.2         0.0         0.8         76.2           Kai and the Pacific         0.1         3.4         59.4         35.4         1.7         0.1         3.8         61.2         32.9         1.9         0.1         3.0         57.3           Itality         0.2         2.3         44.6         52.0         0.9         0.2         2.4         71.7         20.7         20.9         1.1         20.9         71.3         20         71.3         20.9         21.3         20.9         21.3         20.9         20.9         20.7         20.9         20.7         20.9         20.9         20.7         20.9         20.7         20.9         20.7         20.9	Ð	Total	1.8	7.9	59.3	25.9	5.1	2.4	9.6	58.0	23.4	6.6	1.0	5.5	61.1	29.3	
Americas         0.4         1.6         1.2         8.5         0.0         0.4         1.8         1.3         8.42         0.0         0.3         1.3         10.9           Raia and the Pacific         0.0         0.6         6.6.9         32.1         0.4         0.0         0.5         59.6         39.6         0.0         0.8         76.2           Europe and Central Asia         0.1         3.4         59.4         35.4         1.7         0.1         3.8         61.2         3.9         0.9         76.2           Total         0.1         3.4         59.4         35.7         0.1         3.8         61.2         3.9         0.1         3.0         57.3           Italia         0.2         2.3         44.6         52.0         0.9         0.2         2.9         0.1         2.0         4.4           Italica         0.2         1.3         0.1         3.7         0.1         3.7         2.9         4.4         5.3         6.2         4.4         5.4         4.4         4.5         6.2         4.6         4.5         4.5         4.5         4.6         4.5         4.5         4.5         4.5         4.5         4.5																	
Asia and the Pacific         00         0.6         66.9         32.1         0.4         0.0         0.5         59.6         39.6         0.2         0.0         0.8         76.2           Europe and Central Asia         0.1         3.4         59.4         35.4         1.7         0.1         3.8         61.2         32.9         1.9         0.1         3.0         57.3           Total         0.2         2.3         44.6         52.0         0.9         0.2         25.7         44.7         51.7         0.9         0.1         3.0         57.3           Africa         0.2         18.5         44.1         27.7         0.5         11.3         19.7         44.0         24.5         44.5           Africa         0.2         18.5         44.1         27.7         0.5         11.3         19.7         44.0         24.5         3.8         16.2         34.0         14.0         24.5		Americas	0.4	1.6	12.2	85.8	0.0	0.4	1.8	13.5	84.2	0.0	0.3	1.3	10.9	87.6	
Europe and Central Asia         0.1         3.4         59.4         1.7         0.1         3.8         6.12         3.2.9         1.9         0.1         3.0         57.3           Idal         0.2         2.3         44.6         52.0         0.9         0.2         2.5         44.7         51.7         0.9         0.1         2.0         44.5           Africa         0.2         18.5         44.1         27.7         0.5         11.3         19.7         44.0         24.5         0.6         5.3         16.2         44.0           Americas         0.8         5.7         30.2         61.3         2.0         1.1         7.2         32.9         56.8         1.9         0.5         38         26.9           Americas         0.8         5.7         30.2         61.3         2.0         1.1         7.2         32.9         56.8         1.9         0.5         38         26.9           Americas         0.1         4.3         2.0         5.3         40.6         19.5         0.0         2.8         56.9         56.7           Arab States         0.1         4.9         5.8         4.0         0.1         2.8         56.4		Asia and the Pacific	0.0	9.0	6.99	32.1	0.4	0.0	0.5	59.6	39.6	0.2	0.0	0.8	76.2	22.5	
Indal         0.2         2.3         44.6         5.0         0.9         0.2         2.5         44.7         51.7         0.9         0.1         2.0         44.5           Africa         9.2         18.5         44.1         27.7         0.5         11.3         19.7         44.0         24.5         66.5         44.0         24.5         44.0         24.5         44.0         24.5         44.0         24.5         44.0         24.5         44.0         24.5         44.0         24.5         44.0         24.5         24.0         44.0         24.5         44.0         24.5         24.0         24.5         24.0         24.5         24.0         24.5         24.0         10.1         17.7         17.2         23.2         0.0         24.5         24.0         24.5		Europe and Central Asia	0.1	3.4	59.4	35.4	1.7	0.1	3.8	61.2	32.9	1.9	0.1	3.0	57.3	38.2	
Africa9.218.544.127.70.511.319.744.024.50.65.316.244.0Americas0.85.730.261.32.01.17.232.956.81.90.53.826.9Arab States9.424.542.323.70.110.629.240.619.50.02.86.255.7Arab States9.424.542.323.70.110.629.240.619.50.02.86.255.7Arab States9.424.542.323.70.110.629.240.619.50.02.86.255.7Arab States9.424.542.324.05.81.44.961.524.38.00.83.668.9Europe and Central Asia0.25.658.034.91.30.27.359.431.71.40.13.856.5Total1.25.753.636.13.51.66.953.13.94.50.77.359.4		Total	0.2	2.3	44.6	52.0	0.9	0.2	2.5	44.7	51.7	0.9	0.1	2.0	44.5	52.5	
Africa         92         18.5         44.1         27.7         0.5         11.3         19.7         44.0         24.5         0.6         5.3         16.2         44.0           Americas         0.8         5.7         30.2         61.3         2.0         1.1         7.2         32.9         56.8         1.9         0.5         3.8         26.9           Americas         0.8         5.7         30.2         61.3         2.0         1.1         7.2         32.9         56.8         1.9         0.5         3.8         26.9           Arab States         9.4         24.5         42.3         23.7         0.1         10.6         29.2         40.6         19.5         0.5         3.8         26.9           Arab States         1.1         4.3         64.7         24.0         5.8         1.4         4.9         61.5         24.3         8.0         0.8         3.6         68.9         56.7           Furope and Central Asia         0.2         5.8         1.4         4.9         61.5         24.3         3.6         3.6         67.5         3.6         56.5           Total         1.2         5.6         58.0         1.3																	
Americas         0.8         5.7         30.2         61.3         2.0         1.1         7.2         32.9         56.8         1.9         0.5         3.8         26.9           Arab States         9.4         24.5         42.3         23.7         0.1         10.6         29.2         40.6         19.5         0.0         2.8         6.2         55.7           Arab States         9.4         24.5         42.3         23.7         0.1         10.6         29.2         40.6         19.5         0.0         2.8         6.2         55.7           Asia and the Pacific         1.1         4.3         64.7         24.0         5.8         1.4         4.9         61.5         24.3         8.0         0.8         3.6         68.9           Europe and Central Asia         0.2         5.8         1.4         4.9         51.7         1.4         0.1         3.3         4.5         0.1         3.8         56.5           Total         1.2         5.7         53.6         36.1         3.5         1.6         53.1         3.3         4.5         0.7         4.1         54.5		Africa	9.2	18.5	44.1	27.7	0.5	11.3	19.7	44.0	24.5	9.0	5.3	16.2	44.0	34.1	
Arab States       9.4       24.5       42.3       23.7       0.1       10.6       29.2       40.6       19.5       0.0       2.8       6.2       55.7         Asia and the Pacific       1.1       4.3       64.7       24.0       5.8       1.4       4.9       61.5       24.3       8.0       0.8       6.9       68.9         Europe and Central Asia       0.2       5.6       58.0       34.9       1.3       0.2       7.3       59.4       31.7       1.4       0.1       3.8       56.5         Total       1.2       5.7       53.6       36.1       3.5       1.6       6.9       53.1       33.9       4.5       0.7       4.1       54.5		Americas	0.8	5.7	30.2	61.3	2.0	1.1	7.2	32.9	56.8	1.9	0.5	3.8	26.9	66.6	
Asia and the Pacific         1.1         4.3         64.7         24.0         5.8         1.4         4.9         61.5         24.3         8.0         0.8         3.6         68.9           Europe and Central Asia         0.2         5.6         58.0         34.9         1.3         0.2         7.3         59.4         31.7         1.4         0.1         3.8         56.5           Total         1.2         5.7         53.6         36.1         3.5         1.6         6.9         53.1         33.9         4.5         0.7         4.1         54.2	rld	Arab States	9.4	24.5	42.3	23.7	0.1	10.6	29.2	40.6	19.5	0.0	2.8	6.2	55.7	35.2	
0.2         5.6         58.0         34.9         1.3         0.2         7.3         59.4         31.7         1.4         0.1         3.8         56.5           1.2         5.7         53.6         36.1         3.5         1.6         6.9         53.1         33.9         4.5         0.7         4.1         54.2	oW	Asia and the Pacific	1.1	4.3	64.7	24.0	5.8	1.4	4.9	61.5	24.3	8.0	0.8	3.6	68.9	23.9	
1.2 5.7 53.6 36.1 3.5 1.6 6.9 53.1 33.9 4.5 0.7 4.1 54.2		Europe and Central Asia	0.2	5.6	58.0	34.9	1.3	0.2	7.3	59.4	31.7	1.4	0.1	3.8	56.5	38.4	
		Total	1.2	5.7	53.6	36.1	3.5	1.6	6.9	53.1	33.9	4.5	0.7	4.1	54.2	39.0	

42	

1	Ξ	
	æ	
	E	
	vmen	
	<u> </u>	
	Ξ	
	æ	
	_	
	20	
	Ξ	
	5	
١	-	
	_	
	≧	
	3	
	=	
	2	
	=	
	5	
•	Ē	
•		
•	-	
	က	
	æ	
	>	•
	2	
	Ē	
Ļ		
-	_	
-	_	
-		
	ation	
	ication	
-	Iucation	
-	education	
-	r education	
-	of education	
-	el ot education	
-	vel of education	
-	evel of education 1	
-	/ level of education	
-	DV level of education 1	
	s by level of education 1	
	ITS by level of education	•
-	(ers by level of education 1	
	rkers by level of education	•
	Orkers by level of education	•
	workers by level of education	
	IT WORKERS BY LEVEL OT EQUCATION	
	OT WORKERS BY LEVEL OT EQUCATION	
	in ot workers by level of education	
	ion of workers by level of education	
	ition ot workers by level of education	
	ition ot workers by level of education	
	ition ot workers by level of education	
	tribution of workers by level of education	
	tribution of workers by level of education	
	ition ot workers by level of education	
	s Distribution of workers by level of education	
	3 <b>Distribution of workers by level of education</b>	
	tribution of workers by level of education	

Table C.2	Table C.2.3 Distribution of workers by level of education   Employer	of education	Emplo	yers in int	ormai an	s in informal and in formal employment	employm	ent								
				Total					Men					Women		
		м %	% workers with a given level of education	a given lev	el of educa	tion	0M %	% workers with a given level of education	a given lev	el of educa	tion	0M %	% workers with a given level of education	a given lev	el of educa	tion
		None	Primary	Secondary	Tertiary	Other	None	Primary \$	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other
Percentage	Percentage of employees with a given level of education by sex I Informal employment	n by sex <b>I Info</b>	rmal employ	yment												
S	Africa	39.3	25.4	26.1	5.9	3.2	40.7	23.3	27.6	6.2	2.1	30.8	36.2	19.5	4.6	8.9
bné ntrie	Americas	9.7	29.5	43.6	16.3	0.9	9.5	31.6	43.8	14.4	0.8	10.7	23.5	42.9	21.8	1.2
s gni Uoc	Arab States	27.5	48.2	18.0	6.3	0.0	26.7	49.1	17.7	9.9	0.0	50.0	8.9	37.2	3.9	0.0
	Asia and the Pacific	7.7	22.2	61.7	5.4	3.0	7.2	21.9	61.3	6.3	3.3	9.7	22.3	64.1	2.3	1.5
	Europe and Central Asia	0.9	11.8	59.0	28.1	0.2	1.0	14.0	57.9	27.0	0.2	0.6	5.7	61.2	32.4	0.1
)	Total	14.0	23.9	52.3	7.2	2.7	14.1	23.6	52.1	7.5	2.7	13.4	24.3	53.4	6.3	2.6
	Americas	4.7	24.6	47.5	23.0	0.2	4.2	25.2	46.3	23.9	0.3	6.1	22.7	51.2	20.0	0.0
lopeo 2 Seint	Asia and the Pacific	1.5	3.1	71.1	23.4	0.9	0.7	3.1	6.99	28.1	1.2	4.2	3.6	82.0	10.2	0.0
	Europe and Central Asia	0.1	6.7	59.9	30.8	2.5	0.1	6.4	62.6	28.3	2.7	0.0	7.5	53.4	37.0	2.1
	Total	0.5	7.1	60.7	29.5	2.2	0.4	6.9	62.3	28.0	2.4	0.7	7.8	56.0	33.8	1.8
	Africa	39.3	25.4	26.1	5.9	3.2	40.7	23.3	27.6	6.2	2.1	30.8	36.2	19.5	4.6	8.9
	Americas	9.4	29.3	43.8	16.7	0.8	9.2	31.2	43.9	14.9	0.8	10.4	23.4	43.3	21.7	1.1
ph	Arab States	27.5	48.2	18.0	6.3	0.0	26.7	49.1	17.7	9.9	0.0	50.0	8.9	37.2	3.9	0.0
οW	Asia and the Pacific	7.6	21.7	61.9	5.8	2.9	7.1	21.5	61.4	6.8	3.3	9.6	21.9	64.6	2.5	1.4
	Europe and Central Asia	0.2	7.5	59.8	30.4	2.2	0.2	7.6	61.8	28.1	2.3	0.1	7.2	54.5	36.4	1.8
	Total	12.5	22.1	53.2	9.6	2.6	12.7	22.0	53.1	9.5	2.7	11.5	21.8	53.8	10.4	2.5

% workers with a given level of education         % workers with a given level of education         % workers with a given level of education           None         Prinary         Secondary         Tertiary         None         Prinary         Secondary         Tertiary         Secondary         Tertiary<					Total					Men					Women		
IndayTertiaryOtherNoneFrimarySecondaryFertiary $6.3$ $25.0$ $0.9$ $11.7$ $27.1$ $35.0$ $25.4$ $1.9$ $42.8$ $2.3$ $4.0$ $10.7$ $42.6$ $40.6$ $1.9$ $1.32$ $1.3$ $1.0$ $11.6$ $7.1$ $14.8$ $2.7$ $13.2$ $1.3$ $1.0$ $11.6$ $7.1$ $14.8$ $2.7$ $13.2$ $1.3$ $1.0$ $11.6$ $7.4$ $2.7$ $13.2$ $1.3$ $1.0$ $11.6$ $7.4$ $2.7$ $13.2$ $1.3$ $1.0$ $11.6$ $7.4$ $2.7$ $40.1$ $39.5$ $20.7$ $39.4$ $39.5$ $2.8$ $21.4$ $1.4$ $2.2$ $13.1$ $61.8$ $21.5$ $2.9$ $39.5$ $0.0$ $0.5$ $7.7$ $50.9$ $40.1$ $9.4$ $46.3$ $0.9$ $0.1$ $2.7$ $50.7$ $49.5$ $2.8$ $36.7$ $0.8$ $0.0$ $0.1$ $2.9$ $40.1$ $9.4$ $46.3$ $0.9$ $0.1$ $2.9$ $40.1$ $2.6$ $43.4$ $0.9$ $0.1$ $2.9$ $40.1$ $2.8$ $43.4$ $0.9$ $0.1$ $2.9$ $40.1$ $2.8$ $43.4$ $0.9$ $0.1$ $2.9$ $40.1$ $2.9$ $43.4$ $0.9$ $0.1$ $2.9$ $40.1$ $2.9$ $43.4$ $0.9$ $0.1$ $2.9$ $40.1$ $2.9$ $43.4$ $0.9$ $2.1$ $42.7$ $43$			м %	orkers with	a given leve	el of educa	tion	0M %	irkers with	a given lev	el of educa	tion	м %	orkers with	% workers with a given level of education	el of educa	tion
63         25.0         0.9         11.7         27.1         35.0         25.4           19         42.8         2.3         4.0         10.7         42.6         40.6           11         40.4         0.0         22.3         49.3         21.0         7.4           27         13.2         1.3         1.0         11.6         7.1         14.8           27         13.2         1.3         1.0         11.6         71.1         14.8           21.1         40.4         -0.0         0.5         20.7         39.4         39.5           23.1         140.4         1.4         2.2         13.1         61.8         21.5           24         1.4         2.2         13.1         61.8         21.5           29.1         1.4         2.2         13.1         61.8         21.5           20.1         0.1         2.2         13.1         61.8         21.5           21.4         1.4         2.2         13.1         61.8         21.5           21.4         2.2         13.1         61.8         21.5         40.1           21.4         2.3         0.0         0.1         2.7 </th <th></th> <th></th> <th>None</th> <th></th> <th>Secondary</th> <th>Tertiary</th> <th>Other</th> <th>None</th> <th>Primary</th> <th>Secondary</th> <th>Tertiary</th> <th>Other</th> <th>None</th> <th>Primary</th> <th>Secondary</th> <th>Tertiary</th> <th>Other</th>			None		Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other
Africa         108         270         36.3         25.0         0.9         11.7         27.1         35.0         25.4           Americas         34         9.7         41.9         42.8         2.3         40.0         10.7         42.6         40.6           Americas         1.1         11.7         72.7         13.2         1.3         10.0         7.4         40.6           Americas         0.5         18.0         41.1         7.2         13.2         1.0         11.6         7.1         14.8           Europe and Central Naia         0.5         18.0         41.1         40.4         0.0         0.5         13.1         14.8         14.8           Total         2.2         12.7         12.7         13.2         1.4         14.4         2.2         14.9         2.15         14.8         14.8           Total         2.2         12.7         62.3         21.4         14.4         2.2         14.1         14.8         14.8         14.8         14.8         14.8         14.9         14.9         14.9         14.9         14.9         14.9         14.9         14.9         14.9         14.9         14.9         14.9         14.9	Percentage	of employees with a given level of education	by sex I Forr	nal employr	nent												
Americas         3.4         9.7         4.19         4.28         2.3         4.0         10.7         4.26         4.06           Anab States         22.3         48.3         21.9         7.5         0.0         22.3         49.3         21.0         7.4           Rab States         22.3         48.3         21.9         7.5         0.0         22.3         49.3         21.0         7.4           Rab States         0.5         18.0         41.1         72.7         13.2         1.3         1.0         11.6         7.1         14.8           Luope and Central Asia         2.5         18.0         41.1         40.4         -0.0         0.5         20.7         39.4         39.5           Vatal         2.2         12.7         52.9         39.5         1.4         1.4         2.5         40.1         2.1         40.1           Americas         0.1         3.1         4.0         0.5         2.7         40.1         2.5         40.1         2.5         40.1         2.5         40.1         2.5         40.1         2.5         40.1         2.5         40.1         2.5         40.1         2.5         40.1         40.1         40.1<	S	Africa	10.8	27.0	36.3	25.0	0.9	11.7	27.1	35.0	25.4	0.7	8.5	26.6	40.2	23.2	1.5
May States         22.3         48.3         21.9         7.5         0.0         22.3         49.3         21.0         7.4           Ria and the Pacific         1.1         11.7         72.7         13.2         1.3         1.0         11.6         7.1         14.8           Europe and Central Asia         0.5         18.0         41.1         40.4         -0.0         0.5         20.7         39.4         39.5           Total         2.2         12.1         11.7         72.7         13.2         1.4         14.8         39.5           Matricas         0.5         12.7         62.3         21.4         1.4         2.2         13.1         61.8         21.5           Matricas         0.5         12.7         62.3         21.4         1.4         2.2         50.7         39.4         35.5           Matricas         0.1         3.1         52.6         43.4         0.9         0.1         2.9         4.0           Ital         0.1         3.1         52.6         43.4         0.9         0.1         2.9         4.0           Ital         0.1         3.1         52.6         43.4         0.9         0.1         2.9		Americas	3.4	9.7	41.9	42.8	2.3	4.0	10.7	42.6	40.6	2.0	1.7	6.8	40.1	48.4	3.1
Asia and the Pacific         11         117         727         132         13         10         116         711         148           Europe and Central Asia         0.5         18.0         41.1         40.4         -0.0         0.5         20.7         39.4         39.5           Itemperation         2.2         12.7         62.3         21.4         1.4         2.2         13.1         14.8           Americas         0.6         7.0         52.9         39.5         0.0         0.5         7.7         50.9         40.9           Americas         0.6         7.0         52.9         39.5         0.0         0.5         7.1         50.9         40.1           Americas         0.0         2.7         59.8         36.7         0.8         0.0         0.1         2.1         40.1           Compo and Central Asia         0.1         3.1         52.6         43.4         0.9         0.1         2.7         43.5           Ital         141         2.2         13.4         0.9         0.1         2.9         40.1           Ital         3.1         52.6         43.4         0.9         0.1         2.7         43.5		Arab States	22.3	48.3	21.9	7.5	0.0	22.3	49.3	21.0	7.4	0.0	2.8	11.0	65.8	20.4	0.0
Europe and Central Asia         0.5         18.0         4.1.1         4.0.4         -0.0         0.5         20.7         39.4         39.5           Idal         2.2         12.7         62.3         21.4         1.4         2.2         13.1         61.8         21.5           Idal         2.2         12.7         62.3         21.4         1.4         2.2         13.1         61.8         21.5           Idal         0.6         7.0         52.9         39.5         0.0         0.5         7.7         50.9         40.9           Americas         0.0         2.7         63.8         36.7         0.8         0.0         0.7         31         50.9         40.1           Idal         0.1         3.1         52.6         43.4         0.9         0.1         31         50.7         43.5           Idal         0.1         3.1         52.6         43.4         0.9         0.1         21.9         45.1           Idal         1.0         3.1         2.2         43.4         0.9         0.1         2.1         40.1           Idal         2.1         3.1         2.5         0.9         0.1         2.1		Asia and the Pacific	1.1	11.7	72.7	13.2	1.3	1.0	11.6	71.1	14.8	1.4	1.3	11.4	78.7	8.3	0.3
Index         Index <t< th=""><th></th><th>Europe and Central Asia</th><th>0.5</th><th>18.0</th><th>41.1</th><th>40.4</th><th>-0.0</th><th>0.5</th><th>20.7</th><th>39.4</th><th>39.5</th><th>-0.0</th><th>0.5</th><th>6.5</th><th>48.5</th><th>44.6</th><th>0.0</th></t<>		Europe and Central Asia	0.5	18.0	41.1	40.4	-0.0	0.5	20.7	39.4	39.5	-0.0	0.5	6.5	48.5	44.6	0.0
Americas         0.6         7.0         52.9         39.5         0.0         0.5         7.7         50.9         40.9           Americas         Nai and the Pacific         0.0         2.7         59.8         36.7         0.8         0.0         2.2         57.3         40.1           Europe and Central Asia         0.1         3.2         49.4         46.3         0.9         0.1         3.1         50.7         45.1           Total         0.1         3.1         52.6         43.4         0.9         0.1         2.9         45.1           Arrica         0.1         3.1         52.6         43.4         0.9         0.1         2.9         43.5           Mericas         0.1         3.1         52.6         43.4         0.9         0.1         2.9         43.5           Mericas         3.4         9.6         42.7         2.3         49.3         21.0         7.4           Mericas         3.4         9.6         42.7         2.3         40.7         40.7         40.7           Mericas         3.4         9.6         7.4         27.1         42.7         40.7         40.7           Maris and the Pacific	Ð	Total	2.2	12.7	62.3	21.4	1.4	2.2	13.1	61.8	21.5	1.4	1.8	10.7	64.2	22.2	1.1
Americas         0.6         7.0         52.9         39.5         0.0         0.5         7.7         50.9         40.9           Aia and the Pacific         0.0         2.7         59.8         36.7         0.8         0.0         2.2         57.3         40.1           Europe and Central Asia         0.1         3.1         5.4         46.3         0.9         0.1         3.1         50.7         45.1           Total         0.1         3.1         5.2.6         43.4         0.9         0.1         2.1         45.1           Antica         0.1         3.1         52.6         43.4         0.9         0.1         2.9         45.1           Africa         0.1         3.1         52.6         43.4         0.9         0.1         2.9         45.7           Africa         10.8         27.0         36.3         25.0         0.9         10.7         27.1         45.7           Americas         3.4         9.6         42.7         2.3         49.3         27.0         25.4           Americas         27.3         48.3         7.5         0.9         27.3         49.3         21.0         7.4           Arab State																	
Asia and the Pacific         0.0         2.1         59.8         36.7         0.8         0.0         2.2         57.3         40.1           Europe and Central Asia         0.1         3.2         49.4         46.3         0.9         0.1         3.1         50.7         45.1           Total         0.1         3.1         52.6         43.4         0.9         0.1         3.1         50.7         45.1           Antal         0.1         3.1         52.6         43.4         0.9         0.1         2.9         52.7         43.5           Antal         1         3.1         52.6         43.4         0.9         0.1         2.9         52.7         43.5           Anticas         10.8         27.0         36.3         25.0         0.9         11.7         27.1         43.7           Americas         3.4         9.6         42.0         27.9         40.7         40.7         40.7           Ana States         22.3         48.3         71.4         15.5         1.2         0.9         71.4         71.4           Ana States         1.0         10.8         71.4         15.5         12.9         70.9         71.4 <td< th=""><th></th><th>Americas</th><th>9.0</th><th>7.0</th><th>52.9</th><th>39.5</th><th>0.0</th><th>0.5</th><th>7.7</th><th>50.9</th><th>40.9</th><th>0.0</th><th>1.0</th><th>4.9</th><th>58.4</th><th>35.7</th><th>0.0</th></td<>		Americas	9.0	7.0	52.9	39.5	0.0	0.5	7.7	50.9	40.9	0.0	1.0	4.9	58.4	35.7	0.0
Europe and Central Asia         0.1         3.2         49.4         46.3         0.9         0.1         3.1         50.7         45.1           Total         0.1         3.1         52.6         43.4         0.9         0.1         3.1         50.7         45.1           Africa         0.1         3.1         52.6         43.4         0.9         0.1         2.9         53.7         43.5           Africa         10.8         27.0         36.3         25.0         0.9         11.7         27.1         35.0         25.4           Americas         3.4         9.6         42.0         42.7         2.3         40         10.7         42.7         40.7           Arab States         22.3         48.3         21.9         7.5         0.0         22.3         49.3         21.0         7.4           Asia and the Pacific         1.0         10.8         71.4         15.5         1.2         0.9         7.4         40.7           Europe and Central Asia         0.2         7.4         0.9         0.7         0.2         7.4         40.7		Asia and the Pacific	0.0	2.7	59.8	36.7	0.8	0.0	2.2	57.3	40.1	0.4	0.0	4.2	66.8	27.3	1.7
Total         0.1         3.1         52.6         43.4         0.9         0.1         2.9         52.7         43.5           Africa         10.8         27.0         36.3         25.0         0.9         11.7         27.1         35.0         25.4           Americas         3.4         9.6         42.0         42.7         2.3         4.0         10.7         42.7         40.7           Americas         2.3         48.3         21.9         7.5         0.0         22.3         49.3         21.0         7.4           Arab States         2.0         10.8         71.4         15.5         1.2         0.9         10.7         42.7         40.7           Arab States         2.0         10.8         71.4         15.5         1.2         0.9         10.7         7.4           Asia and the Pacific         1.0         10.8         71.4         15.5         1.2         0.9         10.8         17.2           Europe and Central Asia         0.2         7.0         47.8         0.7         0.9         17.2		Europe and Central Asia	0.1	3.2	49.4	46.3	0.9	0.1	3.1	50.7	45.1	1.0	0.1	3.1	45.4	50.6	0.8
Africa10.827.036.325.00.911.727.135.025.4Americas3.49.642.042.72.34.010.742.740.7Arab States22.348.321.97.50.022.349.321.07.4Arab States1.010.871.415.51.20.910.869.817.2Arab states0.27.047.344.80.70.910.874Europe and Central Asia0.27.047.344.80.70.27.943.5		Total	0.1	3.1	52.6	43.4	0.9	0.1	2.9	52.7	43.5	0.8	0.1	3.4	51.4	44.1	1.1
Africa         10.8         27.0         36.3         25.0         0.9         11.7         27.1         35.0         25.4           Americas         3.4         9.6         42.0         42.7         2.3         4.0         10.7         42.7         40.7           Arab States         22.3         48.3         21.9         7.5         0.0         22.3         49.3         21.0         7.4           Asia and the Pacific         1.0         10.8         71.4         15.5         1.2         0.9         10.8         17.2           Europe and Central Asia         0.2         7.0         47.8         0.7         0.2         7.9         43.5																	
Americas         3.4         9.6         42.0         42.7         2.3         4.0         10.7         42.7         40.7           Arab States         22.3         48.3         21.9         7.5         0.0         22.3         49.3         7.4           Arab States         22.3         48.3         21.9         7.5         0.0         22.3         49.3         7.4           Asia and the Pacific         1.0         10.8         71.4         15.5         1.2         0.9         10.8         17.2           Europe and Central Asia         0.2         7.0         47.3         44.8         0.7         0.2         7.9         43.5		Africa	10.8	27.0	36.3	25.0	0.9	11.7	27.1	35.0	25.4	0.7	8.5	26.6	40.2	23.2	1.5
Arab States         22.3         48.3         21.9         7.5         0.0         22.3         49.3         21.0         7.4           Asia and the Pacific         1.0         10.8         71.4         15.5         1.2         0.9         10.8         17.2           Europe and Central Asia         0.2         7.0         47.3         44.8         0.7         0.2         7.9         47.6         43.5		Americas	3.4	9.6	42.0	42.7	2.3	4.0	10.7	42.7	40.7	2.0	1.7	6.8	40.4	48.2	3.0
Asia and the Pacific         1.0         10.8         71.4         15.5         1.2         0.9         10.8         69.8         17.2           Europe and Central Asia         0.2         7.0         47.3         44.8         0.7         0.2         7.9         47.6         43.5	rlq	Arab States	22.3	48.3	21.9	7.5	0.0	22.3	49.3	21.0	7.4	0.0	2.8	11.0	65.8	20.4	0.0
0.2 7.0 47.3 44.8 0.7 0.2 7.9 47.6 43.5	οW	Asia and the Pacific	1.0	10.8	71.4	15.5	1.2	0.9	10.8	69.8	17.2	1.3	1.2	10.6	77.3	10.5	0.4
		Europe and Central Asia	0.2	7.0	47.3	44.8	0.7	0.2	7.9	47.6	43.5	0.7	0.2	3.8	46.1	49.3	0.7
1.8 10.8 60.4 25.7 1.3 1.8 11.2 60.1 25.7		Total	1.8	10.8	60.4	25.7	1.3	1.8	11.2	60.1	25.7	1.3	1.4	9.1	61.3	27.2	1.1

	-
	d)
	=
	E
	-
	=
	-
	-
	-
	പ
	_
	_
	G
	_
	0
	-
	-
	_
	_
	5
	-
	<b>~</b>
	-
	_
	C
	-
	-
	=
	$\sim$
	_
	ď
	-
	1
	>
	>
	-
	-
	0
	ō
	-
	-
	<b>~</b>
	÷
	έ
	Ś
	-uM
•	
•	
•	
-	n Uwn-account workers in informal and in formal employment
-	-uwn uo
-	
•	tion Uwn-
•	ation Uwn-
-	cation Uwn-
•	ication luwn-
	ucation luwn-
	ducation   Uwn-
	education   Uwn-
	education   Uwn-
	t education   Uwn-
	of education   Uwn-
	of education   Uwn-
	I of education   Uwn-
	el ot education   Uwn-
	vel of education   Uwn-
-	svel of education   Uwn-
	level of education   Uwn-
	level of education   Uwn-
	r level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of education
	iv level of educatior
	iv level of educatior
	iv level of educatior
	r level of education

Table C.2	Table C.2.4 Distribution of workers by level of education 1 Own-account workers in informal and in formal employment	of education	1 0wn-a(	scount wo	rkers in i	nformal ar	nd in forms	al employr	nent							
				Total					Men					Women		
		w %	% workers with a giv	a given lev	ren level of education	tion	0M %	% workers with a given level of education	a given lev	el of educa	tion	у WG	% workers with a given level of education	a given lev	el of educat	tion
		None	Primary	Secondary	Tertiary	Other	None	Primary \$	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other
Percentage	Percentage of employees with a given level of education by sex I Informal employment	n by sex <b>I Infor</b>	rmal employ	yment												
S	Africa	48.3	32.2	14.7	1.1	3.7	45.4	33.3	17.0	1.4	2.9	52.5	30.7	11.4	0.7	4.7
	Americas	11.7	28.7	46.8	10.1	2.7	10.7	29.8	47.7	8.8	3.0	13.3	26.8	45.6	12.1	2.1
s gnii Iuoc	Arab States	45.5	33.2	18.7	2.6	0.0	40.4	36.0	20.4	3.2	0.0	74.9	13.9	10.8	0.3	0.0
	Asia and the Pacific	32.3	30.2	31.6	3.5	2.4	30.9	31.5	30.8	4.0	2.9	37.7	25.9	33.9	1.8	0.8
	Europe and Central Asia	5.8	30.1	54.0	8.2	1.9	5.1	32.9	51.3	9.1	1.6	7.6	24.1	60.3	5.6	2.5
Ð	Total	32.7	30.4	30.7	3.7	2.6	30.9	31.6	30.6	4.1	2.8	38.0	26.9	30.7	2.7	1.7
	Americas	2.8	8.4	18.5	70.3	0.0	2.8	8.1	18.1	70.9	0.0	2.8	8.9	19.1	69.3	0.0
lopei tries	Asia and the Pacific	2.2	10.8	72.0	14.7	0.3	1.0	10.1	75.1	13.7	0.0	4.8	12.3	65.0	16.9	0.9
	Europe and Central Asia	0.2	9.3	60.3	27.4	2.8	0.2	9.5	62.8	24.2	3.3	0.2	8.6	55.3	34.0	1.9
	Total	1.3	9.3	50.6	37.2	1.6	1.1	9.3	53.4	34.5	1.8	1.7	9.3	45.4	42.5	1.1
	Africa	48.3	32.2	14.7	1.1	3.7	45.4	33.3	17.0	1.4	2.9	52.5	30.7	11.4	0.7	4.7
	Americas	10.7	26.3	43.5	17.2	2.4	9.7	27.2	44.3	16.1	2.7	12.1	24.7	42.4	19.0	1.9
ph	Arab States	45.5	33.2	18.7	2.6	0.0	40.4	36.0	20.4	3.2	0.0	74.9	13.9	10.8	0.3	0.0
DW	Asia and the Pacific	32.0	30.1	32.0	3.6	2.4	30.7	31.3	31.2	4.1	2.8	37.3	25.7	34.2	1.9	0.8
	Europe and Central Asia	2.0	16.0	58.2	21.2	2.5	1.9	17.4	58.9	19.1	2.7	2.4	13.3	56.8	25.4	2.1
	Total	31.5	29.6	31.4	4.9	2.5	29.9	30.9	31.4	5.1	2.8	36.4	26.1	31.3	4.4	1.7

% workers with a given level of education           avecode         44.3         29.1         21.1         4.6         0.8         4.7         22.1         19.7         30.6         23.6         43.0         23.6         43.0         23.6         43.0         23.6         23.1         23.1         23.6         23.1         23.6         23.1         23.6         23.6         23.6         23.1         23.6         23.1         23.6         23.1         23.6         23.1         23.6         23.6         23.6         23.6         23.6         23.6         23.6         23.6         23.6         23.6         23.1         23.6         23.1					Total					Men					Women		
Idialy         Iertiary         Iertiary         Iertiary         Iertiary         Iertiary         Iertiary         Iertiary         Secondary         Secondar			м %	orkers with	a given lev	el of educa	tion	0M %	orkers with	a given lev	el of educa	tion	м %	orkers with	ı a given lev	el of educa	ition
11         4.6         0.8         47.5         27.1         19.7         4.8         1.0         42.1         30.6           38         27.6         2.0         16.4         19.8         38.2         23.9         1.7         5.3         12.6           39         11.0         -0.0         21.8         44.7         22.7         10.8         0.0         32.5         24.5           39         11.0         -0.0         21.8         44.7         22.7         10.8         0.0         32.5         24.5           39         8.2         1.7         14.1         23.5         57.0         4.5         2.5         13.4         19.8           39         8.2         1.7         14.1         23.8         52.1         7.9         2.1         13.4         19.4           30         94.6         0.0         0.5         0.1         4.4         95.0         0.0         0.0         0.0         0.0           44.4         0.8         0.1         4.4         95.0         -0.0         0.0         0.0         0.0         0.0         0.0         0.0         0.0         0.0         0.0         0.0         0.0         0.0 <th></th> <th></th> <th>None</th> <th></th> <th>Secondary</th> <th>Tertiary</th> <th>Other</th> <th>None</th> <th>Primary</th> <th>Secondary</th> <th>Tertiary</th> <th>Other</th> <th>None</th> <th>Primary</th> <th>Secondary</th> <th>Tertiary</th> <th><b>Other</b></th>			None		Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	<b>Other</b>
Afficia         <	Percentage	of employees with a given level of education	by sex I Fori	nal employr	nent												
Americas         129         177         398         27.6         20         16.4         198         38.2         239         17         53         126           Anab States         238         41.3         239         11.0         -00         21.8         44.7         22.7         108         00         32.5         245           Anab States         23.8         41.3         23.9         11.0         -00         21.8         44.7         22.7         108         00         32.5         245           Europe and Central Asia         0.9         25.8         47.9         25.3         00         12.6         23.5         57.0         45.9         25.7         13.4         198           Intal         23         0.1         5.0         94.6         00         0.5         0.1         44.7         23.7         0.0         13.4         13.4         134         13.4           Americas         0.1         5.0         94.6         0.0         0.5         34.8         1.7         0.1         67           Intal         0.1         3.8         0.1         4.4         0.1         4.4         95.0         0.0         10.0         10.7	S	Africa	44.3	29.1	21.1	4.6	0.8	47.5	27.1	19.7	4.8	1.0	42.1	30.6	22.6	4.1	0.6
And States         23.8         41.3         23.9         11.0 $-0.0$ 21.8         44.7         22.7         10.8         0.0         32.5         24.5           Europe and Central Asia         12.6         22.5         58.9         4.1         1.9         12.6         23.5         57.0         4.5         2.5         13.4         19.8           Indef         13.6         22.5         58.9         4.1         1.9         12.6         23.5         57.0         4.5         2.5         13.4         19.8           Indef         13.6         22.6         53.9         8.2         1.7         14.1         23.8         57.0         4.5         2.5         13.4         19.4           Americas         0.3         5.1         68.8         2.1         14.1         2.3         57.0         4.5         0.0         0.0         0.3         5.4         19.4           Americas         0.1         5.0         94.6         0.0         0.5         4.4         19.4         19.4         19.4         19.4         19.4         10.4         10.4         10.4         10.4         10.4         10.4         10.4         10.4         10.4         10.		Americas	12.9	17.7	39.8	27.6	2.0	16.4	19.8	38.2	23.9	1.7	5.3	12.6	43.0	36.6	2.5
Paid and the Pacific         126         225         589         4.1         19         126         235         57.0         4.5         2.5         134         198           India         136         22.6         539         8.1         1         238         52.1         7.9         2.1         134         134         134           India         136         22.6         539         82         1.1         238         52.1         7.9         2.1         134         134           Anericas         0.3         5.1         6.3         2.6         0.0         0.5         0.1         2.8         134         134           Anis and the Pacific         0.3         5.1         6.3         2.6         0.0         0.5         0.1         4.4         9.5         0.1         134         134           Asia and the Pacific         0.1         3.8         1.4         0.1         4.4         0.1         4.4         0.1         4.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1 </th <th></th> <th>Arab States</th> <th>23.8</th> <th>41.3</th> <th>23.9</th> <th>11.0</th> <th>-0.0</th> <th>21.8</th> <th>44.7</th> <th>22.7</th> <th>10.8</th> <th>0.0</th> <th>32.5</th> <th>24.5</th> <th>22.1</th> <th>20.9</th> <th>0.0</th>		Arab States	23.8	41.3	23.9	11.0	-0.0	21.8	44.7	22.7	10.8	0.0	32.5	24.5	22.1	20.9	0.0
Example and Central Asia         0.9         25.8         4.79         25.3         0.0         1.2         30.5         45.9         22.4         -0.0         0.3         16.3           Idal         13.6         22.6         53.9         8.2         1.7         14.1         23.8         52.1         7.9         2.1         13.4         19.4           Idal         13.6         22.6         53.9         8.2         1.7         14.1         23.8         52.1         7.9         2.1         13.4         19.4           Americas         0.3         5.1         6.8         25.8         0.0         0.5         0.1         4.4         95.0         -0.0         0.3         0.4           Americas         0.1         3.8         1.4         0.1         4.4         95.0         4.5         0.1         0.1         3.4           Idata         0.1         0.3         5.1         4.4         0.8         0.2         3.3         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1         0.1 <t< th=""><th></th><th>Asia and the Pacific</th><th>12.6</th><th>22.5</th><th>58.9</th><th>4.1</th><th>1.9</th><th>12.6</th><th>23.5</th><th>57.0</th><th>4.5</th><th>2.5</th><th>13.4</th><th>19.8</th><th>63.1</th><th>3.3</th><th>0.4</th></t<>		Asia and the Pacific	12.6	22.5	58.9	4.1	1.9	12.6	23.5	57.0	4.5	2.5	13.4	19.8	63.1	3.3	0.4
Total         136         226         539         82         17         141         238         521         79         21         134         194           Matricas         0.3         0.1         5.0         94.6         0.0         0.5         0.1         4.4         95.0         0.0		Europe and Central Asia	0.9	25.8	47.9	25.3	0.0	1.2	30.5	45.9	22.4	-0.0	0.3	16.3	50.5	32.8	0.0
Americas         0.3         0.1         5.0         94.6         0.0         0.5         0.1         4.4         95.0         -0.0         0.0 <th< th=""><th>Ð</th><th>Total</th><th>13.6</th><th>22.6</th><th>53.9</th><th>8.2</th><th>1.7</th><th>14.1</th><th>23.8</th><th>52.1</th><th>7.9</th><th>2.1</th><th>13.4</th><th>19.4</th><th>57.6</th><th>8.9</th><th>0.6</th></th<>	Ð	Total	13.6	22.6	53.9	8.2	1.7	14.1	23.8	52.1	7.9	2.1	13.4	19.4	57.6	8.9	0.6
Mmericas $0.3$ $0.1$ $5.0$ $9.4.6$ $0.0$ $0.5$ $0.0$																	
Asia and the Pacific         0.3         5.1         6.88         25.8         0.0         0.0         4.5         6.8.3         27.2         0.0         1.0         6.7           Europe and Central Asia         0.1         3.8         1.4         0.1         4.0         59.5         34.8         1.7         0.1         3.4           Total         0.1         3.8         5.9         38.8         1.4         0.1         4.0         59.5         34.8         1.7         0.1         3.4           Total         0.2         3.3         50.0         45.6         0.9         0.2         3.1           Africa         1.4         4.8         0.8         1.4         0.8         0.2         3.3         50.0         45.6         0.9         0.1         3.4           Africa         1.4.3         29.1         21.1         4.6         0.8         47.5         27.1         19.7         4.1         9.7           Americas         1.4.3         33.0         40.6         1.6         1.8         4.7         2.7         10.8         1.9         7.6           Arab States         23.8         41.3         23.9         1.4         5.1		Americas	0.3	0.1	5.0	94.6	0.0	0.5	0.1	4.4	95.0	-0.0	0.0	0.0	6.2	93.7	0.0
Europe and Central Asia         0.1         3.8         5.5.9         3.8.8         1.4         0.1         4.0         5.9.5         3.4.8         1.7         0.1         3.4           Idtal         0.2         3.2         47.4         48.4         0.8         0.2         3.3         50.0         45.6         0.9         0.2         3.1           Africa         44.3         29.1         21.1         4.6         0.8         47.5         27.1         19.7         4.8         1.0         30.6           Africa         10.5         14.3         29.1         21.1         4.6         0.8         47.5         27.1         19.7         4.1         30.6           Americas         10.5         14.3         33.0         40.6         1.6         13.6         16.4         32.2         36.4         1.4         9.7           Ana States         23.8         41.3         23.9         11.0         -0.0         21.8         44.7         22.7         10.8         0.0         32.5         24.5           Asia and the Pacific         12.2         21.9         12.0         52.8         51.4         13.0         13.0         13.0         13.6         13.4		Asia and the Pacific	0.3	5.1	68.8	25.8	0.0	0.0	4.5	68.3	27.2	0.0	1.0	6.7	70.3	22.0	0.0
Indal         0.2         3.2         47.4         48.4         0.8         0.2         3.3         50.0         45.6         0.9         0.2         3.1           Africa         Africa         44.3         29.1         21.1         4.6         0.8         47.5         27.1         19.7         4.8         1.0         42.1         30.6           Americas         10.5         14.3         33.0         40.6         1.6         13.6         16.4         32.2         36.4         1.4         9.7           Americas         10.5         14.3         33.0         40.6         1.6         13.6         16.4         32.2         36.4         1.4         9.7           Arab States         23.3         41.3         23.9         11.0         -0.0         21.8         44.7         22.7         10.8         0.1         9.7           Ariab States         23.3         41.0         -0.0         21.8         44.7         22.7         10.8         0.0         32.5         24.5           Ariab States         0.5         14.4         52.1         20.7         17.0         52.8         54.7         13.0         13.0         14.4           Europe		Europe and Central Asia	0.1	3.8	55.9	38.8	1.4	0.1	4.0	59.5	34.8	1.7	0.1	3.4	48.3	47.4	0.8
Africa       44.3       29.1       21.1       4.6       0.8       47.5       27.1       19.7       4.8       1.0       42.1       30.6         Americas       10.5       14.3       33.0       40.6       1.6       13.6       16.4       32.2       36.4       1.4       4.1       9.7         Americas       10.5       14.3       33.0       40.6       1.6       13.6       16.4       32.2       36.4       1.4       4.1       9.7         Arab States       23.8       41.3       23.9       11.0       -0.0       21.8       44.7       22.7       10.8       0.0       32.5       24.5         Asia and the Pacific       12.2       21.9       59.2       4.9       1.8       12.1       22.8       57.4       5.3       2.4       13.0       19.4         Europe and Central Asia       0.5       14.4       52.1       32.3       0.7       0.7       17.0       52.8       5.4       5.4       5.4         Ictual       12.1       20.4       5.3       0.7       0.7       17.0       5.2       2.4       13.0       19.4         Ictual       12.6       1.6       12.6       1.6		Total	0.2	3.2	47.4	48.4	0.8	0.2	3.3	50.0	45.6	0.9	0.2	3.1	41.7	54.4	0.5
Africa         44.3         29.1         21.1         4.6         0.8         47.5         27.1         19.7         4.8         1.0         42.1         30.6           Americas         10.5         14.3         33.0         40.6         1.6         13.6         16.4         32.2         36.4         1.4         4.1         9.7           Arab States         23.8         41.3         23.9         11.0         -0.0         21.8         44.7         22.7         10.8         0.0         32.5         24.5           Arab States         12.2         21.9         59.2         4.9         1.8         12.1         22.7         10.8         0.0         32.5         24.5           Arab States         12.2         21.9         59.2         4.9         1.8         12.1         22.7         10.8         0.0         32.5         24.5           Asia and the Pacific         12.2         21.9         53.2         0.7         0.7         17.0         53.8         7.4         13.0         19.4           Europe and Central Asia         0.5         14.4         52.1         32.3         0.7         0.7         17.0         5.9         24.4         13.0 <td< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th></td<>																	
Americas     10.5     14.3     33.0     40.6     1.6     13.6     16.4     32.2     36.4     1.4     4.1     9.7       Arab States     23.8     41.3     23.9     11.0     -0.0     21.8     44.7     22.7     10.8     0.0     32.5     24.5       Arab States     23.8     41.3     23.9     11.0     -0.0     21.8     44.7     22.7     10.8     0.0     32.5     24.5       Asia and the Pacific     12.2     21.9     59.2     4.9     1.8     12.1     22.8     57.4     5.3     2.4     13.0     19.4       Europe and Central Asia     0.5     14.4     52.1     32.3     0.7     0.7     17.0     52.8     28.7     0.8     0.2     9.4       Total     12.1     20.4     53.2     12.6     1.6     12.6     21.6     51.9     12.0     2.0     11.9     17.6		Africa	44.3	29.1	21.1	4.6	0.8	47.5	27.1	19.7	4.8	1.0	42.1	30.6	22.6	4.1	0.6
Arab States         23.8         41.3         23.9         11.0         -0.0         21.8         44.7         22.7         10.8         0.0         32.5         24.5           Asia and the Pacific         12.2         21.9         59.2         4.9         1.8         12.1         22.8         57.4         5.3         2.4         13.0         19.4           Europe and Central Asia         0.5         14.4         52.1         32.3         0.7         0.7         17.0         52.8         28.7         0.8         0.2         9.4           Total         12.1         20.4         53.2         12.6         1.6         12.6         21.9         17.0         5.0         20         11.9         17.6		Americas	10.5	14.3	33.0	40.6	1.6	13.6	16.4	32.2	36.4	1.4	4.1	9.7	34.6	49.7	1.9
Asia and the Pacific         12.2         21.9         59.2         4.9         1.8         12.1         22.8         57.4         5.3         2.4         13.0         19.4           Europe and Central Asia         0.5         14.4         52.1         32.3         0.7         0.7         17.0         52.8         28.7         0.8         0.2         9.4           Total         12.1         20.4         53.2         12.6         16         12.6         21.6         51.9         12.0         2.0         11.9         17.6	rlq	Arab States	23.8	41.3	23.9	11.0	-0.0	21.8	44.7	22.7	10.8	0.0	32.5	24.5	22.1	20.9	0.0
0.5 14.4 52.1 32.3 0.7 0.7 17.0 52.8 28.7 0.8 0.2 9.4 12.1 20.4 53.2 12.6 1.6 12.6 21.6 51.9 12.0 2.0 11.9 17.6	οW	Asia and the Pacific	12.2	21.9	59.2	4.9	1.8	12.1	22.8	57.4	5.3	2.4	13.0	19.4	63.3	3.9	0.4
12.1 20.4 53.2 12.6 1.6 12.6 21.6 51.9 12.0 2.0 11.9 17.6		Europe and Central Asia	0.5	14.4	52.1	32.3	0.7	0.7	17.0	52.8	28.7	0.8	0.2	9.4	49.4	40.6	0.5
		Total	12.1	20.4	53.2	12.6	1.6	12.6	21.6	51.9	12.0	2.0	11.9	17.6	55.8	14.1	0.6

1		n
	4	h
L	-	U

f	
B	
O V	
d	
en	
a	
E	
월	
i.	
12	
(al	
IS	
Å	
10	
$\leq$	
ï	
fa	
ыg	
uti	
i.	
Ħ	
ដ	
Ľ	
tion	
cation	
ducation	
of education l	
el of education l	
evel of education l	
y level of education l	
s by level of education l	
ers by level of education l	
irkers by level of education l	
workers by level of education l	
of workers by level of education l	
on of workers by level of education l	
ution of workers by level of education l	
ribution of workers by level of education l	
istribution of workers by level of education l	
Distribution of workers by level of education l	
2.5 Distribution of workers by level of education l	
C.2.5 Distribution of workers by level of education l	
le C.2.5 Distribution of workers by level of education l	
Table C.2.5 Distribution of workers by level of education l	

				Tatal					Man					Womon		
																;
		% wor	% workers with a gi		ren level of education	tion	M %	orkers with	% workers with a given level of education	el of educ:	ation	M %	orkers with	% workers with a given level of education	el of educa	ntion
		None	Primary	Secondary	Tertiary	Other	None	Primary	Secondary	Tertiary	<b>Other</b>	None	Primary	Secondary	Tertiary	Other
Percentage	Percentage of employees with a given level of education by sex I Informal employment	y sex <mark>l Inform</mark>	al employ	ment												
S	Africa	45.9	35.0	14.6	0.9	3.6	37.9	38.6	20.4	1.3	1.9	51.4	32.0	11.3	0.7	4.5
	Americas	14.1	29.5	41.7	9.7	5.0	8.9	30.9	42.2	10.0	8.0	17.3	28.7	41.3	9.5	3.2
s gni Inoc	Arab States	25.3	60.7	12.8	1.3	0.0	10.5	72.9	14.5	2.2	0.0	39.0	49.9	10.6	0.4	-0.0
	Asia and the Pacific	24.7	29.1	40.7	3.5	2.0	16.9	29.6	41.0	7.6	4.9	29.6	28.8	39.9	1.3	0.4
	Europe and Central Asia	15.1	31.4	44.2	8.6	0.7	4.3	17.3	64.3	14.0	0.1	19.9	37.7	35.5	6.0	0.9
9	Total	27.8	30.2	36.2	3.4	2.4	20.3	31.2	37.4	6.6	4.4	32.5	29.5	35.1	1.7	1.2
	Americas	5.5	23.2	43.6	27.7	0.0	4.2	26.1	47.9	21.8	-0.0	9.9	20.8	37.4	35.1	0.1
	Asia and the Pacific	1.4	10.8	73.5	14.2	0.0	0.2	6.8	66.8	26.2	0.0	1.8	12.0	76.1	10.1	0.0
unoc  əvə(	Europe and Central Asia	0.4	13.1	72.2	13.0	1.3	0.0	14.8	60.2	23.0	2.0	0.5	12.1	75.4	11.5	0.5
	Total	1.4	12.6	70.7	14.8	0.5	0.6	13.4	60.9	23.9	1.1	1.7	12.6	73.5	12.1	0.2
	Africa	45.9	35.0	14.6	0.9	3.6	37.9	38.6	20.4	1.3	1.9	51.4	32.0	11.3	0.7	4.5
	Americas	13.8	29.3	41.8	10.2	4.9	8.7	30.8	42.4	10.4	7.7	17.0	28.4	41.2	10.2	3.1
rlq	Arab States	25.3	60.7	12.8	1.3	0.0	10.5	72.9	14.5	2.2	0.0	39.0	49.9	10.6	0.4	-0.0
DW	Asia and the Pacific	24.4	28.9	41.1	3.6	2.0	16.8	29.4	41.2	7.7	4.9	29.1	28.5	40.5	1.5	0.4
	Europe and Central Asia	11.1	26.4	51.8	9.8	0.9	3.0	16.5	63.0	16.8	0.7	15.0	31.3	45.6	7.4	0.8
	Total	27.3	29.9	36.8	3.6	2.4	20.0	31.0	37.8	6.9	4.4	31.9	29.1	35.9	1.9	1.2
Note: "" of informal of their leve	Note: "": not available. Global and regional estimates based on data for 107 countries representing 86 per cent of the world employed population and based on country data for the latest available year. Harmonized definition of informal employment and employment and employment in the informal sector. Contributing family workers are by definition all in informal employment independently of the formal or informal nature of the economic units and independently of their contain and environment in the informal sector. Contributing family workers are by definition all in informal employment independently of the formal or informal nature of the economic units and independently of their contained or informal nature of the economic units and independently of their lower of the formal or informal nature of the economic units and independently of their lower of the formal or informal nature of the economic units and independently of their lower of the formal or informal nature of the economic units and independently of their lower of the information of the economic units and independently of the formal or informal nature of the economic units and independently of the formation of the formation of the economic units and independently of the formation of the economic units and independently of the formation of the economic units and independent of the economic units and independent of the economic units and independent of the economic units are economic units and independent of the economic units are economic units and independent of the economic units are	based on data I sector. Cont	a for 107 ributing f	countries r amily worke	epresenting rs are by d	g 86 per cen efinition all	t of the wor in informal	id employe employmer	d population It independ	ר and base ently of th	d on country e formal or ir	data for th∈ ıformal natu	e latest avai ure of the ec	ilable year. I conomic un	Harmonized its and ind	l definition ependently
OI LITEIT LEV.																



Figure C.1 Share of informal employment by level of education, status in employment and region (%, latest available year)

**Note:** Global and regional estimates based on data for 107 countries representing 86 per cent of the world employed population and based on country data for the latest available year. Harmonized definition of informal employment and employment in the informal sector. Contributing family workers are by definition all in informal employment independently of the formal or informal nature of the economic units and independently of their level of education.

				W	Men					Wor	Women				Gen	Gender gap (women-men)	women-m	en)	
				Age groups	(%) sdn					Age groups (%)	(%) sdn				Ag	Age groups (% points)	: (% point:		
		15-24	25-29	30-34	35-54	55-64	65+	15-24	25-29	30-34	35-54	55-64	65+	15-24	25-29	30-34	35-54	55-64	65+
Total employment	yment																		
S	Africa	94.4	83.4	80.3	75.8	81.0	94.7	95.4	87.5	86.1	84.9	88.9	97.9	1.0	4.1	5.8	9.1	7.9	3.2
	Americas	64.5	47.8	46.9	50.3	57.2	76.6	61.6	47.1	49.4	55.2	67.6	83.6	-2.9	-0.7	2.5	4.9	10.4	7.0
s gni Iuoc	Arab States	84.8	69.3	61.4	56.8	60.8	68.1	88.7	68.1	55.8	57.8	71.3	91.6	3.9	-1.2	-5.6	1.0	10.5	23.5
	Asia and the Pacific	90.0	77.6	73.6	68.5	74.5	91.4	83.7	64.1	63.0	62.5	83.7	92.6	-6.3	-13.6	-10.6	-6.0	9.2	1.2
	Europe and Central Asia	50.4	39.8	33.0	33.0	36.7	46.8	47.6	40.3	35.6	33.6	31.4	45.0	-2.8	0.5	2.6	0.5	-5.3	-1.9
Ð	Total	85.9	73.1	69.0	65.6	71.2	88.1	81.6	63.9	62.6	62.1	77.2	89.4	-4.3	-9.3	-6.4	-3.5	6.0	1.3
	Americas	13.7	17.3	18.7	20.0	23.5	27.7	13.2	17.6	18.2	18.8	20.2	22.0	-0.5	0.4	-0.5	-1.2	-3.3	-5.7
lopei tries	Asia and the Pacific	23.2	17.5	12.4	14.4	27.8	52.0	20.9	13.0	14.9	19.4	34.3	60.7	-2.3	-4.5	2.5	5.0	6.5	8.8
	Europe and Central Asia	25.7	17.5	15.3	14.9	17.5	42.6	26.1	15.0	12.3	12.1	13.8	28.9	0.4	-2.5	-3.0	-2.7	-3.6	-13.7
	Total	19.4	17.4	15.9	16.3	21.9	39.3	19.1	15.6	14.7	15.4	20.4	37.5	-0.3	-1.8	-1.3	-0.9	-1.5	-1.8
	Africa	94.4	83.4	80.3	75.8	81.0	94.7	95.4	87.5	86.1	84.9	88.9	97.9	1.0	4.1	5.8	9.1	7.9	3.2
	Americas	48.4	37.7	37.2	39.2	42.5	55.5	43.0	36.0	38.0	40.7	41.6	52.2	-5.4	-1.7	0.8	1.5	-0.9	-3.3
rlq	Arab States	84.8	69.3	61.4	56.8	60.8	68.1	88.7	68.1	55.8	57.8	71.3	91.6	3.9	-1.2	-5.6	1.0	10.5	23.5
oW	Asia and the Pacific	88.8	75.8	71.1	66.0	70.9	86.5	81.6	61.9	60.8	60.3	77.1	85.9	-7.3	-13.9	-10.3	-5.8	6.1	-0.7
	Europe and Central Asia	36.5	27.5	22.6	20.9	23.4	44.3	34.8	25.0	21.0	19.1	19.2	36.2	-1.7	-2.4	-1.6	-1.8	-4.2	-8.1

Table C.3 Share of informal employment by age group, status in employment, sex and region

-3.6

-2.4

-4.7

-7.0

-10.1

-6.1

74.8

58.6

53.1

54.3

56.3

72.8

78.4

61.1

57.9

61.4

66.5

79.0

Total

				Men	=					Women	en				Gend	Gender gap (women-men)	omen-me	(e	
				Age groups	(%) sdr					Age groups (%)	(%) sd				Ago	Age groups (% points)	% points		
		15-24	25-29	30-34	35-54	55-64	65+	15-24	25-29	30-34	35-54	55-64	65+	15-24	25-29	30-34	35-54	55-64	65+
Employees																			
S	Africa	85.4	70.3	66.3	51.3	50.4	68.8	84.4	67.8	63.0	48.8	49.2	79.5	-1.0	-2.5	-3.3	-2.4	-1.2	10.7
	Americas	53.7	36.2	33.3	34.0	38.1	56.4	50.1	35.4	36.1	40.5	49.6	63.7	-3.6	-0.8	2.8	6.5	11.4	7.3
	Arab States	82.9	65.6	55.4	46.6	50.8	78.4	73.3	52.5	34.5	30.0	43.3	91.9	-9.6	-13.1	-20.9	-16.6	-7.4	13.5
qolə <i>ı</i> gnig	Asia and the Pacific	84.3	69.0	62.0	55.8	61.2	87.2	73.8	51.2	46.1	47.2	72.3	90.7	-10.5	-17.8	-16.0	-8.6	11.1	3.5
ner: Der	Europe and Central Asia	41.6	35.0	28.9	28.3	25.3	19.2	34.6	30.6	24.9	24.7	19.3	25.3	-7.0	-4.4	-3.9	-3.7	-6.1	6.0
Ð	Total	75.9	61.8	55.1	50.6	54.2	75.0	67.2	48.3	43.5	44.0	55.1	70.2	-8.8	-13.5	-11.5	-6.5	0.9	-4.7
	Americas	10.2	13.9	12.5	12.2	14.1	15.0	11.6	14.6	14.0	14.3	13.5	13.4	1.4	0.7	1.5	2.1	-0.6	-1.6
lopeo lopeo	Asia and the Pacific	17.4	10.3	5.4	6.7	10.2	19.2	19.6	11.9	9.1	9.8	12.2	24.9	2.2	1.7	3.8	3.1	2.0	5.7
	Europe and Central Asia	17.9	8.5	5.2	4.1	4.1	14.2	19.9	9.9	6.4	5.2	5.0	11.4	2.0	1.4	1.2	1.1	0.9	-2.8
	Total	13.9	10.8	7.8	7.2	0.0	16.1	15.8	11.9	9.4	8.8	9.7	15.8	1.9	1.1	1.6	1.7	0.7	-0.2
	Africa	85.4	70.3	66.3	51.3	50.4	68.8	84.4	67.8	63.0	48.8	49.2	79.5	-1.0	-2.5	-3.3	-2.4	-1.2	10.7
	Americas	37.4	27.8	25.0	24.2	24.5	28.3	32.7	26.4	26.6	27.9	25.1	24.6	-4.6	-1.3	1.7	3.7	0.6	-3.6
hh	Arab States	82.9	65.6	55.4	46.6	50.8	78.4	73.3	52.5	34.5	30.0	43.3	91.9	-9.6	-13.1	-20.9	-16.6	-7.4	13.5
DW	Asia and the Pacific	81.6	65.6	57.5	51.7	54.9	71.2	69.8	48.0	42.9	43.7	56.0	64.1	-11.8	-17.6	-14.6	-8.0	1.1	-7.2
	Europe and Central Asia	27.9	20.2	15.0	12.1	10.5	16.2	25.5	17.7	13.0	11.3	9.2	17.6	-2.5	-2.5	-2.0	-0.8	-1.3	1.4
	Total	64.6	52.4	44.6	40.2	40.0	51.5	54.3	39.5	34.2	33.5	31.3	39.0	-10.3	-12.9	-10.3	-6.7	-8.6	-12.5

				Men	u					Women	len				Gend	der gap (v	Gender gap (women-men)	(u	
				Age groups	(%) sdn					Age groups (%)	lps (%)				Ag	e groups	Age groups (% points)		
		15-24	25-29	30-34	35-54	55-64	65+	15-24	25-29	30-34	35-54	55-64	65+	15-24	25-29	30-34	35-54	55-64	<del>6</del> 5+
Employers																			
S	Africa	81.4	83.5	82.9	81.1	86.8	89.0	84.7	75.9	77.1	75.4	76.2	88.6	3.3	-7.6	-5.8	-5.7	-10.6	-0.4
	Americas	58.0	50.3	46.2	44.6	40.8	43.5	48.0	42.2	41.0	40.8	44.5	56.4	-10.0	-8.1	-5.2	-3.8	3.7	13.0
	Arab States	80.3	62.2	57.0	61.5	54.2	52.3	96.0	88.7	69.6	72.5	62.8	90.6	15.7	26.5	12.6	11.0	8.6	38.3
qolə grig	Asia and the Pacific	70.6	67.9	66.7	58.2	67.2	81.9	75.5	76.5	68.1	56.7	68.6	82.2	4.9	8.6	1.4	-1.5	1.4	0.3
	Europe and Central Asia	29.8	33.1	26.1	23.8	46.1	36.2	81.3	54.0	42.7	33.6	29.7	34.2	51.4	20.9	16.6	9.8	-16.4	-2.0
Ð	Total	70.6	65.7	62.9	57.2	62.9	69.8	72.4	67.9	61.7	53.7	58.0	73.5	1.8	2.2	-1.2	-3.6	-4.9	3.8
	Americas	85.4	47.1	50.6	61.0	56.5	65.1	77.1	72.4	55.9	56.6	57.9	70.7	-8.2	25.3	5.2	-4.4	1.4	5.5
	Asia and the Pacific	100.0	59.0	10.6	17.0	26.0	41.6	68.6	0.3	9.6	12.3	25.3	55.7	-31.4	-58.7	-1.0	-4.7	-0.8	14.0
unoc  əvə(	Europe and Central Asia	50.8	47.9	51.0	40.9	40.1	56.5	69.3	50.1	43.1	47.6	47.5	76.8	18.5	2.2	-8.0	6.7	7.4	20.2
	Total	55.0	48.6	44.3	36.4	36.3	50.7	84.7	47.8	38.9	41.7	43.5	67.4	29.7	-0.8	-5.4	5.2	7.2	16.7
	Africa	81.4	83.5	82.9	81.1	86.8	89.0	48.8	75.9	77.1	75.4	76.2	88.6	-32.5	-7.6	-5.8	-5.7	-10.6	-0.4
	Americas	59.0	50.2	46.3	45.2	41.5	44.4	96.0	43.1	41.3	41.5	45.2	57.1	37.0	-7.1	-5.0	-3.8	3.7	12.7
rlq	Arab States	80.3	62.2	57.0	61.5	54.2	52.3	75.5	88.7	69.6	72.5	62.8	90.6	-4.8	26.5	12.6	11.0	8.6	38.3
DW	Asia and the Pacific	70.6	67.8	63.5	54.5	59.3	75.0	72.1	75.1	64.9	52.4	59.8	75.9	1.6	7.4	1.4	-2.1	0.5	0.8
	Europe and Central Asia	43.7	42.2	43.1	37.1	41.0	53.3	72.1	51.5	43.0	45.0	45.1	74.9	28.4	9.3	-0.1	7.9	4.1	21.6
	Total	69.69	63.9	59.9	52.9	55.8	66.2	0.0	65.4	57.3	50.6	52.7	71.8	-69.6	1.5	-2.6	-2.3	-3.0	5.6

				Z	Mell					women	lell				nen	aciinei gap (wuilleil-illeil)			
				Age gro	Age groups (%)					Age groups (%)	(%) sdn				A	Age groups (% points)	(% point:	()	
		15-24	25-29	30-34	35-54	55-64	65+	15-24	25-29	30-34	35-54	55-64	65+	15-24	25-29	30-34	35-54	55-64	65+
m-accou	Own-account workers																		
S	Africa	95.2	93.6	93.4	94.1	94.7	96.9	95.8	95.0	94.8	96.2	98.0	98.4	0.6	1.4	1.4	2.1	3.3	1.5
	Americas	89.5	83.8	82.1	80.3	79.7	90.8	90.2	84.0	83.7	84.6	86.6	88.9	0.7	0.2	1.6	4.2	6.9	-1.9
	Arab States	77.5	73.7	70.6	71.5	69.6	62.6	95.0	92.0	88.3	89.6	90.5	87.8	17.6	18.4	17.7	18.1	20.9	25.1
qolə gnig	Asia and the Pacific	94.5	91.2	90.0	89.5	91.4	93.6	95.4	88.7	87.0	84.3	87.6	92.9	0.9	-2.5	-3.0	-5.3	-3.8	-0.7
	Europe and Central Asia	78.1	50.5	46.1	48.5	71.7	74.1	72.9	61.3	48.6	49.0	52.9	66.5	-5.2	10.7	2.5	0.5	-18.9	-7.7
ə	Total	93.5	90.1	89.0	88.6	90.2	93.2	94.7	89.4	87.7	85.3	88.4	93.2	1.1	-0.6	-1.3	-3.3	-1.8	0.1
	Americas	87.9	75.2	81.7	73.0	64.3	68.8	80.8	86.8	82.2	72.2	6.9	82.9	-7.0	11.6	0.5	-0.8	5.5	14.0
	Americas	87.9	75.2	81.7	73.0	64.3	68.8	80.8	86.8	82.2	72.2	66.69	82.9	-7.0	11.6	0.5	-0.8	5.5	14.0
	Asia and the Pacific	73.5	40.1	68.5	44.1	64.5	86.6	100.0	89.4	60.9	49.2	65.5	91.3	26.5	49.3	-7.7	5.1	1.0	4.7
unoc Iəvə	Europe and Central Asia	88.3	79.7	73.6	66.3	69.0	83.5	89.1	73.8	68.2	69.3	71.7	87.2	0.8	-5.9	-5.5	2.9	2.8	3.7
	Total	87.7	75.4	75.0	64.4	66.4	81.0	87.0	<i>T.TT</i>	71.5	67.8	69.7	88.1	-0.7	2.4	-3.5	3.3	3.4	7.1
	Africa	95.2	93.6	93.4	94.1	94.7	96.9	95.8	95.0	94.8	96.2	98.0	98.4	0.6	1.4	1.4	2.1	3.3	1.5
	Americas	89.4	83.1	82.0	79.3	76.8	87.3	89.4	84.3	83.5	82.9	82.6	88.0	0.0	1.2	1.5	3.5	5.8	0.6
rld	Arab States	77.5	73.7	70.6	71.5	69.6	62.6	95.0	92.0	88.3	89.6	90.5	87.8	17.6	18.4	17.7	18.1	20.9	25.1
oW	Asia and the Pacific	94.5	91.0	89.9	89.2	90.8	93.3	95.4	88.7	86.9	84.1	87.0	92.8	0.9	-2.4	-3.0	-5.1	-3.8	-0.5
	Europe and Central Asia	84.3	66.5	62.2	59.3	70.0	79.3	82.4	69.7	61.3	61.5	66.3	78.5	-1.9	3.2	-0.9	2.2	-3.7	-0.8
	Total	93.4	89.6	88.6	87.6	88.7	92.3	94.4	89.1	87.1	84.6	86.8	92.8	1.1	-0.6	-1.4	-3.1	-2.0	0.5

 Table C.4
 Informality and working time by sex for total employment and for employees

	employment
1	otal
-	veek by sex l
	X hours a w
-	r more than
;	IN X 0
-	ss tha
•	ing le
-	work
•	. people
-	Percentage of
	lable C.4.1

Noticing less         Noticin					Total					Men					Women		
-48 $-60$ $<15$ $<20$ $<35$ $-48$ $-60$ $<55$ $<48$ $>60$ $<55$ $>48$ $>60$ $<55$ $>48$ $>60$ <t< th=""><th></th><th></th><th>₩</th><th>working le 1an X hours</th><th>SS .c</th><th>% workin than X h</th><th>g more ours</th><th>th th</th><th>vorking les an X hours</th><th>S</th><th>% wo more than</th><th>rking 1 X hours</th><th>tt %</th><th>vorking les an X hours</th><th>SS (</th><th>% workir than X  </th><th>ig more 10urs</th></t<>			₩	working le 1an X hours	SS .c	% workin than X h	g more ours	th th	vorking les an X hours	S	% wo more than	rking 1 X hours	tt %	vorking les an X hours	SS (	% workir than X	ig more 10urs
Indust         Indus         Indus         Indus <th></th> <th></th> <th>&lt;15</th> <th>&lt;20</th> <th></th> <th>&gt;48</th> <th><b>&gt;60</b></th> <th>&lt;15</th> <th>&lt;20</th> <th>&lt;35</th> <th>&gt;48</th> <th>09^</th> <th>&lt;15</th> <th>&lt;20</th> <th>&lt;35</th> <th>&gt;48</th> <th>99&lt;</th>			<15	<20		>48	<b>&gt;60</b>	<15	<20	<35	>48	09^	<15	<20	<35	>48	99<
117 $140$ $9.9$ $14.3$ $32.0$ $37.5$ $16.8$ $16.6$ $24.7$ $24.7$ $227$ $84$ $6.6$ $12.4$ $27.8$ $26.8$ $9.5$ $17.8$ $260$ $51.0$ $170$ $352$ $16.7$ $6.1$ $10.0$ $34.0$ $37.7$ $18.0$ $12.8$ $51.0$ $170$ $552$ $14.9$ $4.2$ $5.3$ $17.0$ $57.3$ $16.8$ $7.2$ $9.4$ $26.9$ $45.7$ $295$ $14.9$ $3.2$ $5.4$ $16.5$ $36.3$ $15.2$ $72$ $94$ $26.9$ $45.7$ $295$ $14.1$ $52.2$ $51.3$ $16.2$ $51.9$ $51.9$ $51.9$ $51.9$ $51.9$ $51.9$ $51.7$ $2112$ $315$ $51.3$ $16.2$ $51.3$ $52.9$ $51.9$ $51.9$ $51.9$ $51.9$ $51.9$ $51.9$ $51.9$ $51.9$ $51.9$ $51.9$	ae of	neonle working less than X or more that	n X hours a w	nours eek hv sex I	hours Informal emn	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours
Americas         I13         194         37.6         227         84         6.6         124         278         9.5         178         290         510         170           Arab States         7.6         11.6         37.6         352         16.7         6.1         100         340         377         180         198         253         67.1         148           Arab States         7.6         1.1         21.5         522         14.9         42         53         16.8         7.2         94         26.9         45.7           Kais and the Pacific         5.7         7.1         21.5         52.2         14.9         42         53         15.2         15.2         16.9         35.2         16.9         35.3         16.1         14.8           Total         7.2         100         25.8         45.7         14.1         52.0         52.1         16.9         13.7         38.2         32.2         34.6         21.6           Americas         7.7         136         27.7         23.5         54.7         14.1         13.9         23.7         35.2         35.4         36.7         36.7         36.7         36.7         36.7	5	Africa	12.5	18.0	39.0	31.7	14.0	9.9	14.3	32.0	37.5	16.8	16.0	22.8	47.5	24.7	10.7
Arab States         7.6         1.6         3.7         6         1.6         3.7         1.6         1.6         3.7         1.6         1.6         3.7         1.8         1.9         2.3         6.71         1.48           Arab States         5.7         7.1         2.15         5.2         1.49         4.2         5.3         1.70         5.73         1.68         7.2         9.4         2.69         4.57           Europe and Central Asia         4.9         7.6         2.17         2.95         1.18         3.2         5.4         16.5         36.3         15.2         17.2         10.5         2.84         2.11           Interies         7.7         1.3         2.3         3.6         1.7         2.35         5.4         10.1         2.17         2.9         3.8         3.32         3.8         3.32         3.8         3.32         3.8         3.3	rries	Americas	11.3	19.4	37.6	22.7	8.4	9.9	12.4	27.8	26.8	9.5	17.8	29.0	51.0	17.0	6.8
Asia and the Pacific         57         71         215         522         149         42         53         170         53         168         72         94         269         457           Total         72         10         23         152         72         105         284         211           Total         72         100         258         457         141         52         72         105         72         105         284         211           Americas         77         136         27.2         202         51.3         16.5         33.3         15.2         72         28.4         21.1           Americas         77         136         27.7         141         52         72         20.2         13.8         23.2         33.2         23.2         23.4         21.1         23.5         24.1         13.1         27.2         23.5         26.7         13.3         25.2         26.7         21.4         21.1         21.2         23.5         26.7         21.3         23.2         23.2         23.2         23.6         23.6         23.6         23.6         23.6         24.5         24.7         24.5         24.7         24.7	unoo	Arab States	7.6	11.6	37.6	35.2	16.7	6.1	10.0	34.0	37.7	18.0	19.8	25.3	67.1	14.8	5.2
Europe and Central Asia         49         7.6         21.7         29.5         11.8         32         54         16.5         36.3         15.2         72         10.5         28.4         21.1           Idal         72         10.0         25.8         45.7         14.1         5.2         72         20.5         13.8         33.2         38.2         38.2           Americas         7.7         13.6         29.4         21.7         5.7         27.3         46         10.5         13.8         33.2         38.2           Americas         7.7         13.6         29.4         21.7         13.0         27.9         33.0         9.3         13.8         23.7         38.2         36.5         14.7           Idation and the Pacific         10.5         17.2         36.9         27.4         41.9         37.3         38.9         37.7         38.9         37.7         38.9         37.7         38.9         37.7         38.9         37.7         38.9         36.9         36.9         36.9         36.9         36.9         36.9         36.9         36.9         36.9         36.9         36.9         36.9         36.9         36.9         36.1         36.9 <th>និមាន</th> <th>Asia and the Pacific</th> <td>5.7</td> <td>7.1</td> <td>21.5</td> <td>52.2</td> <td>14.9</td> <td>4.2</td> <td>5.3</td> <td>17.0</td> <td>57.3</td> <td>16.8</td> <td>7.2</td> <td>9.4</td> <td>26.9</td> <td>45.7</td> <td>12.9</td>	និមាន	Asia and the Pacific	5.7	7.1	21.5	52.2	14.9	4.2	5.3	17.0	57.3	16.8	7.2	9.4	26.9	45.7	12.9
Total         7.2         10.0         25.8         45.1         14.1         5.2         7.2         20.2         51.3         16.2         9.9         13.8         33.2         38.2           Americas         7.1         13.6         29.4         21.2         3.5         5.4         10.1         21.7         27.3         4.6         10.5         17.8         38.8         13.7           Americas         10.5         17.2         36.1         27.7         8.0         7.7         13.0         27.9         33.0         9.3         13.8         22.0         4.5         21.6           Americas         7.1         12.3         28.9         28.7         4.7         2.4         13.0         27.9         35.0         35.0         14.7           Intera         7.1         12.3         28.9         24.7         24.4         13.9         35.2         6.5         17.2         35.9         16.1         17.2         36.9         16.1         17.2         36.9         16.1         17.2         36.9         16.1         17.2         36.9         16.1         16.1         17.9         16.1         17.2         16.9         16.1         17.9         16.1	nerg	Europe and Central Asia	4.9	7.6	21.7	29.5	11.8	3.2	5.4	16.5	36.3	15.2	7.2	10.5	28.4	21.1	7.7
Americas         17         13.6         29.4         21.2         3.5         5.4         10.1         21.7         27.3         4.6         10.5         17.8         38.8         13.7           Americas         10.5         17.2         36.1         27.7         8.0         7.7         13.0         27.9         35.2         6.2         7.2         45.6         21.6           Americas         4.2         7.7         23.5         26.7         4.7         2.4         4.1         13.9         35.2         6.2         7.2         45.6         21.6           Utupe and Central Axia         4.2         7.7         23.5         26.7         4.7         2.4         4.1         13.9         35.2         6.2         7.2         45.6         21.6           Total         11.1         12.3         28.9         24.8         5.1         5.0         8.5         20.2         31.6         16.1         17         20.2         45.6         16.1         16.1         17         20.2         45.6         16.1         16.1         16.2         16.2         16.2         16.1         16.1         16.1         16.2         16.2         16.2         16.2         16.1	a	Total	7.2	10.0	25.8	45.7	14.1	5.2	7.2	20.2	51.3	16.2	9.9	13.8	33.2	38.2	11.7
Americas         7.1         13.6         29.4         21.2         3.5         5.4         10.1         21.7         21.3         4.6         10.5         17.8         38.8         13.7           Akia and the Pacific         10.5         17.2         36.1         27.7         8.0         7.7         13.0         27.9         35.2         6.2         4.5         21.6         21.6           Europe and Central Asia         4.2         7.7         23.5         26.7         4.7         24         4.1         13.9         35.2         6.2         7.2         45.6         21.6           Total         7.1         12.3         28.9         24.8         5.1         5.0         8.5         20.2         31.6         6.2         14.7           Total         7.1         12.3         28.9         24.8         5.1         5.0         8.5         20.2         31.6         6.3         16.7         16.7         16.7         16.7         16.7         16.7         16.7         16.1         16.1         16.1         16.1         16.1         16.1         16.1         16.1         16.1         16.1         16.1         16.1         16.1         16.1         16.1																	
Asia and the Pacific10.517.236.127.78.07.713.027.933.09.313.822.045.621.6Europe and Central Asia4.27.723.526.74.72.44.113.935.26.27.212.936.914.7Total7.112.323.924.85.15.08.520.231.66.310.217.239.916.1Africa7.112.328.924.85.15.08.520.231.66.310.217.239.916.1Africa10.518.039.031.714.09.914.332.037.516.816.723.916.1Africa10.618.336.022.47.56.412.026.726.98.616.426.847.524.7Americas7.611.637.635.216.76.110.034.037.718.019.825.367.114.8Arab States7.611.637.654.817.256.916.77.396.827.345.2Europe and Central Asia4.67.72.42.92.94.917.256.916.77.396.22.114.8Arab States7.611.637.611.84.95.417.256.916.77.396.227.345.2Europe and Central Asia<		Americas	7.7	13.6	29.4	21.2	3.5	5.4	10.1	21.7	27.3	4.6	10.5	17.8	38.8	13.7	2.2
Europe and Central Asia         4.2         7.1         23.5         26.7         4.7         2.4         4.1         13.9         35.2         6.2         7.2         12.9         36.9         14.7           Total         7.1         12.3         28.9         24.8         5.1         5.0         8.5         20.2         31.6         6.3         10.2         17.2         39.9         16.1           Africa         12.5         18.0         39.0         31.7         14.0         9.9         14.3         32.0         37.5         16.8         16.7         39.9         16.1           Africa         12.5         18.0         39.0         31.7         14.0         9.9         14.3         32.0         37.5         16.8         47.5         24.7           Americas         10.6         18.3         36.0         27.4         7.5         6.4         12.0         26.7         26.9         86.6         16.4         26.8         48.6         16.4           Arab States         7.6         11.6         37.7         26.9         86.7         16.4         26.8         47.5         47.5           Asia and the Pacific         5.7         7.9         37.7 <th>sain</th> <th>Asia and the Pacific</th> <td>10.5</td> <td>17.2</td> <td>36.1</td> <td>27.7</td> <td>8.0</td> <td>7.7</td> <td>13.0</td> <td>27.9</td> <td>33.0</td> <td>9.3</td> <td>13.8</td> <td>22.0</td> <td>45.6</td> <td>21.6</td> <td>6.6</td>	sain	Asia and the Pacific	10.5	17.2	36.1	27.7	8.0	7.7	13.0	27.9	33.0	9.3	13.8	22.0	45.6	21.6	6.6
India         7.1         12.3         28.9         24.8         5.1         5.0         8.5         20.2         31.6         6.3         10.2         17.2         39.9         16.1           Africa         12.5         18.0         39.0         31.7         14.0         9.9         14.3         37.5         16.8         16.0         22.8         47.5         24.7           Americas         10.6         18.3         36.0         22.4         7.5         6.4         12.0         26.7         26.9         8.6         16.4         26.8         48.6         16.4         14.8           Arab States         7.6         11.6         37.6         25.4         7.5         6.4         12.0         26.7         26.9         8.6         16.4         26.8         16.4         14.8           Arab States         7.6         11.6         37.6         36.7         16.0         34.0         37.7         18.0         19.8         57.3         67.1         14.8           Asia and the Pacific         5.7         7.3         54.9         16.7         7.3         9.6         27.3         45.2           Europe and Central Asia         4.6         7.0         2.9	unor	Europe and Central Asia	4.2	7.7	23.5	26.7	4.7	2.4	4.1	13.9	35.2	6.2	7.2	12.9	36.9	14.7	2.6
as       12.5       18.0       39.0       31.7       14.0       9.9       14.3       32.0       37.5       16.8       16.0       22.8       47.5       24.7         as       10.6       18.3       36.0       22.4       7.5       6.4       12.0       26.7       26.9       8.6       16.4       26.8       48.6       16.4         tates       7.6       11.6       37.6       35.2       16.7       6.1       10.0       34.0       19.8       25.3       67.1       14.8         dthe Pacific       5.7       7.3       21.7       51.8       14.8       4.3       54       17.2       56.9       16.7       7.3       9.6       27.3       45.2         and Central Asia       4.6       7.7       22.4       28.4       9.0       2.9       45.4       15.4       35.8       11.5       7.2       11.4       31.8       18.6		Total	7.1	12.3	28.9	24.8	5.1	5.0	8.5	20.2	31.6	6.3	10.2	17.2	39.9	16.1	3.5
12.5       18.0       39.0       31.7       14.0       9.9       14.3       32.0       37.5       16.8       16.0       22.8       47.5       24.7         as       10.6       18.3       36.0       22.4       7.5       6.4       12.0       26.7       26.9       8.6       16.4       26.8       48.6       16.4         tates       7.6       11.6       37.6       53.2       16.7       6.1       10.0       34.0       37.7       18.0       19.8       25.3       67.1       14.8         at the Pacific       5.7       7.3       21.7       51.8       14.8       4.3       5.4       17.2       56.9       16.7       7.3       9.6       27.3       45.2         and Central Asia       4.6       7.7       22.4       28.4       9.0       2.9       49.1       15.4       35.8       11.5       7.2       11.4       31.8       18.6																	
10.6       18.3       36.0       22.4       7.5       6.4       12.0       26.7       26.9       8.6       16.4       26.8       48.6       16.4         7.6       11.6       37.6       35.2       16.7       6.1       10.0       34.0       37.7       18.0       19.8       25.3       67.1       14.8         5.7       7.3       21.7       51.8       14.8       4.3       5.4       17.2       56.9       16.7       7.3       9.6       27.3       45.2         4.6       7.7       22.4       28.4       9.0       2.9       4.9       15.4       35.8       11.5       7.2       11.4       31.8       18.6		Africa	12.5	18.0	39.0	31.7	14.0	9.9	14.3	32.0	37.5	16.8	16.0	22.8	47.5	24.7	10.7
7.6         11.6         37.6         35.2         16.7         6.1         10.0         34.0         37.7         18.0         19.8         25.3         67.1         14.8           5.7         7.3         21.7         51.8         14.8         4.3         5.4         17.2         56.9         16.7         7.3         9.6         27.3         45.2           4.6         7.7         22.4         28.4         9.0         2.9         4.9         15.4         35.8         11.5         7.2         11.4         31.8         18.6		Americas	10.6	18.3	36.0	22.4	7.5	6.4	12.0	26.7	26.9	8.6	16.4	26.8	48.6	16.4	5.9
5.7     7.3     21.7     51.8     14.8     4.3     5.4     17.2     56.9     16.7     7.3     9.6     27.3     45.2       4.6     7.7     22.4     28.4     9.0     2.9     4.9     15.4     35.8     11.5     7.2     11.4     31.8     18.6		Arab States	7.6	11.6	37.6	35.2	16.7	6.1	10.0	34.0	37.7	18.0	19.8	25.3	67.1	14.8	5.2
4.6 7.7 22.4 28.4 9.0 2.9 4.9 15.4 35.8 11.5 7.2 11.4 31.8 18.6		Asia and the Pacific	5.7	7.3	21.7	51.8	14.8	4.3	5.4	17.2	56.9	16.7	7.3	9.6	27.3	45.2	12.8
		Europe and Central Asia	4.6	7.7	22.4	28.4	9.0	2.9	4.9	15.4	35.8	11.5	7.2	11.4	31.8	18.6	5.7

				Total					Men					Women		
		* ↓	% working less than X hours	sss	% working mo than X hours	% working more than X hours	th	% working less than X hours	S	% wd more thai	% working more than X hours	tt %	% working less than X hours	ss s	% working more than X hours	ig more 10urs
		<15	<20	<35	>48	>60	<15	<20	<35	>48	>60	<15	<20	<35	>48	09<
		hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours
Percentage oi	Percentage of people working less than X or more than X hours a week by sex I Formal employment	n X hours a v	veek by sex	l Formal emp	loyment											
s	Africa	2.0	3.2	13.5	30.9	13.1	1.7	2.6	11.1	34.5	14.7	2.5	4.2	17.9	23.7	10.0
	Americas	1.3	3.5	12.8	20.0	6.1	1.0	2.4	9.2	24.4	7.9	1.7	5.2	18.1	13.3	3.4
	Arab States	3.2	6.5	42.0	29.2	14.3	2.7	5.2	36.0	33.6	16.5	6.0	14.4	75.4	4.0	1.0
qolə grig	Asia and the Pacific	0.9	1.5	6.1	45.2	11.5	0.9	1.5	0.9	50.0	13.2	0.9	1.6	5.9	39.2	9.1
	Europe and Central Asia	1.1	2.5	9.3	19.3	9.9	0.7	1.6	5.9	25.0	9.1	1.6	3.6	13.4	12.3	3.5
ə	Total	1.1	2.1	8.2	37.4	10.2	1.0	1.7	7.1	42.2	12.1	1.2	2.5	9.5	31.1	7.5
	Americas	5.6	10.9	25.3	18.2	3.0	4.2	7.8	19.2	24.3	4.0	7.1	14.2	31.9	11.5	1.9
	Asia and the Pacific	6.3	11.2	23.9	25.5	5.9	3.2	5.1	11.3	35.3	8.4	10.6	19.7	41.7	11.7	2.3
unoc Ieve(	Europe and Central Asia	2.4	5.2	19.9	12.8	1.8	0.7	1.7	7.4	18.1	2.7	4.3	9.0	33.9	6.8	0.9
	Total	4.2	8.2	22.4	16.9	3.0	2.4	4.4	12.1	23.4	4.2	6.3	12.5	34.5	9.2	1.5
	Africa	2.0	3.2	13.5	30.9	13.1	1.7	2.6	11.1	34.5	14.7	2.5	4.2	17.9	23.7	10.0
	Americas	3.5	7.3	19.3	19.1	4.5	2.6	5.0	14.1	24.4	6.0	4.7	10.2	25.8	12.3	2.6
rlq	Arab States	3.2	6.5	42.0	29.2	14.3	2.7	5.2	36.0	33.6	16.5	6.0	14.4	75.4	4.0	1.0
οW	Asia and the Pacific	1.6	2.7	8.3	42.8	10.8	1.2	1.9	6.6	48.2	12.6	2.1	3.8	10.3	35.8	8.2
	Europe and Central Asia	2.1	4.4	16.8	14.7	3.3	0.7	1.7	7.0	20.2	4.7	3.6	7.6	28.1	8.4	1.6
	Total	2.2	4.2	13.1	30.3	7.7	1.4	2.6	8.7	36.2	9.5	3.1	6.3	18.8	22.9	5.3

1		Λ
	-1	4
P	•	

es
<b>ye</b>
9
Ĕ
X
Se
þ
¥
Ve
a
Š
I
Ĕ
ž
lar
ŧ
J.C
Ĕ
Ŀ
×
an
ţĻ
SS
ĕ
ğ
ķi
õ
3
þ
60
f p
5
<u>6</u>
Ita
ier
Ľ
å
2
C.4.2
_
Table
Га

				101 <b>a</b> 1					Men					Women		
		#	% working less than X hours	S	% working more than X hours	g more ours	% v th	% working less than X hours	ş	% working more than X hours	rking 1 X hours	tt %	% working less than X hours	SS	% working more than X hours	g more 10urs
		<15	<20	<35	>48	>60	<15	<20	<35	>48	>60	<15	<20	<35	>48	>60
		hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours
Percentage of pe	Percentage of people working less than X or more than X hours a week by sex I Informal wage employment	X hours a w	eek by sex I	Informal wage	employm	ent										
	Africa	8.7	14.1	25.7	40.3	18.6	7.1	10.8	21.5	44.3	20.7	12.4	22.3	35.9	32.3	14.1
əirtr	Americas	7.5	14.9	31.3	22.8	7.9	4.5	9.7	23.0	28.1	10.1	11.3	21.8	42.1	15.8	5.2
inoc	Arab States	5.6	10.2	33.7	36.4	16.7	5.4	9.9	32.2	37.5	17.2	10.4	16.1	66.2	12.3	5.1
qol9\ gnig	Asia and the Pacific	1.6	2.6	9.4	56.8	14.4	1.3	2.2	8.3	58.9	15.5	2.1	3.4	11.2	53.1	12.5
nen	Europe and Central Asia	2.3	3.9	11.4	31.6	13.5	1.5	2.9	8.5	37.5	16.3	3.4	5.5	15.4	23.5	9.8
	Total	3.1	5.4	13.9	50.0	14.3	2.3	4.0	11.6	53.2	15.7	4.6	8.1	18.4	44.3	11.6
:	Americas	4.9	10.0	23.5	16.3	1.8	3.3	7.6	15.9	21.7	2.7	6.5	12.5	31.1	11.0	1.0
lopei tries A	Asia and the Pacific	7.5	14.1	32.1	23.3	7.0	4.4	9.9	22.1	28.0	11.0	10.7	18.4	42.1	18.4	6.1
unoo	Europe and Central Asia	5.7	10.6	28.1	9.4	1.5	3.0	6.0	16.1	13.9	1.9	8.3	14.8	39.0	5.4	1.2
	Total	5.7	11.0	26.6	15.7	2.9	3.4	7.6	17.2	20.8	3.8	7.9	14.4	35.7	10.9	2.2
A	Africa	8.7	14.1	25.7	40.3	18.6	7.1	10.8	21.5	44.3	20.7	12.4	22.3	35.9	32.3	14.1
A	Americas	6.9	13.8	29.4	21.2	6.5	4.2	9.2	21.4	26.7	8.5	10.0	19.3	39.2	14.5	4.1
P I	Arab States	5.6	10.2	33.7	36.4	16.7	5.4	9.9	32.2	37.5	17.2	10.4	16.1	66.2	12.3	5.1
	Asia and the Pacific	1.7	2.8	9.8	56.2	14.3	1.3	2.3	8.5	58.5	15.5	2.3	3.8	12.0	52.2	12.3
ш	Europe and Central Asia	3.3	5.8	16.2	25.2	10.7	1.9	3.6	10.3	31.7	13.4	5.0	8.6	23.4	17.5	7.6
F	Total	3.3	5.7	14.7	48.0	13.7	2.4	4.2	11.8	51.7	15.3	4.9	8.6	19.8	41.6	10.9

Image: conditient of the section of the se					Total					Men					Women		
60         <15			#1 %	working le han X hour	ss	% worki than X	ng more hours	th %	vorking les an X hours	ŝ	% wo more thar	rking 1 X hours	tt %	working le: han X hours	SS	% worki than X	ng more hours
Outs         Notis			<ul><li>&lt;15</li></ul>	. <20	<35	48	- <b>0</b> 9	. <15	<20	<35	. >48	09^ .	<15	<20	<35	48	. >60
117 $42$ $47$ $121$ $312$ $12.6$ $7.2$ $80$ $18.4$ $22.4$ $52$ $0.6$ $1.7$ $7.6$ $22.5$ $7.3$ $1.3$ $4.8$ $177$ $113$ $12.5$ $0.6$ $1.7$ $7.6$ $22.5$ $7.3$ $1.3$ $4.8$ $177$ $113$ $12.5$ $0.6$ $0.7$ $5.3$ $3.3$ $300$ $14.6$ $5.8$ $13.7$ $4.0$ $5.4$ $0.5$ $1.3$ $5.2$ $21.3$ $7.5$ $1.5$ $3.3$ $13.1$ $11.4$ $7.8$ $0.9$ $1.4$ $5.8$ $35.7$ $9.6$ $11.4$ $5.8$ $3.7$ $1.2$ $2.4$ $3.5$ $7.4$ $0.7$ $1.7$ $1.7$ $1.2$ $4.7$ $3.5$ $2.66$ $11.4$ $5.8$ $11.4$ $5.8$ $2.14$ $3.5$ $2.14$ $3.7$ $2.14$ $3.7$ $2.14$ $3.7$ $2.11$			hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours	hours
Mitica $51$ $57$ $142$ $283$ $117$ $42$ $47$ $121$ $312$ $126$ $12$ $8$ $177$ $113$ Mericas $09$ $30$ $119$ $177$ $52$ $06$ $17$ $52$ $73$ $40$ $88$ $17$ $113$ $40$ $103$ $30$ $146$ $58$ $137$ $40$ $101$ $25$ $13$ $13$ $10$ $11$ $40$ $51$ $52$ $32$ $40$ $101$ $101$ $101$ $101$ $101$ $101$ $101$ $101$ $101$ $101$ $117$ $52$ $112$ $121$ $1111$ $1111$ $1111$ $1111$	Percentage o	If people working less than X or more tha	n X hours a w	reek by sex	l Formal wag	e employme	ant										
Mericas         0.9         3.0         1.9         1.7         5.2         0.6         1.7         7.6         2.3         7.3         4.8         1.7         11.3           Mericas         3.2         6.7         4.1         2.8         1.2         5.3         38.3         300         146         5.8         137         7.37         4.0           Mab States         3.2         6.7         4.1         2.8         1.2         5.3         38.3         300         146         5.8         137         7.37         4.0           Europe and Central Asia         1.0         2.3         9.0         166         5.4         0.5         1.3         5.5         0.4         9.5         1.1         1.1           Ital         1.1         1.3         5.2         1.3         5.2         5.1         10.3         5.3         33.1         1.1         1.1         1.1           Marcinas         1.1         1.3         5.3         1.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3         5.3	S	Africa	5.1	5.7	14.2	28.3	11.7	4.2	4.7	12.1	31.2	12.6	7.2	8.0	18.4	22.4	10.1
May States         3.2         6.7         4.1         2.5         2.7         5.3         38.3         300         14.6         5.8         13.7         7.37         4.0           Kais and the Pacific         0.4         0.7         4.0         4.06         8.5         0.4         0.5         1.3         7.5         1.5         3.3         13.1         11.4           Total         1.0         2.3         9.0         16.6         5.4         0.5         1.3         7.5         1.5         3.3         13.1         11.4           Total         1.0         2.3         9.0         16.6         5.4         0.5         1.3         7.5         1.5         3.3         13.1         11.4           Americas         5.7         10.9         2.53         1.7         2.8         5.1         1.9         2.4         3.3         3.7         7.2         1.4         3.4         9.5         2.66           Americas         2.4         2.3         4.1         2.3         2.1         4.3         5.8         3.7         7.5         1.4         3.4           Mericas         2.4         3.2         2.4         1.7         2.4         3.7		Americas	0.9	3.0	11.9	17.7	5.2	0.6	1.7	7.6	22.5	7.3	1.3	4.8	17.7	11.3	2.4
Ref         Asia and the Pacific         0.4         0.7         4.0         4.0         8.5         0.4         0.5         1.5         1.2         0.4         0.8         4.4         35.8           Itempore and Central Asia         1.0         2.3         9.0         166         5.4         0.5         1.3         5.2         21.3         7.5         1.5         3.3         13.1         11.4           Itempore and Central Asia         1.0         1.8         7.4         31.8         7.8         0.5         1.9         5.5         21.3         7.5         1.5         3.3         13.1         11.4           Americas         5.7         10.9         25.3         1.74         2.8         0.7         19.5         2.4         9.5         26.6           Itempore and Central Asia         2.4         5.5         1.1         4.3         5.5         1.9         3.2         6.7         1.2         2.4         9.5         26.6           Itempore and Central Asia         2.4         2.5         1.1         4.3         3.4         1.3         1.1           Itempore and Central Asia         2.4         2.5         2.1         7.3         1.4         3.7 <td< th=""><th></th><td>Arab States</td><td>3.2</td><td>6.7</td><td>44.1</td><td>25.8</td><td>12.5</td><td>2.7</td><td>5.3</td><td>38.3</td><td>30.0</td><td>14.6</td><td>5.8</td><td>13.7</td><td>73.7</td><td>4.0</td><td>0.8</td></td<>		Arab States	3.2	6.7	44.1	25.8	12.5	2.7	5.3	38.3	30.0	14.6	5.8	13.7	73.7	4.0	0.8
Employeend Central Asia         10         23         90         166         54         05         13         52         213         75         15         33         131         114           Idal         10         18         74         318         78         09         14         58         357         96         12         24         95         266           Mericas         57         109         123         174         28         41         43         55         119         322         63         331         317         317         317         317           Mericas         57         120         256         233         41         43         55         119         322         63         331         317 <th></th> <td>Asia and the Pacific</td> <td>0.4</td> <td>0.7</td> <td>4.0</td> <td>40.6</td> <td>8.5</td> <td>0.4</td> <td>9.0</td> <td>3.5</td> <td>44.3</td> <td>10.2</td> <td>0.4</td> <td>0.8</td> <td>4.4</td> <td>35.8</td> <td>6.3</td>		Asia and the Pacific	0.4	0.7	4.0	40.6	8.5	0.4	9.0	3.5	44.3	10.2	0.4	0.8	4.4	35.8	6.3
Indation         10         18         74         18         78         0.9         14         5.8         35.7         9.6         12         2.4         9.5         26.6           Americas         5.7         10.9         25.3         17.4         2.8         7.9         19.3         23.3         3.7         7.2         14.2         31.7         11.1           Americas         6.7         12.0         25.6         22.3         4.1         4.3         5.5         11.9         32.3         3.7         7.2         14.2         31.7         9.7           Americas         2.4         5.2         20.1         10.9         14.4         2.8         4.1         7.3         15.3         4.1         9.7         4.1         9.7           Americas         5.1         5.1         10.9         2.5         2.3         2.4         1.7         7.3         19.3         17.7         19.7         17.1         17.7         18.4         17.7         19.7         14.7         17.1         17.7         14.7         17.4         17.4         17.4         17.4         17.4         17.4         17.4         17.4         17.4         17.4         17.4		Europe and Central Asia	1.0	2.3	0.0	16.6	5.4	0.5	1.3	5.2	21.3	7.5	1.5	3.3	13.1	11.4	3.0
Mmericas         57         109         253         174         28         43         79         193         72         142         317         111           Matricas         67         120         256         223         41         4.3         55         11.9         322         6.3         431         97           Matricas         24         5.2         201         109         1.4         0.7         1.7         7.3         16.8         203         431         97           Motion         24         5.2         201         109         1.4         0.7         1.7         7.3         15.3         1.9         32         6.3         338         6.0           Motion         4.3         8.4         2.2         2.1         10.9         1.4         34         6.3         32         3.3         1.2         347         347         84           Motion         4.1         2.2         2.1         4.2         2.4         3.7         2.4         3.4         1.2         3.4         3.4         4.4         4.4         4.4         4.4         1.2         3.4         4.4           Motion         5.7         7.1<	Ð	Total	1.0	1.8	7.4	31.8	7.8	0.9	1.4	5.8	35.7	9.6	1.2	2.4	9.5	26.6	5.3
Americas         5.7         10.9         25.3         17.4         2.8         4.3         7.9         19.3         3.7         7.2         14.2         31.7         11.1           Asia and the Pacific         6.7         12.0         25.6         22.3         4.1         4.3         5.5         11.9         32.2         6.3         43.1         9.7           Interpacific         6.7         12.0         25.6         20.1         10.9         1.4         0.7         1.7         7.3         4.1         9.7           Interpacific         6.7         12.0         26.6         20.1         10.9         1.4         0.7         1.7         7.3         19.2         8.3         4.3         9.7           Interpactific         5.1         5.1         10.9         2.2         2.3         2.1         4.1         3.8         6.0           Africa         5.1         5.7         2.4         1.7         12.1         7.3         12.4         3.4         2.2         4.1         2.4         2.7         8.4           Africa         5.7         7.3         14.2         2.1         4.2         2.4         4.7         10.2         2.5 <td< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th></td<>																	
Kinck         Kinck <th< th=""><th></th><td>Americas</td><td>5.7</td><td>10.9</td><td>25.3</td><td>17.4</td><td>2.8</td><td>4.3</td><td>7.9</td><td>19.3</td><td>23.3</td><td>3.7</td><td>7.2</td><td>14.2</td><td>31.7</td><td>11.1</td><td>1.8</td></th<>		Americas	5.7	10.9	25.3	17.4	2.8	4.3	7.9	19.3	23.3	3.7	7.2	14.2	31.7	11.1	1.8
Europe and Central Asia         2.4         5.2         201         109         1.4         0.7         1.7         7.3         15.3         1.9         4.2         8.9         3.38         6.0           Total         4.3         8.4         22.9         15.2         2.3         2.4         5.1         5.7         34.7         8.4         5.4           Africa         5.1         5.7         14.2         2.3         2.4         12.1         3.4         6.0         8.4           Africa         5.1         5.7         14.2         28.3         11.7         4.2         4.7         12.1         3.4         6.0           Americas         3.5         7.3         19.2         17.6         3.9         2.6         4.1         2.8         13.7         2.1         4.0           Americas         3.5         6.7         44.1         2.58         12.5         2.2         5.3         38.3         30.0         14.6         7.7         4.0           Americas         3.7         1.0         1.5         2.7         5.3         38.3         30.0         14.6         5.8         1.1.2           Anab States         3.7         1.0		Asia and the Pacific	6.7	12.0	25.6	22.3	4.1	4.3	5.5	11.9	32.2	6.3	10.8	20.3	43.1	9.7	1.2
Total         4.3         8.4         22.9         15.2         2.3         2.6         4.6         12.3         2.14         3.4         6.3         12.7         34.7         8.4           Africa         5.1         5.7         14.2         28.3         11.7         4.2         4.7         12.1         31.7         34.7         8.4           Americas         5.1         5.7         14.2         28.3         11.7         4.2         4.7         12.1         31.2         12.7         34.7         8.4           Americas         3.5         7.3         19.2         17.6         3.9         2.6         5.0         13.7         22.9         5.4         4.7         10.2         25.8         11.2           Arab States         3.2         6.7         44.1         25.8         12.5         2.7         5.3         38.3         30.0         14.6         5.8         11.2         4.0           Arab States         1.5         2.7         5.3         38.3         30.0         14.6         5.8         13.7         7.3         4.0           Asia and the Pacific         1.5         2.7         5.3         38.3         30.0         14.6		Europe and Central Asia	2.4	5.2	20.1	10.9	1.4	0.7	1.7	7.3	15.3	1.9	4.2	8.9	33.8	6.0	0.7
Mfrica $5.1$ $5.7$ $14.2$ $28.3$ $11.7$ $4.2$ $4.7$ $12.1$ $31.2$ $12.6$ $7.2$ $8.0$ $18.4$ $22.4$ Americas $3.5$ $7.3$ $19.2$ $17.6$ $3.9$ $2.6$ $5.0$ $13.7$ $22.9$ $5.4$ $4.7$ $10.2$ $25.8$ $11.2$ Arab States $3.2$ $6.7$ $44.1$ $25.8$ $12.5$ $2.7$ $5.3$ $38.3$ $30.0$ $14.6$ $5.8$ $13.7$ $4.0$ Arab States $3.2$ $6.7$ $44.1$ $25.8$ $12.5$ $2.7$ $5.3$ $38.3$ $30.0$ $14.6$ $5.8$ $13.7$ $7.7$ $4.0$ Arab States $3.2$ $6.7$ $44.1$ $25.8$ $12.5$ $2.7$ $5.3$ $38.3$ $30.0$ $14.6$ $5.8$ $13.7$ $7.7$ $4.0$ Arab States $2.7$ $7.8$ $3.73$ $7.7$ $1.0$ $1.5$ $5.0$ $42.1$ $9.5$ $2.2$ $4.2$ $11.2$ $31.7$ Arab States $2.0$ $4.4$ $16.8$ $12.6$ $2.6$ $0.6$ $1.6$ $6.7$ $17.1$ $3.5$ $7.2$ <th></th> <td>Total</td> <td>4.3</td> <td>8.4</td> <td>22.9</td> <td>15.2</td> <td>2.3</td> <td>2.6</td> <td>4.6</td> <td>12.3</td> <td>21.4</td> <td>3.4</td> <td>6.3</td> <td>12.7</td> <td>34.7</td> <td>8.4</td> <td>1.2</td>		Total	4.3	8.4	22.9	15.2	2.3	2.6	4.6	12.3	21.4	3.4	6.3	12.7	34.7	8.4	1.2
Africa $5.1$ $5.7$ $14.2$ $28.3$ $11.7$ $4.2$ $4.7$ $12.1$ $31.2$ $12.6$ $7.2$ $8.0$ $18.4$ $22.4$ $2.4$ Americas $3.5$ $7.3$ $19.2$ $17.6$ $3.9$ $2.6$ $5.0$ $13.7$ $22.9$ $5.4$ $4.7$ $10.2$ $25.8$ $11.2$ Americas $3.2$ $6.7$ $44.1$ $25.8$ $12.5$ $2.7$ $5.3$ $38.3$ $30.0$ $14.6$ $5.8$ $13.7$ $4.0$ Asia and the Pacific $1.5$ $2.7$ $7.8$ $37.3$ $7.7$ $1.0$ $1.5$ $5.0$ $42.1$ $9.5$ $2.2$ $4.2$ $11.2$ Luope and Central Asia $2.0$ $4.4$ $16.8$ $12.6$ $2.6$ $0.6$ $1.6$ $6.7$ $17.1$ $3.6$ $7.6$ $7.6$ $7.6$ Totole and Central Asia $2.4$ $4.5$ $13.6$ $25.1$ $5.6$ $0.6$ $17.1$ $3.6$ $3.5$ $7.4$ $7.8$ $7.6$ Totole and Central Asia $2.9$ $4.4$ $16.8$ $12.6$ $2.6$ $0.6$ $1.6$ $6.7$ $17.1$ $3.6$ $7.6$ $7.8$ $7.6$ Total $2.4$ $4.5$ $13.6$ $25.1$ $5.6$ $2.6$ $8.3$ $30.2$ $7.4$ $27.8$ $7.6$ Total $2.4$ $4.5$ $13.6$ $25.1$ $5.6$ $1.6$ $8.3$ $30.2$ $7.4$ $27.8$ $7.6$ Total $2.4$ $4.5$ $13.6$ $2.5$ $1.5$ $2.6$																	
Americas         3.5         7.3         19.2         17.6         3.9         2.6         5.0         13.7         22.9         5.4         4.7         10.2         25.8         11.2           Arab States         3.2         6.7         44.1         25.8         12.5         2.7         5.3         38.3         30.0         14.6         5.8         13.7         7.3         4.0           Arab States         3.2         6.7         44.1         25.8         12.5         2.7         5.3         38.3         30.0         14.6         5.8         13.7         7.7         4.0           Asia and the Pacific         1.5         2.7         7.8         37.3         7.7         1.0         1.5         5.0         42.1         9.5         2.2         4.2         11.2         31.2         31.2         31.2         31.2         31.2         31.2         31.2         31.2         31.2         31.2         7.4         27.8         7.6		Africa	5.1	5.7	14.2	28.3	11.7	4.2	4.7	12.1	31.2	12.6	7.2	8.0	18.4	22.4	10.1
Arab States       3.2       6.7       4.1       25.8       12.5       2.7       5.3       38.3       30.0       14.6       5.8       13.7       73.7       4.0         Asia and the Pacific       1.5       2.7       7.8       37.3       7.7       1.0       1.5       5.0       42.1       9.5       2.2       4.2       11.2       31.2         Europe and Central Asia       2.0       4.4       16.8       12.6       2.6       0.6       1.6       6.7       17.1       3.6       3.5       7.4       27.8       7.6         Total       2.4       4.5       13.6       25.1       5.6       1.5       2.6       8.3       30.2       7.3       7.4       27.8       7.6         Iotal       2.4       4.5       13.6       25.1       5.6       1.5       2.6       8.3       30.2       7.2       34       6.8       20.2       18.9		Americas	3.5	7.3	19.2	17.6	3.9	2.6	5.0	13.7	22.9	5.4	4.7	10.2	25.8	11.2	2.0
Asia and the Pacific         1.5         2.7         7.8         37.3         7.7         1.0         1.5         5.0         42.1         9.5         2.2         4.2         11.2         31.2           Europe and Central Asia         2.0         4.4         16.8         12.6         2.6         0.6         1.6         6.7         17.1         3.6         3.5         7.4         27.8         7.6           Total         2.4         4.5         13.6         25.1         5.6         1.5         2.6         8.3         30.2         7.2         3.4         6.8         20.2         18.9	rlq	Arab States	3.2	6.7	44.1	25.8	12.5	2.7	5.3	38.3	30.0	14.6	5.8	13.7	73.7	4.0	0.8
pe and Central Asia 2.0 4.4 16.8 12.6 2.6 0.6 1.6 6.7 17.1 3.6 3.5 7.4 27.8 7.6 2.4 4.5 13.6 25.1 5.6 1.5 2.6 8.3 30.2 7.2 3.4 6.8 20.2 18.9	οW	Asia and the Pacific	1.5	2.7	7.8	37.3	7.7	1.0	1.5	5.0	42.1	9.5	2.2	4.2	11.2	31.2	5.4
2.4         4.5         13.6         25.1         5.6         1.5         2.6         8.3         30.2         7.2         3.4         6.8         20.2         18.9		Europe and Central Asia	2.0	4.4	16.8	12.6	2.6	0.6	1.6	6.7	17.1	3.6	3.5	7.4	27.8	7.6	1.4
		Total	2.4	4.5	13.6	25.1	5.6	1.5	2.6	8.3	30.2	7.2	3.4	6.8	20.2	18.9	3.5
	( l																



Figure C.2 Proportion of workers working more than 60 hours a week, depending on the formal or informal nature of their main job

Note: Global and regional estimates based on 106 countries representing 88 per cent of the world employed population and based on country data for the latest available year. Harmonized definition of informal employment and employment in the informal sector.